04 University of Plymouth Research Theses

01 Research Theses Main Collection

2024

Developing an occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD)

Lovegrove, Christopher John

https://pearl.plymouth.ac.uk/handle/10026.1/22594

http://dx.doi.org/10.24382/5227 University of Plymouth

All content in PEARL is protected by copyright law. Author manuscripts are made available in accordance with publisher policies. Please cite only the published version using the details provided on the item record or document. In the absence of an open licence (e.g. Creative Commons), permissions for further reuse of content should be sought from the publisher or author.

COPYRIGHT STATEMENT

Copyright and Moral rights arising from original work in this thesis and (where relevant), any accompanying data, rests with the Author unless stated otherwise¹.

Re-use of the work is allowed under fair dealing exceptions outlined in the Copyright, Designs and Patents Act 1988 (amended)², and the terms of the copyright licence assigned to the thesis by the Author.

In practice, and unless the copyright licence assigned by the author allows for more permissive use, this means,

- that any content or accompanying data cannot be extensively quoted, reproduced or changed without the written permission of the author / rights holder; and
- that the work in whole or part may not be sold commercially in any format or medium without the written permission of the author/rights holder.

Any third-party copyright material in this thesis remains the property of the original owner. Such third party copyright work included in the thesis will be clearly marked and attributed, and the original licence under which it was released will be specified. This material is not covered by the licence or terms assigned to the wider thesis and must be used in accordance with the original licence; or separate permission must be sought from the copyright holder.

The author assigns certain rights to the University of Plymouth including the right to make the thesis accessible and discoverable via the British Library's Electronic Thesis Online Service (EThOS) and the University research repository, and to undertake activities to migrate, preserve and maintain the medium, format and integrity of the deposited file for future discovery and use.

¹ *E.g.* in the example of third party copyright materials reused in the thesis.

² In accordance with best practice principles such as, *Marking/Creators/Marking third party content* (2013). Available from:

https://wiki.creativecommons.org/wiki/Marking/Creators/Marking_third_party_content [accessed 28th February 2022]



DEVELOPING AN OCCUPATION-BASED COMPLEX INTERVENTION FOR LIVING WELL WITH ANXIETY IN PARKINSON'S (OBtAIN-PD)

by

Christopher John Lovegrove

A thesis submitted to the University of Plymouth in partial fulfilment for the degree of

DOCTOR OF PHILOSOPHY

School of Health Professions

March 2024

Acknowledgements

I will always remember the time when I worked on this PhD project as a period of intense professional and personal development. I started this PhD during the COVID-19 pandemic, which presented challenges, upheavals, and uncertainties. Nonetheless, I made it to the finish line! Collaborating with all the inspiring people who accompanied me on this transformative journey has been a privilege.

I want to thank my supervisors: Professor Jonathan Marsden for his academic expertise and commitment to supporting early career researchers and his knowledge of neurology and Parkinson's; Professor Katrina Bannigan for her expertise in occupational therapy, mental health and research, and for inspiring me in this project since 2015; and Dr Ingrid Sturkenboom for her insightful mentorship and expertise in occupational therapy and Parkinson's. Their steadfast support, guidance, and encouragement have been invaluable to my development as a clinical academic occupational therapist and as a person. Thank you to Professor Dave Richards for providing supervision and expertise regarding complex intervention development in the first year of my studies.

To Dr. Niina Kolehmainen, thank you for your mentorship and guidance, which have honed my thinking. I am incredibly proud to have you as a role model.

I acknowledge my colleagues from the University of Plymouth and Glasgow Caledonian University for enriching my journey with their expertise, camaraderie, and kindness. I also thank my colleagues from the Applied Parkinson's Research, Participation in Everyday Life, Rehabilitation, and SYNERGY research groups for welcoming me into their supportive, open,

i

inspiring, and ambitious environments. These have been an excellent example of how serious scientific work can be fun.

Thank you to my colleagues from Royal Devon University Healthcare NHS Foundation Trust for supporting my PhD project, particularly my Renal & Cancer Services and Supportive Care team members. Your compassion, empathy, and gentleness, which you pour into the world daily, are inspirational.

I want to thank my mother, Julia, my father, Graham, and my sister, Amy, for their unwavering support throughout my academic journey. Completing this PhD would not have been possible without their love, care, and encouragement. I am also grateful to my friends, who provided a much-needed break from my studies and helped me balance work and leisure. To all my family and friends, thank you for standing by me through the ups and downs of this journey. Your support and faith in me have been a constant source of strength and motivation.

I express my heartfelt appreciation to coffee for being my constant companion through countless late nights and early mornings. It has often been the necessary jolt I needed to keep my central nervous system going. I am also grateful for our canine family members, Newton, Flynn, Bear, and our cat, Fleur. They have been loyal companions, reminding me to take breaks and enjoy the outdoors, even when buried in work.

I deeply appreciate all the participants who invested their time and shared their opinions and everyday experiences with me during this research.

ii

This thesis is dedicated to my grandparents, Alan Lovegrove (1930-2020) and Peggy Bradford (1936-2020), who were unwavering and ardent in supporting my clinical academic pursuits. I wish they could have read it.

Author's Declaration

At no time during the registration for the Degree of Doctor of Philosophy has the author been registered for any other University award without prior agreement of the Doctoral College Quality Sub-Committee.

Work Submitted for this research degree at the University of Plymouth has not formed part of any other degree either at the University of Plymouth or at another establishment.

This research project was financed with the aid of a Clinical Doctoral Research Fellowship from the National Institute of Health and Care Research (NIHR301565) and supported by the School of Health Professions, University of Plymouth.

A programme of advanced study was undertaken, which included methodological and speciality specific training.

The following external institutions were visited for research and consultation purposes: Glasgow Caledonian University

Disclaimer: The content presents independent research funded by the NIHR. The views expressed are those of the author(s) and not necessarily those of the National Health Service (NHS), the NIHR, of the Department of Health and Social Care.

Word count of the main body of the theses: 49, 601 (80,000 max)

lough

Signed: Jate: 28th March 2024

Abstract

Lovegrove, C.

Developing an occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD)

Background: Parkinson's is the UK's second most common neurodegenerative condition, affecting approximately 175,000 people. Current anxiolytic medications are ineffective in treating anxiety in Parkinson's. The most promising behavioural interventions have inconclusive and mixed results. Occupational therapy is effective at promoting participation in activities of daily living and is recommended in national guidelines. An evidence-based occupational therapy intervention for living well with anxiety in Parkinson's does not exist.

Aim: To design a new occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD) in co-production with people with Parkinson's (PWPs), care partners, and occupational therapists.

Methods: A scoping review, conducted in line with Joanna Briggs Institute guidance, identified what occupational therapy interventions exist for adults with anxiety and the intervention characteristics. A participatory mixed methods research study involving five stages was conducted using online Group Concept Mapping (GCM). Logic modelling was used to co-produce the OBtAIN-PD with people with Parkinson's, carers, and occupational therapists.

Results: The scoping review included 18 studies and identified eight interventions and 31 different outcomes of current occupational therapy interventions for adults living with anxiety. Eighty-three participants took part in the GCM study. Cluster map, pattern match, and 'go-zone' charts were created using multivariate statistical analysis based on their

responses. The final map contained 119 statements with eight clusters (stress value 0.252): exercise, lifestyle changes, self-help, coping, access to information, professional help, peers and groups, support from others. Significant agreement existed between the importance and feasibility rating activities (r = -0.07). 'Go-zone' charts highlighted the priority statements for intervention development. Thirty-two people participated in logic modelling to co-produce the OBtAIN-PD, informed by the findings of previous studies. Resources to support the new intervention ('Inputs') include adequate resourcing, education for professionals and people with Parkinson's, flexibility of delivery methods, and goal setting. The intervention's actions to produce outcomes ('Processes') should include 1:1 support, lifestyle management, providing meaningful information, collaborative goal setting, therapeutic use of everyday activities, and involvement of friends and families. The intended outcomes should include anxiety symptom reduction, increased engagement in meaningful activities, improved understanding of anxiety and PD, better clinical outcomes, and enhanced service level outcomes. The resulting intervention manual, educational material, and training video were reviewed by five occupational therapists for content and comprehension.

Conclusion: The scoping review emphasises the necessity of using occupational therapy outcome measures, appropriate impairment-based measures, and economic evaluations for future interventions and research focused on evaluating the clinical and cost-effectiveness of occupational therapy services for individuals with anxiety. The development of the OBtAIN-PD was informed by a Group Concept Mapping study, which identified priority components. The OBtAIN-PD was co-produced with people with Parkinson's, care partners, and occupational therapists using a logic modelling process. A protocol for a feasibility cluster randomized controlled trial is presented.

vi

Preface

The work presented henceforth was conducted as part of an NIHR Clinical Doctoral Research Fellowship (NIHR301565). This doctoral research thesis adopts a hybrid monograph and article-compilation structure.

Chapters 1, 5, 7, and 8 are monographs authored by Chris Lovegrove.

Chapter 2 consists of a scoping review protocol (published in JMIR Research Protocols) and a scoping review (in review with JMIR Aging at the time of submission).

Chapter 3 presents an article detailing a group concept mapping study published in the Journal of Parkinson's Disease.

Chapter 4 presents an article detailing the logic modelling study used to co-produce the OBtAIN-PD study in peer review with BMJ Open at the time of submission.

Chapter 6 presents an article on the feasibility cluster randomised controlled trial in peer review with BMJ Open at the time of submission.

Chris Lovegrove was the lead author and investigator for the above articles and was responsible for all major areas of concept formation, data collection and analysis, and manuscript composition.

Table of Contents

Chapter 1: Introduction	1.
1.1 Researcher's background	5.
1.2 Anxiety and Parkinson's	8.
1.2.1 What is Parkinson's?	8.
1.2.2 What is anxiety?	11.
1.2.3 Anxiety in Parkinson's	18.
1.2.4 The cost of anxiety in Parkinson's	23.
1.3 Current interventions	26.
1.4 Complex intervention development	30.
1.5 Occupational Therapy	31.
1.6 Patient and Public Involvement and Engagement	35.
1.7 Methodological Orientation	37.
1.8 Aim and Structure	40.
Chapter 2: Scoping Review	43.
2.1 Characteristics of Occupational Therapy Interventions for	
Community-Dwelling Adults With Anxiety: Protocol for a Scoping Review	55.
2.2 Characteristics of occupational therapy interventions for	
community-dwelling adults with anxiety: a scoping review	68.
Chapter 3: Group Concept Mapping Study	99.
3.1 Concept mapping to define components for an	
occupation-based intervention for Parkinson's & anxiety	105.

Chapter 4: Logic Modelling Study		
4.1 Developing an occupation-based complex intervention for living well		
with anxiety and Parkinson's (OBtAIN-PD) using logic modelling	133.	
Chapter 5: Intervention Specification	156	
5.1 Developing the OBtAIN-PD	156.	
5.1.1 Brief name	161.	
5.1.2 Why	161.	
5.1.3 What (material)	162.	
5.1.4 What (procedures)	162.	
5.1.5 Who provided	177.	
5.1.6 How	178.	
5.1.7 Where	179.	
5.1.8 When and how much	179.	
5.1.9 Tailoring	180.	
5.1.10 Modifications (made during the research study)	180.	
5.1.11 How well (planned and actual measurement adherence and fidelity)		
5.2 Summary	182.	
Chapter 6: Feasibility Trial Protocol	183.	
6.1 Evaluating the occupation-based complex intervention for living well		
with anxiety and Parkinson's disease (OBtAIN-PD): A feasibility cluster		
randomised controlled trial protocol	184.	
6.2 Chapter Summary	209.	

Chapter 7: Discussion and Conclusions	
7.1 Scoping review- summary and recommendations	
7.2 Group concept mapping- summary and recommendations	
7.3 Logic modelling- summary and recommendations	
7.4 Feasibility cluster RCT protocol- summary and recommendations	213.
7.5 Complex interventions	214.
7.6 Occupational Therapy	216.
7.7 Developing Evidence	219.
7.8 International Context	220.
7.9 Strengths and Limitations	224.
7.10 Challenges and lessons learned	226.
7.11 Implications for future research	228.
7.12 Conclusion- final thoughts	230.
References	232.
Appendices	312.
Appendix 1: Protocol search (Chapter 2)	312.
Appendix 2: Data extraction tool (Chapter 2)	
Appendix 3: PRISMA-SCr checklist (Chapter 2)	
Appendix 4: Scoping review searches (Chapter 2)	317.
Appendix 5: Abbreviations used in the scoping review (Chapter 2)	321.
Appendix 6: Group Concept Mapping statements (Chapter 3)	323.
Appendix 7: Logic modelling information pack (Chapter 4)	327.
Appendix 8: OBtAIN-PD intervention manual (Chapter 4)	332.

Appendix 9: OBtAIN-PD Parkinson's and Anxiety information sheet(Chapter 5)355.Appendix 10: OBtAIN-PD information sheets (Chapter 5)357.Appendix 11: OBtAIN-PD therapist contact form (Chapter 5)363.Appendix 12: OBtAIN-PD clinician interview schedule (Chapter 5)364.

List of Figures

Figure 1.1: Summary of participants and stakeholders in this research project.

Figure 1.2: International Classification of Functioning, Disability and Health.

Figure 1.3: The metacognitive model of GAD with feedback labels added.

Figure 1.4: Developing an occupation-based complex intervention for living well with

anxiety in Parkinson's (OBtAIN-PD) study in relation to the Medical Research Council

Framework.

Figure 1.5: "Do-Live-Well". A Canadian framework for promoting occupational, health, and wellbeing.

Figure 1.6: The final OBtAIN-PD logo.

Figure 1.7: Overview of the PhD research activities.

Figure 2.1: Search results and study selection and inclusion process.

Figure 3.1: Line graph summarising participation across each GCM activity.

Figure 3.2: Cluster map showing eight priority themes for intervention development.

Figure 3.3: Pattern match comparing the mean importance and feasibility rating scores at the cluster level (upper limit= 5).

Figure 3.4: Go-zone plots for each of the eight clusters.

Figure 4.1: The blank logic model used with participants to co-produce the OBtAIN-PD.

Figure 4.2: Process of aggregating the multiple logic models to develop the OBtAIN-PD.

Figure 4.3: Consolidated logic model for the occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD).

Figure 5.1: Example of a completed goal planning sheet. This example is based on a real-life exemplar provided a person with Parkinson's acting as a PPI stakeholder.

Figure 6.1: The OBtAIN-PD feasibility study questions.

Figure 6.2: The OBtAIN-PD feasibility cluster randomised controlled trial in relation to the

Medical Research Council framework. Adapted from Skivington et al. (2021)

List of Tables

Table 1.1: Parkinson's symptoms.

 Table 2.2: Scoping review data chart. Multiple articles relating to the same study are matching colour-coded.

 Table 2.2: Characteristics of included articles.

Table 2.3: Characteristics of the identified interventions.

Table 2.4: Outcome measures used in the interventions.

 Table 2.5: Observed impacts of the interventions.

Table 3.1: Participant characteristics of respondents to demographic questions.

Table 4.1: Participants demographic characteristics.

Table 4.2: Illustrative quotes from the consolidated logic model.

Table 5.1: Development and reporting of the OBtAIN-PD in relation to the MRC framework

(2008) and the OBtAIN-PD project.

Table 5.2: Approaches to developing complex interventions.

Table 5.3: Summary of the activities of the OBtAIN-PD.

Table 5.4: Examples of the communication skills used during the OBtAIN-PD initial session.

Table 5.5: Summary of the OBtAIN-PD information sheets.

Table 6.1: Outcome measure activities.

Table 6.2: The OBtAIN-PD feasibility cluster RCT progression criteria.

List of Boxes

Box 5.1: Template for Intervention Description and Replication (TIDieR) checklist (Hoffmann

et al., 2014).

List of Abbreviations

- 5-HT: Serotonin
- αSyn: Alpha-synuclein
- A&E: Accident and emergency
- ACS-3: Activity Card Sort
- ADIS-IV: Anxiety Disorders Interview Schedule
- AEs: Adverse events
- APRG: Applied Parkinson's Research Group
- AR: Applied relaxation
- BAI: Beck Anxiety Inventory
- BDI-II: Beck's Depression Inventory
- BLA: Basolateral amygdala
- BMA: Basomedial amygdala
- BNST: Bed nucleus of the stria terminalis
- BR: Breathing retraining
- CA3: Cornu ammonis 3
- CBT: Cognitive behavioural therapy
- CeA: Central amygdala
- CINAHL: Cumulative Index to Nursing and Allied Health Literature
- CoA: Cortical amygdala
- COPD: Chronic obstructive pulmonary disease
- COPM: Canadian Occupational Performance Measure

- CONSORT: Consolidated Standards of Reporting Trials
- COVID-19: Coronavirus disease 2019
- CReDECI: Criteria for reporting the development and evaluation of complex interventions in healthcare
- CRTs: Community Rehabilitation Teams
- CS: Conditioned stimulus
- CSQ: Client Satisfaction Questionnaire
- DALYs: disability-adjusted life years
- DASS: Depression Anxiety Stress Scale
- DAT: Dopamine transporter
- DSM-5-TR: Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition, Text Revision
- EQ-5D: EuroQol-5D
- EQ-VAS: EuroQol visual analogue scale
- FOG: Freezing of gait
- GABA: Gamma-aminobutyric acid
- GAD: Generalised anxiety disorder
- GAD-7: Generalised Anxiety Disorder assessment
- GCM: Group concept mapping
- GHQ-28: General Health Questionnaire-28
- GP: General practitioner
- Haq-II: Health assessment questionnaire
- hd-EEG: High-density electroencephalography
- IAPT: Improving Access to Psychological Therapies
- ICERs: Incremental cost-effectiveness ratios
- ICF: International Classification of Functioning, Disability, and Health
- iRBD: Idiopathic rapid eye movement sleep behaviour disorder
- ISRCTN: International Standard Randomised Controlled Trial Number
- JBI: Joanna Briggs Institute
- JMIR: Journal of Medical Internet Research
- MADRS-S: Montgomery-Åsberg Depression Rating Scale- Self report
- MANSA: Manchester Short Assessment of Quality of Life

- MBIs: Mindfulness-based interventions
- MClinRes: Masters in Clinical Research
- MeA: Medial amygdala
- MEDLINE: Medical Literature Analysis and Retrieval System Online
- MRC: Medical Research Council
- MRI: Magnetic resonance imaging
- NAc: Nucleus accumbens
- NE: Norepinephrine
- NGT: Nominal Group Testing
- NICE: National Institute for Health and Care Excellence
- NIHR: National Institute of Health and Care Research
- NHS: National Health Service
- NMAHP: Nursing, midwifery, and allied health professionals
- NMS: Non-motor symptoms
- OBtAIN-PD: Occupation-based intervention for living well with anxiety in Parkinson's
- Oval-pd: Occupational value assessment with predefined items
- PaG: Periaqueductal gray
- PAG: Patient Advisory Group
- PANAS: Positive and Negative Affect Schedule
- PDQ-39: Parkinson's Disease Questionnaire
- PenCTU: Peninsula Clinical Trials Unit
- PenPRIG: Peninsula Parkinson's Research Interest Group
- PET: Positron emission tomography
- PD: Parkinson's disease
- PFC: Prefrontal cortex
- PIEL: Participation in Everyday Life
- PLOS: Public Library of Science
- PMS: Pearson Mastery Scale
- PPI: Patient & public involvement
- PPIE: Patient & public involvement and engagement
- PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses

- PRISMA-P: Preferred Reporting Items for Systematic Reviews and Meta-Analyses Protocols
- PRISMA-ScR: Preferred Reporting Items for Systematic Reviews and Meta-analyses extension for scoping review
- PSP: Progressive Supranuclear Palsy
- PSS: Perceived Stress Scale
- PWPs: People with Parkinson's
- OBQ: Occupational Balance Questionnaire
- QALYs: Quality-adjusted life years
- QCRI: Qatar Computing Research Institute
- REC: Research Ethics Committee
- RCT: Randomised controlled trial
- ReDO[®]: Redesigning Daily Occupations[®]
- RTT: Referral to treat
- SAEs: Serious adverse events
- SCL-90-r: Symptom Checklist-90-revised
- SDO: Satisfaction with Daily Occupations
- SF-36: 36-item Short Form Survey
- SERT: Serotonin transporter
- SOC: Sense of Coherence Scale
- SPIRIT: Standard Protocol Items: Recommendations for Intervention Trials
- SSMH: Specialist section mental health
- SSNP: Specialist section neurological practice
- STAI: State-Trait Anxiety Inventory
- SYNERGY: Stroke and Neurological Rehabilitation Research Group
- SUMARI: System for the Unified Management, Assessment and Review of Information
- TIDieR: Template for Intervention Description and Replication
- TMG: Trial management group
- TRIP: Turning research into practice
- TSC: Trial steering committee
- TTM[®]: Tree Theme Method[®]

- UCS: Unconditioned stimulus
- UK: United Kingdom
- US: United States
- USD: United States dollar
- VTA: Ventral tegmental area
- WEIS-SR: Work Environment Impact Scale- Self Rating
- WHO: World Health Organization
- WHODAS 2.0: World Health Organization Disability Assessment Schedule
- WRS: Work Readiness Scale
- WSAS: Work and Social Adjustment Scale
- YOPD: Young-onset Parkinson's disease

Chapter 1: Introduction

Parkinson's disease, increasingly known as Parkinson's, is the second most common neurodegenerative condition behind Alzheimer's type dementia (de Lau & Breteler, 2006). Around 10 million people live with Parkinson's globally (Ben-Shlomo et al., 2024; Dorsey, Elbaz, et al., 2018). In the UK, this number is estimated to be 153, 000, likely to increase to 172,000 by 2030 (Parkinson's UK, 2018). The cardinal motor symptoms of Parkinson's include tremors, motor automaticity reduction (loss of automatic movement), bradykinesia (slowed movement), rigidity, and postural instability (Kalia & Lang, 2015). Parkinson's also results in fatigue and sensory, autonomic, cognitive, and behavioural impairments that cause significant disability and impede participation in everyday roles and activities (Barone et al., 2009; Batzu et al., 2024; Benharoch & Wiseman, 2004), leading to reductions in quality of life (Duncan et al., 2014). These problems are collectively known as non-motor symptoms (NMS). Around 98% of people with Parkinson's experience NMS (Barone et al., 2009). Anxiety is a NMS that affects approximately 40-56% of people with Parkinson's (Pontone & Mills, 2021). Reviews indicate no effective pharmacological interventions to reduce anxiety in people with Parkinson's (Pontone & Mills, 2021). Psychological interventions for anxiety in non-Parkinsonian populations are recommended by NICE guidelines and the NHS Talking Therapies for Anxiety and Depression programme, formerly known as the Improving Access to Psychological Therapies (IAPT) programme in the UK (National Collaborating Centre for Mental Health, 2018; NICE, 2011). The NHS Talking Therapies and IAPTs programmes were developed to improve the access and delivery of cost-effective, evidence-based psychological therapies with long-term financial and health benefits (Heuzenroeder et al., 2004). However, evidence for the effectiveness of mindfulness and cognitive behavioural therapy approaches is limited and inconclusive in

people with Parkinson's (Biundo et al., 2016; Hofmann et al., 2010; Johnsen & Friborg, 2015; McLean et al., 2017). Given the shortage of clinical psychologists in the UK, additional approaches are necessary (The King's Fund, 2015).

Individuals with Parkinson's disease have expressed a need for an intervention that helps them live well with the complex experience of anxiety (Lovegrove et al., 2017). This intervention should focus on "doing" or participating in meaningful activities as both a means and a goal of the intervention rather than solely discussing thoughts and feelings (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). With dual training in physical and mental health and expertise in improving wellbeing through participation, occupational therapists are in an ideal position to support people with Parkinson's experiencing anxiety (Christine, 1990).

This thesis reports on the development and co-production of an occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD) using the Medical Research Council (MRC) framework for developing and evaluating complex interventions (Skivington et al., 2021). This work is important because of the need to find a way to best support people with Parkinson's to live well with anxiety using a novel intervention for this population. That is, it does not attempt to take an established intervention (such as cognitive behavioural therapy) and make it 'fit' people with Parkinson's. Instead, it involves people with Parkinson's in co-producing the intervention from the ground up, founded on their experiences and what they feel that they need to live well with anxiety. Care partners and occupational therapists similarly provide their input. The infographic in Figure 1.1 summarises all participants in this research programme. The OBtAIN-PD contributes to original practice by using multiple systematic methodologies to co-produce a new

intervention for living well with anxiety in Parkinson's, delivered by occupational therapists in community settings and utilising lifestyle management, motivational interviewing, and behavioural activation concepts. This thesis reports on the development phase of the MRC framework (Skivington et al., 2021). The thesis concludes with a protocol for a feasibility cluster randomised controlled trial (RCT). Ultimately, this work will lead to said feasibility RCT, which will provide the necessary operational experience and preparatory data to inform the design and implementation of a definitive RCT to test the clinical and costeffectiveness of the OBtAIN-PD should it demonstrate promise as an intervention. The feasibility RCT will address the feasibility phase of the PRC framework (Skivington et al., 2021)

This chapter will present the context of the researcher's position. It will then present the sequelae of anxiety in Parkinson's. An overview of the neurobiology of Parkinson's anxiety will be given, along with an explanation of the personal and societal impacts. Anxiety in Parkinson's is thought to be common due to the dopamine deficiency that is characteristic of the condition. This chapter also describes the limitations of current interventions and justifies a novel approach to living well with anxiety and Parkinson's. An explanation and justification for using an occupation-based approach is provided. Finally, the thesis structure is presented.

Btain-PD**Facts**



Figure 1.1: Summary of participants and stakeholders in this research project.

Language use in this thesis

The term "Parkinson's disease" is used globally. There is an increasing movement to drop the word 'disease' (which has connotations of contagion and infectiousness) and refer to the condition simply as "Parkinson's". This is in keeping with the wishes of people living with Parkinson's. As a result, the condition will be referred to as Parkinson's throughout this thesis without abbreviation. Similarly, the terms "people with Parkinson's" and "care partners" will not be shortened, aligning with the social model of disability (Lawson & Beckett, 2021). Exceptions to these rules will occur in the chapters presenting peerreviewed articles. In these articles, more traditional terminology and abbreviations are used in keeping with the respective journal's scope and word count. People with Parkinson's and care partners acting as Patient & Public Involvement and Engagement (PPIE) stakeholders have deemed the terminology in these articles acceptable, as getting research findings published is a priority for them. Further, these articles will use either UK or US English, depending on the journal's specific guidelines. This will be highlighted in the introductory text to each article chapter. Figures, tables, and boxes in these articles will be sequentially ordered as they appear in this thesis for ease of reading.

1.1 Researcher's background

This section provides a context to the researcher's position on the research topic.

My journey as an occupational therapist began 17 years ago when I started an undergraduate degree in Occupational Therapy. After graduating, I worked in supported housing for homeless young people before achieving my first professional goal: starting a junior rotation in the NHS. During this time, I discovered that neurology was my clinical passion, which I fortunately had an affinity for. What followed was gaining a specialist split role working in stroke rehabilitation and inpatient subacute neurorehabilitation. As my career progressed, I was fortunate to become a team lead occupational therapist on an acute neuromedical ward in 2012.

It was at this time that I started to work routinely with people with Parkinson's and their care partners. Despite being medically optimised and on the best pharmacological

treatment, they remained in the hospital, unable to mobilise and struggling to engage in rehabilitation programmes. I started to work with this group to develop an inpatient Parkinson's Therapy Pathway (Lovegrove & Marsden, 2020). Rapidly, it became apparent that many of the people I was working with were highly anxious. This resulted in substantially prolonged hospital admissions for these people by exacerbating primary symptoms such as postural instability. One of these people described their experience as follows (anonymised):

"It's awful! I can't breathe; I can't move I feel like I'm going to fall over. I feel like I'm going to die!" Neil

I did not know how to support this person best, so I asked my senior colleagues for advice. They were similarly uncertain and allowed me some time to work with the Exeter Health Library to review the literature. With support from the information specialists, it became clear that there was no "gold standard" treatment to help people with Parkinson's affected by anxiety. At the same time, I managed to spend some time with a clinical health psychologist who came to visit Neil on the ward. They explained that, at that time, treatments were limited for this population. Anxiety was perceived as a part of Parkinson's that people had to live with. My internal response was *'absolutely not'*. Remembering Neil's words, I made my first steps towards making a difference.

After a couple of years of being unsure how to get started, I gained a place on an NIHRfunded Masters in Clinical Research (MClinRes) degree at the University of Plymouth. Professor Katrina Bannigan supervised my project, and I am incredibly fortunate that she agreed to be on my PhD supervisory team. The degree course gave me the foundational research skills to challenge the existing evidence base and current treatments to improve

care for people with Parkinson's. Coming from my background as an occupational therapist, I realised that there was an opportunity to work with people with Parkinson's living with anxiety to tailor an intervention to their needs based on their lived experience. This project resulted in two peer-reviewed publications (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). These publications provide the foundation for the research programme outlined in this thesis.

A desire to progress my research towards realisation (a new intervention to help people with Parkinson's to live well with anxiety), as well as my knowledge and skills in clinical research, prompted me to seek supervision from experts in the field. From my MClinRes, I was privileged to work closely with my now Director of Studies, Professor Jon Marsden, a UK expert in neurorehabilitation and Parkinson's. I travelled to Nijmegen, Netherlands, to meet Dr Ingrid Sturkenboom using funding won from a successful grant application to the Royal College of Physicians Edinburgh (being the first NMAHP candidate awarded the Lindsay Bequest and Reid research grant). A recognised world expert in occupational therapy for people with Parkinson's, I was privileged that Dr Sturkenboom joined my supervisory team to support my project and development. The invaluable support of my supervisors and colleagues at the Peninsula Clinical Trials Unit (PenCTU) resulted in my being awarded an NIHR Clinical Doctoral Research Fellowship (NIHR301565) in 2021.

I am fortunate to have been invited to join several research groups. These are the Participation In Everyday Life researcher group (PIEL, University of Plymouth), Applied Parkinson's Research Group (APRG, University of Plymouth), Rehabilitation (University of Plymouth), and the Stroke and Neurological Rehabilitation Research Group (SYNERGY, Glasgow Caledonian University). These multidisciplinary collaborations of academic and

clinical members aim to generate evidence and translate the results from published research to influence and inform everyday clinical practice, focusing on the problems faced by people living with various long-term conditions, including Parkinson's.

These associations have inspired and motivated me to focus on improving the lives of people with Parkinson's living with anxiety.

1.2 Anxiety and Parkinson's

1.2.1 What is Parkinson's?

Parkinson's is a progressive neurodegenerative condition with unknown cause and no cure, resulting from the loss of dopamine-containing cells of the substantia nigra that results in a movement disorder with motor and non-motor symptoms (Emamzadeh & Surguchov, 2018; Jankovic & Tan, 2020; NICE, 2022). It affects dopaminergic and non-dopaminergic brain regions and does not become clinically apparent until at least 50% of dopaminergic cell activity has been lost (Dujardin & Sgambato, 2020; NICE, 2022). This definition will be used throughout this thesis. Environmental and genetic factors increase the risk of developing Parkinson's, with prevalence at 0.3% in industrialised countries and 1% in people above the age of 60 (Cherian & Divya, 2020; Islam et al., 2021; Lee & Gilbert, 2016). It is the second most common neurodegenerative condition after Alzheimer's type dementia and is expected to double in the next ten years (Dorsey et al., 2007; G.B.D. Neurology Collaborators, 2019).

Parkinson's presents complex motor and non-motor symptoms (Bloem et al., 2021) (Table 1.1). Diagnosis is based on asymmetrical bradykinesia, rigidity, and/or resting tremor (Rawat & Pandey, 2022). Postural instability is an advanced-stage cardinal symptom (Jankovic &

Tan, 2020). Early postural instability and poor response to medication may indicate a different diagnosis (Tolosa et al., 2021). Parkinson's is diagnosed when motor symptoms appear, but a premotor phase with NMS features is likely (Kumaresan & Khan, 2021). Parkinson's has no cure yet, and treatments only partially relieve symptoms, which may have side effects (LeWitt & Chaudhuri, 2020). People with Parkinson's must manage the condition and its debilitating symptoms.

	Presentation
Motor Symptoms	Bradykinesia, 4-6 Hz resting tremor, rigidity and postural instability. Impaired balance early in the disease is a contra-indication. Impaired gait pattern (e.g. shuffling gait, freezing), reduced voluntary movement control, postural abnormalities, hypomimia (facial masking), dysarthria, dysphagia.
Non-motor Symptoms	Higher cognitive dysfunction (e.g. executive dysfunction, mental rigidity), dementia, personality change, visuospatial perceptual disorder, apathy, depression, anxiety. Sleep disorders including REM sleep behaviour disorder, insomnia, excessive daytime sleepiness. Sensory problems can include visual dysfunction, proprioceptive dysfunction, pain, impaired smell). Autonomic symptoms include constipation, nocturia, and sexual dysfunction. Fatigue (may be related to both motor and NMS).

Table 1.1: Parkinson's symptoms. The number and severity of symptoms vary between individuals and increase with disease progression. Symptom severity may fluctuate throughout the day depending on medication response.

The International Classification of Functioning, Disability, and Health (ICF) can describe Parkinson's disease's impact on people living with it (World Health Organization, 2001) (Figure 1.2). Parkinson's disease affects people in different ways depending on their body structure and functions, activities, participation, and contextual factors. Coping with the disease in daily life is influenced by personal, physical, social, and environmental factors (Hultqvist et al., 2020). As Parkinson's progresses, people often face more difficulties performing everyday activities and participating in meaningful occupations (Haahr et al., 2021; Read et al., 2022; Rosengren et al., 2021).





Body function is impaired by Parkinson's symptoms and can also be affected by medication side effects, inactivity, falls or maladaptive coping (Pontone et al., 2023) (Table 1.1). Comorbid conditions and ageing contribute to further impairments (Carroll et al., 2024). High cognitive load for complex daily activities hinders motor performance (Nodehi et al., 2021) and executive dysfunction impedes task planning and organisation (Lin et al., 2023). Other NMS, such as visuospatial difficulties and anxiety, also affect daily activities and exacerbate symptoms like tremors and freezing of gait (Chagas et al., 2017; Fernández-Baizán et al., 2022; Ghielen et al., 2020). Environmental restrictions like narrow spaces, crowds or darkness worsen freezing of gait (Nieto-Escamez et al., 2023). Parkinson's can force people to adjust their daily activities due to difficulties and embarrassment (Angulo et al., 2019). As the condition progresses, spontaneous social engagement decreases and paid employment is often stopped earlier than planned (Ahn et al., 2022; Koerts et al., 2016; Read et al., 2022).

1.2.2 What is Anxiety?

Anxiety is a normal, temporary emotional and physical response to stress that can be beneficial in some situations, such as increasing attention or focusing on a work task (Muskin, 2023). Anxiety disorders are not just temporary feelings of anxiousness or nervousness. They involve intense feelings of anxiety or fear that cause clinically significant distress or impairment in daily activities (*Diagnostic and statistical manual of mental disorders : DSM-5™*, 2013; Muskin, 2023). These definitions are used throughout this thesis.

Anxiety is a response to the unknown, while fear is a reaction to a known danger (Steimer, 2002). In normal physiological conditions, both fear and anxiety serve the purpose of anticipating, responding to, and adapting to danger and threats (Steimer, 2002). Increased arousal, expectancy, autonomic and neuroendocrine activation, and specific behaviour patterns are characteristics of both. Although fear and anxiety may appear very similar, it is important to note that they are two separate and distinguishable phenomena. From an evolutionary perspective, fear is a term used to describe the emotions and responses that people experience when the threat is immediate or imminent. In contrast, anxiety is used to describe the emotions people experience when the source of harm is uncertain or distant in space or time (LeDoux & Pine, 2016). Anxiety disorders are a group of psychiatric disorders

characterised by feelings of worry, anxiety or fear that interfere with daily activities (Penninx et al., 2021). Anxiety disorders include separation, selective mutism, specific phobia, social anxiety disorder, and generalised anxiety disorder (Penninx et al., 2021). The prevalence of anxiety disorders in the general population, evidenced in the literature, has varied dramatically between 1980 and 2022, ranging from as low as 2.4% to as high as 31% (Baxter et al., 2013; Dettmann et al., 2022). It was reported by the World Health Organization (WHO) that in 2019, around 301 million people were suffering from anxiety disorders, including 58 million children and adolescents. As per the WHO, anxiety disorders are more prevalent among females, with 4.6% compared to 2.6% in males globally, as of 2015.

Fear is a universal biological response shared by humans and animals (LeDoux, 1998). Pavlovian fear conditioning is commonly used to study fear acquisition and its behavioural and neural mechanisms (Shechner et al., 2014). This is a form of associative memory formation where a conditioned stimulus (CS), such as an auditory tone, is paired with an aversive unconditioned stimulus (UCS), such as a foot shock. As a result, the CS forms a memory that triggers freezing, a behavioural response indicating fear. Fear extinction is a process where repeated exposure to fear-related stimuli lessens defensive reactions without harm (Whittle et al., 2021). This way fear conditioning and fear extinction, a process requiring active involvement of cortical areas, interact with each other and influence behaviour (Shechner et al., 2014). Anxiety disorders cause people to focus on fear-inducing information excessively. This can lead to a detailed processing of such information, which can further worsen their anxiety symptoms (Bystritsky et al., 2013). Generalised anxiety disorders are characterised by excessive and unrealistic worry, while panic attacks are defined by repeated episodes of fear in the absence of a fear stimulus.

Previous research suggests that fear is regulated by a distributed neuronal network spanning brain-wide projection pathways and localised neural connections (Tovote et al., 2015). The amygdala, hippocampus, and cortical regions play a critical role in fear and anxiety responses in the limbic system. People living with anxiety have been found to experience volume reductions in the hippocampus and amygdala regions of the brain (van Mierlo et al., 2015; Vriend et al., 2016). The amygdala is a central brain region responsible for processing emotions, particularly fear and anxiety. This is evidenced by studies investigating animal fear circuits (Han et al., 1997; Lang et al., 2000; LeDoux & Pine, 2016; Zelikowsky et al., 2014). Functional imaging studies have reported amygdala activation during fear conditioning and extinction in humans (Agren et al., 2012; LaBar et al., 1998). The amygdala is a complex structure made up of multiple substructures. These substructures work together to process emotions and include the basolateral amygdala (BLA), the basomedial amygdala (BMA), the central amygdala (CeA), the medial amygdala (MeA), and the cortical amygdala (CoA). The BLA and CoA are crucial in processing anxiety and fear(Beyeler & Dabrowska, 2020; Janak & Tye, 2015). It is known that imminent danger activates the BLA, which, through connections to the CeA, triggers defensive behavioural and physiological reactions that are part of the fear response, such as freezing (Kopchia et al., 1992). It is important to note that the microcircuits present within the CeA play a crucial role in the active process of fear extinction (Whittle et al., 2021). The hippocampal circuitry plays a role in fear, anxiety, and memory (Çalışkan & Stork, 2019; Engin et al., 2016; Sanders et al., 2003). Suppressing principal neurons in the dentate gyrus and cornu ammonis 3 (CA3) hippocampus regions reduces anxiety (Engin et al., 2016). Additionally, the direct projection from the ventral hippocampus to the medial prefrontal cortex is required for anxiety-related behaviours (Padilla-Coreano et al., 2016). It has been demonstrated that various regions of

both the cortex and subcortex play a role in contributing to this circuitry (Alvarez et al., 2008; Gilmartin et al., 2014; Gogolla, 2017; Myers-Schulz & Koenigs, 2012; Rozeske et al., 2015; Shin & Liberzon, 2010). Among these, the prefrontal cortex (PFC) is crucial in regulating fear behaviour and the extinction of established fear memories (Gilmartin et al., 2014). The PFC plays a role in forming emotional memories with complex context (Gilmartin et al., 2014). The PFC plays a role in forming emotional memories with complex context (Gilmartin et al., 2014). Various brain regions contribute to fear and anxiety behavioural responses, including the nucleus accumbens (NAc), the bed nucleus of stria terminalis (BNST), the periaqueductal gray (PAG), brainstem and thalamic nuclei, the superior colliculus, and the pulvinar (Kim et al., 2013; Penzo et al., 2015; Ray et al., 2020; Ren & Tao, 2020). These brain regions are involved in behaviours such as avoidance and increased orientation to and processing potentially adverse stimuli. As highlighted below these circuits are affected in Parkinson's, leading to a higher risk of anxiety compared to people without Parkinson's.

The metacognitive model of general psychopathology proposes that a person's thinking style is a generic causal factor (Wells, 1999, 2011). The process largely consists of perseveration through worry or rumination and arises from metacognition. Metacognition involves monitoring, regulating, and evaluating thinking processes (Flavell, 1979). Metacognitive processes involve knowledge, beliefs, mental regulation, and judgments. It influences and regulates the selection and execution of appraisals, attention, and memory (particularly recall) (Lebuda & Benedek, 2023).

The metacognitive model of generalised anxiety disorder (GAD) describes basic psychological factors and processes that are thought to underlie most types of anxiety disorder (Wells, 1995). The model is shown in Figure 1.3.



Figure 1.3: The metacognitive model of GAD with feedback labels added. Adapted from Bond and Dryden (2002).

A person with GAD worries in response to negative thoughts to anticipate problems, avoid them, or find solutions. This type of worry about oneself and the surrounding world is known as 'Type 1 worry'. Worrying can be an unconscious way of coping with stress, and it is associated with having positive beliefs about one's thinking process (positive metacognitive beliefs). Using the example, positive metacognitive beliefs could be:

> *"If I worry about harm in the future, I will be able to avoid it". "Thinking about bad things that could happen means I will be prepared". "Worrying helps me to cope".*

It is common for people to experience occasional episodes of worry. The metacognitive model suggests that relying too much on worrying or being inflexible in response to negative thoughts may harm emotional self-regulation. Therefore, excessive worrying can contribute to a non-specific vulnerability to emotional symptoms. Regular (Type 1) worrying is not sufficient to cause GAD.

Generalised anxiety disorder develops from negative beliefs about worrying that can form from exposure to information or interpretations of internal events. For example, an individual may re-evaluate their tendency to worry about things after learning from a magazine that anxiety and stress could be harmful to the body and should be controlled. Negative beliefs can cause one to view worrying as uncontrollable and dangerous, leading to a negative appraisal of the situation. These negative appraisals or interpretations are called 'meta-worry' or type-2 worry.

Type-2 worrying can intensify anxiety by making the perceived threat appear more immediate. It can cause the person to feel more anxious and stressed about the situation, making it harder to cope. Negative beliefs about worrying erode confidence in coping and safety. Examples of negative metacognitions could be:

> *"I have no control over my worry." "Worrying will cause me to have a mental breakdown." "Worry can cause a heart attack."*

Two factors that contribute to maintaining the problem are coping behaviours and thought control strategies. People living with GAD vacillate or have conflicting motivations to engage in worry or avoid/control the activity, preferring not to trigger worrying by trying to suppress initial thoughts concerned with worry-topics. This approach is seldom wholly
effective and adds to the tendency to perseverate on a particular topic (Wegner et al., 1987; York et al., 1987). The failure of this approach reinforces the idea that our thoughts are beyond our control, thus leading to negative beliefs about our ability to regulate them. When effective, suppression prevents the person from determining that it is possible not to worry in response to a negative thought. When a person with GAD experiences an uncontrollable thought, they do not interrupt or stop the worrying process. This creates a metacognitive paradox in mental regulation, as increased thinking is used to reduce thinking. This failure to disengage from excessive, prolonged thinking deprives the person experiencing GAD of the opportunity to correct their misconceptions about the uncontrollability of worry.

In addition to patterns of thought control, overt behaviours contribute to problem maintenance (Conner & Norman, 2017). Efforts to prevent worrying often involve seeking reassurance or avoiding triggers. The issue with seeking reassurance is that it surrenders control of worrying to others, preventing individuals from realising their internal control. In certain situations, people may adopt an 'information search' approach as a strategy, trying to find facts that will 'put their mind at rest.' However, they often encounter negative information or generate further questions, triggering more worry (Rachman, 2012). In each instance, the person fails to employ the most suitable or efficient metacognitive control strategy (Wells, 2010). This is because people lack knowledge of alternative strategies and hold positive beliefs about the need to engage in sustained worrying, which is inconsistent with the concept of 'letting thoughts go'. The model includes various feedback loops that function as maintenance mechanisms. These are labelled (a-d) in Figure 1.3. The way these loops function is as follows:

- a) Type 1 worrying can increase anxiety as it heightens threat awareness. However, anxiety can decrease as worry generates ways to cope. Changes in anxiety and other emotional states may indicate the success of worrying and the need for sustained or reduced thinking.
- b) Anxiety symptoms reinforce "worry about worry" (Type 2 worry) because the person negatively interprets symptoms of worry as a sign of mental or physical harm. In turn, Type 2 worry can increase the sense of current threat, leading to greater anxiety and sometimes even panic attacks.
- c) Behaviours such as avoiding triggering situations, using alcohol, or seeking reassurance prevent individuals from discovering they have control over their worry and that it is harmless. Negative beliefs that support type 2 worry continue to persist.
- d) Suppressing thoughts that trigger worry can backfire and worsen anxiety. Type 2 worries (negative appraisals) and beliefs concerning uncontrollability are both sustained. When strategies c and d limit worry, they prevent people from discovering that worrying is harmless, leading to the maintenance of negative beliefs about its possible danger.

In summary, the complexity of anxiety in neuroscience and psychology has been demonstrated. Negative metacognition has been shown to have a relationship with pathological worry.

1.2.3 Anxiety in Parkinson's

There are a range of NMS of Parkinson's. These can be classified as cognitive, neuropsychiatric, gastrointestinal, sleep-related and autonomic (Chaudhuri et al., 2005). Neurotransmitters of the dopaminergic, glutamatergic, cholinergic, serotonergic, and

adrenergic systems are involved in causing these symptoms (Lim et al., 2009; Willis et al., 2012). These systems are impaired by the abnormal accumulation of alpha-synuclein aggregates within neurons (αSyn, a protein that regulates synaptic vesicle trafficking and neurotransmitter release), which causes neuronal damage (Alafuzoff & Hartikainen, 2017). Anxiety is a NMS that affects around 40-56% of people with Parkinson's (Pontone & Mills, 2021). People with Parkinson's are more likely to experience anxiety than the general population and those living with other neurological conditions, such as epilepsy and multiple sclerosis (Broen et al., 2016; Hanna & Cronin-Golomb, 2012). This suggests that the neurobiology of Parkinson's is a key driver for increasing the risk of anxiety for this group. People with Parkinson's frequently experience anxiety associated with the time pressure of performing activities (Wressle et al., 2007). People with Parkinson's are more likely to fall due to fear of falling, which is influenced by anxiety, compared to age-matched controls (Dissanayaka et al., 2014; Farombi et al., 2016).

Parkinson's-associated anxiety is related to changes in the structure of the amygdala and dysfunction of various neuronal types, resulting in an imbalance between excitation and inhibition (Flores-Cuadrado et al., 2017; Floresco & Tse, 2007; Torres et al., 2021). Post-mortem brains of people with Parkinson's show reduced amygdala volume and neuronal cell loss, which is associated with anxiety (Harding et al., 2002). In a recent functional imaging study of 110 people with early-stage Parkinson's, a smaller volume of the left amygdala was found to be associated with symptoms of anxiety (Vriend et al., 2016). The results of these studies align with earlier findings in individuals with anxiety but not Parkinson's (Flores-Cuadrado et al., 2015; Hayano et al., 2009; Milham et al., 2005; van Mierlo et al., 2015). It is interesting to note that people with idiopathic rapid eye movement sleep behaviour disorder (iRBD), which often leads to the development of Parkinson's, have been found to

exhibit reduced gray matter volume in the left amygdala area extending to the hippocampus. This is particularly true for people with iRBD who also have anxiety and not for healthy individuals or those with iRBD without anxiety (Bourgouin et al., 2019). This study showed that structural changes in limbic brain regions, such as the amygdala and hippocampus, are correlated with anxiety prior to the onset of Parkinson's. Importantly, people with Parkinson's have been observed to exhibit α Syn pathology in the amygdala (Flores-Cuadrado et al., 2017). This may represent the structural correlates of anxiety and contribute significantly to the pathophysiological mechanisms that drive anxiety in Parkinson's.

As well as abnormal amygdala size, activity, and connectivity, other neurodegenerative changes associated with different neurotransmitter systems (e.g. dopamine and serotonin) may be linked to anxiety in Parkinson's (Criaud et al., 2021). Dopamine reduction in the caudate putamen is a hallmark of Parkinson's and accounts for the cardinal motor symptoms. However, dysregulation of the dopaminergic system is also linked to NMS of Parkinson's. The exact impact of dopamine on the development of anxiety in Parkinson's is not well comprehended. Striatal dopamine transporter (DAT) reduction is associated with anxiety in people with Parkinson's (Erro et al., 2012; Weintraub et al., 2005). In experimental models, dopaminergic projections from the ventral tegmental area (VTA) to the BA (part of the amygdala) modulate fear and anxiety, potentially implicating the VTA in Parkinson's (Brandão & Coimbra, 2019; de Oliveira et al., 2011; Tang et al., 2020). Dopamine also modulates the inhibitory mechanisms that the medial prefrontal cortex exerts on the anxiogenic output of the amygdala, a brain structure involved in anxiety (Pontone & Mills, 2021; Zarrindast & Khakpai, 2015). Dopamine deficiency thus leads to neuronal hyper-

excitability and amplified responses to perceived adverse threats (Hofmann et al., 2012; Kano et al., 2011; Kienast et al., 2008).

Disturbances in the serotonergic and cholinergic systems may also be involved in the NMS of Parkinson's (Ballanger et al., 2012; Maillet et al., 2016; Reisine et al., 1977). Serotonin (5-HT) is highly expressed in the amygdala and other limbic regions (Bocchio et al., 2016). People in the early stages of Parkinson's have been observed to have changes in their serotonergic system (Maillet et al., 2016). For example, significant alterations were observed in the putamen's cholinergic and serotonin receptors during post-mortem analysis (Reisine et al., 1977). Genetic variants in the promoter regions of the serotonin transporter (SERT) gene were identified, and people with Parkinson's who possess the short allele of the SERT promoter had a significantly higher score on anxiety measures (Menza et al., 1999). Further, people with Parkinson's who experience higher levels of anxiety have lower thalamic serotonin levels in positron emission tomography (PET) studies and people with Parkinson's may have lower levels of serotonin in their brains (Chen & Marsh, 2014). This suggests that serotonergic degeneration may contribute to anxiety symptoms in Parkinson's patients (Joling et al., 2018). Finally, alterations of gamma-aminobutyric acid (GABA) and norepinephrine (NE) systems have also been implicated in anxiety development in Parkinson's (Simuni & Fernandez, 2013).

Neuroimaging analyses in recent years have identified structural and functional changes to interlinked functional networks within the brain that are associated with anxiety related to Parkinson's (Carey et al., 2021; Carey et al., 2020). Functional magnetic resonance imaging (MRI) studies have identified neuronal connectivity changes in multiple brain regions, such as decreased limbic-dorsolateral prefrontal cortex connectivity (Dan et al., 2017). People with Parkinson's living with more impaired neuropsychiatric profiles show reduced functional connectivity in limbic cortico-striatal circuits and increased functional connectivity (the measure of how brain regions interact with one another) between the cerebellum and occipitotemporal regions (Tinaz et al., 2021). Magnetic resonance imaging studies of people in the early stages of Parkinson's also reveal a correlation between decreased structural covariance of left striatal seeds and contralateral caudate nucleus and increased severity of anxiety (Oosterwijk et al., 2018). In people with drug-naive Parkinson's, such abnormal intrinsic connectivity between large-scale networks may represent a neural correlate of anxiety symptoms even without cognitive impairment (De Micco et al., 2021).

In summary, structural, neurochemical and functional activation alterations in large-scale brain networks, including the interconnected amygdala, hippocampus, caudate nucleus, anterior cingulate cortex and prefrontal cortex, are believed to underlie anxiety development in Parkinson's (Betrouni et al., 2022; Criaud et al., 2021; Vriend et al., 2016; Wang et al., 2021; Zhang et al., 2023).

To conclude, anxiety in people with Parkinson's has traditionally been associated with the loss of quality of life and debilitating motor symptoms caused by the condition. However, it is now recognised as a NMS, especially in the early stages. This is due to the pathology in cortico-limbic systems and the corresponding neuro-circuitries and transmitter systems described above. In early disease pathogenesis, fear and anxiety may also be a pathological driver, causing α Syn upregulation in limbic regions or neuronal over-excitation. Further, living with Parkinson's, an incurable neurodegenerative condition, results in a combination of psychological stressors and physiological effects on the brain's neurotransmitter function and structure. As a result, individuals with Parkinson's have a higher risk of anxiety.

1.2.5 The cost of anxiety in Parkinson's

Nonmotor symptoms are common and impact quality of life but are typically less well recognised and addressed by health professionals (Chaudhuri & Schapira, 2009; LeWitt & Chaudhuri, 2020; Shulman et al., 2002; Todorova et al., 2014). A ranked list of research priorities by people with direct and personal experience of Parkinson's identified 'effective stress and anxiety management' as the second greatest research priority (Deane et al., 2014).

Anxiety substantially impacts a person with Parkinson's ability to fully engage in their desired daily activities (Duncan et al., 2014). The neurobiological and psychological stresses related to anxiety decrease energy and motivation for daily activity (Johansson et al., 2012). Poor functioning can persist after even partial recovery from anxiety due to the long-term impact of psychosocial impairment (Rodriguez et al., 2005). The impact of anxiety disorders on everyday life for people with Parkinson's is highlighted in their personal stories:

"Well I sort of, sweaty. And I feel, I feel I go starey. A bit like a rabbit in the headlights. Sometimes. And I just feel that I'm there but I'm not part of what's going on. I feel like I'm an outsider looking in, an observer of a situation rather than being part of it. . ." (Lovegrove & Bannigan, 2021) page 8

Anxiety in Parkinson's can have profound deleterious impacts on the person's lifestyle and identity:

"It ruins a lot of things because I can look forward to something and then get there and it's so awful, I have to leave [...] it limits what I can do." (Blundell et al., 2023) page 5

"You can't help it and then it leaves you feeling as a man, are these things going to be haunting you forever more. Then you can have performance anxiety because of the Parkinson's and not sleep right. Then you've got the business of how does your wife feel about that. You've got the problem of being able to walk with her and hold her hand or be able to lay your hand across her body or something without the hand shaking. And, it-it's really unpleasant and that causes anxiety, definitely. Definitely. You just wish you could, you just wish you could go to sleep and wake up and you're ok." (Lovegrove & Bannigan, 2021)

page 13

People with Parkinson's often describe anxiety as a constant and unpredictable presence that affects their daily lives, according to multiple research studies:

"It's there in the background all of the time, and I was always fearful of it reoccurring, it was a very painful experience having anxiety to the extent that I did. It was, it was just an awful." (Lovegrove et al., 2017) page 498

"And that's an anxiety which is, it is always there but I would say up to now it's background. Background noise. I can see depending on the way the condition goes I could be more and more anxious about me and those around me." (Lovegrove & Bannigan, 2021) page 10

"I just feel it will always be with me." (Blundell et al., 2023) page 4

In the UK, households affected by Parkinson's lose, on average, £16,582 annually due to higher health and social care costs and reduced income (Parkinson's UK, 2017). There is a paucity of research highlighting the economic cost of anxiety in Parkinson's specifically. Estimates suggest that the average cost of admission to an acute hospital for a person with Parkinson's with anxiety is over £5,000 with an average length of stay of 10 days (HSJ Parkinson's UK Excellence Network, no date). In comparison, a person without Parkinson's receiving the lowest level of care and investigation at accident and emergency (A&E) costs £86 (The King's Fund, 2023). The cost of a more complex A&E admission starts at £418 (The King's Fund, 2023). The average length of inpatient hospital stay in the UK is six days (OECD, no date).

Living with anxiety and Parkinson's exacts a profound toll, both personally and economically. The weight of anxiety that comes with Parkinson's can drastically reduce quality of life and impede joy. Whilst scant evidence exists, the economic cost of anxiety in Parkinson's cannot be underestimated. Increased health and social care system utilisation creates an imposing financial burden and places strain on people with Parkinson's and their families. Moreover, the indirect costs, such as diminished work productivity and missed opportunities, further underscore the societal impact of these conditions. Reflecting on the intersection of anxiety and Parkinson's, it becomes increasingly evident that a holistic approach is imperative and a new intervention is warranted.

1.3 Current interventions

Anxiety is a highly prevalent NMS in Parkinson's (Lintel et al., 2021). The recognition and treatment of anxiety in Parkinson's can be complex due to diagnostic criteria overlapping with other Parkinson's symptoms (Ho et al., 2021). The evidence for pharmacological treatments for anxiety in Parkinson's is limited (Pontone & Mills, 2021). In non-Parkinson populations, several non-pharmacological interventions may be used to manage anxiety successfully (Parker et al., 2021). These interventions include psychological approaches such as cognitive behavioural therapy (CBT) and mindfulness, as well as biofeedback approaches such as applied relaxation (AR) or breathing retraining (BR). The effectiveness of these interventions in Parkinson's is limited but rapidly emerging (Pontone & Mills, 2021). This section aims to provide an overview of some of these interventions and how they relate to anxiety in Parkinson's.

Based on reviews, it appears that there are currently no pharmacological interventions that are effective in reducing anxiety among people with Parkinson's (Pontone & Mills, 2021). There is no specific medication for treating anxiety in people with Parkinson's. Benzodiazepines, GABA agonists commonly used as anxiolytics, may cause sedation, cognitive impairment, and increase the risk of falls. Buspirone is a partial serotonin (5-HT) agonist that is used in anxiety disorders. However, Buspirone can worsen motor symptoms in people with Parkinson's (Prediger et al., 2012).

In populations without Parkinson's, psychological treatments for anxiety recommended by NICE guidelines and the NHS Talking Therapies programmes (previously known as Improving Access to Psychological Therapies, or IAPTs) are cost-effective and provide long-term financial and health benefits (Heuzenroeder et al., 2004; National Collaborating Centre for

Mental Health, 2018; NICE, 2011). IAPT services vary across the UK due to a shortage of clinical psychologists (Clark, 2018; The King's Fund, 2015). The NICE guidelines recommend mindfulness-based cognitive therapies and cognitive behavioural therapy (CBT) for anxiety as part of the IAPT program (National Collaborating Centre for Mental Health, 2018; NICE, 2011). The evidence base for using psychological therapy to help people with Parkinson's manage anxiety is limited. It has been suggested that theoretical models and interventions for anxiety should be expanded to address issues specific to people with Parkinson's (Egan et al., 2015). In 2015, Egan et al. developed an augmented Cognitive Behavioural model for anxiety and depression in Parkinson's disease. This augmented CBT model is part of similar models in clinical neuropsychology and older peoples' psychology (Broomfield et al., 2011; Gracey et al., 2016; Kishita & Laidlaw, 2017). The augmented model incorporates wellestablished predisposing and maintaining factors of anxiety and depression using CBT techniques. These factors include core beliefs (Beck, 2020), rumination (Ehring & Watkins, 2008), hypervigilance relating to physical symptoms (Clark, 1986), counterproductive behaviours (Salkovskis et al., 1999) and avoidance (Farchione et al., 2012). However, the approach is adapted to incorporate specific components of Parkinson's, including categories related to cognition, illness beliefs, and cohort beliefs about disability. The long-term sustainability of quality improvements following CBT has been questioned (Johnsen & Friborg, 2015). This is particularly relevant for people with Parkinson's as cognitive impairment is a common NMS and may mean that the long-term effects of CBT for this group are reduced (Biundo et al., 2017; Biundo et al., 2016). People with Parkinson's express a desire for interventions that address everyday problems rather than psychological therapy (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). Stakeholders also reiterated this as part of this project's patient and public involvement and engagement (PPIE) consultation.

Mindfulness is the psychological process of bringing one's attention to present experiences compassionately and non-judgmentally and has been incorporated into psychological therapies (Willekens et al., 2018). In a meta-analysis of mindfulness-based training for anxiety (not related to Parkinson's), an average effect size of 0.67 (95%CI [0.47, 0.87]) in favour of mindfulness was found across 39 studies (n=1,570). In their meta-analysis, Hofmann et al. (2010) identified mindfulness as a promising intervention for treating anxiety. More recent studies suggest that mindfulness techniques may have overall positive effects on Parkinson's with the the strongest results seen in anxiety and depression symptoms (van der Heide, Speckens, et al., 2021). Mindfulness interventions are also considered feasible and acceptable by people with Parkinson's (Bogosian et al., 2022). Potential mechanisms underlying mindfulness are unclear but may include reducing neurodegeneration and enhancing cerebral mechanisms (Goltz et al., 2024; van der Heide, Meinders, et al., 2021). These results, though not definitive, suggest mindfulness as a promising anxiety intervention.

Applied relaxation (AR) focuses on relaxing the muscles in a particular way during situations that usually cause anxiety (Ost & Breitholtz, 2000). Hayes-Skelton et al. (2013) identify AR as a preferred and effective intervention for anxiety in the general population and recommend directions for future research. Methodological flaws reduce the robustness and reliability of their conclusions. For example, there is no description of the search process; thus, no transparency or rationale regarding study selection is provided. Hayes-Skelton et al. (2013) clearly state their bias towards AR in the introduction and conclusion yet fail to address this inherent bias in the methodology. Due to the identified issues, the findings have limited credibility and cannot be transferred to people with Parkinson's. Likewise, a search of multiple databases yielded no results on AR's use and efficacy with people with Parkinson's.

Currently, there is insufficient evidence to support the use of AR as an effective treatment for anxiety in individuals with Parkinson's disease.

Breathing retraining (BR) focuses on gaining control over breathing to slow a person's respiration rate and reduce physiological anxiety (Tweeddale et al., 1994). Breathing retraining has been suggested as an option to manage and reduce anxiety in people with GAD and can be delivered by various professionals with training (Tolin et al., 2020). A search of multiple databases as part of this PhD project yielded numerous results for BR's use and efficacy with respiratory conditions such as chronic obstructive pulmonary disease (COPD) but no results for BR in people with Parkinson's. There is currently not enough evidence to support the effectiveness of BR as a treatment for anxiety in people with Parkinson's.

To summarise, psychological interventions recommended by the UK's NICE guidelines and NHS Talking Therapies (previously known as IAPTs) for non-Parkinsonian anxiety are costeffective and offer long-term benefits for health and finances (Heuzenroeder et al., 2004; National Collaborating Centre for Mental Health, 2018; NICE, 2011). There is emerging evidence on the effectiveness of mindfulness and cognitive behavioural therapy approaches in people with Parkinson's (Biundo et al., 2016; Hofmann et al., 2010; Johnsen & Friborg, 2015; McLean et al., 2017). This section has identified a lack of credible evidence for the use of AR or BR in managing anxiety in Parkinson's. Previous research has highlighted that people with Parkinson's living with anxiety place less value on group work (the typical format for many anxiety treatments), with more importance being placed on participation in meaningful roles and activities (Lovegrove et al., 2017). Additional approaches are also warranted, given the shortage of clinical psychologists in the UK (The King's Fund, 2015) and the central role of restricted participation in contributing to increased chronic anxiety

(Coventry et al., 2014; CSDH, 2008). Complex interventions, commonly used in health and social care services and other public health and economic policy areas where there are health consequences, could offer promising avenues for addressing these challenges.

1.4 Complex intervention development

Complex interventions are implemented and assessed across various individual and societal levels. An intervention could be classified as complex due to its various associated factors. These factors may include the number of components involved, the range of behaviours targeted, the expertise and skills needed by those delivering and receiving the intervention, the number of groups, settings, or levels targeted, or the level of flexibility of the intervention or its components. The Medical Research Council (MRC) have published a framework that outlines the process of developing, evaluating, and implementing complex interventions (Medical Research Council, 2008; Skivington et al., 2021). The framework comprises four main stages: development, feasibility/piloting, evaluation, and implementation. There are interactions between the phases that are not necessarily sequential, and each stage is not completely independent (Medical Research Council, 2008). Although the MRC framework is a well-cited and flexible approach to intervention development, other more prescriptive approaches are available. For example, there is a classification system for approaches to intervention development that identifies eight different categories of approach (O'Cathain et al., 2019). Alternative approaches to developing complex interventions are available, but the choice of approach should depend on the developer's context, needs, and values (Moore et al., 2019b; O'Cathain et al., 2019). Occupational therapy typically involves complexities not present in drug or medical trials (Nelson & Mathiowetz, 2004).

The aim of this thesis was considered with the guidance to ensure the application of the most appropriate research methods. This process involves identifying relevant existing evidence through a systematic review (Medical Research Council, 2008; Skivington et al., 2021). Figure 1.5 shows how the OBtAIN-PD research project relates to the MRC framework for developing complex interventions.



Figure 1.4: Developing an occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD) study relating to the Medical Research Council Framework. Adapted from (Skivington et al., 2021).

1.5 Occupational Therapy in Parkinson's

Occupational therapy was founded on the idea that humans require meaningful activity (occupation) to occupy their time (Yerxa, 1992). The power of occupation to transform the lives of people affected by illness and trauma is a core concept (Kielhofner, 2007). Adolf Meyer, a founder of the occupational therapy profession, stated in the 1920s that human health is promoted through occupation, i.e., work, play, rest, and sleep (Meyer, 1922). This idea that meaningful engagement in occupation is strongly related to human development, health, and wellbeing is one of the underpinning assumptions of occupational therapy (Jonsson, 2008; Meyer, 1922). The profession is rooted in medical, social-behavioural, psychological, psychosocial, and occupational sciences (World Federation of Occupational Therapists, 2019). This foundation prepares occupational therapists to work collaboratively with people in various settings, including communities, groups, and individuals (World Federation of Occupational Therapists, 2019). Occupational therapy aims to help people improve their ability to participate in daily activities by addressing personal, environmental, or combined factors (World Federation of Occupational Therapists, 2019).

Fisher and Marterella (2019) define occupation as the tasks and activities a person wants, needs, or is expected to complete. Regarding occupational therapy, three fundamental beliefs form the philosophical foundation of the profession and guide the therapeutic process. Firstly, occupational therapists view people as beings who engage in various occupations and may face challenges related to these activities. Secondly, they acknowledge occupation's substantial role in people's lives. Finally, occupational therapists consider incorporating occupation in all phases of the therapeutic process to help their clients achieve optimal outcomes (Fisher & Marterella, 2019). It serves as a starting point for what occupational therapists do and a driver of implementing occupational therapy in clinical practice (Fisher & Marterella, 2019).

The 'Do-Live-Well' framework (Moll et al., 2015) illustrates occupational therapy's focus on participation. 'Do-Live-Well' provides a structure for promoting wellbeing that reframes participation as 'doing' to reinforce the fundamental message that 'what you do every day matters' (Moll et al., 2015). It provides a health promotion tool centred on four interdependent concepts (Moll et al., 2015) (Figure 1.5), underpinned by theoretical and empirical evidence (Moll et al., 2015). Occupational therapists are experts in analysing and enabling people to optimise their occupational performance through engaging in meaningful occupations.





The role of occupational therapy with people with Parkinson's is to enable the person to engage in meaningful roles, tasks, and activities at home and in the community (Royal College of Occupational Therapists, 2018; Wood et al., 2022). This engagement in activities and roles is called *occupational performance* (Baum & Law, 1997). Occupational therapy for Parkinson's involves enhancing occupational performance, safety, and efficiency in daily activities (Royal College of Occupational Therapists, 2018). In the early stages, it aims to help people with Parkinson's prevent giving up activities, while in later stages, it focuses on enabling adapted participation. Occupational therapy also supports care partners in assisting in daily activities while considering their well-being (Prado et al., 2020; Sturkenboom et al., 2014; Wood et al., 2022). Occupational therapy is a mainstay of clinical guidelines for people with Parkinson's (National Collaborating Centre for Chronic, 2006; NICE, 2018, 2022). Interventions delivered by occupational therapists that provide a space for self-rediscovery, identity formation, and community participation are often valued over acute, assessment-based psychological interventions in non-Parkinson populations (Wimpenny et al., 2014).

Occupational therapists can collaborate with people with Parkinson's to evaluate and modify their activity planning and execution to minimise time constraints and enhance their self-efficacy (Creighton, 1992). The impact of these strategies on anxiety-related participation limitations has yet to be investigated. Occupational therapy interventions for people with Parkinson's are described in professional guidelines (Royal College of Occupational Therapists, 2018; Sturkenboom et al., 2011; Wood et al., 2022). Limited, moderate-quality evidence exists within the occupational therapy literature for addressing anxiety symptoms in the general population as well as Parkinson's, and the variety of interventions used and study methodologies means that the effectiveness of occupational therapy interventions for this population are not yet determined (Fox et al., 2019; Pisegna et al., 2022). Due to the specific needs of people with Parkinson's, these interventions must be modified for this population (Bogosian et al., 2020; Mendorf et al., 2023). Interventions provided by occupational therapists can be occupation-based (using occupation as the foundation) or occupation-focused (focusing a person's attention on occupation) (Fisher, 2013). The findings of previous research and the views of the PPIE stakeholders of this project reinforce the need for an occupation-based intervention for anxiety in Parkinson's (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). Occupational therapists, with their expertise in improving wellbeing through participation and training in physical and mental

health, are well-positioned to support individuals with Parkinson's whose participation is impacted by anxiety (Christine, 1990).

1.6 Patient and Public Involvement and Engagement

Patient and public involvement and engagement (PPIE) in research positively benefits recruitment, design, dissemination, and implementation of findings (Boote et al., 2011; Renedo et al., 2015). This research project was completed as part of an NIHR Clinical Doctoral Research Fellowship (NIHR301565). A PPIE consultation was conducted with people with Parkinson's (n=15) recruited through the charity Parkinson's UK. This took the form of PPIE stakeholders reviewing the draft proposal with a follow-up one-to-one telephone discussion. The NIHR Involve Briefing Notes for Researchers: public involvement in NHS, public health and social care research (National Institute for Health Research, 2012) was used as a framework for the consultation process. Two focus groups were completed at the requests of the PPIE stakeholders as they had discussed the study proposal between themselves and thought a group discussion would be valuable. These groups were conducted as video calls on Zoom ("CYPH, INC. v. ZOOM VIDEO COMMUNICATIONS, INC," 2022). The point of the calls was to clarify and elaborate on any points or ideas within the proposal that the PPIE stakeholders had. The research topic was strongly endorsed as meaningful and relevant by the PPIE stakeholders, who commented:

"Thank you for taking on research into anxiety and Parkinson's. It is a symptom that has dogged me for years, with no obvious interventions available."

"Yes, this is needed."

"Anxiety research is absolutely valuable."

"As a person living with Parkinson's, this research is important to me"

Furthermore, PPIE stakeholders stated that they felt the order of the study process was sensible, and it would be a study of interest to people with Parkinson's. Other stakeholder recommendations, such as making the language more accessible and being more transparent about the potential benefits to people with Parkinson's, have been integrated throughout the OBtAIN-PD project. The involvement of care partners in the group concept mapping study (chapter 3) was included. A separate small group of PPIE stakeholders composed of people with Parkinson's (n=5) reviewed the plain English summaries for each study and made recommendations to improve their readability.

Additionally, a PPIE consultation was conducted with occupational therapists (n=3) from a range of clinical backgrounds (including neurology, mental health, and inpatient services) who all have experience working with people with Parkinson's and/or anxiety. This consultation aimed to check that the proposed project's approach, aims, and objectives were meaningful and realistic for occupational therapists. All occupational therapists agreed that the project was meaningful and realistic.

The OBtAIN-PD acronym was co-produced and chosen by people with Parkinson's (n=5) and care partners (n=3). This consultation was conducted as an online focus group using video conferencing ("CYPH, INC. v. ZOOM VIDEO COMMUNICATIONS, INC," 2022). A list of computer-generated acronyms was presented to the group, who were also given the choice to bring their own ideas to the session. Following the discussion, the group chose the

acronym OBtAIN-PD because they felt that the word 'obtain' had positive and empowering connotations. They believed it conveyed the idea of acquiring control over their anxiety and, consequently, their lives, through proactive effort and support.

The OBtAIN-PD logo (figure 1.6) was co-produced with the project's Patient Advisory Group (PAG), which consisted of people with Parkinson's (n=4) and care partners (n=2) in collaboration with a graphic designer at the University of Plymouth Design and Print Centre. As well as being easy to read, the PAG strongly felt that the logo should symbolise support and that OBtAIN-PD is inclusive of everyone. This is evoked by the 'O' which symbolises a group of people in a circle holding hands or with their arms around one another with a rainbow colourway that is symbolic of the LGBTQIA+ pride flag.

Btain-PD

Figure 1.6: The final OBtAIN-PD logo.

Each study involved stakeholders, including people with Parkinson's, care partners, and occupational therapists. The respective chapters detail their involvement.

1.7 Methodological orientation

Informed by the PPIE consultations described in section 1.6 and in line with the Medical Research Council guidance on developing and evaluating complex interventions (Skivington

et al., 2021), it was evident that the use of one research method would not be effective in developing the content of the OBtAIN-PD and determining the feasibility of conducting a trial. Qualitative methods to achieve consensus would enable the development of the intervention and enable a deeper exploration of participants' experiences of participating in a trial. Consequently, a mixed-methods design comprising four phases, including a scoping review, group concept mapping, logic modelling, and a feasibility RCT with embedded qualitative study, would provide a more holistic view when integrating and interpreting findings. This would enable comprehensive conclusions to be drawn when evaluating complex healthcare interventions (Craig et al., 2008).

Plano Clark and Ivankova (2016) describe mixed-methods research as:

"...the intentional integration of qualitative and quantitative research methods to best address a problem". (p. 4)

Since the 1980s, the combination of research methods has become increasingly accepted, although there are differing views on integrating quantitative and qualitative methods (Greene, 2008; Johnson et al., 2007). Greenhalgh and Papoutsi (2018) argue that many evaluation studies that test complex interventions fail to consider the intervention's complexity fully. They suggest that complex healthcare systems have unclear boundaries, where clinicians operate under internal rules and adapt and improve those systems iteratively. Randomised controlled trials alone cannot fully address questions as much reallife healthcare is uncontrollable (Subbiah, 2023). Therefore, more in-depth mixed-methods approaches are required to explore the dynamic nature of testing an intervention's feasibility and implementation. The Medical Research Council Framework for the

development and testing of complex interventions (Skivington et al., 2021) emphasised that there is a need for more studies that use mixed methods and process-based approaches to explore the challenges of non-linearity and iterative local adaptations (Moore et al., 2015). The OBtAIN-PD project utilised mixed-methods research principles throughout the research process, adopting an integrated approach to conceptualising and interpreting the findings.

When it comes to research, using both quantitative and qualitative methods can be complicated because the underlying philosophies of these two approaches seem to contradict each other. Quantitative research is traditionally based on positivism, which asserts that scientific inquiry is the only way to establish reliable knowledge. This experimental method was first introduced by August Comte (1798-1857), who is considered the first philosopher of modern science (Crotty, 1998). In the early 19th century, philosophers like Karl Popper introduced post-positivism philosophy. It valued objectivity and recognised that researchers' values and backgrounds can influence their research. This influence can create a potential source of bias that should be acknowledged and managed (Popper, 1959). The post-positivist movement in the 1980s enabled the development of qualitative methodologies and mixed research methods (Giddings & Grant, 2007). Constructivist qualitative research aims to interpret meaning and context from multiple realities, including the researcher's experience (Lincoln & Guba, 1985).

According to some authors, there is no necessary connection between specific philosophies and research methods (Cook, 1979; Greene et al., 1989). Greenhalgh suggested a departure from traditional paradigms and challenged "prevailing assumptions and methodological rules" to advance the knowledge base (Greenhalgh, 2013, p. 92). Integrating research methods in mixed-methods research and their underlying conventions can be challenging for researchers. The researcher may attempt to combine differing paradigms within one study. Additionally, the investigator will operate within their personal and professional contexts, adding another complexity layer. However, this reflexivity is considered an essential and valued contribution to qualitative research ontology (Giddings & Grant, 2007).

I am an occupational therapist who uses quantitative and qualitative research data to inform clinical practice. Quantitative data may provide a higher level of objectivity of results. On the other hand, qualitative data can offer an effective way of exploring attitudes, feelings, and behaviours in a more detailed and profound manner.

The OBtAIN-PD project was designed using a pragmatic approach to consider the practicality of a wide range of designs and their applicability to the complex healthcare evaluation field with awareness of the underlying ideologies. The design of the OBtAIN-PD project aimed to avoid conflicts between different methods. This approach avoids the domination of one method over others. If there is any difference in findings across methods, it provides an opportunity to refocus the research questions and original research assumptions. Lather (2006) suggests that disagreements between paradigms should not lead to conflict or require reconciliation; instead, there should be "disjunctive affirmation" (Lather, 2006, p. 52). This means that researchers accept that there are multiple valid ways of approaching research and that one methodology must be 'false' as the other is 'true'.

1.8 Thesis Aim and Structure

This thesis aims to co-produce an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD). This will be achieved using a complex intervention development process consisting of a scoping review, group concept mapping,

and logic modelling. As a researcher and practising occupational therapist, I intend to ensure that this research is clinically relevant and meaningful to people with Parkinson's and their care partners. This thesis is derived from and informed by people with Parkinson's, care partners, occupational therapists, academics, methodologists, and PPIE.

This chapter presents an introduction and background to the substantial burden of anxiety in Parkinson's. An overview of interventions used with people with Parkinson's living with anxiety is presented. An overview of Occupational Therapy is given, and its suitability in addressing the challenges presented by anxiety in Parkinson's is established. The process of developing a complex intervention is discussed. The PPIE has been discussed, and the research philosophy underpinning the project has been outlined. Chapter two presents a scoping review protocol and the resulting review that identifies current occupational therapy interventions for community-dwelling adults with anxiety. Chapter three describes a study where people with Parkinson's, care partners, and occupational therapists identify critical components for an occupation- and community-based intervention for people with Parkinson's with anxiety. Chapter four details a logic modelling study in which people with Parkinson's, care partners, and occupational therapists worked together to develop OBtAIN-PD. This resultant intervention aims to help individuals with Parkinson's manage their anxiety and improve their quality of life. The specification of the OBtAIN-PD is given in chapter five. Chapter six presents a protocol for a feasibility cluster randomised controlled trial of the OBtAIN-PD. Chapter seven discusses the research project, while chapter eight provides recommendations and conclusions for future studies. An overview of the PhD timeline, indicating where research activities were conducted in parallel and how they flow together, is shown in Figure 1.7.

			2	2020)								20)21											20	22											20	23						2	2024	
	June	July	Aug	Sept	Oct	Nov	Dec	January	Feb	March	April	Mav	June	July	Aug	Sept	Oct	Nov	Dec	January	Feb	March	April	May	June	July	Aug	Sept	Oct	Nov	Dec	January	Feb	March	April	May	June	July	Aug	Sept	Oct	Nov	Dec	January	Feb	March
АРР																																														
PPIE																																														
SR (ch.2)																																														
GCM (ch.3)																																														
LM (ch.4)																																														
RCT (ch.6)																																														
Thesis																																														

Кеу

APP- Application process to NIHR Clinical Doctoral Research Fellowship

PPIE- Patient and Public Involvement and Engagement

GCM- Group concept mapping study (chapter 2)

LM- Logic modelling study (chapter 4)

RCT- Randomised controlled trial protocol development and ethics application (chapter 6)

Figure 1.7: Overview of the PhD research activities.

Chapter 2: Scoping Review

In the previous chapter, the need for a new intervention to help people with Parkinson's live well with anxiety was outlined. The first step of the MRC's framework when developing and evaluating a complex intervention involves identifying the evidence base (Figure 1.5).

Identifying features of antecedent interventions is part of the development phase of the MRC framework (Skivington et al., 2021). This allows those features or characteristics to be adapted for a new intervention, such as the OBtAIN-PD (Skivington et al., 2021). This chapter presents a protocol followed by the subsequent scoping review (Lovegrove et al., 2023; Lovegrove, Marsden, et al., in press). Due to the paucity of evidence in the area of anxiety interventions used by occupational therapists to help people with Parkinson's with anxiety, it was decided to widen the search to the population of all adults with anxiety. Since there is still a lack of evidence in this area, it was considered valuable to identify specific characteristics and concepts in papers or studies and to map, report, and discuss these characteristics and concepts as potential components for the OBtAIN. In these cases, a scoping review is a better choice compared to a systematic review (Munn et al., 2018). Therefore, a scoping review was selected as the methodology for this study.

The scoping review findings informed the co-production of the OBtAIN-PD in the logic modelling study (Chapter 4). The scoping review data chart of the included studies is presented in Table 2.1. Articles that relate to the same study are colour-coded for clarity. This scoping review revealed that occupational interventions for anxiety often use outcome measures that are not specific to occupational therapy. The review indicates that future interventions and research should use occupational therapy outcome measures and appropriate impairment-based measures. This will demonstrate the success and

effectiveness of occupational therapy services in improving the occupational performance and participation of individuals with anxiety. Most interventions were group-based, but individual occupational therapy sessions could offer unique benefits. Occupational therapy interventions for adults with anxiety should use goal setting and tailoring to enhance effectiveness and impact. Rigorously designed clinical trials are needed for occupational therapy interventions in adults with anxiety, including economic evaluations to demonstrate cost-effectiveness.

Reference	Brief name	Why	What	Procedures	Who provided	How	Where	When, how much	Tailoring	Modifications	Planned	Actual
Chugh- Gupta <i>et al.,</i> 2013	Yoga	Facilitate slower and deeper respiration, increase O2 consumption, reduce BP and HR, increase muscle tone and blood flow, positively influence GABA production, enhance strength and flexibility	Described in figure 3	Hatha yoga- physical postures, breathing, meditation Lyengar yoga- Alignment, equipment to achieve poses Tibetan yoga- specific routines Ashtanga yoga- physical postures	N=9 RCTs taught by expert or "trained" yoga instructors	N=14 RCTs, group sessions N=2 RCTs, individual sessions All non-RCTs included group sessions	N=1 non-RCT had home practice sessions. No others described location	No. of sessions: 1- 60 (range) No. of weeks: 1-12 (range)	Not discussed	Not discussed	Not discussed	Not discussed
Eklund, 2013	Redesigning Daily Occupations (ReDO)	Reduce burdens and increase motivation and energy through balance of everyday occupations	No details provided	 Three phases: Five weeks, focus on self-analysis of everyday activity patterns. Five weeks, goal setting and strategies to accomplish desired changes. Six weeks, period of work placement at women's ordinary workplaces. Included three group support/mentoring sessions. 	Each group led by two occupational therapists with specific training for leading the ReDO program	Group sessions and a work placement	Primary health care centres	6 weeks in three phases. Phases 1 and 2= two sessions per week, each lasting 2.5 hours Phase 3= three group sessions but unclear if they would be for the same length of time	Phase 3 delivered in workplace	Not discussed	Not discussed	Not discussed
Eklund and Erlandsson, 2013	Redesigning Daily Occupations- program (ReDO)	Achieve better balance between everyday occupations leading to better wellbeing and increased ability to return to work	Not discussed	 Phase 1 focuses on occupational self-analysis (five weeks). Phase 2 focuses on goal setting and strategies for accomplishing change in patterns of everyday occupations (five weeks). 	Two licensed occupational therapists who were specifically trained to lead the ReDO programme	Group sessions and a work placement	Sweden, two towns in selected district	Over 16 weeks, three phases. First five weeks phase 1, second five weeks phase 2, then 6 week work	Not discussed	Not discussed	Not discussed	Not discussed

				Phase 3; 6-week work placement, meet three times as a group				placement in phase 3 Two 2.5 hour sessions per week except phase 3 when group meets three times				
Eklund et al., 2013	Redesigning Daily Occupations- program (ReDO)	Based on occupational therapy and occupational science principles. Focus on patterns of daily occupations and balance between them. Avoid detrimental occupations and engage in positive ones to increase motivation and energy.	Not discussed	Considered whether the woman would have the time and energy to engage in the ReDO programme. Reflecting on own occupational history. Identifying interests. Stresses and rewards Mapping how time is used. Becoming aware of value and meaning found in everyday occupations. Strengthening the women's own abilities to analyse their everyday occupations and find strategies to obtain balance is also emphasised.	Not discussed	Three phases. Phases one and two; group work. Phases three; work practice and groups.	Primary healthcare centres	16 weeks, 10 groups total. Phases one and two 5 weeks each. During this phases, group meets twice a week. Phase 3; 6 week work practice. Group meets three times to offer support.	Not discussed	Not discussed	Not discussed	Not discussed
Erlandsson, 2013	Redesigning Daily	Based on people's experiences,	Tools and exercises for	1.1 Individual meeting with interview	Two specially trained	Occupational therapy	Southwest of Sweden	6-8 participants	Not discussed	Not discussed	Not discussed	Not discussed
	Occupations-	ability to return to	self-analysis,	1.2 Program intro	occupational	group		per group,				
	program (ReDO)	on ability to re-	personal	1.2 Program Intro	therapists	treatment		over 16				
		adapt routines,	reflection,	1.3 What is an occupation?				weeks.				
		change in patterns	24-nour circle, fruit,	perspectives				Phases 1 & 2				
		of daily	music,					are five				
		occupations,	scrapbooks,	1.4 Occupational balance				weeks each.				

	domonstrato	farowoll						
	increased words	laieweii	4 5	Detterne of deiler		Dhasa 2 (iah		
	Increased work	party	1.5	Patterns of daily		Phase 3 (Job		
	capacity, improve			occupation 1		placement)		
	health		1.6	patterns of daily		is six weeks.		
				occupation 2				
				-		Two and half		
			17	Stress bassles and		hours per		
			1.7	unlifts in doily life		group		
				upints in daily life		group.		
			1.8	hassles and uplifts in				
				day-to-day life				
			19	Goal setting 1				
			2.0	eeu eeu g				
			1 10	Cool cotting 2				
			1.10	Goal Setting 2				
				-				
			2.1	Sleep as a prerequisite				
				for occupation and				
				health				
			2.2	Work-occupational				
				history				
				This con y				
			2.3	Evening seminar for				
				selected relatives,				
				colleagues, and/or				
				employer				
				. ,				
			21					
			2.4					
			2.5					
			2.5	Preparations for Job				
				placement				
			2.6	Hassles at work				
			2.7	2.7 Occupational				
				balance including work				
				T				
			_					
			2.8	occupational balance				
				including work 2				
			2.9	conclusion of phases 1				
				and 2				
			2 10	Ending phases 1 and 2				
			2.10	Enuming phases 1 and 2				

				3. Work placement								
Fox et al., 2019	Occupational therapy; led intervention, using cognitive- behavioural therapy, using health behaviour change theory, using skill-building theory, using occupation science theory, using neurological and sensory theories	Improving functioning and mental health outcomes. Improve anxiety symptoms. Target difficulties in completing ADLs.	Neurological and sensory intervention s; cranial electro therapy stimulation device administered by the participant themselves	Oversights only in keeping with article type	Intervention defined as occupational therapy, or led/designed/fa cilitated by an occupational therapist All interventions occupational therapy-led	Group, individual, and combination	Not discussed	Sessions range 3-18 Hours per session range 2-4 Weeks range 4-16	Not discussed	Not discussed	Not discussed	Not discussed
Fox <i>et al.,</i> 2021			Manual with session guidance, slides, worksheets, and additional reading. Worksheets given to participants.	Individual meeting with occupational therapist then group sessions: 1. Intro 2. Occupational history 3. Occupational balance 4. Patterns of daily occupations and time (1) 5. Patterns of daily occupations and time (2) 6. Hassles and uplifts in daily life 7. Goal setting 8. Occupational value 9. Evening seminar	Facilitators completed 3- day training course and certification. Facilitators were occupational therapists.	Pre-group individual sessions, then groups.	City-based primary care centre.	Group sessions over 10 weeks.	Some adaptatio ns made by treating therapists but the interventi on itself was not tailored, described as having a rigid manual.	Changing content order (fidelity maintained). Adapting occupation- based sessions. Homework done within session. Cut content reintroduced into ReDO-10.	Therapists complete d a written reflection after each session on how closely they had adhered to the manual, any adjustme nts made and notes on their facilitatio n	Unclear if fidelity fully maintaine d.

Gunnarsson et al., 2006	The Tree Theme Method (TTM)	Creative activities like painting a tree may be used to stimulate occupational storytelling and making. Life story narrative to understand how one acts and actions. Promote self- image development and how person relates to the environment.	Sheets of 40x48cm paper, watercolours , and oil crayons.	 10. Goals and strategies 11. Concluding 12. Follow-up 1 (after 1 month) Occupational therapist introduces theme and leads relaxation exercise. Questions to focus reflection. Painting intervention sessions: Present life situation Childhood Teenage years Future tree Future tree Follow-up interview three years later Follow-up TTM Follow-up TTM 	Occupational therapist.	Individual sessions with occupational therapist.	Mental health care in different clinics in Sweden. Psychiatric outpatient clinic.	Five sessions, then a follow-up stage three years later (one interview and two TTM sessions) Unclear what time frame the first five sessions were delivered over	Not discussed	Not discussed	Not discussed	Not discussed
Gunnarsson et al., 2015	The Tree Theme Method (TTM)	Based on activities to achieve a satisfying everyday life. Intervention that can be used to compliment CBT and pharmacological treatment. Based on art therapy and life storytelling. Increase ability to cope with everyday life by	Not discussed.	Paint pictures of trees that represent certain periods of life in five sessions. First four start with progressive relaxation and person paints a symbolic tree. Picture used as a starting point to discuss different points of life with focus on everyday ADLs. Theme of tree differs from one session to the next and each tree represents a particular period of life: the present, childhood, adolescence, adulthood.	Occupational therapists trained in TTM. Also training in study procedures.	Individual sessions.	Primary health care centres and general outpatient mental health care units in three counties in the south of Sweden	Five sessions of 60 mins each during 6-9 week period.	Not discussed	Not discussed	Fidelity forms complete d after each session by treating occupatio nal therapists in both groups	Not discussed

Gunnarsson et al., 2018	The Tree Theme Method (TTM)	developing strategies. Based on art therapy and life storytelling with a focus on activities in everyday life. Telling life story allows opportunity to reflect on how to act and explain why they act in a specific situation and context.	Not discussed.	Fifth session focus is on story- making and shaping future plans. Patient and therapist decide on tasks for completion between sessions. Sessions start with progressive relaxation. Reflective dialogue between person and therapist followed by painting tree representing different life periods: 1. Present life situation 2. Childhood 3. Adulthood 4. Not stated 5. Future tree Final session includes dialogue of how to incorporate change. New tree painted each session.	Occupational therapists specially trained in the specific frames and techniques of TTM.	Not made clear if individual or group	Primary health care centres and general outpatient mental healthcare units in three counties in Sweden	Five sessions. No indication of session length or frequency.	Not discussed	Not discussed	After each complete d interventi on the therapists reported the content in each session to assess adherenc e to the TTM procedure s	Not discussed
Kitchiner et al., 2009	Cognitive behavioural therapy adult education evening class (Stress Control), anxiety managemen t group	As some individuals uncomfortable in a group setting, so a generic management strategy using stress control principles that could be delivered to up to 60 people at a time delivered	PowerPoint slide presentation s, therapist /patient manual. Written materials for participants to take home	Stress control (SC)- anxiety education, self- assess techniques, techniques to control responses, discussion of personal problems. Anxiety management (AM)- more interactive with broad psycho-educational approach. CBT model used with emphasis on activity scheduling and planning. Training in a variety of applied relaxation techniques. Groups used to engender a self-help ethos in a safe environment.	SC courses delivered by two experienced mental health nurses with experience in using CBT. AM group facilitated by occupational therapists with 15-20 years of delivering anxiety	Group sessions presented as a taught course	University Hospital of Wales. Groups held in the late afternoon in a formal classroom setting.	Six two hour sessions held at weekly intervals	Not discussed	Not discussed	Random selection of three video or audio tapes for each condition assessed for fidelity by experts.	Judged to adhere closely to manuals.

					management groups.							
					Trained to use the manual.							
Lambert <i>et</i> <i>al.,</i> 2007	Occupational therapy-led lifestyle approach.	Potential role of lifestyle in the prevention of mental health problems recognised. Modifying habitual lifestyle behaviours in a health protective direction will reduce the frequency and severity of anxiety and panic symptoms.	Lifestyle diaries.	Lifestyle review using self- report and lifestyle diaries. Education to increase awareness of the potential negative benefits of some lifestyle behaviours. Specific lifestyle changes negotiated. Monitoring and review of the agreed lifestyle changes and any changes in symptoms.	Occupational therapists.	Individual appointment s.	Two primary care trusts in the east of England.	10 intervention sessions over a 16-week period. 3x1h at weekly intervals, 3x0.5h at weekly intervals, 3x0.5h at fortnightly intervals, 1x1h at monthly intervals. Delivered in four stages.	Not discussed	Not discussed	Not discussed	Not discussed
Lambert <i>et</i> <i>al.,</i> 2008	16-week occupational therapy-led lifestyle intervention.	Habitual lifestyle behaviours can affect health both negatively and positively. Modifiable lifestyle behaviours contribute to anxiety and depression, therefore influencing these may have beneficial effects.	Not discussed	 Lifestyle review. Education to increase awareness of behaviour health effects. Specific lifestyle goals negotiated between therapist and person. Monitoring and review of the agreed lifestyle changes and any consequent change in symptoms. 	Occupational therapists.	Individual lifestyle intervention sessions.	GP practice, primary care.	10 lifestyle intervention sessions over a 16-week period. Three 60 minute appointment s at weekly intervals, three 30 min appointment s at weekly intervals, three 30 minute appointment s at	Not discussed	Not discussed	Not discussed	Not discussed

Lambert <i>et</i> <i>al.,</i> 2010	Occupational therapy-led lifestyle	Lifestyle approach may be at least as effective as usual	Not discussed.	Lifestyle review using self- report diaries.	Trained research occupational	Not clear if individual or group	Primary care.	fortnightly intervals, and a final 60 minute appointment after a month. 10 intervention sessions over	Not discussed	Not discussed	Not discussed	Not discussed
	approacn.	care when treating panic disorder.		Education to increase patient awareness of negative health effects of some behaviours. Specific lifestyle changes negotiated. Monitoring and review of agreed changes.	therapist.	sessions.		a 16-week period.				
Lork <i>et al.,</i> 2019	Work- directed rehabilitatio n provided by OTs and PTs to promote work capacity in CMDs	Efforts to increase work capacity need to be directed towards specific work context. Person- environment- occupation (PEO) model assumes occupational performance is an interaction between these factors.	Not discussed	First session- work-related resources and problems identified with occupational therapist to agree on more specific goals and tentative content to develop strategies. Occupational therapy module- vocational behavioural strategies inspired by ReDO aiming at increasing balance in everyday life by changing activity patterns.	Occupational therapist (n=1) and/or a PT (n=2) specialised in treating patients with mental health problems, 5-20 years' experience in this field. Two seminars prior to study with PI to discuss and learn more about the approach	Not discussed.	Swedish primary care, rehabilitatio n centre.	Intervention period of 8 weeks with 4-16 visits.	Tailored with participan t and adapted top their needs. Patients chose OT module, PT module, or both.	Not discussed	Not discussed	Not discussed
Prior, 1998a.	Short-term anxiety managemen t course.	Cognitive behavioural group using stress management and assertiveness techniques. Detailed	Handouts.		Not discussed	Group sessions	Mental health day hospital.	Six sessions over six weeks.	Not discussed	Not discussed	Not discussed	Not discussed
Prior, 1998b.	Short-term anxiety	education on anxiety, stress reduction techniques, progressive relaxation, and problem solving. Not discussed	Not dicsussed	Not discussed	Not discussed	Not discussed	Mental health day	Not discussed	Not discussed	Not discussed	Not discussed	Not discussed
-------------------------------	----------------------------------	---	--------------------	---	--	--	-------------------------------------	---	--	---------------	------------------	------------------
	managemen t course.						hospital in West London					
Rosier <i>et al.,</i> 1998	Anxiety managemen t groups	Anxiety management group developed by occupational therapists based on cognitive behavioural approach. Objectives were to recognise personal symptoms, identify triggers, identify triggers, identify lifestyle factors, understand management techniques, share personal experience, facilitate movement to other appropriate treatments following extended assessment.	Homework forms.	Week 1- introduction, identification of somatic symptoms and coping strategies, relaxation techniques. Homework; anxiety symptoms & frequency checklist. Week 2- Review homework, anxiety spiral, nervous system education, lifestyle factors and assessment, goal formulation, relaxation techniques. Homework; how I use my time. Week 3- review homework and goals, negative thinking patterns. Guided imagery relaxation techniques. Homework; recording negative thoughts. Week 4- review homework and goals, challenging 'thinking errors'. Guided imagery relaxation techniques. Homework; negative thought patterns. Week 5- review homework and goals, panic attacks and hyperventilation. Musical	Facilitators who receive weekly supervision.	Group sessions, in person. Maximum 10 people per group and two facilitators. Facilitators have weekly supervision with a psychodyna mic therapist.	Community mental health team.	Seven weekly 1.5 hours sessions.	Week 6- people can choose own relaxation technique	Not discussed	Not discussed	Not discussed

relaxation techniques. Homework; advise on case
study.
Week 6- review case study
relaxation techniques of
choice.
Week 7- review goals,
complete evaluation form,
Followed-up 5-6 weeks later,
invited to group reunion.

Table 2.1: Scoping review data chart. Articles relating to the same study are matching colour-code

The scoping review protocol presented in this chapter is the Word version accepted for publication in *JMIR Research Protocols* (Lovegrove et al., 2023). The scoping review is the accepted Word version submitted for publication in *JMIR Aging* (Lovegrove, Marsden, et al., in press). In both articles, the authors are referred to intext with initials when delineating their role in the review. The authors are Christopher Lovegrove (CL), Ingrid Sturkenboom (IS), Mary Smith (MS, information specialist, Exeter Health Library), Jonathan Marsden (JM) and Katrina Bannigan (KB). The full list of references for both articles paper is included in the main thesis bibliography to improve readability.

2.1 Characteristics of Occupational Therapy Interventions for Community-Dwelling Adults With Anxiety: Protocol for a Scoping Review

Background: Anxiety is linked to decreased life quality and well-being. Living with an anxiety disorder results in higher personal and societal financial expenditure. Occupational therapists work with people living with anxiety in a variety of settings. A preliminary database search was conducted, and no current or underway systematic or scoping reviews on the topic were identified. Developing an overview of studies of occupational therapy interventions for people with anxiety is a necessary first step to understanding the existing knowledge and to increase the impact of future interventions. This scoping review will build on the findings of a previously conducted systematic review.

Objective: This scoping review will identify the following: (1) what occupational therapy interventions exist for adults with anxiety and (2) the intervention characteristics including outcomes used and impact observed.

Methods: This protocol was reviewed by an occupational therapist as part of a patient and public involvement consultation. The review will consider all studies and publications of occupational therapy that include community-dwelling adults with a diagnosis of anxiety who are aged 18 years and older and diagnosed with an anxiety disorder (Diagnostic and Statistical Manual of Mental Disorders [DSM-5-TR] criteria). Databases to be searched are MEDLINE, CINAHL, Cochrane Library, Science Direct, PsycArticles, Psychology & Behavioural Sciences Collection, Embase, PubMed, TRIP, Proquest, Social Care Online, JBI EBP database, OpenGrey, and OALster. Titles and abstracts will be screened against the inclusion criteria using Rayyan Qatar Computing Research Institute. Potentially relevant studies will be retrieved in full and assessed against the inclusion criteria. Articles published in English will be included. No date limiters will be used. Study selection will be completed by 2 independent reviewers. Data will be extracted using a data extraction tool. **Results:** Data will be charted using the Template for Intervention Description and

Replication (TIDieR) checklist in alignment with the review objectives. The scoping review will be reported in accordance with the Preferred Reporting Items for Systematic review and Meta-Analysis Protocols statement.

Conclusions: This scoping review will produce valuable information about community-based interventions used to improve participation, life quality, and well-being for adults with anxiety to support the development of occupational therapy interventions. The findings will be disseminated through professional and National Health Service bodies, employer organizations, conferences, and research articles. The findings will be of value to health care professionals and researchers working to improve the lives of people living with anxiety. **Trial registration:** Open Science Framework DOI 10.17605/OSF.IO/JS549;

https://osf.io/js549/.

International registered report identifier (irrid): DERR1-10.2196/41230.

Keywords: activities of daily living; adult; anxiety; community; development; health; intervention; life quality; occupational therapy; quality of life; research; support; therapy; well-being.

Background

Anxiety and stress-related disorders, conditions that involve excessive fear or worry about a current or anticipated future event, are the ninth-highest global cause of disability (Vos et al., 2015). Anxiety disorders are highly persistent over a person's lifetime and are associated with elevated levels of disability, reduced quality of life, and decreased participation (Kessler et al., 2007). Living with an anxiety disorder results in higher personal and societal financial expenditure. For example, in the United Kingdom, the mean annual care costs for people living with anxiety are £71 million (US \$84.2 million) (Parkinson's UK, 2017). In the United States, this figure is approximately US \$52 billion (Yang et al., 2019).

Anxiety disorders, no matter the type or origin of the disorder, have a profound influence on an individual's ability to participate in their chosen activities in daily life to a level with which they are satisfied (Duncan et al., 2014). The stress associated with anxiety disorders reduces energy and the motivation to complete activities of daily living (Johansson et al., 2012) and is linked to both resilience and occupational adaptivity, that is, any adjustment or behavior change in response to the demands of living (Lopez, 2011). Focusing on the biochemical or cognitive aspects of anxiety as part of a health intervention may not necessarily contribute to an improvement in an individual's participation in life activities. Additionally, poor functioning can continue after full or partial recovery from an anxiety disorder due to the longitudinal impact of psychosocial impairment, suggesting that participation in daily life appears to be independent of symptoms of anxiety (Rodriguez et

al., 2005). It is also known that participation and functioning are a greater predictor of health care usage and cost than the severity of either anxiety or depressive symptoms (Twomey et al., 2017). These findings point to a need to identify the characteristics of interventions that focus on how individuals with anxiety participate in daily activities, as this will contribute to the development of more targeted and effective interventions in the future. Furthermore, interventions that are targeted more appropriately may potentially be more cost-effective (Laxminarayan et al., 2006).

The stories of those living with anxiety disorders emphasizes the impact that this condition has on people's everyday lives. This includes the effect that anxiety has on social relationships and feelings of shame and embarrassment:

I go to the bathroom and when I come back he says "where have you been?" I've been to the toilet. "You have been a long time." So trying to hide... (the attacks). He thinks it's stupid. It makes you feel ashamed. (Kadam et al., 2001)

The experience of anxiety is a complex phenomenon that affects people across cultures. In a study of adults living with anxiety in Iran, the participants explained the experience as "*it is like I am in a cage*" and fearing being both rejected or neglected by others, such as being mocked or losing popularity (Kadam et al., 2001). In the United Kingdom, some have explained the experience of anxiety and its impact as "*My head says I'm under attack and physically I feel like I'm under attack*" and "*it has prevented me from doing a lot of things*" (Mohammadi et al., 2019). This range of experiences further reinforces the need to explore the characteristics of interventions for people.

Occupational therapists have historically worked with people experiencing anxiety in a variety of settings. In a systematic review and narrative synthesis of occupational therapy interventions for people with anxiety disorders, Fox et al. (2019) were unable to make a judgment on overall clinical effectiveness due to the heterogeneity of the identified studies. As it was an effectiveness review, the authors highlighted the potential for replicability of the interventions in future research and as treatments, the characteristics of the identified studies, and their interventions were not charted in detail. It is also unclear if the authors adhered to the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) statement for reporting a systematic review, which the limits replicability and transparency of their review. In their conclusion, (Fox et al., 2019) highlight that future research should report on the impact of occupational therapy interventions for people living with anxiety. The proposed scoping review seeks to further the authors' findings (Fox et al., 2019) by updating the search and exploring and understanding the current landscape of intervention characteristics of the interventions identified to aid the development of future interventions and research. A search of the CINAHL database has identified that studies have been conducted since 2019 that would add to the data.

The identification of potential intervention characteristics is crucial to the development of new complex interventions (those interventions with several interacting components) (O'Cathain et al., 2019). If too little time is spent on identifying characteristics in the early stage of the complex intervention development cycle, this may result in a flawed intervention that is lacking in clinical and cost-effectiveness (O'Cathain et al., 2019). Furthermore, identifying the characteristics of previous occupational therapy studies will benefit the design and future evaluation of new interventions by examining how the research is conducted and identifying knowledge gaps (O'Cathain et al., 2019; Skivington et

al., 2021). This approach aligns with the development stage of the "developmentevaluation-implementation process" for complex interventions as outlined by the Medical Research Council (O'Cathain et al., 2019; Skivington et al., 2021). Adopting this approach will help to ensure that future interventions have a better chance of being effective when evaluated and of implementation, that is, being adopted in real-world settings (O'Cathain et al., 2019).

A preliminary search of MEDLINE, the Cochrane Database of Systematic Reviews, and *JBI Evidence Synthesis* was conducted, and no current or underway systematic reviews or scoping reviews on the topic were identified. Thus, the objective of this scoping review is to examine the range and characteristics of occupational therapy interventions that have been reported to improve the participation, quality of life, and well-being of community-dwelling adults living with anxiety. This scoping review will create a map of community-based interventions used to improve participation, quality of life, and well-being for adults with anxiety. It is hoped that the findings of the review will support the development of occupational therapy interventions to empower people to live well with their condition by improving their quality of life, mood, and participation in meaningful occupations.

Review Questions

What Occupational Therapy Interventions Exist Globally for Adults with Anxiety? Specifically, this review will collect and chart data to address the following subquestions: (1) what are the characteristics of the occupational therapy interventions identified? (2) What are the outcome measures used to evaluate the occupational therapy interventions identified? (3) What was the reported impact of occupational therapy interventions for

adults with anxiety? (4) What are the implications for practice and future research of the occupational therapy interventions identified?

Methods

Study Design

The proposed scoping review will be conducted in accordance with the Joanna Briggs Institute (JBI) methodology for scoping reviews (Peters, Godrey, et al., 2020). This scoping review protocol is registered on the Open Science Framework (Lovegrove, 2021) and will be reported in accordance with the Preferred Reporting Items for Systematic Reviews and Meta-Analyses Protocols (PRISMA-P) statement (Moher et al., 2015). This scoping review protocol will be conducted using the JBI guidelines for scoping reviews to ensure a systematic methodology that can be replicated (Peters, Godrey, et al., 2020). The planned review will be reported in line with the PRISMA Extension for scoping reviews (Tricco et al., 2018).

Eligibility Criteria

Participants

The review will consider studies of occupational therapy that (1) include adults with a diagnosis of anxiety who are aged 18 years and older and are resident in their own home or a care setting and (2) include diagnoses such as generalized anxiety disorder, panic disorder, and anxiety disorder not otherwise specified as defined by DSM-5-TR (or earlier versions) criteria. Studies for exclusion will be those with people without a confirmed diagnosis of an anxiety disorder, those concerning pharmacological interventions, researches focused on diagnostics, researches focused on stress or stress-related disorders.

Concept

The concept being mapped within this scoping review will be occupational therapy interventions used to support the participation, well-being, and quality of life of people living with anxiety. The terms anxiety and stress are often used interchangeably in research. As a result, these terms require clarification. Bystritsky and Kronemyer (2014) explain that stress tends to be an external stimulus arising from the environment. Alternatively, Hallion and Ruscio (2013) describe anxiety as a persistent internal feeling of fear and worry that is intrusive in daily life. Clarifying these definitions suggests that it is more appropriate to focus solely on anxiety in this scoping review, rather than a combination of terms.

Occupation is viewed as the performance of chosen daily life tasks that provide desirable levels of pleasure, productivity, and restoration (Fisher, 2013). Examples of meaningful occupations may include (but is not exclusive to) walking in nature, interacting with the community (eg, leisure groups), or going to work (paid and unpaid). Occupational therapy interventions can be characterized as occupation based or occupation focused (Fisher, 2013). Occupation-based interventions therapeutically use the individual's participation in a chosen meaningful occupation in the context as it would unfold in that person's everyday life, for example, running, working, cooking, and as the method for evaluation. Occupationfocused interventions focus on training the specific skills required for the successful performance of an individual's chosen occupation, for example, the provision of compensatory or adaptive equipment or teaching alternative strategies during a personal care occupation such as toileting (Fisher, 2013). Interventions that are focused on changing a person's basal physiology are neither occupation based nor occupation focused.

The specific items of interest are the characteristics, outcome measures, impact, and implications for the practice of the identified interventions. Participation can be defined as the ability of a person to meaningfully engage and be satisfied with their activities of daily life and has been identified by the World Health Organization as essential to health (CSDH, 2008; Larsson-Lund & Nyman, 2017). For the purposes of this review, all types of occupational therapy interventions that support participation in meaningful occupation as the method for evaluation and intervention will be included. All modes of delivery will be incorporated, and any dosage or frequency of intervention will be considered. Examples of types of interventions include self-management, social support, and psychosocial interventions and their variants, and examples of modes of delivery include support groups, one-to-one sessions, telehealth, and digital delivery models.

Context

This scoping review will consider studies that specifically include community-based occupational therapy interventions in all countries for adults living with anxiety. This could include people with anxiety living in their own homes or other care settings.

Types of Sources

This scoping review will consider both experimental and quasi-experimental study designs including randomized controlled trials, nonrandomized controlled trials, before and after studies, and interrupted time-series studies. In addition, analytical observational studies including prospective and retrospective cohort studies, case-control studies, and analytical cross-sectional studies will be considered for inclusion. This review will also consider descriptive observational study designs including case series, individual case reports, and descriptive cross-sectional studies for inclusion.

Qualitative studies will also be considered that focus on qualitative data including, but not limited to, designs such as phenomenology, grounded theory, ethnography, qualitative description, action research, and feminist research. In addition, systematic reviews that meet the inclusion criteria will also be considered, depending on the research question. Text and opinion papers will also be considered for inclusion in this scoping review.

Review Team

The review is being conducted by a team comprised of a National Institute for Health and Care Research Clinical Doctoral Research Fellow (CL), academics (KB, IS, and JM), and an information specialist (MS).

Patient and Public Involvement

The scoping review protocol has been reviewed by an occupational therapist (AP) with experience of working in a community setting.

Search Strategy

The search strategy will aim to locate both published and unpublished studies. An initial limited search of MEDLINE, CINAHL, and PsycINFO was undertaken to identify articles on the topic. The text words contained in the titles and abstracts of relevant articles, and the index terms used to describe the articles were used to develop a full search strategy for PubMed, MEDLINE, CINAHL, JBI EBP database, PLOS, Social Care Online, The Cochrane Library, Science Direct, Embase, PsycArticles, Psychology & Behavioural Sciences Collection, TRIP database, OpenGrey, OALster, and ProQuest Dissertations & Theses (Appendix 1). The search strategy, including all identified keywords and index terms, will be adapted for each included

database and information source. The reference list of all included sources of evidence will be screened for additional studies. The searches will be conducted by MS and CL.

Studies published in English will be included, as the resources for translation are not available, and no date limiters will be used.

Study or Source of Evidence Selection

Following the search, all identified citations will be collated and uploaded into the bibliographic citation management system, EndNote 20.2.1 (Clarivate Analytics), and duplicates removed (The EndNote Team, 2013). The citations will be transferred to Rayyan Qatar Computing Research Institute to manage the screening process (Ouzzani et al., 2016). Following a pilot test, titles and abstracts will then be screened by two or more independent reviewers (CL and KB) for assessment against the inclusion criteria for the review. Potentially relevant sources will be retrieved in full, and their citation details imported into the JBI System for the Unified Management, Assessment, and Review of Information (Munn et al., 2019). The full text of selected citations will be assessed in detail against the inclusion criteria by 2 independent reviewers (CL and KB). The reasons for the exclusion of sources of evidence at full text that do not meet the inclusion criteria will be recorded and reported in the scoping review. Any disagreements that arise between the reviewers at each stage of the selection process will be resolved through discussion, or with an additional reviewer (JM or IS). The results of the search and the study inclusion process will be reported in full in the final scoping review and presented in a Preferred Reporting Items for Systematic Reviews and Meta-analyses extension for scoping review (PRISMA-ScR) flow diagram (Tricco et al., 2018).

Data Extraction

A data extraction form is provided (Appendix 2). The Template for Intervention Description and Replication (TIDieR) checklist will be used for data extraction (Hoffmann et al., 2014). This internationally recognized checklist for describing interventions will enable a systematic description of the interventions and their characteristics in sufficient detail to allow future replication (if required). Data will be extracted from papers included in the scoping review by 1 reviewer (CL) using the TIDieR checklist. A second reviewer (KB) will check the extracted data. The data extraction process will first be piloted by the 2 reviewers initially on 3 papers to ensure that TIDieR checklist is used consistently. The data extracted, in line with the TIDieR checklist items, will include specific details about the intervention rationale, materials, procedures, provider, delivery mode, location, dosage, modifications, and fidelity. Any disagreements that arise between the reviewers will be resolved through discussion, or with an additional reviewer. If appropriate, the authors of papers will be contacted to request missing or additional data, where required.

Ethical Considerations

Since the scoping review methodology consists of reviewing and collecting data from publicly accessible material, this study does not require ethical approval.

Results

The extracted data will be charted in tabular form using the TIDieR checklist (Hoffmann et al., 2014). A narrative summary will accompany the charted results and will describe how the results relate to the review's objective and question. The results will be presented following the TIDieR format and the main conceptual categories used in the extraction too,

as well as gaps in the literature. These will be presented in relation to the question of this scoping review.

Discussion

This scoping review will produce a map of the current body of research and communitybased occupational therapy interventions that are used to improve participation, quality of life, and well-being of adults living with anxiety by building on the findings of a previously published systematic review (Fox et al., 2019).

To the best of our knowledge, this is the first scoping review to map the characteristics of community-based occupational therapy interventions for community-dwelling adults living with anxiety. The comprehensive information about the characteristics of such interventions, such as the outcome measures used and impact, will make an important contribution to the development of new occupational therapy interventions and future research. Identifying the characteristics and components of currently used interventions and published research is useful in optimizing the clinical and cost-effectiveness of future treatments and studies (O'Cathain et al., 2019). As anxiety is a global health issue, these findings will be of interest not only to clinicians and researchers in the United Kingdom but also to those in the international community working in this area.

As with any review, there are several limitations to conducting the proposed scoping review. While a rigorous identification and inclusion strategy has been outlined, there remains a risk that some data which may have provided further insights are inadvertently omitted. This risk is enhanced by the challenges of searching gray literature. In addition, as scoping reviews explore the breadth of a topic and not the depth, the reviewers cannot comment on the quality of the studies included in the scoping review. The proposed scoping review will synthesize the study limitations and findings as reported by the included literature. A methodological appraisal of the included studies will not be performed. Thus, the limitations and findings reported in the proposed review are not necessarily exhaustive or interrogated beyond the peer-review process of the published article. Finally, restricting the search strategy to articles only published in English means that articles that would be eligible for inclusion have the potential to be missed.

The review authors will use the findings of this scoping review to contribute to the coproduction of a new complex intervention to help people with Parkinson to live well with anxiety, embedded within the UK National Health Service. Similarly, these findings will be of value to a wider audience of health care professionals and researchers involved in service development or research projects that aim to improve the lives of people living with anxiety.

2.2 Characteristics of occupational therapy interventions for community-dwelling adults with anxiety: a scoping review

ABSTRACT

Background: Anxiety is a global health issue associated with reduced life quality and wellbeing. Developing an overview of studies of occupational therapy interventions for people with anxiety is an essential first step to understanding the existing knowledge. This will increase the impact of future interventions.

Objective: This scoping review will identify: i) what occupational therapy interventions exist for adults with anxiety; ii) the intervention characteristics, including outcomes used and impact observed.

Methods: We considered all studies and publications of occupational therapy that include community-dwelling adults with a diagnosis of anxiety who are 18 years and older and diagnosed with an anxiety disorder (DSM criteria). Databases searched were PubMed, MEDLINE, CINAHL, JBI EBP database, PLoS, Social Care Online, The Cochrane Library, Science Direct, Embase, PsycArticles, Psychology & Behavioural Sciences Collection, TRIP database, OpenGrey, OAlster, and ProQuest Dissertations & Theses. Titles and abstracts were screened against the inclusion criteria using Rayyan QCRI. Potentially relevant studies were fully retrieved and assessed against the inclusion criteria. Two independent reviewers completed the study selection. Data were extracted using a data extraction tool. Data were charted using the Template for Intervention Description and Replication (TIDieR) checklist in alignment with the review objectives.

Findings: Eighteen studies were included in the review. Articles were published in the northern hemisphere and were equally split between occupational therapy-specific and non-specific journals. Two RCTs (n=2, 11%) were included with the remainder being quasi-experimental, non-RCT designs (n=16, 89%). Articles reported multiple interventions and data collection methods in various practice settings. Eight interventions with a range of characteristics were identified. Thirty-one different outcomes were reported. Twenty-four individual impacts were observed.

Interpretation: Future research should report on the impact of occupational therapy interventions in activity as a key outcome of interest, as opposed to impairment-based outcome measures. This scoping review reinforces previous research findings and highlights the need for further high-quality research into occupation-based or occupation-focused interventions for community-welling adults living with anxiety.

Funding: The National Institute for Health and Care Research (NIHR), grant number NIHR301565, supported this work.

Systematic review registration: Open Science Framework (osf.io/js549).

KEYWORDS

Occupational Therapy; Anxiety; Adult; Health; Activities of Daily Living

INTRODUCTION

Anxiety and stress-related disorders are the ninth-highest global cause of disability (Vos et al., 2015). These disorders comprise excessive fear or worry about a current or anticipated future event (Vos et al., 2015). Anxiety disorders are highly persistent over an individual's life course, associated with higher levels of disability, reduced quality of life, and decreased participation in meaningful activity (Kessler et al., 2007). Living with an anxiety disorder leads to higher personal and societal financial expenditure. The mean annual care costs for people with anxiety in the UK is £71m (Parkinson's UK, 2017). This figure is around \$52 billion in the US (Yang et al., 2019). Global financial expenditure related to anxiety disorders (combined with depression) is estimated to be approximately USD 1 trillion and is forecast to reach USD 16 trillion by 2030 (Patel et al., 2018).

Anxiety disorders profoundly influence a person's ability to participate in their chosen activities in daily life to a level with which they are satisfied, no matter the type of disorder or its origin (Duncan et al., 2014). The neurobiological and psychological stresses associated with anxiety disorders reduce both energy and motivation to complete activities of daily living (Johansson et al., 2012). This is linked to resilience and occupational adaptivity, i.e., any behaviour adjustments in response to the demands of living (Lopez, 2011). Focusing on anxiety's biochemical or cognitive aspects as part of a health intervention may not necessarily improve a person's participation in life activities. Additionally, poor functioning can continue after full or partial recovery from an anxiety disorder due to the longitudinal

impact of psychosocial impairment (Rodriguez et al., 2005). The continuation of poor functioning post-recovery suggests that participation in daily life is independent of anxiety symptoms (Rodriguez et al., 2005). Participation and functioning are also stronger predictors of healthcare usage and cost than the severity of anxiety or depressive symptoms (Twomey et al., 2017). These findings highlight a need to identify the characteristics of interventions that focus on how individuals with anxiety participate in daily activities. Identifying such features will contribute to developing more targeted and effective future interventions. Furthermore, interventions targeted more appropriately may be more cost-effective (Laxminarayan et al., 2006).

The stories of those living with anxiety disorders accentuate the impact of anxiety disorders on their everyday lives. The experience of anxiety is a complex phenomenon that affects people across cultures. In a study of adults with anxiety in Iran, participants describe the experience as *"it is like I am in a cage"* and fearing rejection or neglect by others, such as mocked or losing popularity (Mohammadi et al., 2019). In the UK, the experience of anxiety and its impact has been described as:

"My head says I'm under attack and physically I feel like I'm under attack" and "it has prevented me from doing a lot of things." (Mental Health Foundation, 2014) pg.19 & 29

The impact of anxiety disorders on social relationships and feelings of shame and embarrassment was shared by participants in a qualitative study of GP patients experiencing anxiety, depression, or both: 'I go to the bathroom and when I come back he says "where have you been?" I've been to the toilet. "You have been a long time." So trying to hide... (the attacks). He thinks it's stupid. It makes you feel ashamed.' (Kadam et al., 2001) pg.378

Other long-term conditions do not prevent a person from developing an anxiety disorder. Anxiety is a common experience that negatively affects wellbeing in various long-term conditions, including Parkinson's disease, Multiple Sclerosis, and Chronic Obstructive Pulmonary Disease (Naylor et al., 2012). For example, in qualitative studies of anxiety in people with Parkinson's participants shared the following:

"Well I sort of, sweaty. And I feel, I feel I go starey. A bit like a rabbit in the headlights. Sometimes. And I just feel that I'm there but I'm not part of what's going on. I feel like I'm an outsider looking in, an observer of a situation rather than being part of it. . ." (Lovegrove & Bannigan, 2021) pg.8

These experiences, ranging from the ubiquitous nature of anxiety to the negative impacts on daily life, reinforce the necessity of exploring the characteristics of interventions for people living with anxiety to inform intervention development.

Occupational therapists have historically worked with people experiencing anxiety in various settings, including (but not limited to) acute hospitals, outpatient clinics, and community settings (Bryant, 2022). In a systematic review and narrative synthesis of occupational therapy interventions for people with anxiety disorders, Fox, Erlandsson, and Shiel (2019) could not judge overall clinical effectiveness due to the heterogeneity of the identified studies. The authors highlighted the potential for replicability of the interventions in future

research and as treatments. The identified studies and intervention characteristics were not charted in detail, as the review's primary aims focused on effectiveness. It is unclear if the authors adhered to the PRISMA statement for reporting a systematic review, thus limiting the review's replicability and transparency. In conclusion, Fox, Erlandsson, and Shiel (2019) highlight that future research should report on occupational therapy interventions' impact on people living with anxiety. This scoping review seeks to build on Fox, Erlandsson, and Shiel (2019) findings by updating the search and exploring and understanding the intervention characteristics. The findings of this review will aid the development of future interventions and research.

Identifying potential intervention characteristics is critical to developing new complex interventions, i.e. those with several interacting components (Skivington et al., 2021). Spending too little time on identifying attributes in the early stage of the complex intervention development cycle may lead to a flawed intervention that is lacking in clinical and cost-effectiveness (Skivington et al., 2021). Likewise, identifying the characteristics of previous occupational therapy studies will benefit the design and evaluation of new interventions by identifying knowledge gaps (Peters, Godrey, et al., 2020; Skivington et al., 2021). This approach aligns with the development stage of the "development-evaluationimplementation process" for complex interventions outlined by the Medical Research Council (Skivington et al., 2021). This approach will help ensure that future interventions are more effective when evaluated and adopted in real-world settings (Skivington et al., 2021).

A preliminary search of MEDLINE, the Cochrane Database of Systematic Reviews and *JBI Evidence Synthesis* was conducted. No current or underway systematic or scoping reviews on the topic were identified. This scoping review aims to examine and map the range and characteristics of occupational therapy interventions that have been reported to improve the participation, quality of life, and wellbeing of community-dwelling adults living with anxiety. The review findings will support the development of occupational therapy interventions to empower people to live well with their condition by improving participation in meaningful occupation, thus improving quality of life.

Review Questions

This review aims to answer the question: what occupational therapy interventions exist globally for adults with anxiety?

Specifically, this review collects and charts data to address the following sub-questions:

i) What are the characteristics of occupational therapy interventions identified?

ii) What outcome measures are used to evaluate identified occupational therapy interventions?

iii) What was the reported impact of occupational therapy interventions for adults with anxiety?

iv) What are the implications for practice and future research of the occupational therapy interventions identified?

METHODS

This scoping review followed the JBI methodology for scoping reviews (Peters, Godrey, et al., 2020). It is registered on the Open Science Framework (Lovegrove, 2021). In line with best practice, review reporting used the updated Preferred Reporting Items Systematic Reviews and Meta-Analyses Extension for Scoping Reviews checklist (PRISMA-ScR) (Appendix 3) (Skivington et al., 2021). The review followed an a priori protocol (Lovegrove et al., 2023). Any deviations from the protocol are reported and justified.

Eligibility Criteria

Participants

The review considered studies of occupational therapy that include:

- Adults with a diagnosis of anxiety who are aged 18 years and older and are resident in their own home or a care setting;
- diagnoses such as generalised anxiety disorder (GAD), panic disorder, and anxiety disorder not otherwise specified as defined by DSM criteria.

Studies for exclusion were:

- those with people without a confirmed diagnosis of an anxiety disorder;
- those concerning pharmacological interventions;
- research focused on diagnostics;
- research focused on stress or stress-related disorders.

Concept

The concept mapped within this scoping review is occupational therapy interventions used to support the participation, wellbeing, and quality of life of people with anxiety. The terms anxiety and stress have been used interchangeably in research. Consequently, these terms require clarification. Stress tends to be an external stimulus arising from the environment, whereas anxiety is a persistent internal feeling of fear and worry that is intrusive in daily life (Hallion & Ruscio, 2013). Clarifying these definitions suggests that focusing solely on anxiety in this scoping review rather than a combination of terms was more appropriate.

Occupation is regarded as performing chosen daily life tasks that provide desirable pleasure, productivity, and restoration (Larsson-Lund & Nyman, 2017). Examples of meaningful occupations may include (but not exclusively) walking in nature, interacting with the community (e.g., leisure groups), or working (paid and unpaid). Occupational therapy interventions can be occupation-based or occupation-focused (Fisher, 2013). Occupationbased interventions therapeutically use the person's participation in a chosen meaningful occupation (e.g., learning, gardening, cooking) in the context as it would unfold in their everyday life and as the method for evaluation. Occupation-focused interventions focus on training the specific skills required for the successful performance of a chosen occupation (Fisher, 2013). Examples of such interventions could be providing compensatory or adaptive equipment, teaching alternative strategies during showering, or providing fatigue management education. Interventions focused on changing a person's underlying physiology are neither occupation-based nor occupation-focused.

This review included all types of occupational therapy interventions that support participation in meaningful occupation, such as the evaluation method, intervention

method, or both. All modes of delivery were included. The specific items of interest were the characteristics, outcome measures, impact, and implications for the practice of the identified interventions. Any intervention dosage or frequency was considered.

Context

This scoping review considers studies that specifically include community-based occupational therapy interventions for adults living with anxiety in all countries. These studies could have included people with anxiety residing in their own homes or other care settings.

Types of Sources

This scoping review considered both experimental and quasi-experimental study designs. These include randomised controlled trials, non-randomised controlled trials, before and after studies, and interrupted time-series studies. Analytical observational studies included prospective and retrospective cohort studies, case-control studies, and analytical crosssectional studies. This review also included descriptive observational study designs such as case series, individual case reports, and descriptive cross-sectional studies.

This review considered studies focusing on qualitative data including, but not limited to, designs such as phenomenology, grounded theory, ethnography, qualitative description, action research, and feminist research. Text and opinion papers were considered for inclusion in this scoping review if they contributed to answering the research question.

Review team

The review team comprised of an NIHR Clinical Doctoral Research Fellow (CL), academics (KB, IS, JM), and an information specialist (MS).

An occupational therapist (AP) with experience working in a community setting reviewed the a priori protocol (Lovegrove et al., 2023). Their input aided in shaping the review aims, objectives, and inclusion criteria.

Search strategy

An information specialist (MS) was involved in designing and implementing the search strategy to support its accuracy. The search strategy aimed to locate published and unpublished primary studies, reviews, and text and opinion pieces. Two research team members (CL, MS) completed an initial limited search of MEDLINE, CINAHL, and PsycINFO to identify articles on the topic. A complete search strategy was developed using the text words in the titles and abstracts of relevant articles and their index terms. Databases searched were PubMed, MEDLINE, CINAHL, JBI EBP database, PLoS, Social Care Online, The Cochrane Library, Science Direct, Embase, PsycArticles, Psychology & Behavioural Sciences Collection, TRIP database, OpenGrey, OALster, and ProQuest Dissertations & Theses (Appendix 4). The search strategy, including all identified keywords and index terms, was adapted for each included database and information source. One team member (CL) screened the reference list of all included sources of evidence for additional studies. Two research team members completed the searches (CL, MS). Searches were completed in 2022 and repeated in 2023.

This review only includes studies published in English, as the translation resources were unavailable. The search did not use data limiters.

Study/Source of Evidence selection

Following search completion, all identified citations were collated and uploaded into the bibliographic citation management system, EndNote 20.2.1 (Clarivate Analytics, PA, USA) (The EndNote Team, 2013). Duplicates were removed. The research team made the decision not to use the JBI System for the Unified Management, Assessment and Review of Information (JBI SUMARI; Adelaide, Australia) as JBI SUMARI does not offer modifiable data extraction templates, which was needed for this review (Munn et al., 2019). Citations were transferred to Rayyan QCRI (Qatar Computing Research Institute [Data Analytics], Doha, Qatar), a systematic review web application to manage the independent relevance checking process (Ouzzani et al., 2016).

Two reviewers (CL, KB) piloted a screening tool on a sample of studies. Minor adjustments were made until a consensus was reached to enhance clarity before continuing the screening process. The screening tool ensured that reviewers consistently applied the inclusion criteria and that all decisions were recorded on Rayyan QCRI. Two reviewers (CL, KB) screened the title and abstracts against the inclusion criteria. Potentially relevant papers were retrieved in full and assessed in detail against the inclusion criteria by the same two independent reviewers. Full-text studies that did not meet the inclusion criteria were excluded. All exclusion reasons were recorded. No disagreements that required discussion with the independent third reviewer (IS) arose. Per scoping review guidance (Peters, Godrey, et al., 2020), studies were not quality assessed, as this scoping review aimed to map available evidence rather than understanding methodological quality.

Data Extraction

Alongside data about impact and outcomes, data about intervention characteristics were extracted using The Template for Intervention Description and Replication (TIDieR) checklist (Appendix 2) (Hoffmann et al., 2014). This internationally recognised checklist for describing interventions enabled a systematic description of the interventions and their characteristics in sufficient detail to allow future replication (if required). Two independent reviewers (CL, KB) piloted the TIDieR on a sample of three articles to ensure its consistent use. One reviewer (CL) extracted data from papers included in the scoping review using the TIDieR checklist. A second reviewer (KB) checked the extracted data. Discrepancies were addressed and clarified without involving the third reviewer (IS). In line with the TIDieR checklist items, the extracted data includes specific details about the intervention rationale, materials, procedures, provider, delivery mode, location, dosage, modifications, and fidelity. One reviewer (CL) contacted the authors of papers to request missing or additional data where required.

Data Presentation

As specified in the protocol and recommended in the JBI scoping review guidance (Lovegrove et al., 2023; Peters, Godrey, et al., 2020), the extracted data is presented in diagrammatic and tabular form. A PRISMA flow diagram was generated using an online tool (Haddaway et al., 2022). The extracted data is charted in tabular form using the TIDieR checklist (Hoffmann et al., 2014). A narrative summary accompanies the charted results and describes how the results relate to the review objectives and questions. A mapping approach to analysis was used as the scoping review's objective is to collate the range of existing evidence and describe the methodological characteristics of existing research rather

than synthesising or appraising the evidence. The results are presented following the TIDieR format, the main conceptual categories used in the extraction, and gaps in the literature.

FINDINGS

Database and secondary searching returned 9,913 records (Figure 2.1). Due to the number and range of databases searched, a high number of duplicates were identified. After removing duplicates, 2,058 underwent title and abstract screening with 2,037 articles excluded. Full-text screening led to the exclusion of a further three articles. Most studies were excluded due to not having an occupational therapy focus or for including anxiety as a secondary outcome. A final total of 18 records were included in the review (Chugh-Gupta et al., 2013; Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Lork et al., 2021; Prior, 1998a, 1998b; Rosier et al., 1998). Full charting of the included articles is included as a supplementary file (Appendix 5). The characteristics of the included studies are presented in Table 2.1. Across the 18 included studies, articles were published predominantly in northern hemisphere countries (n=16/18, 89%) with the remainder not stated (Table 2.1). Articles were published equally in occupational therapyspecific and non-specific journals (n=9/18; 50%). Most studies adopted a non-RCT/quasiexperimental design (n=6/18; 33%); studies reporting RCTs were also highly represented (n=5/18; 28%) (Table 2.2). Studies took place in a variety of practice settings and have used multiple interventions and multiple data collection methods, including interviews and observation and multiple types of outcome data. The included articles will now be mapped to answer each question of this review.



Figure 2.1: Search results and study selection and inclusion process (30, 48)

Article Characteristics	Empirical (n=18)
Publication year	
1998	3
2006	2
2008	1
2009	1
2010	1
2013	4
2014	1
2015	1
2018	1
2019	2
2021	1
Publication type	
Journal article	18
Methodology	
RCT	5 (1 report)
Non-RCT/ quasi-experimental	6
Systematic review	2
Case study research	1
Economic evaluation	1
Intervention development	1
Intervention description	1
Mixed method	0
Qualitative	1
Intervention Agreed	
Lifestyle	9
Psychological anxiety management	4
Art	3
Exercise	1
Not specified (systematic review)	1
Geographical context	
Sweden	8
England	6
Ireland	1
Wales	1
Not stated (but author locations stated)	2
Setting	
Participant's home	0
Primary healthcare centre	9
Mental healthcare centre (inc. Community team)	3
Dual primary and mental healthcare	0
Outpatients	1
Secondary healthcare	0
University	1
Not explicitly stated	4

Table 2.2: Characteristics of included articles.

What are the characteristics of occupational therapy interventions identified?

There was a range of characteristics across the studies (Table 2.3). Eight interventions were identified across the 18 studies. These were Yoga, Redesigning Daily Occupations (ReDO), The Tree Theme Method (TTM), anxiety management group (psycho-educational), occupational therapy-led lifestyle approach, work-directed rehabilitation, short-term anxiety management course, and an anxiety management group (community mental health). Studies were collated and analysed together if they used the same intervention protocol, such as the ReDO programme. Group work was the predominant delivery method (n=5/8 interventions, 63%) (Chugh-Gupta et al., 2013; Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Prior, 1998a, 1998b; Rosier et al., 1998). Three interventions were delivered 1:1 (n=3/8 interventions, 25%) (Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Lork et al., 2021). One intervention used both groups and 1:1 sessions (n=1/8 interventions, 12%) (Chugh-Gupta et al., 2013). Seven of the interventions were delivered by an occupational therapist (n=7/8 interventions, 88%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Lork et al., 2021; Prior, 1998a, 1998b), with the remainder by trained instructors (n=1/8, 12%) (Chugh-Gupta et al., 2013). Seven interventions used a structured approach (n=7/8 interventions, 88%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Lork et al., 2021;

Prior, 1998a, 1998b). Goal setting was used in four interventions (n=4/8 interventions, 50%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2021; Lork et al., 2021; Prior, 1998a, 1998b; Rosier et al., 1998). Tailoring, where the intervention content is adapted for an individual, was used in two interventions (n=2/8 interventions, 25%) (Lork et al., 2021; Rosier et al., 1998). Two of the interventions used follow-ups (n=2/8 interventions, 25%) (Lork et al., 2021; Rosier et al., 1998). Two of the interventions used follow-ups (n=2/8 interventions, 25%) (Lork et al., 2021; Rosier et al., 1998). Cognitive behavioural therapy (CBT) was used in three interventions (n=3/8 interventions, 38%) (Fox et al., 2019; Kitchiner et al., 2009; Prior, 1998a, 1998b; Rosier et al., 1998). Lifestyle redesign® or management approaches were used in three interventions (n=3/8 interventions, 38%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021). Modifications and the involvement of family, loved ones, or carers were each used in one intervention (n=1/8 interventions, 12%) (Eklund, 2013; Eklund & Erlandsson, 2013; Erlandsson, 2013).

	Characteristics											
Source	Name of intervention	Delivered by an occupational therapist	Group	1:1	Structured	Goal setting	Tailoring	Modifications	Family/ loved ones/ carers involved?	Follow-up	Cognitive behavioural therapy	Lifestyle redesign®/ management
Chugh- Gupta <i>et al.,</i> 2013	Yoga		x	x								
Eklund and Erlandsson, 2013; Eklund <i>et al.</i> 2013; Erlandsson, 2013; Eklund, 2014; Fox <i>et al.,</i> 2019; Fox <i>et al.,</i> 2021)	Redesigning Daily Occupations (ReDO)	x	x		x	x		x	x			x
Gunnarsson <i>et</i> <i>al.,</i> 2006; 2015; 2018	The Tree Theme Method (TTM)	x		x	x							
Kitchiner <i>et</i> <i>al.,</i> 2009; Fox <i>et al.</i> 2019	Anxiety management group	x	x		x						x	
Lambert <i>et al.,</i> 2007; 2008; 2010; Fox <i>et</i> <i>al.,</i> 2019	Occupational therapy-led lifestyle approach	x	x		x							x
Lork <i>et al.,</i> 2019	Work-directed rehabilitation provided by OTs and PTs to promote work capacity in CMDs	x		x	x	x	x			x		x
Prior 1998a.; 1998b.; Fox <i>et</i> <i>al.,</i> 2019	Short-term anxiety management course	x	x		x	x					x	
Rosier <i>et al.,</i> 1998	Anxiety management groups	x	x		x	x	x			x	x	

Table 2.3. Characteristics of the identified interventions.

What outcome measures are used to evaluate identified occupational therapy

interventions?

Thirty-one different outcomes were identified (Table 2.4). Two interventions did not report

outcome measures reported (n=2/8 interventions, 25%). Four outcomes were occupational

therapy-specific measures (n=4/31, 13%; Canadian Occupational Performance Measure,

COPM; Occupational value assessment with predefined items, Oval-pd; Satisfation with

Daily Occupations, SDO; Occupational Balance Questionairre, OBQ) and were used in one of the eight interventions (n=1/8 interventions, 12%) (Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018). Most measures were specifically used to measure anxiety (n=9/31, 29%) and were used in five interventions (n=5/8 interventions, 63%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Prior, 1998a, 1998b). These were the Positive and Negative Affect Schedule (PANAS), State-Trait Anxiety Inventory (STAI), Depression Anxiety Stress Scale (DASS), Hospital Anxiety and Depression Scale (HADS), Anxiety Disorders Interview Schedule (ADIS-IV), Beck Anxiety Inventory (BAI), Fear Questionnaire, and Performance Anxiety Questionnaire. Three quality of life measures were used (n=3/31 measures, 10%) in three interventions (n=3/8 interventions, 38%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010). These were the 36-Item Short Form Survey (SF-36), Manchester Short Assessment of Quality of Life (MANSA), the EuroQol-5D (EQ-5D), and the EuroQol visual analogue scale (EQ-VAS). Three work-specific outcome measures (n=3/31, measures 10%) were used in three interventions (n=3/8 interventions, 38%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Kitchiner et al., 2009). These were the Work and Social Adjustment Scale (WSAS), Work Readiness Scale (WRS), and the Work Environment Impact Scale- Self Rating (WEIS-SR). Two life satisfaction measures (n=2/31 measures, 6%), the Client Satisfaction Questionnaire (CSQ) and the Sense of Coherence Scale (SOC), and were used in three interventions (n=3/8 interventions, 38%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Prior, 1998a, 1998b). Two measures, the Beck's

Depression Inventory (BDI-II) and Montgomery-Åsberg Depression Rating Scale- Self report (MADRS-S), were used to specifically measure depression (n=2/31 measures, 6%) and were used in three interventions (n=3/8 interventions, 38%) (Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010). A stress-specific measure (Perceived Stress Scale, PSS) was used in two interventions (n=2/8 interventions, 25%) (Chugh-Gupta et al., 2013; Eklund, 2013). Measures used to assess economic (quality-adjusted life years, QALYs; Incremental Cost-Effectiveness Ratios, ICERs), physiological, psychopathology (Symptom Checklist-90-Revised, SCL-90-r), therapeutic relationships (Health assessment questionnaire-II, Haq-II), disability (WHO Disability Assessment Schedule, WHODAS 2.0), mastery (Pearlin Mastery Scale, PMS), and general health (General Health Questionnaire-28, GHQ-28) were used once each (n=1/31 measures, 3%; total n=8/31 measures, 26%), and were used across five interventions (n=5/8, interventions 63%) (Chugh-Gupta et al., 2013; Fox et al., 2021; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Kitchiner et al., 2009; Lambert et al., 2010).
uo							(Ou	Itc	on	ne	m	eas	sui	res	s u	se	d	in	th	e i	nc	lu	de	d s	stu	ıdi	ies	5					
Interventio	PANAS	STAI	HADS	PSS	MANSA	PMS	csq	WRS	WEIS-SR	GHQ-28	COPM	DASS	WHODAS 2.0	OVal-pd	PMS	EQ-5D-5L/ EQ-	BAI	ADIS-IV	SF-36	BDI-II	QALYs/ ICERs	SCL-90-r	MADRS-S	SDO	OBQ	soc	HAq-II	Life and Social	Adjustment	Fear	Destionnaire	Anxiety	Ouectionnaire	Physiological
Yoga	x	х	х	x																												x		x
Redesigning Daily Occupations			x	x	x	x	x	x	x			x	x	x	x	x																		
The Tree Theme Method (TTM)					x		x				x												x	x	x	x								
Anxiety management group										x										x									х	x				
Occupational therapy-led lifestyle approach																x	x	x	x	x	x	x								x				
Work- directed rehabilitation provided by OTs and PTs to promote work capacity in CMDs																																		
Short-term anxiety management course			x				x																						x	x				
Anxiety management groups																																		

 Table 2.4. Outcome measures used in the interventions.

What was the reported impact of occupational therapy interventions for adults with

anxiety?

Twenty-four individual impacts were observed across the eight interventions (Table 2.5). Impacts were defined as any identifiable positive or negative influence for the person's health or functioning, service delivery, or costs (Cruz Rivera et al., 2017). Anxiety reduction was most common across seven interventions (n=7/8 interventions, 86%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Lork et al., 2021; Prior, 1998a, 1998b; Rosier et al., 1998). Greater behaviour awareness was an observed impact in five interventions (n=5/8 interventions, 63%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Kitchiner et al., 2009; Lambert et al., 2008; Lambert et al., 2007; Lork et al., 2021; Rosier et al., 1998). Other observed impacts were at the process level. Intervention satisfaction (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Fox et al., 2019; Fox et al., 2021; International handbook of occupational therapy interventions, 2015; Prior, 1998a, 1998b; Rosier et al., 1998), acceptability (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Prior, 1998a, 1998b), delivered as intended (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018), and increased understanding (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Lambert et al., 2008; Lambert et al., 2007) were each observed three times (n=3/8 interventions, 38%). Increased quality of life (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Fox et al., 2019; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010), enhanced satisfaction (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Rosier et al., 1998), improved coping (Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Rosier et al., 1998), general health improvement (Chugh-Gupta et al., 2013; Lambert et al., 2008; Lambert et al., 2007), extra workload (Fox et al., 2021; Rosier et al., 1998), depression reduction (Eklund, 2013; Kitchiner et al., 2009), timeconsuming (Lambert et al., 2008; Lambert et al., 2007; Prior, 1998a, 1998b), and follow-ups (Prior, 1998a, 1998b; Rosier et al., 1998) were observed twice each (n=2/8 interventions, 25%). The remaining impacts were observed once each (12%) (Eklund, 2013; Eklund &

Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Lambert et al., 2008; Lambert et al., 2007; Prior, 1998a). The need for further research on impact was identified across six interventions (n=6/8 interventions, 75%) (Chugh-Gupta et al., 2013; Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Kitchiner et al., 2009; Lork et al., 2021; Prior, 1998a, 1998b).

Regarding individual interventions, the Redesigning Daily Occupations (ReDO) programme had the most observed impacts at fourteen (n=14/24, 58%) (Eklund, 2013; Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2019; Fox et al., 2021). The Tree Theme Method (TTM), occupational therapy-led lifestyle approaches, and short-term anxiety management course had seven observed impacts (n=7/24, 29%) (Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Prior, 1998a; Prior & Campbell, 2018). The community anxiety management groups observed six impacts (n=6/24, 25%) (Rosier et al., 1998). The taught anxiety management group had five observed impacts (n=5/24, 21%) (Kitchiner et al., 2009). Finally, work-directed rehabilitation and yoga had three observed impacts (n=3/24, 13%) (Chugh-Gupta et al., 2013; Lork et al., 2021).

	Impact observed																							
Intervention	Anxiety reduction	Improved quality of life	Enhanced satisfaction	Improved coping	General health improvement	Long term effects	Intervention satisfaction	Increased paid work engagement/ reduced sick leave	Acceptable	Extra workload	Delivered as intended	Value for money	Research needed	Self-Mastery	Depression reduction	Increased understanding	Helps assist client to tell life story	Panic reduction	Lifestyle review	Reduced fear	Greater behaviours awareness	Time-consuming	Followed up	Redesigning daily occupations
Yoga (Chugh- Gupta <i>et al.,</i> 2013)	x				x								x											
Redesigning Daily Occupations (ReDO) (Eklund and Erlandsson, 2013; Eklund <i>et al.</i> 2013; Erlandsson, 2013; Eklund, 2014; Fox <i>et al.,</i> 2019; 2021)	x	x	x				x	x	x	x	x		x	x	x	x					x			x
The Tree Theme Method (TTM) (Gunnarsson <i>et al.,</i> 2006; 2015; 2018)			x	x					x		x		x			x	x							
Anxiety management group (Kitchiner <i>et al.,</i> 2009; Fox <i>et al.</i> 2019)	x										x		x		x						x			
Occupational therapy-led lifestyle approach (Lambert <i>et al.,</i> 2007; 2008; 2010; Fox <i>et al.,</i> 2019)	х	x			x							x				x			x	x	x	x		
Work-directed rehabilitation provided by OTs and PTs to promote work capacity in CMDs (Lork <i>et</i> <i>al., 2019</i>)	x												x								x			
Short-term anxiety management course (Prior 1998a.; 1998b.; Fox <i>et al.,</i> 2019)	x					x	x		x				x									x	x	
Anxiety management groups (Rosier <i>et</i> <i>al.,</i> 1998)	x			x			x			x											x		x	

Table 2.5. Observed impacts of the interventions.

DISCUSSION

This scoping review explored what occupational therapy interventions exist globally for community-dwelling adults with anxiety. It extends previous research (Fox et al., 2019) to identify the characteristics of these interventions, the outcome measures used to evaluate them, and their reported impact. Eighteen articles were identified that report on eight interventions with a range of characteristics, outcomes, and impacts reported. Interventions were delivered in a variety of settings using different methods.

Group therapy was the main method of delivering interventions to adults living with anxiety (n=5/8 interventions, 63%). Group therapy is associated with various beneficial therapeutic factors, such as universality, cohesiveness, and service-level benefits (Lambert et al., 2010; Malhotra & Baker, 2023). Participants often find groups acceptable, finding it helpful to share experiences and provide peer support, which can be instrumental in supporting change (Fox et al., 2019; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007). Whilst group work generally involves additional workload, occupational therapists in one study found this acceptable, and there are benefits in treating more people at a time (Fox et al., 2021). Despite the seemingly overwhelming benefits, group therapy has potential disadvantages that need consideration. Group delivery can mean that there is less opportunity for clinicians to spend time with a person and to tailor interventions to their individual needs (Whitfield, 2010). When discussing intimate experiences such as anxiety, groups may seem intimidating, and there may be a preference to do this on an individual basis (Lovegrove et al., 2017). Other interventions (n=3/8 interventions) reported positive impacts from 1:1 individual therapy with similar reports of acceptability (Chugh-Gupta et al., 2013; Gunnarsson et al., 2006; Gunnarsson et al., 2015; Gunnarsson et al., 2018). Most interventions reported using a structured approach (n=7/8). This is not surprising as

structured interventions that account for context and complexity are both effective and recommended (Hawe, 2015; Skivington et al., 2021). Future occupational therapy research and interventions should be vigilant in exploring the best way of addressing people's needs and not assume that this will automatically be via group therapy work.

Goal setting was used in four interventions (n=4/8 interventions, 50%). This is not surprising given the evidence for using goal setting in occupational therapy (Arbesman & Logsdon, 2011; Hand et al., 2011; Scobbie et al., 2011). In anxiety interventions, goal setting is effective via mechanisms such as open communication, collaboration, and hope (Jacob et al., 2022; Richardson, 2023). Goal setting is often used to tailor interventions to a person's specific needs, yet tailoring (n=2/8 interventions, 25%) and modification (n=1/8 interventions, 12%) were not routinely used in the reported interventions. This seems at odds with the person-centred ethos of occupational therapy and the focus on meaningful activity, particularly considering the effectiveness of tailored interventions and their impact on reducing health disparities (Baker et al., 2015; Campbell & Quintiliani, 2006). This finding underscores the importance of using a checklist such as the TIDieR so that aspects that likely contribute to the mechanisms of change are not missed. Research using methodologies such as realist evaluation is needed to enhance understanding of such working mechanisms (Lemire et al., 2020).

The involvement of family, loved ones, or carers was only reported in one intervention (n=1/8 interventions, 12%) (Eklund & Erlandsson, 2013; Eklund et al., 2013; Erlandsson, 2013; Fox et al., 2021). This was unexpected considering that the want for said involvement has been clearly expressed by various patient groups (Dijkman et al., 2022; Lovegrove et al.,

2017; Prior & Campbell, 2018). Future occupational therapy and interventions should utilise goal setting and intervention tailoring to enhance effectiveness where feasible.

Outcome measures are important to determine the effectiveness of an intervention and are used widely in healthcare, including occupational therapy (Laver Fawcett, 2014). The ReDO programmes described the most observed impacts (n=14/24, 58%). This is perhaps to be expected as the studies evaluating this intervention also reported the most outcome measures (n=12/27, 44%) and thus had the most data from which to report. Because the interventions were designed to address anxiety, it is not surprising that anxiety-specific measures were widely used. Fortunately, anxiety reduction was also the most observed impact (n=7/8 interventions, 86%). Quality of life measures were also used for similar reasons (Barbalat et al., 2019; Mercieca-Bebber et al., 2018), despite only being used in three interventions (n=3/8 interventions, 37%). Only one intervention (n=1/8 interventions, 12%) reported using occupational therapy-specific outcome measures(Eklund, 2013; Eklund et al., 2013; Fox et al., 2021). Therefore, it could be argued that the impact of occupational therapy on occupational performance in meaningful activity is only measured in one of the reviewed interventions.

Despite the growing need for health professions to demonstrate cost-effectiveness in an increasingly financially constrained climate, only one intervention was economically evaluated (Lambert et al., 2010). This finding reflects the current low number of health-economic evaluations of occupational therapy interventions, which will be important to address as global financial challenges continue (Green & Lambert, 2017). Future research studies and interventions should include validated occupational therapy outcomes and

economic evaluations to document and evaluate their contribution to a person's health outcomes (Leland et al., 2015).

What are the implications for practice and future research of the occupational therapy interventions identified?

Future research should report on the impact of occupational therapy interventions on activity as a key outcome of interest, as in conjunction with select impairment-based outcome measures. Participation or function was not reported as an outcome in seven interventions included in this review, including in two randomised controlled trials (n=7/8 interventions, 86%) (Kitchiner et al., 2009; Lambert et al., 2007). We recommended that functioning and participation in meaningful occupations be measured as core outcomes of occupational therapy interventions for anxiety in future clinical trials. These should be included alongside impairment-based measures to facilitate more comprehensive measurement of people's health.

The need for further research was noted in six interventions (n=6/8 interventions, 75%). The structure of the interventions is not clearly described in all the reported interventions, which makes replication difficult (Klein et al., 2000). Rigorous, well-powered studies need to be conducted to test occupational therapy interventions for adults living with anxiety. The studies included in this review demonstrate that it is possible to complete RCTs and quasi-experimental studies in this clinical area. The need for more high-quality research in occupational therapy, particularly in mental health, has been well documented in the literature (Bannigan et al., 2008; Kirsh et al., 2019). This review reinforces previous research and highlights the need for high-quality research into occupation-based or occupation-

focused interventions for community-welling adults living with anxiety that include economic evaluations (Fox et al., 2019; Mackenzie et al., 2017).

Strengths and limitations

This review searched published and grey literature to conduct a comprehensive review. It followed a peer-reviewed protocol with systematic and transparent processes (Lovegrove et al., 2023). JBI methodology for conducting scoping reviews was followed (Peters, Godrey, et al., 2020; Peters, Marnie, et al., 2020). Bibliographic software (Endnote) and systematic review software (Rayyan) was used to robustly manage citations and the screening process (Ouzzani et al., 2016; The EndNote Team, 2013).

Fifteen databases were searched. No further relevant articles were identified through websites or citation searching, confirming that a thorough search was conducted. The research team made some decisions to balance a comprehensive search with available resources that may have affected the inclusivity of this review. English language limits and Western-dominant databases were used because of translation resource availability within the research team. All included articles originate from Northern Hemisphere countries, which may reflect this (Table 2.1). Thus, the research team may have missed articles published in other languages and eligible for inclusion, which means we may have missed valuable insights.

Scoping reviews explore the breadth of a topic and not the depth. As such, the research team cannot comment on the quality of the included studies. Methodological appraisal of the included studies was not performed. Consequently, the limitations and findings reported in the proposed review are not exhaustive or examined beyond the peer-review process of the published article.

CONCLUSION

Anxiety is a common global cause of disability (Vos et al., 2015). Occupational therapists work to enable people to participate in everyday life activities to a level they are satisfied with (World Federation of Occupational Therapists, 2019). This scoping review aimed to determine what occupational therapy interventions exist globally for adults with anxiety. Eighteen studies were included in the review, which report on eight different interventions for this group of people. A range of outcome measures were used but only one intervention employed occupational therapy-specific measures. Future interventions and research should be vigilant to use occupational therapy outcome measures alongside appropriate impairment-based measures. This will allow proper evaluation of the effectiveness of occupational therapy services that aim to improve occupational performance and participation of people living with anxiety. The majority of interventions were delivered in groups, but both the potential benefits of individual occupational therapy sessions should not be ignored. Goal setting and tailoring should be used to enhance the effectiveness and impact of occupational therapy interventions for adults living with anxiety. Rigorously designed clinical trials of occupational therapy interventions for adults living with anxiety are needed; these should include economic evaluations to demonstrate cost-effectiveness.

Acknowledgements

The authors thank Anna Paisey (AP) for contributing to developing this scoping review's protocol. A list of abbreviations is provided in Appendix 6.

Chapter 3: Group Concept Mapping Study

The scoping review contributed to the thesis by identifying the evidence; part of the 'development' stage of the MRC complex intervention framework (Skivington et al., 2021). The next component of this stage is identifying and developing theory (Figure 1.5). A consensus study using group concept mapping was conducted to identify the key components required for the co-production of the OBtAIN-PD. This chapter presents the peer-reviewed article of this study. This novel participatory study highlighted priority components that provide starting points for future development of an occupation- and community-based intervention for people with PD-related anxiety.

Understanding and addressing the priorities of relevant stakeholders is crucial in healthcare decision-making, especially in situations with limited evidence or resources (Arakawa & Bader, 2022; Carter et al., 2023). Consensus methods help prioritise research questions, trial outcomes, clinical guidelines, medical education topics, and resource allocation (Arakawa & Bader, 2022; Carter et al., 2023). They also play a key role in complex intervention development by combining evidence with expert opinion to prioritise components for inclusion (Farquhar et al., 2013; Johnston et al., 2020). Common features of consensus methods include standardised materials, moderated interactions, private polling, and iterative feedback from the group (Arakawa & Bader, 2022).

Andre Delbecq and Andrew Van de Ven developed nominal group testing (NGT) in the late 1960s to generate ideas and reach consensus (Van de Ven & Delbecq, 1972). The process involves four steps: brainstorming, idea reduction, idea prioritisation, and consensus building. Participants generate ideas, then reduce the list, rank the remaining ideas, and work together to reach a consensus. Nominal Group Testing is effective for idea generation, especially when many stakeholders have diverse opinions, or when the topic is sensitive and people are reluctant to share their views publicly (Maguire et al., 2022; Vahedian-Shahroodi et al., 2023). However, it may not be suitable for complex problems that require critical thinking and analysis (Vahedian-Shahroodi et al., 2023).

One risk of the NGT process is that opinions may not converge in the voting process, making the process more time-consuming for participants and necessitating additional rounds (Maguire et al., 2022; McMillan et al., 2016). This may make NGT more difficult for those managing complex long-term health conditions, such as Parkinson's. Additionally, due to the structure of NGT, there is a risk that the cross-fertilization of ideas may be reduced or lost, potentially resulting in the loss of innovative ideas (Manera et al., 2019; McMillan et al., 2014).

The Delphi method, also known as the Delphi process or Delphi technique, is a communication method used to obtain a group consensus from panels of experts. It involves using one or more rounds of questionnaires shared anonymously with participants, with controlled feedback following each round (Khodyakov, Grant, Kroger, & Bauman, 2023). This method ensures that decision-making is based on the convergence of expert opinion. It was originally developed by the RAND Corporation in the 1950s to survey expert opinion, gather information, and forecast the impact of new technologies on future defence policy (Khodyakov, Grant, Kroger, & Bauman, 2023). More recently, the Delphi consensus method has been a popular way to gain controlled feedback and group opinion and to define best practices in various fields, including healthcare (Niederberger et al., 2021).

The Delphi method has numerous strengths. First, it allows experts to communicate through a mediator (a researcher), facilitating rational academic debate by eliminating obstacles (Ismail & Taliep, 2023). Therefore, this method prevents direct confrontations among experts and connects existing knowledge and areas of agreement/disagreement. The Delphi method enables anonymity, promoting creativity, honesty, and balanced consideration of ideas while reducing the risk of negative group dynamics affecting outcomes (Ismail & Taliep, 2023). It has been argued that downsides associated with group dynamics, such as manipulation or coercion to conform to a certain viewpoint, can be minimized (Ismail & Taliep, 2023). This reduces the effect of noise, which distorts data and focuses on group and/or individual interests rather than problem-solving (Naisola-Ruiter, 2022). Lastly, the Delphi method enables group communication without geographical constraints, saving time and money by reducing travel costs (Naisola-Ruiter, 2022; Niederberger et al., 2021).

The Delphi method lacks clear guidance and agreed-upon standards for interpreting and analyzing results, as well as universally accepted definitions of consensus and criteria for selecting participants (Khodyakov, Grant, Kroger, Gadwah-Meaden, et al., 2023; Niederberger et al., 2021). This can make the process time-consuming and laborious for both researchers and participants, leading to drop-outs. Factors contributing to drop-outs include long-term commitment, distractions between rounds, and participant dissatisfaction with the process (Donohoe & Needham, 2009). The use of monetary payments or moral persuasion to convince participants may introduce bias into the results (Iqbal & Pipon-Young, 2009; Ismail & Taliep, 2023).

While the Delphi method's anonymity has its strengths, it can also lead to less ownership of ideas (Iqbal & Pipon-Young, 2009). Methodological limitations include challenges in generalizing results to a wider population due to sample size, varying response rates among invited experts, limited perspectives, uneven distribution of expertise among participants, and participants' specific agendas or geographic/cultural locations (Hsu & Sandford, 2007; Ismail & Taliep, 2023; Nambisan et al., 1999; Niederman et al., 1991).

A potential weakness of the Delphi method is the considerable discretion given to researchers in its application. As the quality of feedback determines the procedure's success, thorough analysis of responses becomes a significant responsibility for the researcher (Cuhls, 2023; Ismail & Taliep, 2023). Furthermore, it can be challenging to define what constitutes sufficient consensus in the Delphi procedure (Donohoe & Needham, 2009). The independence of experts' responses in the Delphi process may be compromised, especially when the experts are in contact with one another (Devaney & Henchion, 2018; Taylor, 2020).

Group concept mapping is a consensus method that helps a group organise their ideas on any topic and represent them visually in interconnected maps (Kane & Trochim, 2007). It combines qualitative and quantitative data collection and analysis approaches, making it an integrative mixed method (Kane & Trochim, 2007; Trochim & McLinden, 2017). This process allows for collaboration among groups of any size and with diverse participants. It combines the ideas of diverse participants to show what the group thinks and values in relation to the specific topic of interest. It is a structured conceptualization used by groups to develop a conceptual framework, often to help guide evaluation and planning efforts (Kane & Trochim, 2007). Group concept mapping is participatory in nature, allowing participants to

have an equal voice and to contribute through various methods. As participants can see the statements of other stakeholders in the sorting and rating stage, the cross-fertilization of ideas is enabled (Napier-Raman et al., 2023). For these reasons, group concept mapping was selected over other consensus methods for the development of the OBtAIN-PD.

A group concept map visually represents all of a group's ideas and how they relate to each other. Depending on the scale, it shows which ideas are more relevant, important, or feasible (Rosas, 2017). Group concept mapping has been widely used in various fields, such as community and public health, social work, health care, human services, biomedical research and evaluation, and complex intervention development (Rosas, 2020).

Group concept mapping is a structured multi-step process. It includes brainstorming, sorting and rating, multidimensional scaling, cluster analysis, and the generation and interpretation of multiple maps (Kane & Trochim, 2007; Rosas, 2017). The first step requires participants to brainstorm a large set of statements relevant to the topic of interest, usually in response to a focus prompt. Participants are then asked to individually sort those statements into categories based on their perceived similarity and rate each statement on one or more scales, such as importance or feasibility.

The data is analyzed using 'Groupwisdom[™]' software (*The Concept System*[®] groupwisdom[™], 2020). This software creates a series of interrelated maps by using multidimensional scaling of the sort data, hierarchical clustering of the multidimensional scaling coordinates applying Ward's method (Ward, 1963), and calculating average ratings for each statement and cluster of statements. The resulting maps display the individual statements in a two-dimensional space. Statements that are more similar are located closer to each other, and

the statements are grouped into clusters that partition the space on the map. The 'Groupwisdom[™]'software also generates other maps that show the statements in each cluster rated on one or more scales, and absolute or relative cluster ratings between two cluster sets. Lastly, participants go through a structured interpretation session to better understand and label all the maps.

The article in this chapter is the Word version accepted for publication in *Journal of Parkinson's Disease* (Lovegrove et al., 2024). The main thesis bibliography includes the full list of references for the article to improve readability. The article uses language in keeping with the journal's style, including US English and terms such as "Parkinson's disease". **3.1** Concept mapping to define components for an occupation-based intervention for Parkinson's & Anxiety

Abstract

Background: Anxiety, a common symptom of Parkinson's disease (PD), results in reduced life quality, reduced participation in meaningful roles and daily activities, and increased health burden. There are no evidence-based interventions to reduce the impact of anxiety in PD on participation.

Objectives: This study aimed to identify the key components required for the co-production of an occupation- and community-based intervention for people with PD-related anxiety.

Methods: A participatory mixed-methods research study was conducted using online Group Concept Mapping methodology that included five stages: brainstorming, idea synthesis, sorting activity, rating activity, and analysis. A cluster map, pattern match, and 'go-zone' charts were created through multivariate statistical analysis based on participants' responses. The stages were guided by questions generated by the research team working with stakeholders.

Results: Eighty-three people participated, with 64 taking part in more than one activity. Participants included people with PD (n=72), care partners (n=6), and occupational therapists (n=5). The final map contained 119 statements with eight clusters (stress value 0.252): exercise, lifestyle changes, self-help, coping, access to information, professional help, peers and groups, support from others. Significant agreement existed between the importance and feasibility rating activities (r = -0.07). 'Go-zone' charts highlighted the priority statements for intervention development.

Conclusions: This novel participatory study highlighted priority components that provide starting points for future development of an occupation- and community-based intervention for people with PD-related anxiety.

Introduction

Parkinson's disease (PD) is the second-most common neurodegenerative condition in the UK, affecting approximately 145,000 people (Parkinson's UK, 2014). In addition to the cardinal motor symptoms of bradykinesia, rigidity, and tremor, PD also results in a wide range of sensory, cognitive, and psychological non-motor symptoms (Kalia & Lang, 2015). These can cause significant disability, interfere with participation in everyday roles and activities, and reduce quality of life (Duncan et al., 2014; Kalia & Lang, 2015).

Anxiety is a non-motor symptom that affects around 40-56% of people with PD (Pontone & Mills, 2021). People with PD fall more than age-matched controls, and risk factors for falling include fear of falling, which is influenced by anxiety (Dissanayaka et al., 2014; Farombi et al., 2016). Reviews indicate no effective pharmacological interventions to reduce anxiety in people with PD (Pontone & Mills, 2021). Psychological interventions for anxiety in non-Parkinsonian populations, recommended in the UK as part of NICE guidelines and the Improving Access to Psychological Therapies (IAPTs) programme (National Collaborating Centre for Mental Health, 2018; NICE, 2011), are cost-effective and have long-term financial and health benefits (Heuzenroeder et al., 2004). However, evidence for the effectiveness of mindfulness and cognitive behavioural therapy approaches is limited and inconclusive in

people with PD (Biundo et al., 2016; Hofmann et al., 2010; Johnsen & Friborg, 2015; McLean et al., 2017). Additional approaches are also warranted, given the shortage of clinical psychologists in the UK (The King's Fund, 2015) and the central role of restricted participation in contributing to increased chronic anxiety (Coventry et al., 2014; CSDH, 2008)[15, 16].

People with PD have expressed a desire for an intervention aimed at living well with the complex experience of anxiety with PD, focused on 'doing' (i.e., participation in meaningful occupation) as both a means and a goal of intervention rather than only talking about thoughts and feelings (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). With their dual training in physical and mental health and expertise in improving wellbeing through participation, occupational therapists are in an ideal position to support people with PD experiencing anxiety (Christine, 1990).

Occupational therapy focuses primarily on enabling participation in a person's chosen meaningful roles and activities, such as getting dressed, cooking a family dinner, or playing the piano in a jazz band (World Federation of Occupational Therapists, 2019). Access to occupational therapy is recommended in clinical guidelines to support symptom management, promote independence, and reduce hospital admissions. Research suggests that occupational therapy focused on the person's prioritized meaningful goals in activities in daily living and participation, is an effective intervention for people with PD (Radder et al., 2016; Sturkenboom et al., 2014). No identified best-practice occupational therapy intervention exists for people with PD with anxiety-related participation problems.

With a high prevalence of anxiety in people with PD, but little research into its management, it is essential to establish a sound scientific basis for developing an intervention based on the Medical Research Council's guidance for complex interventions (Skivington et al., 2021). As a first step in intervention development, this study involved all stakeholders as a precursor to a logic modeling process. It aimed to identify the components needed to co-produce a new occupation-based intervention to help people with PD live well with anxiety.

Methods

Study design

Consensus methods, such as nominal group technique and the Delphi methodology, are commonly used to generate and/or validate the components of a health intervention (Cantrill et al., 1996). However, these approaches struggle to engage diverse groups, particularly rarely heard groups (Cook & Bergeron, 2019). In contrast, Group Concept Mapping (GCM)

"(1) generates group wisdom by creating consensus on key ideas and supports systems thinking by eliciting individual experiences to identify group conceptualization of an issue; (2) equalizes or lessens power differentials among participants; (3) engages patients as both study participants and collaborative decision makers throughout the research process; and (4) includes the availability of specific web-based software to engage participants, provide project management, and generate concept maps" (Cook & Bergeron, 2019) [page 2]

Therefore, GCM was used to identify the components needed for a new occupation-based complex intervention. It involved qualitative and quantitative methods in a structured,

sequential process to organize the ideas of a group about a topic—people with PD living well with anxiety—and involved brainstorming, sorting, and rating (qualitative processes) combined with multivariate statistical analysis (quantitative) (Kane & Trochim, 2007). These ideas are then visually represented using two-dimensional concept maps (Kane & Trochim, 2007). This versatile method can be implemented in paper form or using a secure online platform called 'Groupwisdom[™]' (*The Concept System® groupwisdom[™]*, 2020), enabling participants to contribute at their convenience (Cook & Bergeron, 2019).

Ethical approval was secured from the University of Plymouth Faculty of Health Staff Research Ethics and Integrity Committee (Reference 2473).

Participants and Recruitment

Sample size in GCM studies varies depending on the question and data collection method (Nielsen et al., 2018). The recruitment target of n=40 was based on previous studies that have reported attrition of 10% between the GCM stages and reaching data saturation in stage 5 (Nielsen et al., 2018; Sjodahl Hammarlund et al., 2014; Trochim & McLinden, 2017). The sample was divided into three groups— people with PD, care partners, and occupational therapists (13-14 participants per group)— to avoid potential power differences between group participants. Occupational therapists were included based on feedback from patient and public involvement (PPI) stakeholders who contributed to the study design and felt that it was important to include therapists' perspectives. Twelve weeks were allocated for recruitment.

Purposive sampling was used and was targeted to ensure a diverse range of respondents recruitment materials targeted age (for all groups) and diagnosis duration (for people with

PD). Inclusion criteria for our target sample were people with PD with a formal diagnosis of PD with self-reported Parkinson's-related anxiety, any care partner of a person with PDrelated anxiety, and occupational therapists working with people with PD or with expertise in working with people with anxiety. Formal assessment of anxiety status was not used to avoid missing rich experiential data.

People with PD and care partners were recruited via the Parkinson's UK Research Support Network and the Peninsula Parkinson's Research Interest Group (PenPRIG). Occupational therapists were recruited via the Royal College of Occupational Therapists specialist sections for neurological practice (SSNP) and mental health (SSMH). A member of each organisation acted as a gatekeeper and sent out email advertisements to their distribution lists. The gatekeeper protected the autonomy of prospective participants and allowed them to decide freely whether they wished to participate. The email invitations included a copy of the participant information sheet and the lead researcher's (CL) contact details. Interested participants contacted the lead researcher, who answered any questions presented and then sent a weblink to 'Groupwisdom™', which required participants to complete an informed consent form.

Participants were asked to engage in three out of five stages of the GCM: brainstorming, sorting, and rating (see below). Each stage was open for two to four weeks; an invitation and two email reminders were sent to participants at each stage.

Data collection

There were five specific stages: *brainstorming* ideas in response to the focus prompt, using open text (stage 1), *idea synthesis* by the research team (stage 2), participants *sorting* the

brainstormed statements into groups that made sense to them (stage 3), participants rating each statement on its importance and feasibility (stage 4), and data analysis (stage 5) (Kane & Trochim, 2007; Trochim & McLinden, 2017). Participants could access 'Groupwisdom[™]' at a convenient time and place during each stage and contribute as much data as they wished; no data limiters were used. Participants initially provided demographic information, including their role in the research (i.e., if they were a person with PD, care partner, or occupational therapist), gender, age, and how long they had been diagnosed with PD (if appropriate).

Stage 1: Brainstorming

Participants were asked to generate statements based on their thoughts and experiences in response to the focus prompt *"one thing that would help a person with Parkinson's to live well with anxiety might be…"*. There was a two-week window for participants to complete the brainstorming stage.

Stage 2: Idea Synthesis

The list of the participants' raw statements was exported to a spreadsheet, interrogated, and edited for grammar and readability without changing the wording. Statements expressing multiple ideas were split into single-idea statements; these were the only edits performed. All statements were assigned a keyword generated from the participants' language to express their meaning. Statements with similar keywords were scrutinised together, and duplicates were removed. The result was a final set of refined, unique statements for use in the later stages (Kane & Trochim, 2007).

Stage 3: Sorting Activity

Each statement in the final set was assigned a number and listed in random order on the 'Groupwisdom[™]' software. Each participant was asked to sort statements into groups that were meaningful to them by dragging and dropping them onto a blank workspace in the software. Participants were also asked to give each group a relevant name. For the data to be approved for further analysis, participants must have sorted ≥75% of the statements into meaningful themes (Kane & Trochim, 2007).

Stage 4: Rating Activity

Stage 4 ran simultaneously with stage 3 to reduce participant time burden. They were asked to rate each idea on a five-point ordinal scale for importance and feasibility to facilitate the identification of priority items. Importance ranged from 1 (*'not at all important'*) to 5 (*'very important'*), and feasibility from 1 (*'not realistic at all'*) to 5 (*'very realistic'*). Participants were asked to rate each statement with all the others to ensure that the full range of the scale was used. Similar rating questions have been used in other GCM studies (Murray-Mendes et al., 2021). Both activities were closed after three weeks.

Stage 5: Analysis

Qualitative and quantitative techniques were used to analyse the data. The 'Groupwisdom[™]' software combined the sorted data into a similarity matrix, showing each statement as a position on an X-Y axis indicating how participants sorted statements. The similarity matrix produced a two-dimensional point map using a multidimensional scaling process. The closer points represented statements that were more often grouped together because the participants thought they were conceptually similar (Kane & Trochim, 2007)[27]. The software generated a stress value that indicated the degree to which the distances between each point on the map were representative of the sorting data. Lower

stress values suggest greater stability and goodness-of-fit. Stress values below 0.365 are desirable for GCM research studies (Kane & Trochim, 2007; Trochim & McLinden, 2017).

Hierarchical cluster analysis was applied to the data using Ward's method to group statements with similar meanings into clusters representing how participants grouped their ideas (Ward, 1963). This statistical technique was performed in the 'GroupwisdomTM' software and involves drawing boundaries onto the point map to produce a cluster map. Clusters are successively merged, resulting in several viable cluster maps. A final cluster 'solution' is then selected through qualitative interpretative analysis (Kane & Trochim, 2007). In this research study, the first author (CL) consulted with the rest of the research team (KB, JM, IS) to evaluate maps of six to nine clusters to determine the final cluster solution. This process involved checking whether the statements within each cluster represented the overall theme and comparing the cluster maps to each other to check for similarities and discrepancies between cluster statements and titles.

Cluster labels were suggested by the software for the final cluster solution based on the verbatim statements made by the participants during the sorting activity. The research team refined the cluster labels for grammar and readability, but the wording was unchanged.

The rated data were then organized into a pattern match as a visual representation of the data. This allowed comparison of each cluster's mean importance and feasibility (Kane & Trochim, 2007) and helped highlight priority themes for future intervention development. Individual go-zone graphs were created for each cluster, indicating which statements within the cluster were the most important and feasible. Go-zones are bivariate graphs divided into four quadrants by the mean scores generated for importance and feasibility (Kane &

Trochim, 2007). The upper right quadrant contained the statements that scored above the mean for importance and feasibility. Statements in the 'go-zone' were considered priority statements for intervention development (Appendix 7). These visual representations of the data are intended to aid in the appraisal and subsequent use of the results in a future intervention development study.

Rigor and Validity

The participatory approach of the GCM methodology promoted both the credibility and real-world applicability of the findings by ensuring that the results directly represented the participants' experiences and ideas. A senior researcher (KB) peer reviewed the cluster map to ensure that the headings represented the participants' language.

Data Sharing

The data supporting the results of this study are available from the corresponding author upon reasonable request.

Results

Study population

The final sample consisted of 83 who participated in at least one activity and 64 who participated in more than one activity. The final sample was predominantly female. The recruitment target for people with PD was exceeded (n=72). The care partner (n=6) and occupational therapist (n=5) groups were under-recruited. There was attrition between the study phases (Figure 1) with 38% (n=27) of people with PD, 33% (n=2) of care partners, and 80% (n=4) participating in the final phase.

Figure 3.1 shows how many participants from each sample group took part in each activity. During brainstorming, participants provided 198 ideas in response to the focus prompt which were synthesized into a list of 119 unique statements (Appendix 7). One participant's sorting data was rejected because less than 75% of statements were sorted. Forty-four data sets from the sorting activity were approved for analysis. From the rating activity, one participant's feasibility rating data set was rejected due to rating less than 75% of the statements, with 33 approved for analysis.

Because the participants used the 'Groupwisdom^{TM'} software at different times for the activities open for specified periods, some participants skipped, did not have the opportunity, or chose not to answer the demographic questions (n=12). Participant characteristics of respondents to demographic questions are shown in Table 3.1.

Table 3.1: Participant characteristics of respondents to demographic questions.										
	Characteristic	All participants (n=83)	People with Parkinson's (n=72)	Care Partners (n=6)	Occupational Therapists (n=5)					
	Male	34 (41%)	32 (44%)	1 (17%)	1 (20%)					
	Female	46 (55%)	37 (52%)	5 (83%)	4 (80%)					
er	Other (including transgender)	3 (4%)	3 (4%)	0	0					
send(35-44* years	3 (4%)	1 (1%)	0	2 (40%)					
0	45-64 years	34 (41%)	28 (39%)	3 (50%)	3 (60%)					
	65-74 years	34 (41%)	33 (46%)	1 (17%)	0					
	75 years and above	12 (14%)	10 (14%)	2 (33%)	0					
	Median years PD diagnosis duration (range)	n/a	5 (1.0-21.0)	n/a	n/a					
*No	*No participants younger than this were involved in the study									



Figure 3.1: Line graph summarising participation across each GCM activity.

Concept Map

Multidimensional scaling of the data from all participant groups resulted in a point map with a stress value of 0.252. This score indicates the goodness of fit of the sorting data, i.e., the sample data are likely to represent the data that would be expected in the actual population (Maydeu-Olivares & Garcia-Forero, 2010). After the hierarchical cluster analysis, several cluster maps were generated and reviewed by the research team and a patient and public involvement (PPI) representative. Consensus was reached on an eight-cluster solution with the following cluster names taken directly from the participants' contributions: Access to information, Professional help, Peers and groups, Support from others, Coping, Self-help, Lifestyle changes, and Exercise. The cluster map is shown in Figure 3.2. A point on the map represents each numbered statement (see Appendix 7).



Figure 3.2: Cluster map showing eight priority themes for intervention development. The map shows the distribution of each statement as a numbered point.

Pattern Match

A pattern match compares the eight clusters' mean rating scores for importance and feasibility (Figure 3.3). The clusters are ranked along the vertical lines representing importance (left) and feasibility (right). The pattern match shows that three clusters were rated higher for importance than feasibility (*Professional Help, Peers and groups, Support from others*). The remaining clusters were rated higher for feasibility than importance (*Self-help, Lifestyle changes, Access to information, Coping, Exercise*). The overall correlation between the two rating patterns shows no to negligible relationship between importance and feasibility ratings across clusters (r = -0.07).



Figure 3.3: Pattern match comparing the mean importance and feasibility rating scores at the cluster level (upper limit= 5). Higher mean scores are closer to the pattern match's top (r = -0.07). Importance (n=46); feasibility (n=33).

Go-Zone Graphs and Statements

Go-zone graphs that were generated for each cluster are presented in Figure 3.4. Statements in the go-zones are identified as actionable priorities. Those with high importance and high feasibility were of particular interest, as they are considered priority themes for intervention development. Those with high importance but slightly lower feasibility are areas for future intervention development but are not immediate priorities.



Figure 3.4: Go-zone plots for each of the eight clusters. Statements are shown as numbered points (see appendix 7). The upper right quadrant of each plot represents statements considered priorities for intervention development. The upper left quadrant represents areas for future development. Importance (n=46); feasibility (n=33).

The eight clusters represent a common understanding among participants of the key

components for future intervention development.

"Professional Help"

This cluster was rated as the most important (Figure 3.3). There are 18 statements in this cluster. A common theme within this cluster is increasing clinicians' condition-specific knowledge, such as:

"Understanding how PD anxiety is different from other anxiety." (Statement 102)

Participants felt it was important and feasible for clinicians to understand anxiety in PD, consistent with previous research on clinical expertise in care teams and observational data (Talebi et al., 2023).

"Peers and groups"

This cluster was ranked second for importance but last for feasibility (Figure 3.3) and consists of 11 statements. Access to people and groups with experience of PD-related anxiety was a common theme:

"A matched and paired "Buddy" to discuss developments with" (Statement 57)

"Having a structured support network with a list of who is responsible for what." (Statement

40)

"Support from others"

While superficially like the previous cluster, the 11 statements in this cluster highlight the specific qualities that people with PD would like to see people, in their social context, exhibit to support them in their anxiety:

"Encouragement to share anxiety with partners and close friends." (Statement 92)

"One to one support to improve self-confidence." (Statement 18)

The participants identified this cluster as important and feasible (Figure 3.3) which suggests that educating people in their social context about people with PD living with anxiety should be a core component of the intervention.

"Self help"

The 17 statements in this cluster relate to ways in which people with PD can help themselves to manage their anxiety so that they can live well with anxiety and PD. This cluster was rated fourth in importance and first in feasibility (Figure 3.3), suggesting that the intervention should include a substantial self-help component in order to:

"To simplify your life." (Statement 1)

"Exploring different ways to relax." (Statement 13)

Relaxation techniques that are easy and quick to implement were identified in the go-zone. These findings suggest that relaxation techniques commonly used for anxiety (e.g., box breathing exercises) should be central to the new intervention (Manzoni et al., 2008).

"Coping"

The concept of coping as a strategy for reducing the unpleasant feelings associated with anxiety is embodied in this cluster. The 16 statements in this cluster focused on coping with the emotional impact of anxiety, for example:

"Finding out what triggers your anxiety." (Statement 107)

"Accepting my condition rather than being frustrated by what I can't do." (Statement 111)

Addressing these pressures may provide a viable method of coping with anxiety for people with PD. Several statements in this cluster (statements 45, 108, 111) highlight learning to accept living with a long-term incurable illness as an important and feasible component of the proposed intervention (Figure 3.3).

"Access to information"

Statements in this cluster relate to people with PD not only having access to information about their condition when they need it but also being educated about the condition that they are living with, for example:

"Access to up to date research." (Statement 79)

"Better information and support regarding the later stages of Parkinson's Disease."

(Statement 81)

Statements in this cluster suggest the provision of written information as important and feasible (Figure 3.4).

"Lifestyle changes"

This cluster ranked seventh in importance and second in feasibility (Figure 3.3) and contains 21 statements. The importance of engaging in meaningful activities coupled with lifestyle adjustments (such as improving diet and increasing physical activity) to manage anxiety is woven throughout the cluster, for example:

"Maintaining social interests e.g. sports." (Statement 78)

"Having a balanced diet." (Statement 80)

"Exercise"

The 10 statements in this cluster relate to physical exercise such as dancing (statement 6), yoga (statement 113), and tai chi (statement 8). However, exercise was rated the least important and sixth most feasible for an intervention to support people with PD s with anxiety (Figure 3.3).

Discussion

It could be argued that all clusters generated in this study are essential components of the new intervention, as no clusters or individual statements were rated lower than 3 (slightly important) on importance. The inclusion of feasibility ratings, therefore, highlighted clusters that should be targeted as a priority. Participation in meaningful roles and activities is included in several clusters (*"Exercise", "Lifestyle changes", "Self help"*) but not in clusters such as *"Coping"* and *"Support from others"*. These clusters highlight broader individual and social factors that support participation in meaningful occupation and will need to be considered in intervention development.

"Professional Help" was rated as the most important cluster by participants. People with PD identify a lack of understanding among professionals about PD and PD-related anxiety as a barrier to quality healthcare, which has been reflected in national campaigns in the UK (Parkinson's UK, 2014; Read et al., 2019). This reinforces the notion that there is still much work to be done in this area and that clinician training in PD will be critical to the success of the new intervention. Improving clinicians' understanding of PD-related anxiety would likely strengthen a positive therapeutic relationship (Horton et al., 2021). Establishing positive relationships with healthcare providers, in this case occupational therapists, can support maintaining a degree of normality and identity for people with PD (Read et al., 2019). This improvement in the relationship with healthcare providers may also reduce the 'treatment burden' for people with PD by optimizing treatment to work toward individual goals and minimize disruptions (Tan et al., 2023). This is consistent with research exploring patientcentered management of people with PD (Bloem et al., 2020).

The cluster *"Peers and groups"* highlights that participants consider timely access to peer support important, consistent with research exploring patient-centered management of PD (Bloem et al., 2020). Peer support can enhance social connectedness and help people with PD lead meaningful and satisfying lives. This is reflected in research exploring the lived experience of anxiety in PD and will be an important consideration for the next stage of intervention development (Lovegrove & Bannigan, 2021). Other research has identified that people with PD value support from people who understand what triggers their problems (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). Fostering a 'partnership' between people with PD and their peers will help to create a safe environment for people with PD to share their experiences (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). This
partnership approach has been used in other interventions for long-term conditions (Apostol et al., 2021).

Peer support offers unique benefits from shared lived experiences (Gerritzen et al., 2022). Nevertheless, several potential barriers exist to accessing peer support (Gerritzen et al., 2022). This is possibly reflected in the participant's low feasibility rating of the cluster. Similar findings were highlighted in the *"Support from others"* cluster. This cluster focuses on fostering partnerships with people without PD who understand the triggers of people with PD, which is consistent with studies highlighting the need for emotional support (Uebelacker et al., 2014). It will be important to explore how such barriers to accessing peer support and support from others can be mitigated for individuals developing a new intervention.

"Self help" and *"Coping"* were identified as separate clusters by participants but are closely related. This is evidenced by the physical proximity and similar size of the clusters on the map (Figure 3.2) and similar importance ratings (Figure 3.3). Both clusters include selfmanagement statements. Self-management strategies are effective in PD, but many focus primarily on motor symptoms (Milne-lves et al., 2022). Although this may not seem applicable to a new intervention focusing on anxiety, several studies report that motor symptoms are closely related to anxiety in people with PD (Blundell et al., 2023; Lovegrove & Bannigan, 2021). Time pressure has been associated with increased anxiety, increased motor symptoms such as freezing of gait, and decreased quality of life in PD (Duncan et al., 2014; Lovegrove & Bannigan, 2021). Self-management techniques can provide a sense of empowerment through mechanisms such as participation in decision-making (Kang et al., 2022). Including self-management techniques addressing anxiety related to motor

symptoms, and vice versa, in a new intervention may be valuable. While people with PD have expressed that they do not want another psychological therapy, a new intervention based on a collaborative relationship with a healthcare professional may include psychological therapy concepts (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). Acceptance and commitment therapy is a form of psychotherapy that enables people to focus on the present moment and accept their thoughts and feelings without judgment; it can be included under the umbrella term 'talking therapies' (Graham et al., 2016). This therapy has promising evidence of initial effectiveness in the self-management of long-term conditions including PD and provides an underlying concept for intervention development (Graham et al., 2016).

People with PD use fewer coping strategies than people with other long-term conditions (Costa et al., 2016; Pusswald et al., 2012). Many tend to use "self-control" (Costa et al., 2016; Pusswald et al., 2012). Women with PD tend to have more problem-focused coping strategies than men, but people with PD generally use both problem-focused and emotionfocused coping strategies (Anzaldi & Shifren, 2019). In addition, cognitive impairments, common in PD, are associated with decreased use of task-related coping (Fang et al., 2020; Hurt et al., 2012). Feelings of shame related to having PD can negatively affect coping and self-management and exacerbate anxiety symptoms (Angulo et al., 2019). Any new intervention will need to incorporate tailoring and individualization of treatment to account for these variations in coping strategies and the impact of cognitive impairment and emotional state (Ambrosio et al., 2014).

Participants rated having "Access to information" as a feasible component to include in the new intervention. This is consistent with previous research highlighting this need for

education (McLaughlin et al., 2011; Uebelacker et al., 2014). Disease-specific education interventions have evidenced improvement in people with PD quality of life, potentially through patient empowerment (Rajan et al., 2020). Ensuring that patient education is delivered in a way that is both culturally and individually specific will be crucial to the successful delivery of the intervention and will likely involve a range of media (Friedman et al., 2011). Any information provided in a new intervention will need to be concise, approachable, and non-threatening to promote engagement, particularly as people with PD report "wanting to know but not wanting to know" (Read et al., 2022).

"Lifestyle changes" were similarly highlighted as feasible by participants, including maintaining social interests, engaging with hobbies, and integrating daily exercise and physical activity (Figure 3.4). The value of using preventative lifestyle programmes has been demonstrated to be a cost-effective way to improve health and wellbeing in myriad studies, including occupational therapy and provides a concept for future intervention (Clark et al., 2012). Similar concepts have been used in other lifestyle management interventions (Forsyth et al., 2015; Piotrowski et al., 2021). Participants also reported that maintaining a healthy diet is both important and feasible for living well with anxiety, which was also reported in other studies (Piotrowski et al., 2021; Pyatak et al., 2022). This links closely with the cluster *"Exercise"*.

Exercise was rated the least important and sixth most feasible for an intervention supporting people with anxiety with PD (Figure 3.3). As previously discussed, this may reflect a lack of understanding and education about PD. Exercise is an evidence-based intervention in PD that has demonstrated benefits on the motor symptoms of PD, but there is only emerging evidence on the effect on non-motor symptoms such as anxiety (Ng et al., 2021). Exercise has been shown to decrease anxiety in people with PD and is not restricted to a specific type of exercise (Ferreira et al., 2018; Kwok et al., 2019). Despite this, many barriers prevent sustained engagement in exercise and physical activity despite people with PD recognising its importance (Ellis et al., 2013; Hunter et al., 2019). Given the benefits of exercise on quality of life and anxiety in non-Parkinsonian populations, incorporating regular exercise into the daily routine may be a helpful component of the new intervention (Kandola & Stubbs, 2020). The new intervention will likely need to integrate these wider lifestyle components to promote sustained engagement in meaningful roles and activities to promote living well with anxiety.

In the next stage of intervention development, we will consider to what extent and in what way the key components can be included in a community- and occupation-based occupational therapy intervention. This will require a suitable method to meaningfully involve stakeholders (people with PD, care partners, and occupational therapists) in a coproduction process.

Strengths and Limitations

A strength of the GCM methodology is its structured and systematic procedure to data collection that places the voice of the participants, most of whom were people with PD with anxiety, at the center. The eight clusters portray what the participants feel is important to include in a new intervention to help people with PD to live well with anxiety. In addition, the large number of statements suggests a range of thoughts among the participants, not just those of dominant group members. Online delivery using the 'Groupwisdom[™] platform allowed participants to contribute to the study at their convenience, thus facilitating participation.

This study is not without limitations. The sample size exceeded our target (n=40) and was heavily weighted towards people with PD (n=72). Attrition occurred despite the research team's efforts to maintain study participation, such as automated email and text message reminders when requested by participants. This likely resulted in additional attrition bias that skewed the results and compromised the study's external validity.

Anxiety tends to occur earlier in PD and may be an early symptom of neurodegeneration (Seritan et al., 2019). This is reflected in the average disease duration of the participants (median 5 years). Evidence suggests that people living with young-onset PD (YOPD, diagnosed before 50 years) are more likely to experience anxiety than people with a lateonset of the disease (Dissanayaka et al., 2010). It would be reasonable to expect this to be reflected in the characteristics of the participants. However, we can only be confident that 1% of participants with PD meet the YOPD criterion (Table 3.1). The next age group (45-64) accounted for 39% of the sample. As these age groups are based on the UK census, we cannot determine how many participants would be considered YOPD. Although our sample would still represent those aged 65 and above (60%), it would be helpful to use adjusted age groups in future research to distinguish between early and late-onset PD.

Small numbers of care partners and occupational therapists participated compared to people with PD. Thus, we cannot be confident that our findings represent all the groups included in this GCM study as they are likely weighted towards the views and experiences of people with PD. The under-recruitment in both care partner and occupational therapist groups may partly be attributed to the COVID-19 pandemic, which has had a noted impact on non-COVID-19 research due to increased service demands (Bratan et al., 2021; Park et

al., 2021). Our sample was predominantly female, including for the group of people with PD. This finding was unexpected, particularly as PD tends to be more prevalent in men and men are over-represented in most research (Baldereschi et al., 2000). Resultantly, our findings may not fully represent the views of male people with PD, care partners, or occupational therapists. Future research will likely need to change its recruitment strategy and sampling frame to address this imbalance.

Moreover, GCM does not claim to be a definitive methodology, so studies using different methods may be needed. Although accounted for in the design of this study and the GCM methodology, the loss of participants between stages means that potential data was lost. Finally, we only collected basic demographic data to identify gender, age, and diagnosis duration (if applicable). As we lack data on areas such as cultural background and socioeconomic status, we cannot be confident any one group was not over-represented, limiting the external validity of this study to the UK population, including those from seldom-heard groups.

While digital delivery provided a convenient means of conducting this research, the impact on potential participant engagement in this study cannot be ignored, as it is likely that those who are less digitally literate will not have participated (Office for National Statistics, 2019).

Future research directions

The findings highlight eight priority components, presented as clusters, for consideration in the future intervention. Individual ideas from within the clusters rated highly on importance and feasibility, are suggested as starting points for future intervention development. Clusters, such as "*Coping*" and "*Support from others*" highlight broader individual and social

factors that support participation in meaningful roles and activities and need to be considered in intervention development. The ability to access timely support from peers and the development of individual coping strategies will be important considerations for the next stage of intervention development (Lovegrove & Bannigan, 2021). Future research, such as co-producing the intervention with stakeholders, is needed to integrate the components identified in this study into a new intervention to help people with PD to live well with anxiety. Future studies should seek to remove barriers to research participation for care partners and occupational therapists, as providing convenient data collection is unlikely to be sufficient.

Conclusion

This novel participatory study used an online GCM methodology to ask people with PD, care partners, and occupational therapists to highlight priority components that provide starting points for the future development of an occupation- and community-based intervention for people with PD-related anxiety. The eight clusters identified in the findings are exercise, lifestyle changes, self-help, coping, access to information, professional help, peers and groups, and support from others. Statements in each cluster rated highly on importance and feasibility will be prioritized in intervention development. We plan to use these findings to co-produce an occupation-based intervention for living well with anxiety in PD.

Chapter 4: Logic Modelling Study

The scoping review (Chapter 2) and group concept mapping study (Chapter 3) both contribute to the 'development' stage of the MRC framework (Figure 1.3). Before moving on to the feasibility phase, the final step is modelling processes and outcomes (Skivington et al., 2021). This chapter presents the findings of a logic modelling study where the findings of the previous chapters (and pre-PhD research) are used as 'inputs' for the systematic coproduction of the OBtAIN-PD. This provides the basis for the next study in which this intervention will be tested for feasibility.

The article in this chapter is the Word version submitted for publication in *BMJ Open* (Lovegrove, Bannigan, Marsden, et al., in press). To improve readability, the full list of references is included in the main thesis bibliography. The article uses language in keeping with the journal's style, such as the terms "Parkinson's disease" and the abbreviation "PD."

4.1 Developing an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD) using logic modelling

Background: Anxiety is a common non-motor symptom of Parkinson's disease (PD). There is no specific pharmacological intervention for people with PD who experience anxiety. Current non-pharmacological treatments have mixed or inconclusive results. Based on the contemporary published literature, there does not appear to be a non-pharmacological intervention for people with PD and anxiety that focuses on activity and participation. **Objective:** To co-produce an occupation-focused complex intervention suitable for delivery by occupational therapists to help people with PD live well with anxiety.

Design: Six-stage complex intervention development using logic modelling and a participatory approach to organise the new intervention's key inputs, processes, and outcomes.

Participants: 34 participants were recruited (People with PD n=14, care partners n=9, occupational therapists n=11) for the online logic modelling sessions.

Key Results: Resources to support the new intervention ('Inputs') include adequate resourcing, education for professionals and people with PD, flexibility of delivery methods, and goal setting. The intervention's actions to produce outcomes ('Processes') should include 1:1 support, lifestyle management, providing meaningful information, collaborative goal setting, therapeutic use of everyday activities, and involvement of friends and families. The intended results ('Outcomes') should include reductions in anxiety symptoms, people with PD enjoying more meaningful activities, increased understanding of anxiety and PD, improvement in clinical outcomes, and improvement of service level outcomes. These key aspects were incorporated into an intervention manual, educational material, and training video.

Conclusion: We have systematically co-produced a new occupation-focused complex intervention to help people with PD to live well with anxiety. This provides the basis for the next project in which this intervention will be tested for feasibility.

Keywords: Parkinson's disease; health care stakeholder engagement, knowledge coproduction

- What is already known on this topic: Anxiety is a common non-motor symptom of Parkinson's disease. Current pharmacological and non-pharmacological interventions for this symptom have inconclusive or mixed results on anxiety for people living with Parkinson's disease.
- What this study adds: A novel complex intervention has been developed that is focused on activity and participation to help people with Parkinson's disease live well with anxiety. This study involves people with Parkinson's, care partners, and occupational therapists in a multistakeholder co-production.
- How this study might affect research, practice or policy: This new intervention provides the basis for a feasibility randomised controlled trial, leading to new insights into this intervention's acceptability and potential impact for this population.

Introduction

Parkinson's disease (PD) is the second most common neurodegenerative condition worldwide (de Lau & Breteler, 2006). Globally, the prevalence of PD has doubled in the past 25 years and now affects more than 10 million (Dorsey, Sherer, et al., 2018). Recent estimates suggest that PD resulted in 5.8 million disability adjusted life years (DALYs) in 2019 (G.B.D. Neurology Collaborators, 2019). Most people with PD (98.6%) live with nonmotor symptoms (Barone et al., 2009; Pontone et al., 2019). Between 43-56% of PWPs experience anxiety (Barone et al., 2009; Pontone et al., 2019). Reviews have found a higher prevalence of anxiety in PWPs compared to the general population (Walsh & Bennett, 2001). People with Parkinson's who experience anxiety have a greater risk of falling, live with greater health burden, and have reduced quality of life, independence, and social roles (Dissanayaka et al., 2014; Fineberg et al., 2013). Heightened anxiety in PWPs leads to increased fluctuations in motor symptom presentation (Ehgoetz Martens et al., 2014; Fernie et al., 2019; Pontone et al., 2019).

Anxiety is a continual internal feeling of worry and fear that intrudes on everyday life (Hallion & Ruscio, 2013). People with PD experience more anxiety than those with other neurodegenerative conditions and fear negative public perceptions (Walsh & Bennett, 2001). People with PD may be more at risk of anxiety than those with other long-term conditions due to the dopamine deficiency that is characteristic of PD (Kienast et al., 2008). Dopamine modulates the inhibitory mechanisms that the medial prefrontal cortex exerts on the anxiogenic output of the amygdala, a brain structure involved in anxiety (Pontone & Mills, 2021; Zarrindast & Khakpai, 2015). Dopamine deficiency leads to neuronal hyperexcitability and amplified responses to perceived adverse threats (Hofmann et al., 2012; Kano et al., 2011; Kienast et al., 2008). Dopamine-replacement medication is a primary

intervention in PD, but people with PD in the later stages can experience a notable increase in symptoms when the medication effects wear between doses (Richard et al., 2004). Psychological stressors associated with long-term conditions experienced by people with PD may increase anxiety further (Delgadillo et al., 2017). This contributes to the maintenance of the hyper-excited neuronal anxiety circuit (Hofmann et al., 2012; Kano et al., 2011). This suggests that people with PD experience anxiety differently from other populations because of the neurobiology specific to PD combined with the psychological stressors of living with an incurable long-term condition. This is important for clinicians and researchers to consider when developing complex interventions, especially as there is no specific pharmacological intervention for PWPs experiencing anxiety (Pontone & Mills, 2021; Skivington et al., 2021).

The presence of anxiety in PD has a negative effect on PD motor symptoms. Anxiety has been shown to increase the freezing of gait (FOG) episodes in 'freezers' and higher motor symptom severity when compared to 'non-freezers', implying that anxiety is an important mechanism underpinning FOG (Ehgoetz Martens et al., 2014). This means that if people with PD are anxious, this will affect their functional mobility in daily life and consequently reduce their quality of life (Ehgoetz Martens et al., 2014; Hanna & Cronin-Golomb, 2012; Pontone et al., 2019). Cognitive impairment associated with anxiety impacts quality of life (Dissanayaka et al., 2017; Hanna & Cronin-Golomb, 2012). Other potential sources of stress that contribute to anxiety for all people, not just people with PD, include (but are not limited to) health, family, finances, and work, all of which can change over the life course (Kessler et al., 2007). Research suggests that anxiety is not experienced in isolation but as a set of complex multi-dimensional interactions with the world that may require a bespoke intervention (Blundell et al., 2023; Lovegrove & Bannigan, 2021).

The charity Parkinson's UK identifies anxiety as a research priority (Deane et al., 2014). Previous studies have contributed to understanding the experience of anxiety for people with PD and what sorts of concepts may be included in a new complex intervention (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017; Lovegrove et al., 2023; Lovegrove, Sturkenboom, et al., in press). People with PD have stated that they want an intervention aimed at living well with the complex experience of anxiety with PD, focused on 'doing' (i.e., participation in meaningful occupation) as the means and goal of intervention rather than only talking about their thoughts and feelings (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017).

Occupational therapy focuses on enabling people with PD to participate in their chosen meaningful roles and activities (World Federation of Occupational Therapists, 2019). Access to occupational therapy is recommended in clinical guidelines to support symptom management, promote independence, and reduce hospital admissions (NICE, 2018). Research suggests that occupational therapy focused on the person's prioritised meaningful goals in activities in daily living and participation is an effective intervention for people with PD (Radder et al., 2016; Sturkenboom et al., 2014) There is no occupational therapy intervention specific for people with PD with anxiety-related participation problems.

This study aimed to systematically co-produce an intervention that occupational therapists can deliver to help people with PD live well with anxiety.

Materials and method

We used the Medical Research Council's guidance for developing and evaluating complex interventions as a framework (Skivington et al., 2021). This study focused on the

development stage. We used the method of logic modelling and a participatory approach to organise the new intervention's key inputs, processes, and outcomes for intervention development.

A logic model is a hypothesised description of the chain of causes and effects that result in an outcome of interest. In this research, the end outcome is enabling people with PD to live well with anxiety by increasing their participation in meaningful activities. Logic modelling can involve representative stakeholders in a structured process to support complex interventions' design, planning, communication, and evaluation (Moore et al., 2019a). There were six stages in developing the occupation-based complex intervention for living well with anxiety and Parkinson's disease (OBtAIN-PD) using the logic model (Sermeus, 2015).

Stage 1: Installing a project team and formulating key objectives

The core project team consisted of the research team (CL, KB, JM, IS) and a patient & public involvement (PPI) representative (NC).

Stages 2 and 3: Component consensus and clustering of clinical activities into a process flow Clinical activities, characteristics, and group consensus identified in previous research were used as input for modelling (Lovegrove & Bannigan, 2021; Lovegrove et al., 2023; Lovegrove, Sturkenboom, et al., in press). Participants were given an electronic or printed information pack (depending on participant choice) to inform their thinking. The information pack contained summaries of the findings of previously completed studies by the research team (Lovegrove & Bannigan, 2021; Lovegrove et al., 2023; Lovegrove et al., 2024) (Appendix 8). The information pack had been refined in collaboration with PPI representatives.

Stages 2 and 3 co-occurred within the same session to reduce the time burden on participants attending multiple sessions. A researcher (CL) facilitated the logic modelling process, starting with long-term outcomes and moving backward towards inputs. During the sessions, the participants were invited to group clinical activities based on their coherence (stage 2). The participant's ideas were mapped directly into a process flow diagram to illustrate the interdependency of the individual components by participants (stage 3). Participants were allowed to complete sessions in person or via an online platform (Zoom) ("CYPH, INC. v. ZOOM VIDEO COMMUNICATIONS, INC," 2022). Sessions were recorded as it was likely that the planned outputs (logic models) produced by the groups would have variations requiring further analysis by the research team. Individual logic models were completed at each group session and then combined into an overarching model by the research team with PPI involvement (Schell et al., 2013; Sermeus, 2015; Skivington et al., 2021). The blank logic model that was used with participants in each session is shown in Figure 4.1.



Figure 4.1: The blank logic model used with participants to co-produce the OBtAIN-PD.

Participants who felt unable to participate in a group session were offered a one-to-one session. All sessions were completed at a time, date, and in a setting convenient for the participants. The people with PD, care partners, and occupational therapists met separately to avoid potential bias caused by power dynamics (Robinson, 2014). There was no set order for the different participant sessions to be completed, as each session produced an individual logic model.

Stage 4: Process organisation

Stage 3 resulted in fourteen separate logic models. These logic models were aggregated to create a unified, more robust model. Three focus groups, one for each participant group, were conducted by the lead researcher (CL) to review the existing models to identify common elements and differences (Mills et al., 2019) through discussion of each component (i.e. the inputs, outputs, and outcomes). Each participant was sent a copy of the

logic models to familiarise themselves with them before the focus group session. This process resulted in three separate logic models. The lead researcher developed a draft consolidated logic model using the data from the three models in collaboration with the research team. This process involved combining each model into a final version, considering individual assumptions and potential external factors. The draft consolidated logic model was presented to participants in a focus group for validation attended by people from each participant group (n=8). Feedback was gathered, and adjustments were made during the session to reflect the group's collective input accurately. The final consolidated logic model was checked with two National Health Service (NHS) community therapy service managers to evaluate the organisational impact of the intervention. This took the form of a researcher (CL) presenting the logic model to the managers and providing the opportunity for them to comment on its feasibility and implementation within an NHS setting. This stage was conducted remotely for convenience and is presented in Figure 4.2. This information was used in subsequent stages.



Figure 4.2. Process of aggregating the multiple logic models to develop the OBtAIN-PD.

Stage 5: Detailed description of key interventions

A detailed protocol for the OBtAIN-PD intervention was written based on the data collected from stages 2-4. Individual components- the activities, inputs, and outcomes identified by participants- were reconsidered (Sermeus, 2015). This was completed by the research team with input from PPI representatives.

Stage 6: Translation into a set of process and outcome indicators

Process and outcome indicators were developed to help verify compliance with the intervention during fidelity testing (Berry et al., 2009; Nolan & Berwick, 2006). This was completed by the research team with PPI input and took the form of an intervention manual. A final review with the occupational therapist group was performed to check the intervention manual's content and comprehension and inform a final version.

The UK Health Research Authority granted ethical approval for this study (22/NW/0021).

Participants

The co-production was with all potential end users: people with PD, care partners, and occupational therapists. The recruitment target was n=30, with the sample split equally between people with PD, care partners, and occupational therapists (10 participants per group). Twelve weeks were allocated for recruitment.

People with PD and care partners were recruited purposively using a maximum variation strategy to maximise sample diversity across the stages of disease progression: diagnosis, maintenance, complex, and palliative stages (MacMahon & Thomas, 1998; Robinson, 2014). Snowball sampling helped access participants from under-represented groups who use health or social care services but are seldom heard by these services and decision-makers (Emerson, 2015). We collected demographic data based on the UK Census to verify participant diversity (Office for National Statistics, 2020). Both groups were recruited via Parkinson's UK's 'Research Support Network'.

Occupational therapists were primarily recruited via the Royal College of Occupational Therapists Specialist Sections for 'Neurological Practice' and 'Mental Health'. In addition, snowball sampling was used to recruit occupational therapists who may not be part of these groups but may have relevant experience in either PD or anxiety-related OT interventions (Emerson, 2015).

Results

The final sample was n=34. The people with PD and care partners participated in *stages 2* and *3*. Occupational therapists participated in *stages 2, 3,* and *6*. The recruitment target of 10 in each stakeholder group was exceeded for people with PD (n=14) and occupational therapists (n=11) and below the target for care partners (n=9). Sixteen participants (people with PD n=6, care partners n=5, occupational therapists n=5) consolidated the raw logic model. The demographic data of the participants are shown in Table 4.1. There was notably less gender diversity in the occupational therapist group.

Table 4.1: Partic	<i>le 4.1:</i> Participants demographic characteristics.				
Characteristic		All participants (n=34)	People with Parkinson's (n=14)	Care Partners (n=9)	Occupational Therapists (n=11)
Gender	Male (including transgender men)	11 (32%)	6 (43%)	4 (44%)	1 (9%)
	Female (including transgender women)	22 (65%)	7 (50%)	5 (56%)	10 (91%)
	Prefer not to say	1 (3%)	1 (7%)	0	0
łge	25-34 years*	4 (12%)	0	1 (11%)	3 (27%)
	35-44 years	13 (38%)	3 (21%)	5 (56%)	5 (45%)
	45-64 years	9 (26%)	5 (36%)	1 (11%)	3 (27%)
-	65-74 years	6 (18%)	5 (36%)	1 (11%)	0
	75 years and above	2 (6%)	1 (7%)	1 (11%)	0
	Median years PD diagnosis duration (range)	N/A	5 (1-22)	N/A	N/A
	Working as an employee	19 (56%)	5 (36%)	4 (44%)	10 (91%)
ent	Self-employed or freelance	3 (9%)	2 (14%)	1 (11%)	0
oym	Retired	9 (26%)	6 (43%)	3 (21%)	0
Emplo	Long-term sick or disabled	2 (6%)	1 (7%)	1 (11%)	0
	On maternity or paternity leave	1 (3%)	0	0	1 (9%)
	£20,001 to £30,000	3 (9%)	0	1 (11%)	2 (18%)
me	£30,001 to £40,000	4 (12%)	1 (7%)	1 (11%)	2 (18%)
nco 3P)	£40,001 to £50,000	7 (20%)	2 (14%)	1 (11%)	4 (36%)
al ir EGB	£50,001 to £60,000	1 (3%)	1 (7%)	0	0
) יעער	£70, 001 to £80,000	2 (6%)	1 (7%)	0	1 (9%)
A	Prefer not to say	17 (50%)	9 (64%)	6 (67%)	2 (18%)
_	Single/ never married	5 (15%)	2 (14%)	3 (33%)	0
/arital status	Married/ domestic partnership	23 (67%)	10 (71%)	4 (44%)	9 (82%)
E	Divorced	6 (18%)	2 (14%)	2 (22%)	2 (18%)
× ÷	Straight or Heterosexual	30 (88%)	12 (86%)	7 (78%)	11 (100%)
Sex Ori	Gay or Lesbian	4 (12%)	2 (14%)	2 (22%)	0
	No religion	17 (50%)	5 (36%)	5 (56%)	7 (64%)
ц.	Christian	11 (32%)	5 (36%)	2 (22%)	4 (36%)
ligid	Jewish	3 (9%)	2 (14%)	1 (11%)	0
Rel	Muslim	2 (6%)	1 (7%)	1 (11%)	0
	Any other (not described)	1 (3%)	1 (7%)	0	0
	White- English, Welsh, Scottish, Nothern Irish, or British	27 (79%)	9 (64%)	7 (78%)	11 (100%)
icity	Asian or Asian British	2 (6%)	1 (7%)	1 (11%)	0
Ethnic	Black, African, Caribbean, or Black British- African	3 (9%)	2 (14%)	1 (11%)	0
	Other ethnic group- Arab	1 (3%)	1 (7%)	0	0
	Any other ethnic group	1 (3%)	1 (7%)	0	0
*No participants	s younger than this were involved	in the study			

Inputs for the intervention

Participants identified the resources needed to support the intervention and generated 58 raw statements. These were consolidated into five inputs: "Resources", i.e the need for proper funding and staffing, physical resources, and time; "Education for professionals" about PD; "Education for people with Parkinson's" about anxiety and how it affects PD; "Flexible delivery" of the intervention in a variety of settings, including in-person and remotely and by occupational therapists; and "Goal setting" by participants from each group.

Processes of the intervention

The processes included what needs to be done as part of the intervention (activities) and what and who should be targeted by the intervention (participation). Participants from each group generated 103 raw statements identifying the processes and activities they felt should be included in the new intervention. Participants from each group consolidated these statements into six aspects: "1:1 support first" meaning 1:1 support should be prioritised over group therapy; "Lifestyle changes" towards a healthier lifestyle to live well with anxiety;, "Providing meaningful information" that is concise and up-to-date, "Collaborative goal setting" rather than the therapist leading, "Use everyday activities", i.e. using the person's meaningful activities therapeutically (rather than abstract exercises or talking therapy); and "Involve friends and families" (where needed).

Outcomes of the interventions

Participants from each group generated 153 total statements on the short, medium and long-term outcomes they felt the new intervention should produce. Participants from each group consolidated these statements into "Reduce anxiety symptoms" on validated

measures; the person with PD "Enjoying more meaningful activities" to them; "Better understanding" of one's anxiety and how to manage it; "Improvement of clinical outcomes" using validated occupational therapy measures; and "Service level outcomes" such as reductions in new or re-referrals into services.

The consolidated logic model is presented in Figure 4.3. Illustrative verbatim quotes are shown in Table 4.2.



Figure 4.3. Consolidated logic model for the occupation-based complex intervention for

living well with anxiety and Parkinson's (OBtAIN-PD).

Table 4.2: Illustrative quotes from the consolidated logic model.					
Inputs	Resources	"Staffing availability as a resource. Physical resources; easy access to equipment, services, groups." (OT)			
	Education for professionals	"We need more people with more knowledge. Help in more areas. Increased knowledge amongst those people who have most contact with PWPs." (person with PD)			
	Education for people with Parkinson's	"Education. Information that anxiety makes the physical symptoms of Parkinson's worse. Acknowledging it is okay to be anxious." (person with PD)			
	Flexible delivery	"There needs to be options for delivery. Some people don't have online access, some do." (care partner)			
	Goal setting	"Goal setting resources will be important." (care partner)			
Processes	1:1 support first	"1:1 basis for the sessions to be delivered." (person with PD)			
	Lifestyle changes	"Developing understanding between the links between lifestyle and anxiety. Understanding that you won't get everyone. You need to see the benefits/ feel-good factor of what you are doing. Developing a positive snowball effect. Using real-life examples to demonstrate benefits." (person with PD)			
	Providing meaningful information	"Working with the persons partner to help the PWP do things so they do not become inactive, and building up the confidence of the carer." (person with PD)			
	Collaborative goal setting	"Goal setting- SMART goals used in o/p service- this works as a 'contract' as well between the service and patient." (OT)			
	Use everyday activities	"The intervention has to focus on what is important to the person. Everyone is individual." (care partner)			
	Involve friends and families	"Working with the persons partner to help the PWP do things so they do not become inactive, and building up the confidence of the carer." (person with PD)			
Outcomes (Short, Medium, and Long)	Reduce anxiety symptoms	"Really important to measure anxiety symptoms, important to show the person that it has reduced and I have a reading to show that." (OT)			
	Enjoying more meaningful activities	"Participating more in activities, and hobbies, that they enjoy doing." (care partner)			
	Better understanding	"Have understanding of what anxiety is, had an opportunity to practise some coping strategies and then call on those a year down the line. The carers would be able to draw on these skills as well, and be able to draw on support when needed and not start at the beginning of the process." (person with PD)			
	Improvement of clinical outcomes	"COPM improvement- includes goals and breaks them down in a really specific way. Not a massive additional time constraint." (OT)			
	Service level outcomes	"No referral back to the service at the 6 month timepoint- may have actually referred on to another service (long term conditions counselling team)." (OT)			

Process flow

The consolidated logic model was presented to three NHS managers leading community rehabilitation teams. They felt the proposed model was feasible within current practice, and no further changes were suggested.

Intervention manual

Following the review with NHS managers, the research team drafted an intervention manual, information sheets, and collaborative goal-planning document, all with PPI input and based on the consolidated logic model. A review was completed with occupational therapists from the sample (n=5) to check the manual content and readability. Based on feedback, a training video was developed to accompany the intervention manual (Appendix 9).

Discussion

This study aimed to systematically co-produce an intervention that occupational therapists can deliver to help people with PD live well with anxiety using a logic modelling approach (Sermeus, 2015). The Medical Research Council's guidance for complex interventions guided the intervention development by providing a framework for the whole project, in which this study focused specifically on the development stage (Skivington et al., 2021). As primary stakeholders, people with PD, carers, and occupational therapists contributed to coproducing a logic model for the OBtAIN-PD. This logic model was checked with NHS managers. The resulting intervention manual was refined with occupational therapists.

All participant groups stated that having access to adequate resources, such as time, staffing, or physical materials, was necessary for the OBtAIN-PD. Resource limitations are

associated with changes in care delivery and poorer clinical outcomes in some cases (Anesi & Kerlin, 2021). Lack of resources is an identified barrier to intervention implementation in PD and other neurological conditions (Hanssen et al., 2021; Laar et al., 2023). The NHS is recognised as a resource-constrained environment, especially community health services (Iliffe et al., 2021). The OBtAIN-PD must have a low-resource burden to be successfully implemented using only existing resources (Gesell et al., 2021; Ramaswamy et al., 2018). This may include using existing outcome measures, minimising the impact on clinicians' time, and providing flexibility on how the intervention can be delivered (Mathieson et al., 2019). For example, all participants highlighted providing adequate education to people with PD and occupational therapists, as necessary. Training and education provision can be a time-consuming endeavour if not delivered in a way that is appropriate to the context, thus introducing an implementation barrier (Mathieson et al., 2019). Any training related to the new intervention, either for the people with PD receiving it or the occupational therapists giving it, must be concise and convenient, such as e-learning (Wearne et al., 2011). The occupational therapists and managers felt that e-learning would be an appropriate training method, but face-to-face delivery would be preferred if the intervention included highly advanced concepts. There was no indication from the group of how much training would be required.

The participants of this study agreed with the concept of flexibility in the OBtAIN-PD. Previous research has identified flexibility in how interventions can be delivered as a facilitator for complex interventions, as community-based clinicians are responsible for their caseload organisation (Mathieson et al., 2019). Ensuring that the intervention can be delivered flexibly but with fidelity (by providing an intervention manual), supporting occupational therapists' caseload demands, and the needs of people with PD will facilitate the successful implementation of OBtAIN-PD.

The occupational therapists did not have a preference between group or individual intervention sessions, which is interesting given the benefits of group therapy in managing service demands. People with PD and care partners who participated in this study identified that OBtAIN-PD should be delivered primarily as a tailored individual intervention rather than in a group setting. Previous studies have reflected on the treatment preferences of people with PD (Lovegrove et al., 2017; Uebelacker et al., 2014; Weernink et al., 2017). This suggests that the OBtAIN-PD should be delivered as a tailored individual intervention.

Participants from all groups felt that including lifestyle changes and management would be necessary in the OBtAIN-PD and that initiating lifestyle changes would help them to live well with anxiety by reducing their anxiety symptoms. Preliminary research exists on lifestyle management approaches for people with PD, and more research is needed (Lister, 2020; Nag & Jelinek, 2019). Lifestyle management approaches focus on educating and motivating people to improve their quality of life through healthier habits and everyday behaviours (Clark et al., 2012). Lifestyle management approaches have been utilised in various contexts for supporting engagement in meaningful activity (Clark et al., 2012; Pyatak et al., 2022). Similar approaches have been used in occupational therapy interventions, including for adults with generalised anxiety disorder (Clark et al., 2012; Lambert et al., 2008; Lambert et al., 2007). This approach aligns with our study participants' opinion that the OBtAIN-PD should include providing meaningful information, collaborative goal setting, and the therapeutic use of everyday activity. These concepts are pillars of lifestyle management approaches (Clark et al., 2012; Pyatak et al., 2022). Furthermore, lifestyle management

approaches for long-term conditions often involve collaboration with care partners such as friends and families (Frost et al., 2020; Pyatak et al., 2022). The fact that lifestyle management approaches resonate closely with our participant's opinions affirms the inclusion of these concepts in the OBtAIN-PD.

Outcome measure selection is essential for comprehensively evaluating complex interventions (Skivington et al., 2021). Establishing what outcomes an intervention aims to achieve allows the selection of appropriate outcome measurement tools (Datta & Petticrew, 2013). Participants from all groups stated that anxiety reduction is an essential outcome of the intervention and that they felt this was part of 'living well' with anxiety. The occupational therapy group reported that the Generalised Anxiety Disorder Assessment (GAD-7) would be a sensible tool to measure this as it is already established in clinical practice. The GAD-7 is a valid and reliable 7-item instrument used to assess the severity of generalised anxiety disorder and is commonly used in the NHS (Ruiz et al., 2011; Spitzer et al., 2006). As well as being valid and reliable, the GAD-7 is quick and easy to administer, which is an important facilitator (Mathieson et al., 2019). For these reasons, the GAD-7 would make it an appropriate measure for the OBtAIN-PD. People with PD and care partners in this study stated that enjoying more meaningful activities was an important outcome for them. The occupational therapists report that improvement in clinical measures is an essential outcome for the OBtAIN-PD, and multiple suggested using the Canadian Occupational Performance Measure (COPM). The COPM is a validated measure of a person's self-perception of performance in everyday living (Chesworth et al., 2002; Law et al., 1994; Tuntland et al., 2016). It has been recommended for people with PD (Bianchini et al., 2022; Kobayashi et al., 2023). The COPM has been used as a primary outcome measure in clinical trials of people with neurological conditions, including Parkinson's (Sturkenboom

et al., 2014; Veenhuizen et al., 2019). Furthermore, it has been shown to detect improvements with occupational therapy in people with mental health disorders, young people, and older adults (Cup et al., 2003; Dedding et al., 2004; Eyssen et al., 2005; Horkoff, 2012). This suggests that the COPM would be an appropriate outcome measure for use with OBtAIN-PD. Regarding service level outcomes, participating people with PD and occupational therapists reported that prevention of re-referring to a community rehabilitation service within six months for the same problem will be an important outcome. Whilst a standardised measure for this outcome does not appear to exist, this data is routinely collected by services and thus would be feasible to monitor as part of the OBtAIN-PD.

Strengths and Limitations of the study

The structured process to which participants contribute directly is a strength of the coproduction approach using logic modelling. The model portrays how the participants feel the intervention should be structured, including the resources needed and the desired outcomes. Online sessions allowed for a convenient method of participating, thus facilitating participation. We were able to tailor our engagement within the sessions to the individual groups for this particular research project, such as delivering group or individual sessions at convenient times for participants. How these approaches might need to be modified for other organisational contexts and research studies cannot be determined from an instrumental case study. This study may provide a helpful foundation for future coproduction processes to be developed and assessed. Nonetheless, care should be taken when generalising specific aspects of our approach to other studies. The sample size exceeded our target (n=34). Fewer care partners participated (n=9) than people with Parkinson's (n=14) and occupational therapists (n=11), of whom we overrecruited. Though not substantial, the under-recruitment of care partners likely reflects the barriers faced by care partners in taking part in research, such as time constraints due to caring responsibilities (Malm et al., 2021). Future studies should aim to address these barriers more comprehensively, as providing a convenient participation method is likely not enough (Malm et al., 2021; Robinson, 2014). The online sessions were convenient for participants and researchers, but the impact on recruitment cannot be ignored. Those who are less digitally literate may not have participated (Office for National Statistics, 2019).

Our sample was predominantly female but this figure is skewed by the occupational therapy sample which reflects the 90% female workforce (Ledgerd & World Federation of Occupational Therapists, 2020). The sample was more evenly matched for people with PD. This was unexpected, especially as PD tends to be more prevalent in men, and men are over-represented in most research (Baldereschi et al., 2000). Our findings may not fully represent the views of male people with PD, care partners, or occupational therapists. It could be argued that this is less relevant for occupational therapists, as the intervention is delivered on patient needs and is not dependent on the gender of the clinician. Future studies should consider amending their recruitment strategy to address this imbalance. Despite our efforts, our sample across all groups mainly consisted of white English people from Christian or non-religious backgrounds. Other ethnic and religious denominations are represented in this study in small numbers, so the external validity to other groups is limited. Future studies should design recruitment strategies to actively reach people from seldom-heard groups, as relying on peer referral methods like snowball sampling will likely yield limited results.

Conclusion

This study has systematically co-produced an intervention that occupational therapists can deliver to help people with PD live well with anxiety (OBtAIN-PD). The logic modelling approach provided a unique opportunity to co-produce the OBtAIN-PD with diverse participant stakeholders. These stakeholders included people living with PD (including care partners) and healthcare professionals. The identified desired inputs, processes and outputs for OBtAIN-PD resulted in a consolidated logic model. This provided the input for an intervention manual, information sheets, and goal-planning documentation.

The Logic model framework was be a helpful tool for identifying the inputs needed for a successful intervention, the key processes essential for implementation, and the outcomes that can be considered when determining the intervention's impact. The OBtAIN-PD will undergo initial testing in a feasibility randomised controlled trial.

Acknowledgements: The authors would like to thank the participants for sharing their experiences and their time, the patient and public involvement consultees for contributing to the design of this study, and Parkinson's UK and the Royal College of Occupational Therapists for their support with the study recruitment.

Chapter 5: Intervention Specification

This chapter aims to describe the OBtAIN-PD in more detail following the identification of components in chapters three and four. Using the MRC's framework for developing and evaluating complex interventions, the first stage involved identifying the characteristics of current occupational therapy interventions for community-dwelling adults with anxiety to identify potential active ingredients. The components for the new intervention were identified with input from participant stakeholders in the second stage. The third stage involved working closely with participant stakeholders to model the new intervention, using insights from the author's previous research (including this thesis) to co-create the OBtAIN-PD.

5.1 Developing the OBtAIN-PD

The MRC framework for developing and evaluating complex interventions and how the OBtAIN-PD relates to this is described in section 1.4. Developing a complex intervention involves three key functions and activities (Medical Research Council, 2008; Skivington et al., 2021) (Table 5.1). This research programme developed the intervention using these three key functions. The first stage of the criteria for reporting the development and evaluation of complex interventions in healthcare (CReDECI) was used to report the development of the intervention specification transparently (Möhler et al., 2012). The six items of the CReDECI first stage are summarised in Table 5.1. The table demonstrates how they relate to the MRC guidance (2008) and the OBtAIN-PD project.

Development of complex interventions, key functions and activities (MRC, 2008).	Development of complex interventions; criteria for reporting the development and evaluation of complex interventions in healthcare (CReDECI) (Möhler et al., 2012)	Relation to the OBtAIN- PD project
 Identifying the evidence base. 	 Description of the interventions underlying theoretical considerations. 	Chapter 2
 Identifying/ developing appropriate theory. 		Chapter 3
3. Modelling	2. Description of all components of the	
process and	intervention.	Chapter 4
outcomes.	3. Rationale for the selection of the	
	interventions components.	Chapter 6
	4. Illustration of any intended	
	interactions between different	
	components.	
	5. Rationale for the aim/ essential	
	functions of the intervention's	
	Components.	
	and determinants of the setting in the	
	modelling of the intervention.	

Table 5.1: Development and reporting of the OBtAIN-PD in relation to the MRC framework(2008) and the OBtAIN-PD project.

The CREDECI was used to report the development and evaluation of complex interventions in healthcare (Möhler et al., 2012). Since the 2012 paper, the guidelines have been revised following a consensus process (Möhler et al., 2015). This has created a first-stage "Development" process with 4 stages. Here, stages 2 and 3 are combined, as well as stages 5 and 6, to produce a 4-stage process. Future work could report using the updated CREDECI2, although it was felt that the original 6-item description was more directive as to the steps an early career researcher with less experience in this method should follow. The MRC does not provide specific expectations for detailing interventions. The Template for Intervention Description and Replication (TIDieR) checklist allows for structured intervention reporting (Hoffmann et al., 2014). The 12-item checklist extends the intervention-related aspects of CONSORT 2010 and SPIRIT 2013 statements (Chan et al., 2013; Schulz et al., 2010). The TIDieR checklist items are outlined briefly in Box 5.1.

- 1. Brief name
- 2. Why
- 3. What (materials)
- 4. What (procedures)
- 5. Who provided
- 6. How
- 7. Where
- 8. When and how much
- 9. Tailoring
- **10.** Modifications (made in the course of the research study)
- **11.** How well (planned measurement adherence and fidelity)
- **12.** How well (actual adherence and fidelity)

Box 5.1: Template for Intervention Description and Replication (TIDieR) checklist (Hoffmann et al., 2014)

Developing complex interventions is not a straightforward process (MRC 2008). However, several frameworks are available that guide how to manage the intervention mapping process. These frameworks are listed in Table 5.2. Sermeus (2015) critiqued various models for building complex interventions and identified that this informs their general approach. Subsequently, Sermeus (2015) and Abraham et al. (2015) proposed six-stepped processes for modelling complex interventions and intervention mapping, respectively. These processes are briefly outlined in Table 5.2. Both approaches were scrutinized in relation to the key steps for the development and reporting of the OBtAIN-PD, as outlined in Table 5.1.

The outcome of this appraisal identified that each approach encompassed the key functions and activities of developing complex interventions stated by the MRC (2008; Skivington et al., 2021). These two approaches sum up the considerations and decisions taken in developing the occupational therapy intervention specification for this study. An outline of the specific approach used to develop the OBtAIN-PD is also briefly given in Table 5.2. Whilst comparing the approaches to developing complex interventions, three main components emerge (see Table 5.2):

- Developing and describing the processes and outcomes of the complex intervention.
- Implementation considerations of the complex intervention.
- Evaluation of the processes and outcomes of the complex intervention.

Components of	Six stages of	Six step modelling	Approaches to	
complex	intervention	for complex	developing the	
intervention	intervention mapping (Abraham		OBtAIN-PD	
development	et al., 2015.)	(Sermeus, 2015)		
Developing and describing	 Needs assessment (e.g., what is the 	 Installing a project team and formulating 	 Completing the scoping review to identify 	
outcomes	 health problem). 2. Objective setting (primary and secondary outcome objectives). 3. Identification of change 	 key objectives of the complex intervention. 2. Getting consensus on 	potential intervention characteristics (Chapter 2) (Lovegrove et al. 2023a, 2023b) 2. Group concept mapping to identify	
	change mechanisms and techniques (processes and techniques which are active ingredients of the intervention).	 consensus on the components of the complex intervention. Clustering of clinical activities into key interventions and building a process flow. 	intervention components (Chapter 3) (Lovegrove et al., 2023c)	
Implementation considerations	 Delivery methods (How best to deliver intervention). Implementation (How will the intervention be delivered in practice). 	 4. Get the process organised and allocate resources. 5. Detailed description of the key interventions. 	 Logic modelling with stakeholders (Chapter 4) (Lovegrove et al., 2023d) Applying the <i>Template for</i> <i>Intervention</i> <i>Description and</i> <i>Replication</i> (<i>TIDieR</i>) checklist (Hoffmann et al 2014). 	
Evaluation of the processes and outcomes	 Evaluation (outcome and processes). 	 Translation into a set of process and outcome indicators. 	 The OBtAIN-PD feasibility cluster RCT (Chapter 6) (Lovegrove et al., 2023e). 	

Table 5.2: Approaches to developing complex interventions.
Ultimately, Sermeus's (2015) six-step modelling for complex interventions was used for intervention modelling as it aligned more clearly with the OBtAIN-PD project (Chapter 4) (Lovegrove, Bannigan, Marsden, et al., in press). The following sections detail the OBtAIN-PD in the context of the TIDieR checklist (Box 5.1) (Hoffmann et al., 2014).

5.1.1 Brief name

The full name for the intervention is the 'Occupation-based complex intervention for living with anxiety in PD'. The brief name for the intervention is the OBtAIN-PD.

5.1.2 Why

The OBtAIN-PD aims to help people with Parkinson's live well with anxiety. People with Parkinson's have expressed that they want an intervention to help them manage anxiety specifically and that this is a research priority (Deane et al., 2014; Lovegrove et al., 2017). The OBtAIN-PD intervention aims to restore and maintain engagement in meaningful social and habitual roles important to people with Parkinson's, such as participating in a club or pursuing a hobby (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). In contrast, traditional NHS occupational therapy interventions tend to focus on compensatory techniques, such as providing equipment to maintain personal care. This forms the usual care group in the feasibility cluster RCT. People with Parkinson's with anxiety have identified this as less of a priority (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). The OBtAIN-PD incorporates lifestyle modification (the altering of habits to develop and maintain longterm healthy behaviours) to help people with Parkinson's engage in meaningful activities and manage their anxiety effectively. A potential working mechanism of OBtAIN-PD may be the mediating role of personal control in managing their condition. Higher perceived personal control has been shown to have a positive effect on a person's psychosocial and

physical wellbeing (Ashley et al., 2015; Tamagawa et al., 2012). In an RCT of web-based selfmanagement intervention for cancer survivors, personal control had a statistically significant effect on depression and fatigue levels (Willems et al., 2017). Similar mechanisms have been reported in other occupation-based interventions and may work through stimulating occupational adaptation (the interaction between the person, occupation, and environment in response to challenge) (George-Paschal et al., 2022; Gilboa & Helmer, 2020; Johansson & Björklund, 2016; Walder et al., 2021).

5.1.3 What (materials)

The OBtAIN-PD intervention requires very few additional materials to reduce implementation barriers, especially in resource-constrained environments like the NHS. These additional materials are goal planning and information sheets, alongside the intervention manual. Providing an intervention manual will support flexible delivery of the OBtAIN-PD while promoting content fidelity, and supporting occupational therapists' caseload demands and the needs of people with Parkinson's. Occupational therapists are free to provide the intervention-specific goal-setting and information documentation either electronically or as hard copies, depending on the preference of the person with Parkinson's. Usual care occupational therapy can be provided alongside the OBtAIN-PD, but the OBtAIN-PD can be delivered without this.

5.1.4 What (procedures)

Intervention may involve education, coaching, skill practice, and feedback. Table 5.3 provides an overview of the proposed activities performed during the planned feasibility cluster RCT for participants receiving OBtAIN-PD. The table also outlines the procedure for delivering the OBtAIN-PD. As the trial protocol (chapter 6) outlines, the suggested eight one-

hour sessions will be conducted over 10 weeks (Lovegrove, Bannigan, Hayward, et al., in press). The table also indicates who will be responsible for delivering each activity. The activities are colour-coded based on the person who will perform them. The community rehabilitation team occupational therapists will perform the green-shaded activities, while the chief investigator (or delegated research team members) will perform the ambershaded activities. More detailed descriptions of the activities will be provided in the following sections.

Session/ Activity	Time	Activity (confirmed & suggested)	Delivered by?
Referral sent to the community rehabilitation team (CRT)			Normal referral routes- consultants, Parkinson's specialist nurses, clinics, other AHPs
Referral triaged as per local protocol	As per local protocol	Referral triaged and prioritized. First appointment booked. Study and intervention details discussed.	CRT clinician
Screening log completed, information pack sent to participant	Complete screening log. If participant interested, send email of postal information pack (includes details of what is involved in OBtAIN-PD).	CRT clinician	
Details sent to CI	1-2 minutes	Participant details emailed to Chief Investigator (CI), or CRT clinician sends details to CI	Participant or CRT clinician
Screening and informed consent process completed	30 minutes	Questions answered. Informed consent documentation completed. Participant details entered to trial database.	CI (Chris Lovegrove)
Baseline outcome measures taken	One week prior to first appointment. Clinical outcomes measures (60 minutes). Patient- reported outcome measures (15 minutes)	Canadian Occupational Performance Measure, Activity Card Sort, Generalised Anxiety Disorder assessment (GAD-7), Parkinson's Disease Questionnaire (PDQ- 39), EQ-5D-5L, Barthel Index, Baseline clinical outcomes sent to the treating clinician.	Cl. Participant- Patient-reported outcome measures
Session 1	60 minutes (or as required), complete the therapist contact form (max. 5 minutes)	Initial assessment and goal discussion. Complete the OBtAIN-PD goal planning sheet. Provide 'Parkinson's & Anxiety' information sheet. Complete Therapist contact sheet. Report adverse events (if needed).	CRT occupational therapist
Session 2 - 7	60 minutes (or as required), therapist contact form (max. 5 minutes)	Review goal planning. Treatment. Provide necessary information sheet (exercise, sleep, or diet & nutrition) if indicated. Amend goal sheet. Complete Therapist contact sheet. Report adverse events (if needed.	CRT occupational therapist
Sesson 8	60 minutes (or as required), therapist contact form (max. 5 minutes)	As per sessions 2-8. Complete final session procedure as per OBtAIN-PD manual.	CRT occupational therapist
12-week (from baseline assessments) follow-up	Clinical outcomes measures (60 minutes). Patient-reported outcome measures (15 minutes)	As per baseline	CI. Participant- Patient-reported outcome measures
24-week (from baseline assessments) follow-up	Clinical outcomes measures (60 minutes). Patient-reported outcome measures (15 minutes)	As per baseline	CI. Participant- Patient-reported outcome measures
Qualitative interview	60- 90 minutes	Semi-structured interview at a date and location convenient to the participant/ clinician	CI

Table 5.3: Summary of the activities of the OBtAIN-PD.

The goal planning sheet for OBtAIN-PD should be completed during the first session (Figure 5.1). The person with Parkinson's should complete this sheet (if possible) and keep it with them. The purpose of this sheet is to encourage the person receiving the OBtAIN-PD to take ownership of their goals and use it as a tool for future therapy sessions. Promoting ownership in both approaches has been shown to support positive and sustained behaviour change (Morton et al., 2015; Yıldız, 2020). Ownership is an important concept in acceptance and commitment therapy (ACT) and motivational interviewing (Coto-Lesmes et al., 2020; Wagner & Ingersoll, 2012), and this concept is included in the OBtAIN-PD. The goals should be related to managing anxiety and engaging in meaningful activities. The goal planning sheet should be reviewed at the beginning and end of each session and adjusted accordingly. If necessary, a new sheet should be provided.



Figure 5.1: Example of a completed goal planning sheet. This example is based on a real-life exemplar provided a person with Parkinson's acting as a PPI stakeholder.

During the first assessment and goal-setting discussion, it is important to communicate in a compassionate manner, i.e with empathy, kindness, and understanding, that guides

individuals towards identifying the steps they need to take to achieve their goals. The aim is

to inspire motivation within the person to make positive changes (Morton et al., 2015;

Wagner & Ingersoll, 2012). The occupational therapist uses open-ended questions to learn

more about a person with Parkinson's perspectives and ideas about how to change. Using affirmation, i.e. statements that recognise the person's strengths and acknowledge positive behaviours, enables the occupational therapist to acknowledge a person's abilities in a nondemeaning way, which supports therapeutic relationship formation (Street & Richard, 2013). Research has shown that using affirmation can help acknowledge the recipient's emotions and feelings (Fosha, 2000). Reflective listening, also known as active listening, i.e. purposefully paying attention to what people are saying and demonstrating this, can help to clarify what the recipient is saying while simultaneously showing them that they are being heard. Reflective listening can motivate a person to consider change, thus preparing them to act towards it (Miller & Rollnick, 2012). Summarising, an application of reflective listening to link together and reinforce information that has been discussed should be used to reflect what the person has told the occupational therapist using their own words. Hearing their own words can effectively motivate people to consider change (Miller & Rollnick, 2012; Vansteenkiste & Sheldon, 2006). Examples of these communication approaches are shown in Table 5.4.

Examples					
Open-ended questions	 "What makes you think it might be time for a change?" "What brought you here today?" "What happens when you [insert risky/problem/unhealthy behaviour]?" "What was that like for you?" "What's different about (quitting smoking, improving your exercise, diet, etc.) this time?" 				
Affirming	 "That's a good suggestion." "I appreciate that you are willing to meet with me today." "It sounds like it has been very hard to cut down on coffee." "It's clear that you're really aware of how important it is to get enough sleep." 				
Reflective listening	 "It sounds like" "It seems as if" "What I hear you saying" "I get the sense that" "I get the sense that this has been difficult" "What I hear you saying is that your [insert risky/problem/unhealthy behaviour] is affecting you by" 				
Summarising	 "It sounds like you are concerned about your anxiety because it is stopping you from spending quality time with your grandchildren." "You have been talking about improving your diet and losing weight as you feel it will help your anxiety. It seems you have started to recognise the less good things about being overweight. It's easy to understand why you want to work on this." 				

Table 5.4: Examples of the communication skills used during the OBtAIN-PD initial session.

During the initial session, the occupational therapist provides an information sheet titled "Parkinson's & Anxiety V1" to the person with Parkinson's (Appendix 10). This sheet provides introductory information about anxiety and Parkinson's based on current evidence

(Lovegrove & Bannigan, 2021; Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove,

Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). During intervention development, people with Parkinson's, care partners, and occupational therapists felt it was important to receive information about why anxiety occurs in Parkinson's and the problems it causes (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). Behaviour change theories suggest that educating people about their experiences is crucial to increase a person's intent to change (Ryan et al., 2008; Sinaise et al., 2023). It is not enough to provide education alone to change behaviours. It is important to explain to people why behavioural changes are necessary (Arlinghaus & Johnston, 2018).

The number of occupational therapy sessions provided depends on the person with Parkinson's goals, reason for referral, and interventions used. The treating occupational therapist makes this decision and it is not a prescription. When implementing complex interventions, supporting clinician autonomy can serve as a facilitator (Mathieson et al., 2019). The treatment delivered in the OBtAIN-PD sessions should be tailored to address the individual goals of the person with Parkinson's. For instance, the example mentioned in the sample goal planning sheet:

"To be able to walk my dog along the seafront without having a panic attack." (Figure 5.1)

There are multiple ways to approach the example goal:

- Exploring and identifying triggers that cause anxiety and finding ways to reduce them (Chapter 3, pages 108-110, Appendix 7; and 4, pages 132-134).
- Encouraging individuals to complete daily activities they are already doing to increase their sense of mastery and confidence (Chapter 3, pages 108-111, Appendix 7; and 4, pages 133-134).

- Supporting individuals to plan ahead can reduce potential stressors, such as timing walks when fewer people are around (Chapter 3, pages 108-109, Appendix 7).
- Grading activity involves gradually increasing engagement in an activity, such as a walk at the seafront, alongside the individual (Chapter 3, page 109, Appendix 7).
- Situational use of relaxation techniques, such as breathing exercises, can prevent anxiety from escalating during an activity (Chapter 3, pages 108-109, Appendix 7; and 4, pages 133-134).
- Cueing strategies, such as auditory or visual cueing, can be used to manage freezing episodes or initiate movement (Chapter 4, pages 108-109).
- Cognitive support strategies can include setting reminders or prompts on mobile devices and planning new and interesting activities to promote cognitive initiation (Chapter 3, pages 109, Appendix 7; and 4, pages 132-134).

This is not an exhaustive list of the interventions that can be provided by occupational therapists delivering the OBtAIN-PD. The list provides suggestions of the sorts of treatment activities that an occupational therapist may provide when working with a person with Parkinson's with anxiety.

During the intervention, the occupational therapist may need to provide the person with extra information to help them adapt their roles, routines, and lifestyles to help them live well with anxiety. This is consistent with other interventions that utilise lifestyle management strategies (Clark et al., 2012; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007). As part of the OBtAIN-PD project, three information sheets were created on exercise, sleep, and diet and nutrition (Appendix 11). A summary of each information sheet is provided in Table 5.5. The information sheets have been created based on the input and feedback from people with Parkinson's, care partners, and occupational therapists (Lovegrove et al., 2017; Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). They have been designed to include the necessary information for the OBtAIN-PD and are based on the current NHS guidance. The training emphasises that the information sheets should only be provided if the person with Parkinson's has identified the topic and should not be considered a sole intervention. People with Parkinson's have previously expressed dissatisfaction at being provided information sheets as the sole treatment and the long-term impacts of such an approach are questionable (Hughes & Burch, 2020; Keyworth et al., 2020; Lovegrove, Bannigan, Marsden, et al., in press; Miller & Rollnick, 2012). These information sheets are designed to assist the delivery of OBtAIN-PD by providing information and facilitating discussion between the person and the occupational therapist. An active therapeutic partnership, where information is discussed collaboratively, has been identified as a key active ingredient in other complex interventions (McAllister et al., 2021).

Lifestyle interventions used by occupational therapists encompass a range of activities and strategies to promote health and well-being (Pyatak et al., 2022). Whilst occupational therapy has a long history of targeting lifestyle through everyday occupation to improve human health and wellbeing, lifestyle interventions tend to be structured programs using approaches such as health education and behaviour change theory (Bradley et al., 2022; Nielsen & Christensen, 2018; Uyeshiro Simon & Collins, 2017). Therefore, occupational therapists may use lifestyle interventions, but lifestyle interventions are not necessarily occupational therapy and may be delivered by various professionals (Pyatak et al., 2022; Younossi et al., 2023). These interventions are not limited to a specific diagnosis or

population like people with Parkinson's. Instead, they are applicable across diverse settings and client profiles. Lifestyle interventions in occupational therapy can include several key components.

Occupational therapists work with clients to modify activities based on their abilities and needs. This approach is commonly known as occupational or activity adaptation (Grajo et al., 2018; Johansson & Björklund, 2016). This may involve reorganising activities, using assistive devices, or altering the task itself to enable greater participation. In communitydwelling older adults, activity adaptation can improve performance in activities of daily living (Liu et al., 2018). A systematic review of interventions modifying activities for individuals with Alzheimer's-type and other dementias indicates strong evidence for this approach in addressing self-care and leisure occupations (Padilla, 2011). Establishing and modifying routines and habits is a fundamental aspect of lifestyle interventions in occupational therapy (Epley et al., 2021). By helping individuals create and maintain positive daily routines, therapists contribute to developing healthy lifestyle patterns (Arlinghaus & Johnston, 2019). In occupational therapy, the ReDO program equips individuals with the tools to analyse their occupational patterns, leading to a more fulfilling and healthy daily life (Erlandsson, 2013). It should be noted that the ReDO program is designed for women and aims to improve return-to-work outcomes (Eklund & Erlandsson, 2013; Erlandsson, 2013; Fox et al., 2021). Thus, the transferability to people with Parkinson's at different life course and condition stages is limited. On the other hand, the Well Elderly Lifestyle Redesign® intervention used similar concepts and effectively improved health outcomes for participants experiencing various health disparities in different community settings (Pyatak et al., 2022). Furthermore, the Lifestyle Redesign[®] approach has been successfully replicated in other settings and populations, such as chronic pain management and diabetes

(Pyatak et al., 2019; Uyeshiro Simon & Collins, 2017). Occupational therapists are key in occupation-based and -focused health promotion and education (Morris & Jenkins, 2018). In the OBtAIN-PD intervention, occupational therapists provide people with Parkinson's with information and skills to make informed choices about their health that emphasise preventive measures and lifestyle modifications to routines and habits (Morris & Jenkins, 2018).

A common component of occupational therapy is stress management. Occupational therapists can assist people in developing effective stress management and coping strategies (Hogan et al., 2023). These may include mindfulness techniques, relaxation exercises, and incorporating enjoyable and stress-reducing activities into daily life (Hardison & Roll, 2016; Scott et al., 2001). Though an adjunct psychological approach, mindfulnessbased stress reduction techniques have been used in occupational therapy interventions and may have a positive effect in reducing anxiety symptoms (Hardison & Roll, 2016; Hardison et al., 2021; Thompson, 2009). In Parkinson's, the evidence for mindfulness techniques in managing anxiety is inconclusive (McLean et al., 2017). Mindfulness was not specifically incorporated into the OBtAIN-PD programme for this reason.

The integration of lifestyle interventions in occupational therapy is supported by a growing body of evidence that highlights the positive impact of these approaches on health outcomes (Clark et al., 2012; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010; Pyatak et al., 2022). Research studies and clinical trials provide insights into the effectiveness of various lifestyle interventions in addressing a range of health conditions and improving overall well-being. Studies have shown that lifestyle interventions, including activity modification and dietary changes, can be used to effectively manage

diabetes (Galaviz et al., 2018). A key active ingredient for occupational therapists seems to be working collaboratively with individuals to develop personalised strategies for incorporating physical activity and healthy eating habits into their daily lives (Reitz et al., 2020). In cardiovascular disease, lifestyle interventions, such as promoting regular physical activity and stress reduction techniques, contribute to improved cardiovascular health (Rippe, 2019). Lifestyle interventions focused on balance training, home modifications, and exercise have proven effective in preventing falls among older adults (Clark et al., 2012; Sherrington et al., 2020). Occupational therapists work to identify and address factors that contribute to fall risk, promoting safety and independence (Walker & Howland, 1991). This suggests that including lifestyle management interventions in OBtAIN-PD is reasonable. This is especially important due to the increased falls risk experienced by people with Parkinson's and the benefits of regular physical activity for this population (Bhalsing et al., 2018; Pelicioni et al., 2019).

There is increasing evidence that lifestyle interventions can be an effective way to manage mental illness and improve mental and physical health, as well as overall quality of life (Manger, 2019). This can help to reduce the gap in life expectancy between individuals with severe mental illness and the general population (Manger, 2019). Lifestyle interventions can be effective in the management of anxiety (Correll et al., 2023). An occupational therapy-led lifestyle intervention was demonstrated to be both clinically and cost-effective in reducing symptoms of anxiety and depression (Lambert et al., 2008; Lambert et al., 2007; Lambert et al., 2010). The lifestyle review included in the intervention may have helped people to recognise their specific sensitivities and thus supported behaviour change (Lambert et al., 2008; Lambert et al., 2007). This is particularly valuable for inclusion in the OBtAIN-PD. People with Parkinson's often experience anxiety as a standalone symptom, but depression

is a common comorbid NMS alongside other NMS such as cognitive impairment (Blundell et al., 2023; Lovegrove & Bannigan, 2021; Pontone & Mills, 2021).

Lifestyle interventions incorporating cognitive stimulation activities and routines have demonstrated positive outcomes for individuals with cognitive impairment (Logsdon et al., 2007; Wu et al., 2023). Lifestyle interventions can have a moderate effect in people with cognitive impairment that is detectable for up to a year beyond facilitated sessions (Whitty et al., 2020). In a recent single-blind RCT, a cognitive stimulation intervention tailored to a person's individual needs and interests reduced anxiety levels (Gómez-Soria et al., 2023). As cognitive impairment can be a symptom of Parkinson's, including in the early stages of the condition, lifestyle interventions with cognitive stimulation properties warrant inclusion in the OBtAIN-PD and will be supported by the tailored nature of the intervention (Aarsland et al., 2021; James et al., 2021; Pfeiffer et al., 2014; Whitty et al., 2020).

Information sheet	Summary
Exercise	 Explains what exercise and physical activity is. Provides evidence-based recommendations for the amount per week (2.5 hours, 150 minutes). Gives examples of non-exercise activity
Sleep	 Explains what sleep is and why it is important. Provides NHS recommendations of how much sleep is needed per night (seven to nine hours). Provides suggestions on how to improve sleep for the person to refer to.
Diet & Nutrition	 Provides general rules for following a healthy diet based on NHS recommendations. Provides information on medication, adapted cutlery, and swallowing problems to help inform the person's thinking.

Table 5.5: Summary of the OBtAIN-PD information sheets.

The final part of each treatment session should involve reviewing what occurred during the session, assessing progress towards the participant's goal, and planning the next session if necessary. Based on the example in Figure 5.1, this process would include:

- Reviewing the person with Parkinson's experience of the session.
- Using previously identified communication strategies (Table 5.4) to help identify what went well and what did not.
- Collaborate with the person to plan the next session. For instance, if they walked their dog for 5 minutes, they can aim for 10 minutes in the next session.
- Update the goal planning sheet if required.
- Confirm the date and time for the next treatment session.

Collaborating with the person with Parkinson's to agree on a plan for the next treatment session can evoke motivation towards positive lifestyle changes and enhance the likelihood of achieving their goal (Arbuckle et al., 2020; Frost et al., 2018).

After a person completes their goal and no further intervention is required, the goal planning sheet undergoes a review. During the final session, the occupational therapist has a discussion with the person with Parkinson's about the steps they have taken to achieve their goal. It is also discussed how they have dealt with any potential barriers that may have hindered their progress towards achieving their goal. Goal formation is a key component in fostering behaviour change; closing the intervention by focusing on goal accomplishment can be a supportive factor in maintaining behaviour change (Bailey, 2019; Swann et al., 2021). To help maintain behaviour change, discussion of the following is encouraged:

- What are some habits that prevent you from doing the things that you want to?
- How do you react when you are not able to do the things that you want to?
- What helps you to develop new habits and routines?
- What are some barriers to making new habits?
- What are some unexpected things that may interfere with forming new habits and stop you from doing what you want to do? What steps can you take to prepare for such things?

5.1.5 Who provided

The OBtAIN-PD program is designed to be administered by registered occupational therapists with experience of working with people with Parkinson's. According to the GCM study and logic modelling study, people with Parkinson's and their caregivers prefer to receive treatment from a registered therapist rather than a therapy assistant (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). This aligns with other studies exploring the care preferences of people with Parkinson's (Uebelacker et al., 2014; Weernink et al., 2017).

To ensure the success of an intervention, the clinicians delivering it must be properly trained in using it (Gallo & Barlow, 2012). All participants emphasized the importance of providing occupational therapists with training as part of the intervention (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). Adequately trained clinicians leads to better adoption of interventions and contribute to improved outcomes and patient confidence (Gallo & Barlow, 2012). However, training and education provision can be time-consuming and may introduce implementation barriers if not delivered appropriately (Mathieson et al., 2019). The UK's NHS is a resource-constrained environment, particularly in community health services (Iliffe et al., 2021). Therefore, the OBtAIN-PD needed a low-resource burden to be successfully implemented using only existing resources (Gesell et al., 2021; Ramaswamy et al., 2018). To achieve this, occupational therapist training must be concise and convenient, such as e-learning (Wearne et al., 2011). The occupational therapists and managers agreed that e-learning would be an appropriate training method (Lovegrove, Bannigan, Marsden, et al., in press). E-learning approaches have been effectively used to teach clinicians to deliver other similar interventions (Hurley et al., 2019; Jones et al., 2023). As a response, the lead researcher created a 45-minute training video. The occupational therapist can access this video at a convenient time, date, and place. The learning objective of this training is to enable occupational therapist's to deliver the OBtAIN-PD competently. An OBtAIN-PD intervention manual is also provided to support the training (Appendix 12). Before delivering the OBtAIN-PD, the occupational therapists must complete the training video and read the intervention manual.

5.1.6 How

Based on findings from group concept mapping and logic modelling studies, the OBtAIN-PD intervention is delivered through individual sessions instead of group therapy, (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). This approach reflects the findings of previous studies that have examined the treatment preferences of people with Parkinson's (Lovegrove et al., 2017; Uebelacker et al., 2014; Weernink et al., 2017). Whilst group therapy groups are as effective as individual therapy in a range of modalities, listening to the preferences of people receiving interventions can facilitate engagement (Dilgul et al., 2018). Furthermore, group therapy may be less appropriate for individuals experiencing high levels of emotional distress caused by anxiety (Malhotra &

Baker, 2023). This will also allow the OBtAIN-PD to be delivered alongside usual occupational therapy care if required.

5.1.7 Where

The OBtAIN-PD can be administered in various settings for people with Parkinson's, such as the recipient's home, community, or outpatient clinics. Previous research has shown that flexible delivery methods can facilitate the implementation of complex interventions, particularly as community-based clinicians are responsible for organising their caseloads (Mathieson et al., 2019). The participants of the OBtAIN-PD project suggested that there should be some level of flexibility in the OBtAIN-PD program (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove et al., 2024) (Chapter 4, pages 131-133).

5.1.8 When and how much

There is currently no accepted or researched dosage for OBtAIN-PD. Based on previous research studies of occupational therapy with people with Parkinson's, it is estimated that OBtAIN-PD will last 30 minutes per session over 10 weeks (Clark et al., 2012; Lambert et al., 2008; Lambert et al., 2007; Nag & Jelinek, 2019). The intervention sessions will include 30 minutes of usual care occupational therapy, i.e., 60 minutes. Furthermore, the estimated frequency and dosage align with other interventions utilising lifestyle management and modification techniques. The Redesigning Daily Occupations (ReDO[™]) programme consists of 11 group sessions over 10 weeks, each lasting 2.5 hours (Fox et al., 2021). This may suggest that the estimated dosage of the OBtAIN-PD may be too low. However, a similar occupational therapy-led lifestyle approach used 10 sessions over 16 weeks with sessions ranging from 0.5-1 hour in length has been effective for people with anxiety (Lambert et al., 2008; Lambert et al., 2007). The dosage of OBtAIN-PD is also similar to those used in other

behaviour change interventions that focus on exercise self-efficacy and adherence in people with Parkinson's (Ahern et al., 2024). These interventions resulted in positive changes in quality of life and symptom profiles, suggesting the estimated frequency and dosage of the OBtAIN-PD may be appropriate (Fox et al., 2019; Fox et al., 2021; Lambert et al., 2008; Lambert et al., 2007).

5.1.9 Tailoring

People with Parkinson's emphasized the importance of adapting OBtAIN-PD to their individual needs, enabling them to re-engage in meaningful occupations (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). The OBtAIN-PD was tailored to meet the individual occupational needs of people with Parkinson's, allowing for variation in the frequency of key activities utilised. All parts of the OBtAIN-PD trial manual must be completed, but the frequency of key activities may vary.

5.1.10 Modifications (made during the research study)

Any modifications made to the OBtAIN-PD during the feasibility cluster RCT will be collected via the *Therapist contact form* and during the interviews with occupational therapists.

5.1.11 How well (planned and actual measurement adherence and fidelity)

The feasibility cluster RCT will evaluate the fidelity to content using process evaluation (Lovegrove, Bannigan, Hayward, et al., in press) Treatment fidelity refers to whether the treatment was delivered as intended (Bellg et al., 2004). In contrast, treatment adherence refers to how much a participant's behaviours comply with medical or health advice (Persch & Page, 2013). During the scoping review, it was found that most of the studies included did not provide information on how occupational therapy intervention's fidelity and adherence were supported and measured (Chapter 2) (Lovegrove, Marsden, et al., in press). As part of the feasibility study, measuring fidelity and adherence to occupational therapy will be important. Understanding the processes is crucial to assessing complex interventions (Medical Research Council, 2008). It is recommended to conduct process evaluations of the study with the same high methodological standards and thorough reporting as the evaluation of outcomes (Medical Research Council, 2008).

A *Therapist contact form* was designed for the occupational therapists to complete after each occupational therapy session (Appendix 13). This means that the additional burden on occupational therapists was kept to a minimum, again an important barrier to minimise (Mathieson et al., 2019). Since the occupational therapists will already have a caseload, it is important to balance the need for information with their workload in the feasibility cluster RCT. In a previous protocol for an RCT evaluating the effectiveness of occupational therapy in Parkinson's, a process evaluation summary sheet was utilised to collect data (Sturkenboom, Graff, Borm, Adang, et al., 2013). A similar format was used for this study. The information requested in the *Therapist contact form* pertains to the TIDieR checklist questions (Hoffmann et al., 2014). A free text box allows occupational therapists to express their views and opinions.

In the feasibility cluster RCT, the interviews with occupational therapists aims to explore their experiences with the OBtAIN-PD trial and intervention. An inductive approach was used to generate information for the research objectives. An extensive interview schedule consisting of open-ended questions was created (see Appendix 14).

5.2 Summary

This chapter provides a detailed specification of the OBtAIN-PD intervention using the TIDier checklist (Hoffmann et al., 2014). The details of the OBtAIN-PD intervention are provided in the individual checklist items. The OBtAIN-PD will undergo initial testing as part of a feasibility cluster RCT.

Chapter 6: Feasibility Trial Protocol

The previous chapters outline the 'development' stages of developing an intervention following the MRC framework (Figure 1.3) (Medical Research Council, 2008; Skivington et al., 2021). This comprises identifying the evidence base (Chapter 2, scoping review), identifying and developing theory (Chapter 3, GCM study), and modelling processes and outcomes (Chapter 4, logic modelling study). The next stage of the framework is 'feasibility'. A feasibility RCT should be designed and delivered to assess predefined progression criteria related to the evaluation design (data collection and retention) or the intervention itself (such as optimal content and delivery) (Arain et al., 2010; Medical Research Council, 2008; Skivington et al., 2021). This chapter presents the feasibility cluster randomised controlled trial (RCT) protocol. The feasibility cluster RCT protocol indicates the transition from the 'development' to the 'feasibility' phase of the MRC framework. This cluster RCT aims to test the feasibility and fidelity of the OBtAIN-PD. Trial recruitment started in April 2023.

The article in this chapter is the Word version submitted for publication in *BMJ Open* (Lovegrove, Bannigan, Hayward, et al., in press). The complete list of references for both articles paper is included in the main thesis bibliography to improve readability. The article uses language in keeping with the journal's style.

6.1 Evaluating the occupation-based complex intervention for living well with anxiety and Parkinson's disease (OBtAIN-PD) in community rehabilitation teams in the UK: A feasibility cluster randomised controlled trial protocol

ABSTRACT

Introduction

Anxiety is a common non-motor symptom of Parkinson's that is associated with reduced life quality, independence, and health outcomes. Current anxiolytic medications and the most promising behavioural interventions have inconclusive and mixed results. Occupational therapy is effective at promoting participation in activities of daily living and is recommended in national guidelines. This cluster randomised controlled trial aims to test the feasibility and fidelity of a new occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD). No such evidence-based intervention currently exists.

Methods and analysis

Fifty people with Parkinson's will be recruited from Devon, UK, to undertake the OBtAIN-PD or usual care delivered by community-based occupational therapists across two National Health Service sites. Recruitment, attrition rates, and feasibility of proposed outcome measures (COPM, GAD-7, PDQ-39, Activity Card Sort, Barthel Index, fall logs) will be tested. Resource data will be collected to aid in the feasibility assessment. Fidelity to content will be assessed using process evaluation. Subjective experiences will be explored qualitatively (10 participants, occupational therapists, and decliners).

Ethics and dissemination

This trial has been registered with the ISRCTN registry. Ethical approval has been obtained from the Health Research Authority (REC reference 23/NE/0027) before data collection. Participants will receive a summary of the results at the end of the data analysis. We will publish the results in a peer-reviewed journal and on institution websites.

Trial registration number

ISRCTN registry (ISRCTN62762494).

Strengths and limitations of this study

- Collecting resource-use data will assist in planning a full economic evaluation of the intervention in a definitive trial.
- 2. The cluster design may introduce an imbalance between trial arms, but this should also reduce contamination and improve the study's internal validity.
- Assessor blinding might be challenging to achieve in a real-world practice setting, thus introducing the possibility of assessment bias.
- 4. Follow-up is limited to 24 weeks, which may not capture the long-term effects of the intervention.

INTRODUCTION

Background and rationale

Parkinson's disease, commonly referred to as Parkinson's, is the second-most common neurodegenerative condition in the UK, affecting approximately 145,000 people (Parkinson's UK, 2014). The cardinal symptoms include tremors, loss of automatic movement, and slowed movement (Kalia & Lang, 2015). Parkinson's also results in sensory, cognitive, and psychological impairments that cause significant disability and impede participation in everyday roles and activities, resulting in reduced quality of life (Benharoch & Wiseman, 2004; Duncan et al., 2014). Anxiety affects up to half of people with Parkinson's (PWPs) (Dissanayaka et al., 2014). Higher anxiety makes PWPs more prone than agematched controls to falling and losing independence; it reduces their quality of life, leads to social role dysfunction, reduced participation, and increases health burden (Coventry et al., 2014; Pontone et al., 2019).

The psychological stressors associated with long-term conditions that PWPs experience can increase anxiety (Kano et al., 2011). Furthermore, PWPs may be more susceptible to anxiety than other long-term conditions due to Parkinson's associated dopamine deficiency, alongside other factors (Kienast et al., 2008). Dopamine is a modulator in the amygdala, a brain structure involved in fear and anxiety (Rosen & Schulkin, 1998). When dopamine is deficient, this produces neuronal hyper-excitability and exaggerated responses to perceived threats (Kienast et al., 2008; Rosen & Schulkin, 1998; Zarrindast & Khakpai, 2015). Although PWPs can be treated with dopamine-replacement medication they often experience a marked increase in symptoms as the medication wears off throughout the day (Richard et al., 2004). Primary (or type-1) worry can, in turn, rapidly progress to type-2 worry (metaworry, or 'worry about worry') (Wells, 2010). This can further increase anxiety symptoms, contributing to the maintenance of the hyper-excited neuronal anxiety circuit (Hofmann et al., 2012). Thus, as highlighted by the research team's previous research, living with anxiety in Parkinson's is a complex experience shaped by neurobiology, individual experiences, and life context that restricts participation in meaningful activities (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017).

Previous research has highlighted that PWPs living with anxiety place more importance on participation in meaningful roles and activities than group work (the common format for many anxiety treatments) (Lovegrove & Bannigan, 2021). Restricted participation in everyday activities has a central contributing role to heightened chronic anxiety (Coventry et al., 2014). Thus, a participation-focused approach to managing anxiety in PWPs is warranted. Approaches such as cognitive behavioural therapy (commonly known as CBT) have evidence of effectiveness and acceptability in the Parkinson's population (Reynolds et al., 2020; Roper et al., 2024). Due to the high incidence of cognitive impairment in PWPs, approaches such as CBT often require tailoring to individual PWPs (Roper et al., 2024). People with Parkinson's have expressed a preference for interventions focused on meaningful activity and 'doing' rather than psychotherapy (Lovegrove & Bannigan, 2021; Shah et al., 2022). In response to these issues, a novel occupation-based complex intervention to help people with Parkinson's to live well with anxiety (OBtAIN-PD) has been co-produced with PWPs, care partners, and occupational therapists as part of a logic modelling process using Medical Research Council (MRC) guidance for complex interventions (Skivington et al., 2021). During intervention co-production, participants stated that participation in meaningful activity should be the proposed primary outcome, with anxiety symptoms as a secondary outcome (Lovegrove & Bannigan, 2021; Lovegrove et al., 2024). The OBtAIN-PD intervention can be considered complex because of the number of components involved, the setting, the permitted degree of flexibility, and the training needed by the delivering occupational therapists (Skivington et al., 2021). This intervention provides a manualised treatment package targeting modifying an individual's lifestyle to remove barriers to engaging in activities that are important for PWPs and is delivered on a one-to-one basis.

As this is a new intervention, no current studies examine the feasibility, acceptability, or clinical or cost-effectiveness of the OBtAIN-PD intervention in PWPs.

STUDY AIMS AND OBJECTIVES

Our aim is to conduct a cluster randomised feasibility trial of the OBtAIN-PD in a real-world practice setting. We aim to provide high-quality data to facilitate designing and planning a future definitive trial by answering the feasibility study questions (Bowen et al., 2009). Our study objectives are to estimate the feasibility and acceptability of the trial procedures by answering a range of feasibility study questions (Figure 6.1). This trial aligns with the MRC framework for developing complex interventions (Fig 6.2) (Skivington et al., 2021).

- **1** Are the eligibility criteria suitable?
- 2 What are the numbers of eligible and interested participants from the target population? Specifically, conversion rates by recruitment method (invitation letter, personal contact with clinician).
- 3 What is the ability of clinicians to recruit participants and understand any differences between this for different referral pathways (for example GP vs Community Rehabilitation Teams vs consultant outpatient clinics vs self-referral)?
- 4 Are community rehabilitation teams willing to be randomised?
- 5 What are the retention rates as participants move through the trial?
- **6** What is the intervention fidelity between sites?
- **7** What is the feasibility and acceptability of the OBtAIN-PD intervention (adherence) for both participants and clinicians?
- 8 What is the completion rate and performance of self-report outcome measures, including completion rates, baseline scores, distributional properties and standard deviations, and responsiveness to refine the selection of the primary and secondary outcome measures for the definitive trial?
- **9** Facilitate a definition of occupational therapy treatment as usual for use in a future definitive trial.
- **10** What are the baseline factors, if any, associated with outcomes?
- 11 Inform estimates of the correlation between baseline and follow-up outcome measures to inform future sample size calculation.
- **12** What is the feasibility of collecting data to estimate intervention resource requirements and costs?
- **13** What is the feasibility of collecting data to estimate health, social care, and broader societal resource use and costs?
- 14 What is the feasibility of aligning the trial with existing clinical services i.e., referral date, first treatment date?
- 15 What are the total resources required for the full trial?

Figure 6.1: The OBtAIN-PD feasibility study questions.



Figure 6.2: The OBtAIN-PD feasibility cluster randomised controlled trial in relation to the Medical Research Council framework. Adapted from Skivington et al. (2021).

STUDY DESIGN

This is a pragmatic feasibility cluster randomised controlled trial, with assessor-blinded outcome assessment, cluster randomising Community Rehabilitation Teams (CRTs) to implement either the OBtAIN-PD and usual occupational therapy care (intervention) or usual occupational therapy care alone (usual care). A cluster randomised design was chosen to help control for potential contamination between trial arms (Moberg & Kramer, 2015).

Four CRTs across two sites (therefore two CRTs at each with one intervention and one usual care) will be allocated on a 1:1 basis to implement the OBtAIN-PD intervention plus usual care, or usual occupational therapy care alone, stratified by site with a block size of two. This will be logistically convenient and accommodate the current pressures on National Health Services, while providing a control for contamination. Group allocation will be stratified by site.

METHODS

Participants, interventions, and outcomes

Study setting

This trial will be conducted in two National Health Service trusts in the Southwest of England, UK. Each trust will have two community rehabilitation teams (CRTs) involved in the trial. Each CRT covers a distinct geographical area within their respective Trusts. All CRTs will implement the protocol in the same manner, apart from the treatment they provide, depending on allocation. Occupational therapists within these teams will deliver either usual care as part of their routine National Health Service role or the OBtAIN-PD intervention, which includes usual care, depending on treatment allocation. The unit of allocation, or cluster, is therefore at the CRT level; two of the four CRTs will be allocated to the intervention group (OBtAIN-PD plus standard care), and two will be allocated to the usual care group (standard care alone). The OBtAIN-PD trial began recruitment in April 2023 and ended recruitment in October 2023. The final 24-week follow-up will be completed in April 2024.

Eligibility criteria

The following criteria must be met to be enrolled in the study:

- They are over 18 years of age.
- The person has a diagnosis of idiopathic Parkinson's, as diagnosed by a neurologist or movement disorder consultant.
- Experiences anxiety measured as 'moderate' (≥10) by the Generalised Anxiety
 Disorder Assessment (GAD-7) as part of the screening process.
- Willing and able to undertake eight intervention sessions over 10 weeks.

• Able to give informed consent.

Exclusion criteria for this RCT are:

- Participants unable to give informed consent.
- People who are unable to physically complete self-report forms and do not have someone to assist them.
- PWPs experiencing anxiety measured as 'mild' (9 or less) by the GAD-7.
- PWPs with a severe cognitive deficit that affects their ability to follow instructions assessed using the Montreal Cognitive Assessment (<23).
- 'End-of-life stage' Parkinson's or other potentially life-limiting condition which is likely to be the main source for anxiety, e.g. cancer, heart failure, advanced lung disease.
- PWPs currently participating in a research study testing an intervention for anxiety
 or receiving another clinician-delivered non-pharmacological intervention for anxiety
 that has started in the last six months.

Interventions

USUAL CARE

The usual care group will receive 'treatment as usual' occupational therapy care based on that delivered in previous pragmatic trials of occupational therapy for PWPs (Carl et al., 2016; Sturkenboom, Graff, Borm, Veenhuizen, et al., 2013) and guided by both NICE guidelines (NICE, 2018) and the Royal College of Occupational Therapists 'Occupational Therapy for People with Parkinson's' (Royal College of Occupational Therapists, 2018) guidelines. Usual care will be delivered by a community-based occupational therapist at each site (n=2). The site file will provide descriptions of a selection of areas that could be targeted and approaches used. This will include equipment provision, personal care practice, and addressing falls. In line with occupational therapy practice, the exact approaches used will be flexible to provide therapist autonomy based on patient needs. The usual care therapy input received by the participants will be recorded to further facilitate a definition of treatment as usual for a future study via intervention logs as part of the case report form. This form will collect the session date, time, duration, location, and content. Total participants contact time and administration time will be collected to ascertain the impact on the occupational therapists' time. Treatment as usual duration will last 60 minutes per session with an estimated eight sessions over a 10-week period based on patient need.

OBtAIN-PD

OBtAIN-PD is an intervention that focuses on re-establishing and maintaining engagement in meaningful social and habitual roles, like attending a club or engaging in a hobby that PWPs value (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). In contrast, traditional National Health Service occupational therapy interventions that form the usual care group tend to focus on compensatory techniques (e.g., equipment to maintain personal care) that PWPs with anxiety have identified as less of a priority (Lovegrove & Bannigan, 2021; Lovegrove et al., 2017). The OBtAIN-PD has been co-produced with PWPs, care partners, and occupational therapists as part of the development phase of the MRC framework for developing and evaluating complex interventions (Skivington et al., 2021). The development of OBtAIN-PD is explained in more detail in previous articles (Lovegrove & Bannigan, 2021; Lovegrove et al., 2023; Lovegrove et al., 2024). The OBtAIN-PD will include lifestyle modification underpinned by Acceptance and Commitment Therapy concepts to help PWPs engage in the meaningful activities that they want to and to live well with their anxiety. The meaningful activities the participants choose to engage with will be individual and may

include activities such as walking the dog or attending a social club. Collaborative goalsetting and motivational interviewing concepts will be part of the OBtAIN-PD to promote engagement and adherence. OBtAIN-PD will be delivered by a community-based occupational therapist at each site (N=2).

The intervention will be delivered in community settings, and the occupational therapists delivering OBtAIN-PD will receive specific training in its use. The training consists of a 30minute training video supported by an intervention manual and completed examples of the patient goal and information sheets. Questions will be directed to a research team member (JM) who will act as a gatekeeper to prevent the lead researcher and assessor (CL) from becoming unblinded. The occupational therapists in both trial arms will be based in separate teams and will not share a clinical supervisor to reduce contamination risk. It is estimated that OBtAIN-PD will last 30 minutes per session, with eight sessions over a 10-week period (n=8) (Carl et al., 2016; Ghielen et al., 2017; Sturkenboom, Graff, Borm, Veenhuizen, et al., 2013). The intervention will include an additional 30 minutes of usual care, i.e., 60 minutes in total. The usual care component will include (but is not limited to) the ordering and provision of aids/adaptations to support participation in care and follow-up contact after delivery to ensure the safe use of the equipment (Sturkenboom et al., 2014). The dosage of usual care occupational therapy and OBtAIN-PD are based on published pragmatic trials of occupational therapy for Parkinson's (Carl et al., 2016; Sturkenboom, Graff, Borm, Veenhuizen, et al., 2013; Sturkenboom et al., 2014). In summary, OBtAIN-PD will run alongside usual care and provide a novel means of delivering occupational therapy targeting anxiety-related issues in performance. To minimise inter-therapist contamination, therapists will be advised to not discuss details of OBtAIN-PD with colleagues.

Participants in both groups will continue to receive usual medical and therapy management, such as physiotherapy. This will be monitored via a health, social care, and personal resource-use questionnaire.

Outcomes

We will conduct both a quantitative and qualitative evaluation. All participants will be requested to complete standardised, validated patient self-reported questionnaires. Electronic, web-based delivery will be utilised where possible. Paper versions of the questionnaires will be provided with stamped addressed envelopes to return by post for participants who do not have internet access. Email reminders will be used, with telephone calls used when required. The online patient-reported outcome forms will use appropriate completion rules to ensure data completeness.

All measures will be undertaken at baseline and post-intervention (12- and 24-weeks following baseline assessment). The follow-up assessments will be anchored to the baseline. This timeframe is anticipated to be adequate for delivery intervention, including potential wait times. The 12- and 24-week follow-ups are important to comprehensively assess potential benefits and the maintenance of any observed effect. All outcome measure activity is summarised in Table 6.1.

Table 6.1: Outcome measure activities							
Outcome Measured	Instrument used	Screening	Baseline	12-weeks	24-weeks		
Self-perceived	Canadian						
satisfaction in everyday	Performance						
activities	Measure (COPM)						
Participation	Activity Card Sort						
	(ACS3)		~	~	V		
Anxiety symptoms	Generalised Anxiety						
	Disorder Assessment	V	~	~	V		
	(GAD-7)						
Health-related quality	The Parkinson's						
of life	Disease		V	V	V		
	39)						
Quality of life	EQ-5D-5L						
			~	\checkmark	\checkmark		
Independence in	Barthel Index						
everyday self-care			~	~	~		
activities							
Subjective experience	Qualitative						
	interviews				V		

Baseline assessment

Baseline outcome measures will be completed in the week before treatment sessions commence to allow for relevant therapy goal-related information (derived from the COPM) to be transferred to the treating occupational therapist. If the GAD-7 has been repeated at this point and the participant score is ≤ 6 , indicating a minimal clinically important change (Toussaint et al., 2020) of four points from the inclusion criteria score, the participant will be informed of this and withdrawn from the study. Due to the nature of anxiety in Parkinson's (Kano et al., 2011), it is anticipated that this will be a rare occurrence.
Proposed primary outcome

The Canadian Occupational Performance Measure (COPM) is the proposed primary outcome measure for the future definitive trial undergoing testing. The findings of this study will determine the suitability of the COPM for the future trial. The standard COPM is a valid measure of a person's self-perception of performance in everyday living (Cup et al., 2003; Dedding et al., 2004; Eyssen et al., 2011; Eyssen et al., 2005) and has been recommended for PWPs (Gaudet, 2002). The COPM is a client-centred outcome measure for individuals to identify and prioritise everyday issues restricting their participation in everyday living. This measure focuses on occupational performance in all areas of life, including self-care, leisure, and productivity, and is measured using performance and satisfaction scores. The COPM has been used as a primary outcome in clinical trials of people with neurological conditions including Parkinson's (Sturkenboom, Graff, Borm, Veenhuizen, et al., 2013; Sturkenboom et al., 2014; Veenhuizen et al., 2019). The COPM has a minimum possible score of 1 and a maximum possible score of 10 for performance and satisfaction respectively. An improvement of 0.9 for performance and 1.9 for satisfaction is regarded as clinically significant (Eyssen et al., 2005). The COPM has adequate content and construct validity, with moderate responsiveness to change in mixed populations of older home-dwelling adults (Tuntland et al., 2016). This outcome measure will be administered via a secure webbased application or by clinical research teams in person or over the phone if the participant does not have internet access. During the development of this feasibility trial and the OBtAIN-PD, patient and public involvement (PPI) stakeholders felt that participating in meaningful activity was more important than reducing anxiety symptoms alone. Therefore, the COPM was selected as this feasibility RCT's proposed primary outcome measure.

Secondary outcomes

The Activity Card Sort (ACS) is an assessment of a person's perceived level of participation that has demonstrated applications in clinical practice and research (Duncan & Earhart, 2011). It will be administered in the same session as the COPM. The ACS is a valid and reliable tool for assessing participation, with excellent intra and interrater reliability (ICC inter-rater = 0.85; ICC intra-rater = 0.89) (Katz et al., 2003; Poerbodipoero et al., 2016).

The following patient-reported outcome measures will be sent to the participant once informed consent has been gained. These measures will include:

- GAD-7: a valid and reliable 7-item instrument used to assess the severity of generalised anxiety disorder (Ruiz et al., 2011; Spitzer et al., 2006). The GAD-7 has demonstrated validity and reliability in various populations, with high internal consistency (α = 0.89) (Johnson et al., 2019; Löwe et al., 2008; Rutter & Brown, 2017). The GAD-7 is commonly used in the 'Improving Access to Psychological Therapies' pathway and will provide a viable way of comparing the OBtAIN-PD against other interventions in the future. A score of ≥10 identifies a level of anxiety that has a direct impact on the quality of life and is a recommended cut-off that identifies a need for further clinical evaluation (Locke et al., 2015). The GAD-7 is used as both a screening tool and an outcome measure in this trial and will be re-sent to the participant if their first treatment session is not within one week of gaining consent.
- PDQ-39: This condition-specific questionnaire is a patient-reported measure of health status and quality of life and has been psychometrically evaluated in this population. It comprises 39 questions related to activities and symptoms. It takes approximately three minutes to complete. Minimally important changes are

available [>25-point change total score, > 20-point change symptom subscale] (Horváth et al., 2017).

- *EQ-5D-5L:* Evaluation of health-related quality of life. This measure has been used within clinical trials with PWPs and has been psychometrically validated for this population (Alvarado-Bolaños et al., 2015). The EQ-5D-5L can be used to calculate quality adjusted life-years (QALYs), enabling cost-utility analyses.
- Barthel Index: A common scale used to measure a person's performance in self-care activities of daily living and a widely used measure in research to determine the effect of an intervention on independence in these activities (Morley et al., 2012; Sainsbury et al., 2005).

At monthly intervals throughout the trial, participants will complete a log of falls, resource use, and any discomforts they believe to be associated with the OBtAIN-PD study (online or paper-based depending on participant requirements). This log will also collect medication details and any changes to inform analysis, such as introducing anxiolytic medication. The monthly serial collection of data enables the impact of the OBtAIN-PD on these areas to be identified and to assess the suitability of this data collection method, as well as monitor any adverse events. All patient-reported outcome measures included in this study are validated for use in Parkinson's Disease clinical care and research. The research team designed the falls and resource use log for this study.

Qualitative evaluation

An embedded qualitative research study will examine the intervention to explore mechanisms and components for subsequent testing. Ten PWPs and the trial occupational therapists will be interviewed. The participants will be asked if they want to take part in an interview at their final 24-week outcome measure session. If the participant wants to take part in an interview, a separate PIS for the qualitative interviews and a consent form will be provided. A provisional time and date for the interview will be set with the participant. The informed consent form will be completed with the participant immediately prior to the interview. The interview will be conducted online or on the telephone. Purposeful sampling will ensure the demographic representation of PWPs based on sex, location, and baseline GAD-7 score.

The specific aims of this will be to investigate:

- Acceptability of trial methods across both trial arms.
- Acceptability of the OBtAIN-PD.
- Impact that the OBtAIN-PD had on the lives of PWPs.
- Potential recruitment and retention barriers.
- Outcome measure suitability and burden.

Sample size

The target sample size is 50. The number of PWPs in the site catchment areas is estimated as 1800 with 70% (1300) known to the Parkinson's services. Around 50% are estimated to have anxiety (n=650) (Barone et al., 2009; Pontone et al., 2009). The presence of a severe cognitive impairment in Parkinson's that is likely to impair mental capacity has an estimated point prevalence of 5-10% (Aarsland & Kurz, 2010; Garcia-Ptacek & Kramberger, 2016; Hanagasi et al., 2017). Thus, the participant pool is approximately estimated with a precision of at least ±4% (i.e., width of 95% confidence interval). Assuming a retention rate of 80% at follow-up, this sample size would allow estimation of the overall retention rate with precision of ±11%. Assuming a non-differential follow-up rate of 80%, it is anticipated to follow up a minimum of 20 participants in each of the two treatment groups, which would provide data to help inform indicative sample size calculations for a definitive RCT.

Recruitment

Participants will be recruited through usual clinical pathways by staff unaware of allocation to reflect usual National Health Service practice as closely as possible whilst avoiding overwhelming the CRTs with an additional workload. Potential participants will be given an information pack by the clinician who has made contact. The information pack will contain an introductory letter, participant information sheet (PIS), GAD-7 for screening purposes, a reply slip (including consent to contact form), a stamped addressed envelope, and the lead researcher's telephone and email contact details. This information can also be emailed to participants at their request. The PIS and reply slip will encourage those who do not want to take part to return the reply slip stating their reasons for non-interest, but it will be emphasised that there is no obligation. This will provide valuable information for designing the main trial. Information packs sent out will be followed up by CRN research teams one week later via telephone to encourage responses.

Given the total sessions in the intervention and usual care group (8 sessions in 10 weeks), this recruitment rate results in an estimated 2-2.5 hrs/week of occupational therapy per site for each occupational therapist dedicated to this project (n=2 per site). This reflects the current workload allocation (<5%) for PWPs for community occupational therapists.

Allocation, concealment, and blinding

Randomisation will be undertaken by the Plymouth Clinical Trials Unit (CTU) to allow the lead researcher to remain blinded to group allocation. The CRTs will be provided with a unique, anonymous code. Within each site, the CRTs will be randomly allocated (1:1) into the usual care [60 minutes of usual occupational therapy management only per session] or intervention [30 minutes OBtAIN-PD plus 30 minutes of usual occupational therapy management per session]. The statistics team generated allocations. These were sent to the co-investigator (JM) who informed the four CRTs of their treatment allocations. Recruitment will commence within one to two weeks of cluster allocation. Participants will then be booked, and treatment provided as per arm allocation. The randomisation list and the program that generated it will be stored in a secure network location within the PenCTU, accessible only to those responsible for its provision. PenCTU staff, independent of the trial, will verify the integrity of the randomisation system throughout the trial according to established written protocols.

Access to the code/list will be confined to the PenCTU data programmer. No one else in the trial team will be aware of allocated trial arms until randomisation is completed, maintaining effective concealment. Following randomisation, only the individuals described above will be aware of the allocations to intervention or usual care arm; the blinded lead OBtAIN-PD researcher will NOT have access to treatment allocation.

Data will be collected on the following to aid in developing potential minimisation factors for a future definitive trial:

- Referral to Treat (RTT) length of time.
- Access to specialist Parkinson's rehabilitation services such as home-based care pathways and specialist Parkinson's therapists.
- Organisation size.
- Size of population served.

The trial participants cannot be blinded in this trial due to the nature of the intervention they are receiving. Similarly, the National Health Service treating occupational therapists cannot be blinded. The occupational therapists will be aware of their allocation following the randomisation of their CRT. Participants will be aware of the treatment they receive from their first treatment session with the occupational therapist, such as through the receipt of OBtAIN-PD branded materials. The recruiting clinicians (consultants, Parkinson's nurses, allied health professionals, and CRN research nurses) will not be informed of community rehabilitation teams' group allocation. The only exception will be the occupational therapists in the community rehabilitation teams who will identify potential participants in their usual referral triage process.

The OBtAIN-PD lead researcher conducting all eligibility checks, screening assessments, consent, and outcome assessments will be blinded to the participants' allocated group until database lockdown, along with the team statistician. A second, unblinded statistician generated the group allocation script. Clinical outcomes will be taken remotely wherever possible, and participants will be asked to not reveal their geographic location or the treatment they have received to preserve assessor blinding. At each time-point, all outcome measures are patient-reported assessments, thereby minimising the opportunity for the researcher to influence the outcome assessment. Every effort will be made throughout the trial to maintain blinding of the OBtAIN-PD lead researcher, for example by reminding participants not to discuss their treatment with them.

Assessor blinding will be monitored and tested by recording 'guess' participation group allocation at each time point. The blinded OBtAIN-PD researcher will be asked to record on the electronic database any cases of inadvertent unblinding to group allocation at the end of the trial. If this occurs, they will be asked to explain how this unblinding happened. The research team (including the trial statistician) will finally be unblinded after the creation of a locked analysis data set and analysis has been undertaken.

Data collection, management, and analysis

A web-based system developed by PenCTU will be used to electronically capture participant-level data and for general trial management. This consists of a bespoke online system for participant screening, consent, randomisation, and management integrated with REDCap Cloud (Harris et al., 2019), which will capture electronic case report forms.

When registered on the data collection website, the OBtAIN-PD Research team will allocate each participant a unique trial number. PenCTU Data Management staff will monitor the completeness and quality of data recorded in the database and correspond regularly with the OBtAIN-PD research team to capture any missing data where possible and ensure continuous, high-quality data. All data will be collected and stored in accordance with the UK Data Protection Act 2018 and the General Data Protection Regulation 2018. Reporting of the trial will be in accordance with the CONSORT guidance for pilot and feasibility studies and a detailed statistical analysis plan will be written and signed off prior to database lock. A CONSORT diagram will display data from screening, recruitment and follow-up logs and be used to generate estimates of eligibility, recruitment, consent, and follow-up rates. Reasons for withdrawal will be reported where available. Completion rates will be estimated for outcome measures at each time point, including the health, social and wider care resource-use data. 95% confidence intervals will accompany recruitment and retention rates to inform assumptions for planning the definitive trial. Adherence data will contribute to the evaluation of intervention acceptability and feasibility. As a feasibility trial, it is not powered to detect differences in outcomes between the groups – as such, no inferential statistical testing will be undertaken, and the analysis will be descriptive, using appropriate summary statistics and plots to illustrate any potential between-group differences.

The demographic and clinical characteristics of the sample at baseline will be summarised overall and by allocated group, using appropriate descriptive statistics (e.g. means and standard deviations for normally distributed data, numbers, and percentages for categorical variables) to informally check for balance between the groups.

The variables derived from participant reported outcomes will be calculated in accordance with published guidance. The amount of missing data will be summarised, but no imputation will be carried out unless in accordance with published guidance. At each follow-up, all participant reported outcomes will be summarised overall and by the allocated group with appropriate descriptive statistics. The change in these outcomes between baseline and follow-up will also be calculated and descriptive statistics for these reported overall and by allocated group, on an intention-to-treat basis. The between-group differences in change between baseline and follow-up will be reported with 95% confidence intervals. Estimates of the correlation between baseline and follow-up outcome measures will be used to inform a future sample size calculation for a definitive RCT.

Qualitative evaluation

The qualitative data for analysis will include verbatim transcripts from the one-to-one interviews of participants and National Health Service occupational therapy staff. Ten purposively (using the Hoehn & Yar score) sampled participants from the trial will include five trial participants randomised to the usual care group and five participants from the intervention group. All trial occupational therapists will be offered an interview. These will be run through individual one-off semi-structured interviews using an interview schedule. The interviews will be conducted at the end of the trial period at a time, date, and method (face-to-face or remote) convenient for the participant. Interviews will be recorded using a secure digital or web application (Microsoft Teams) to support transcription. The recordings will be securely deleted once the interviews have been transcribed and anonymised using pseudonyms. The anonymised transcribed data will be uploaded into NVivo 12 software for organisation and analysis (QSR International, 2012). Data will be analysed using thematic analysis adopting Braun and Clarke's six-phase process of (i) data familiarisation; (ii) generating initial codes; (iii) searching for themes; (iv) reviewing themes; (v) defining and naming themes and (vi) writing up to identify patterns of meaning within the data sources (Braun & Clarke, 2014; Braun & Clarke, 2023). Two researchers will refine initial themes to

maximise credibility and dependability. The occupational therapist interviews will be completed between December 2023 and March 2024. The participant interviews are scheduled to begin in April 2024.

Interviewees will be invited to review a draft of the analysis as part of a member-checking process to ensure their experiences are accurately represented.

Economic Evaluation

The resources required to provide the intervention will be assessed, and a framework will be established for future cost-effectiveness analysis alongside a full RCT. Data on intervention resources will be collected via within-trial reporting, including participant-level contact and non-contact time and training for delivery staff. Participants will self-report health, social and wider care resource use using a resource use questionnaire adapted for this trial. Participants will complete the EQ-5D-5L (the anticipated primary economic outcome measure in a full trial) and assess the feasibility of estimating QALYs over the follow-up period. The economic evaluation methods will be developed to provide a future policy-relevant cost-effectiveness analysis of the intervention in the context of the UK National Health Service and Social Services.

MONITORING

The trial monitoring plan has been developed and agreed upon by the Trial Management Group (TMG) and Trial Steering Committee (TSC). The TMG will meet monthly. The TSC will meet three times over the 24-month project period, with the first meeting taking place prior to the start of study recruitment. The TSC will include members independent from the trial and an independent chair and statistician.

The likelihood of participants being harmed by the interventions in either trial arm is very low. As such, the collection and reporting of adverse events in this trial is restricted only to those events that are classified as Serious Adverse Events (SAEs). In the context of clinical care and in accordance with local practice, adverse events (AEs) detected by the treating occupational therapists will be recorded in an electronic case report form. PenCTU will immediately notify a research team member (JM) of any reported AEs/SAEs. This team member will then complete an assessment of causal relationship.

ETHICS AND DISSEMINATION

We obtained ethical approval from our institution's Faculty of Health Research Ethics and Integrity Committee (Project ID 3757) and the Health Research Authority (REC reference 23/NE/0027). Participants can withdraw from the trial at any time without giving a reason; this will not affect their ongoing or future care. Data will be available upon request from the corresponding author upon trial completion. We will store data securely on an encrypted institution server. Data will be anonymised before archiving. We will publish the results in a peer-reviewed scientific journal. Participants will receive a summary of the findings at the end of the data analysis. The results will also be published on institution websites, as well as presented at conferences and to professional and patient groups.

PATIENT AND PUBLIC INVOLVEMENT

Lay members, including PWPs, care partners, and occupational therapists have provided patient and public Involvement (PPI) input. Our PPI has provided input into key aspects of study design. Discussion with our PPI representatives has led the trial to implement the recording of self-report measures via a web-based app to minimise burden, and

interviewing decliners. PPI representatives will be involved in the development of topic guides for the qualitative component of the trial. In addition, an options appraisal was conducted at the 2022 Royal College of Occupational Therapists annual conference. It was attended by 124 occupational therapists from various clinical backgrounds including (but not exclusively) acute, community, mental health, and academic services. Recommendations from this appraisal have been incorporated into this protocol and include the flexibility of treatment session delivery method, competency assessment for the occupational therapists involved in the trial, and regular contact/ support for the trial occupational therapists in addition to the provided training. A Patient Advisory Group (PAG) has been recruited to support the research team with patient-relevant advice and guidance on the delivery and progression of the study.

ETHICS STATEMENTS

Patient consent for publication

Not applicable.

6.2 Chapter Summary

This chapter has outlined a protocol for a cluster feasibility randomised controlled trial. This trial aims to assess the feasibility and acceptability of conducting a trial of the OBtAIN-PD in a real-world clinical practice setting. During the initial TSC meeting, criteria for progression were established, which are based on a traffic light system: green (proceed), amber (make changes), and red (halt). These progression criteria can be found in Table 6.2. If these criteria are not met, it indicates that a full trial may not be feasible unless the accompanying qualitative study suggests clear ways for improvement. A list of recommendations will be created to help refine the full RCT.

PROGESSION CRITERION	NOTES	RED	AMBER	GREEN
		%	%	%
Participant recruitment feasibility	Proportion of approached patients providing consent within the 10- month recruitment window	≤ 50%	51-69%	≥ 70%
Participant recruitment rate (overall)	Actual total recruitment compared to target total recruitment	≤ 50%	51-69%	≥ 70%
Participant recruitment rate (monthly)	Actual maximum achieved monthly recruitment rate compared to target monthly recruitment rate	≤ 50%	51-69%	≥ 70%
Participant retention rate	Proportion of randomised participants providing final study follow up visit data	≤ 40%	41-79%	≥ 80%
Intervention fidelity	Proportion of interventions delivered as per protocol	≤ 60%	61-74%	≥ 75%
Primary outcome data completeness	Proportion of randomised participants providing sufficient data at <primary outcome<br="">assessment timepoint> to support analysis*</primary>	≤ 60%	61-79%	≥ 80%
Interview completion	Proportion of interviews completed with participants and trial occupational therapists	≤ 50%	62-75%	≥ 75%
Site willingness	Willingness of the four individual CRTs to participate in the full cluster RCT	≤ 25%	50%	≥ 75%

Table 6.2: The OBtAIN-PD feasibility cluster RCT progression criteria.

Chapter 7: Discussion and Conclusions

The purpose of this final chapter is to summarise the findings included in this thesis and interpret these in the context of what is already known to demonstrate how they have addressed the thesis aim and added to existing knowledge. This is followed by discussion of strengths and limitations of this thesis, reflections on the challenges encountered, recommendations for future research, and concluding the thesis.

7.1 Scoping review- summary and recommendations

Anxiety is a widespread cause of disability worldwide that impedes participation in daily life, and occupational therapists aim to help people participate in daily life activities to their satisfaction. To determine what occupational therapy interventions are available globally for adults with anxiety, the scoping review in this thesis identified and analysed 18 studies that reported on eight different interventions. Although various outcome measures were used, only one intervention included occupational therapy-specific measures. It is suggested that future interventions and research should incorporate occupational therapy outcome measures and appropriate impairment-based measures to demonstrate the effectiveness of occupational therapy services. While most interventions are delivered in groups, individual occupational therapy sessions can also be beneficial. Goal setting and tailoring should be used to enhance the effectiveness and impact of occupational therapy interventions for adults living with anxiety. Rigorously designed clinical trials of occupational therapy interventions for adults living with anxiety are needed, including economic evaluations to demonstrate cost-effectiveness.

7.2 Group concept mapping- summary and recommendations

This study identified eight priority components for future intervention, presented as clusters. Individual ideas within these clusters were rated highly on importance and feasibility and could serve as starting points for intervention development. The clusters, such as "Coping" and "Support from others", highlighted the individual and social factors supporting participation in meaningful roles and activities that needed to be considered in intervention development. Access to timely peer support and the development of coping strategies were highlighted as important considerations for the next phase of intervention development. To integrate the components identified in this study into a new intervention to help people with Parkinson's live well with anxiety, future research should involve stakeholders in co-producing the intervention. Additionally, removing barriers to research participation for care partners and occupational therapists will be important as providing convenient data collection will likely not suffice.

This novel study involved people with Parkinson's, their care partners, and occupational therapists. They were asked to identify the most important components for developing an occupation- and community-based intervention to address Parkinson 's-related anxiety. This was done through an online GCM methodology that involved key stakeholders, as per the MRC framework (Skivington et al., 2021). The study identified eight clusters that served as starting points for intervention development. These clusters include exercise, lifestyle changes, self-help, coping, access to information, professional help, peers and groups, and support from others. The statements within each cluster rated highly on importance and feasibility were prioritised in developing the intervention. The study's findings were used to co-produce the OBtAIN-PD intervention.

7.3 Logic modelling- summary and recommendations

In this study, a new intervention called OBtAIN-PD was developed for occupational therapists to help people with Parkinson's live well with anxiety. The intervention was coproduced with the help of diverse stakeholders, including people with Parkinson's, care partners, and occupational therapists. A logic model was created by identifying the necessary inputs, processes, and outputs, which served as a foundation for an intervention manual, information sheets, and goal-planning documentation.

Logic model frameworks can help determine what is needed to make an intervention successful, including the essential processes and outcomes. The comprehensive and staged approach to developing OBtAIN-PD using the MRC framework will strengthen the viability of the intervention and increase the likelihood that it will benefit people with Parkinson's (Bleijenberg et al., 2018). Others who want to co-produce similar interventions with stakeholders could use this same framework. The OBtAIN-PD will undergo initial testing in a feasibility RCT.

7.4 Feasibility cluster RCT protocol- summary

Chapter 6 introduced the protocol for a feasibility cluster RCT of the OBtAIN-PD. The intervention, specified in Chapter 5, will be standardised and tested in a real-world practice setting to assess its feasibility and acceptability. The study also aims to define occupational therapy as usual care for people with Parkinson's who experience anxiety. Resource-use data will be collected to assist in planning a complete economic evaluation of the intervention in a definitive trial.

A cluster design may introduce an imbalance between trial arms, but this can also reduce contamination and improve the study's internal validity. However, assessor blinding may be challenging in a real-world practice setting, which could introduce the possibility of assessment bias. The follow-up period is limited to 24 weeks, which may not capture the long-term effects of the intervention.

This chapter (Chapter 7) provides a discussion of the thesis, the aim of which was to coproduce an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD) (Chapter 1). The results and findings are brought together and interpreted from the various phases of the research and related to the broader literature (Chapter 5). Recommendations for future research are presented in a later section (Section 7.10).

7.5 Complex interventions

Informed by the MRC framework for developing complex interventions, OBtAIN-PD was developed to help people with Parkinson's live well with anxiety. Since 2006, several significant conceptual, methodological, and theoretical advancements have occurred with the MRC framework. These advancements have been integrated into a new framework commissioned by the National Institute of Health Research (NIHR) and the MRC (Skivington et al., 2021). The framework's objective is to assist researchers in collaborating with other stakeholders to pinpoint the key questions about complex interventions and to design and conduct research with diverse perspectives and appropriate methods. The scoping review, group concept mapping, and logic modelling study just described were used to co-produce the OBtAIN-PD. Using the MRC framework as a structured approach allows reporting of the OBtAIN-PD according to the TIDieR checklist, which provides transparency and enables

future replication (Hoffmann et al., 2014). Complex intervention research involves more than evaluating whether an intervention works. It also includes identifying its impact, assessing its value, theorising the working mechanisms, considering its context, analysing how it contributes to system change, and using evidence to support decision-making. These areas are detailed in Chapters 2, 3, 4, and 5, demonstrating the extent to which a comprehensive development phase was achieved (Skivington et al., 2021).

Outcomes following complex interventions can be uncertain due to their intricate components, making prediction of outcome difficult (Levati et al., 2016). Multiple factors contributed by various participants across several studies may influence the outcome of OBtAIN-PD (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). Therefore, interpreting results is complicated and could be better assessed using a process evaluation. A process evaluation is included in the OBtAIN-PD feasibility cluster RCT to determine whether the OBtAIN-PD are implemented as intended and result in certain outputs (Lovegrove, Bannigan, Hayward, et al., in press) (Chapter 6). During the OBtAIN-PD project, participants emphasised the importance of involving care partners, friends, and family in the intervention (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove, Sturkenboom, et al., in press). This is in keeping with other studies investigating occupational therapy for people with Parkinson's (Sturkenboom et al., 2014; Welsby et al., 2019). Other occupational therapy interventions using lifestyle management approaches also use this sort of involvement in their delivery (Clark et al., 2012; Fox et al., 2021; Lambert et al., 2008). Research from other long-term conditions suggests that the support of family and friends is closely linked to the self-efficacy of individuals with diabetes. This support can help them maintain lifestyle changes and optimise diabetes management outside of formal care (Rintala et al., 2013; Rosland et al., 2008). Improved clinical outcomes

in people with type 2 diabetes are associated with higher levels of social support (Strom & Egede, 2012). In individuals with chronic obstructive pulmonary disorder, higher levels of social support are linked to increased physical activity and greater participation in pulmonary rehabilitation programs (Chen et al., 2017). It will be important to quantify adherence and self-care practices among participants in the future feasibility RCT and analyse if there is an effect of this on the primary outcome between intervention groups.

7.6 Occupational Therapy

The OBtAIN-PD program aims to assist individuals with Parkinson's disease in managing anxiety by encouraging involvement in meaningful activities. Engaging in occupational activities is necessary for all individuals, including those living with Parkinson's, as it promotes good health and wellbeing (Bar & Jarus, 2015; Reitz et al., 2020). This perspective on individuals with Parkinson's disease and the role of occupational therapists in supporting their engagement in daily activities embodies Yerxa's (1998) belief that occupational therapists are *"search engines for potential"* (pg. 413). Specifically, Yerxa proposed, "Occupational therapists discover a person's resources and emphasize what that person can or might be able to do instead of the person's incapacities; what's right instead of what's wrong" (p. 413). This quote seems to align with the aims and objectives of the OBtAIN-PD. Participants in group concept mapping (Chapter 3) and logic modelling studies (Chapter 4) highlight the importance of adapting their lifestyle to promote better habits and routines to improve quality of life (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove et al., 2024). The participant's ideas regarding establishing time usage patterns for occupational engagement are supported by a long-standing philosophy of occupational therapy. In his classic work on this philosophy, Adolph Meyer (1922) explained that it is important to live

life with a rhythm, particularly in balancing work, play, rest, and sleep. To achieve this balance, one must actually engage in these activities:

"Our body is not merely so many pounds of flesh and bone figuring as a machine, with an abstract mind or soul added to it. It is throughout a live organism pulsating with its rhythm of rest, and activity, beating time (as we might say) in ever so many ways, most readily intelligible and in the full bloom of its nature when it feels itself as one of those great selfguiding energy-transformers which constitute the real world of living beings. Our conception of man is that of an organism that maintains and balances itself in the world of reality and actuality by being in active life and active use, i.e., using and living and acting its time in harmony with its own nature and the nature about it. It is the use that we make of ourselves that gives the ultimate stamp to our every organ." (p.5)

The participants in the OBtAIN-PD project study agreed that people with Parkinson's disease still strive to lead an active and fulfilling life (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove et al., 2024). For example, participants in the group concept mapping study stated it was important to engage in meaningful activities regularly, and the logic modelling participants felt it was important to engage in activities that the person enjoys (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove et al., 2024). Similarly, the PPIE stakeholders for this project felt that a focus on 'doing' was important. A person's life is made up of both moments of engagement and disengagement in different activities. However, OBtAIN-PD aims to help people with Parkinson's live well with anxiety by promoting a balance in one's lifestyle. According to Matuska and Christiansen (2008), lifestyle balance means having patterns of engagement in activities that meet psychological needs rather than just an equal distribution of different types of activities: *"a satisfying pattern of daily occupation that is*

healthful, meaningful, and sustainable to an individual within the context of his or her current life circumstances" (p.11). Occupational patterns must fulfil certain psychological needs such as maintaining biological health and physical safety, establishing rewarding and self-affirming relationships, feeling engaged, challenged, and competent, and creating meaning and a positive identity. For example, Kitwood (2011) identified specific psychological needs for individuals with dementia (a neurodegenerative condition with some similarities to the advanced stages of Parkinson's) (van der Steen et al., 2019), including attachment, comfort, inclusion, identity, occupation, and ultimately, love. People with Parkinson's receiving OBtAIN-PD may satisfy a proportion of their psychological needs through occupational engagement and promote a balanced lifestyle.

The participants of the logic modelling study (Chapter 4) describe the possibility that people living with Parkinson's could achieve more in life than just biological health and physical safety as long as they receive adequate support (Lovegrove, Bannigan, Marsden, et al., in press). Through their discussions and views, participants seemed to suggest that occupational therapists working to create an occupationally enlivening environment can promote rewarding relationships, opportunities to be challenged and feel competent, create meaning, and maintain personal goals. Whilst not the original aim or focus of OBtAIN-PD, this may provide insight into a potential working mechanism that warrants further research. In this way, the participants in these studies were inspirational; they set the bar high for occupational therapists working with people with Parkinson's who are experiencing anxiety. The goal for people with Parkinson's must no longer be stagnant in safety and physical health; the overarching goal can be lifestyle balance through occupational engagement. The aim of OBtAIN-PD is to help people with Parkinson's live well with anxiety, so this could provide an avenue for further research. Although this view may

seem a stretch in the resource-strained UK public services, the participants suggest an occupational therapy practice that lives up to the profession's core beliefs and warrants further research.

7.7 Developing Evidence

The scoping review presented in Chapter 2 suggested that future research should focus on reporting the impact of occupational therapy interventions on activity outcomes. Research is an accepted component of evidence-based occupational therapy practice in the UK (Royal College of Occupational Therapists, 2019). It makes sense to incorporate research capacity into evidence-based practice frameworks, particularly within the National Health Service (NHS), as it recognises the importance of research in informing practice decisions. However, the findings from the scoping review (Chapter 2) suggest that there may be a need to enhance the profession's research capacity to ensure inclusivity across all occupational therapy practitioners (Di Bona et al., 2017; Slade et al., 2018), thereby promoting highquality research with increased emphasis on evaluating effectiveness through participation and occupation focused outcome measures (Lovegrove, Marsden, et al., in press). It is imperative to conduct rigorous and well-designed studies to evaluate the effectiveness of occupational therapy interventions for adults with anxiety and people with Parkinson's. The studies reviewed in the scoping review (chapter 2) suggest that conducting randomised controlled trials (RCTs) and quasi-experimental studies in this clinical field is feasible. However, the number of such studies is still limited. The scoping review highlights the urgent need for more high-quality research in occupational therapy, particularly in mental health (Bannigan et al., 2008; Kirsh et al., 2019). In order to enhance the research capabilities of occupational therapists and make the most of their knowledge and skills, including those of early career professionals, we need to reshape the way we view research.

While research utilisation is an essential part of the job for all occupational therapists, only some occupational therapists will be researchers. Ensuring that occupational therapy researchers have adequate and meaningful support will be paramount to developing the evidence for occupation-based interventions. Pighills et al. (2013) frame research capacity as a continuum. The continuum encompasses the consumption of research produced by others; in the UK, this is commonly referred to as evidence-based practice, which refers to using high-quality research findings to support and underpin clinical practice (Pighills et al., 2013). It could benefit the profession to adopt a research capacity framework based on a continuum such as the one outlined by Pighills et al. (2013). This could help broaden the perspective of what research capacity involves, making it more accessible for the majority of members to identify ways in which they can contribute to it.

7.8 International Context

Anxiety is a common human emotion. Whatever country a person lives in, feelings of worry, fear, and apprehension are part of our shared psychological landscape (Grogans et al., 2023). However, how these emotions are expressed, understood, and managed can differ substantially based on cultural context (Pugh et al., 2022). Alongside healthcare system variations between different cultures, this may impact the feasibility or effectiveness of OBtAIN-PD in these international systems (Evans & Stoddart, 2017; Nilsen & Bernhardsson, 2019). The articles identified in the scoping review (chapter 2) are predominantly from Western cultures with established public-funded healthcare systems (Lovegrove, Marsden, et al., in press)It is important to Consider how international variations in healthcare may affect the delivery of OBtAIN-PD when planning future research.

Different languages have unique ways of describing emotional experiences (Lindquist et al., 2015). For instance, some cultures might have specific words or phrases to express anxiety that don't directly translate into English (Jackson et al., 2019). These linguistic nuances can affect how individuals communicate their distress. Cultures also have varying norms regarding emotional expression, often referred to as emotional display rules (Hwang & Matsumoto, 2020; Zhou et al., 2023). Some societies encourage open emotional expression, while others emphasise stoicism (Moore et al., 2013). These norms impact how anxiety symptoms are displayed and perceived (Jackson et al., 2019). With this in mind, the language and examples used in OBtAIN-PD will likely need considerable adaptation when researching the implementation of OBtAIN-PD in different countries and cultures.

Healthcare interventions often require cultural adaptation for successful scaling and implementation (Heim et al., 2019). For example, researchers have explored the effectiveness of CBT in non-Western cultures. When delivering CBT to culturally diverse clients, clinicians often make modest adaptations. These might include incorporating culturally relevant metaphors, adjusting homework assignments, or considering cultural beliefs about emotions (Jankowska, 2019). Case studies have shown promising outcomes when CBT is culturally adapted (Wallace et al., 2021)Furthermore, mindfulness has similarly been adapted for different cultures. Cultural adaptation of mindfulness has involved modifying mindfulness-based interventions (MBIs) to align with the cultural values, beliefs, and practices of different populations. This process ensures that mindfulness practices are more accessible and effective across diverse cultural contexts. For example, a systematic review and meta-analysis of MBIs for Hispanic populations found that culturally adapted MBIs were associated with significant improvements in depression symptoms, stress reduction, and chronic illness management (Castellanos et al., 2020). The study highlighted

the importance of incorporating cultural elements to enhance the relevance and acceptance of mindfulness practices among Hispanic individuals (Castellanos et al., 2020). Similarly, mindfulness practices have been adapted for use in various cultural settings, including Asian, African, and Indigenous communities (Karl et al., 2022; Kirmayer, 2015). These adaptations often involve integrating traditional cultural practices and languages, which can help individuals connect more deeply with mindfulness techniques (Karl et al., 2022; Kirmayer, 2015). Overall, the cultural adaptation of the OBtAIN-PD will be crucial in ensuring that it is effective and meaningful for people from diverse backgrounds. For example, the OBtAIN-PD is currently an individual 1:1 treatment, but the delivery may be more effective as a group-based intervention involving trusted community leaders in a collectivist culture. Similarly, these cultural adaptations will need to be reflected in the training for the healthcare professionals delivering OBtAIN-PD (Mahon, 2023)

Differences in infrastructure across countries greatly impact the delivery of healthcare interventions, creating barriers that can hinder access to quality care (Alderwick et al., 2021). These disparities are often rooted in economic, political, and social factors, leading to varied healthcare outcomes globally (Alderwick et al., 2021). Economic disparities are a primary factor influencing infrastructure quality. High-income countries typically have welldeveloped healthcare infrastructure, including advanced medical facilities, robust supply chains, and extensive healthcare networks. In contrast, low- and middle-income countries often struggle with inadequate infrastructure due to limited financial resources. This can result in poorly equipped hospitals, insufficient medical supplies, and a lack of trained healthcare professionals (Al-Worafi, 2023b). For instance, a study on healthcare facilities in developing countries highlights that limited resources and workforce shortages are significant challenges (Al-Worafi, 2023b). These limitations can lead to overcrowded

services, long waiting times, and inadequate patient care, which ultimately affect health outcomes and will need consideration for the OBtAIN-PD.

Geographic barriers also play a crucial role in healthcare delivery. In rural and remote areas, the lack of infrastructure, such as roads, transportation, and communication networks, can severely limit access to healthcare services. People in these regions may have to travel long distances to reach the nearest healthcare facility, which can be particularly challenging in emergencies (Edelman et al., 2021). For example, in many parts of Sub-Saharan Africa and across the Philippines, poor road conditions and limited transportation options make it difficult for patients to access healthcare services (Geleto et al., 2018; Gregorio et al., 2023; Varela et al., 2019). This geographic isolation can lead to delays in receiving care, exacerbating health issues and increasing mortality rates (Edelman et al., 2021). Future adaptation of the OBtAIN-PD will need to account for these geographical barriers, especially as it has been developed in the context of UK infrastructure and the NHS.

Technological advancements in healthcare, such as telemedicine and electronic health records, have the potential to improve healthcare delivery. The OBtAIN-PD utilises various delivery methods, including telephone calls and video conferencing. However, the adoption of these technologies varies widely between countries. High-income countries often have the necessary infrastructure to support these technologies, while low- and middle-income countries may lack the required digital infrastructure and technical expertise (Al-Worafi, 2023a). A review of telemedicine implementation in developing countries found that inadequate digital infrastructure and limited internet connectivity are major barriers (Al-Worafi, 2023a). These technological gaps can prevent the effective use of telemedicine,

limiting access to remote consultations and digital health services, and will need to be accounted for in the future development of the OBtAIN-PD for international populations.

While considerable effort was made to ensure diverse representation in the co-production of the OBtAIN-PD, it is undoubtedly limited to the UK's health and social care context. With this in mind, a likely future avenue of research will be exploring the OBtAIN-PD's cultural adaptation across diverse populations. This could include collaborating globally with researchers, clinicians, and policymakers to share knowledge and develop culturally sensitive adaptations of the OBtAIN-PD.

7.9 Strengths and Limitations

Developing the OBtAIN-PD using the MRC framework for complex interventions is a key strength of this thesis. Collaborative approaches were employed to design the OBtAIN-PD, which aims to address the challenges associated with living with anxiety and Parkinson's disease. A structured approach to intervention design can increase the potential for successful implementation and minimize the risk of subjects being exposed to ineffective interventions (Bleijenberg et al., 2018). A feasibility RCT design with an embedded qualitative study will ensure a realistic assessment of the capability to conduct a future RCT. The feasibility cluster RCT will use well-established validated outcome measures and will be conducted in multiple centres to assess the intervention in different contexts (Craig et al., 2018). Semi-structured interviews will provide substantial insight into the experiences of participants and occupational therapists in delivering and receiving the OBtAIN-PD. The feasibility cluster RCT design means that it will not be possible to determine whether the OBtAIN-PD is clinically or cost-effective. This is frustrating, considering it takes an average of

five years to design and implement a definitive RCT and 17 years for new understanding and knowledge to be incorporated into clinical practice (Morris et al., 2011).

The lack of diversity in some of the baseline characteristics of the population recruited for the group concept mapping and logic modelling studies raises concerns about whether they truly represent people with Parkinson's and care partner populations in the UK. This is a limitation. Although participants from all nations of the UK were recruited, the samples are predominantly white British from Christian backgrounds. While people with Parkinson's and care partners from different ethnic backgrounds were also recruited, their numbers are relatively low. This is important to ensure that the causal effects of the intervention on the study's population are truly generalizable to the real-world setting (He et al., 2020). Therefore, the next steps of the OBtAIN-PD research project, including the feasibility cluster RCT and future definitive RCT, must address barriers to participation for people from different ethnic backgrounds and seldom-heard groups to ensure that the results are generalisable. For example, using the PROGRESS-Plus framework to identify the characteristics that stratify health opportunities and outcomes may help consider equity when planning these studies and collecting data (Karran et al., 2023; O'Neill et al., 2014)

We recruited fewer occupational therapists during the intervention development studies than people with Parkinson's and care partners. This could be considered a limitation of the study. However, it is not surprising given the substantial clinical pressures on occupational therapists in the UK, particularly in the NHS (Walker, 2023). On the other hand, the demographics of the occupational therapist group, specifically gender, could be argued to be representative of the occupational therapy workforce. The sample was predominantly white British, similar to the other participants, resulting in similar transferability issues. To

address the barriers to participation, the next steps of the OBtAIN-PD project will need to consider more than just providing convenient methods for participation such as recruiting through community organisations and non-probability approaches (Valerio et al., 2016).

7.10 Challenges and lessons learned

Reflecting on my experiences conducting the research work for this thesis, there were several challenges to carrying out this research as a clinical academic occupational therapist working across clinical practice and research. Consequently, lessons can be learned for the future.

Striking a balance between the ideal research methodology and design and the practical aspects of conducting research in clinical settings can create tensions, especially when designing research with busy clinical settings and clinicians in mind. In the OBtAIN-PD project, occupational therapists expressed their research interest informally and reported barriers to service-level participation. The lessons learned from this experience highlight the importance of engaging with those most likely to facilitate and engage in research in practice during the research design and development stage (Boaz et al., 2018; Boaz et al., 2015).

The project aimed to co-produce a new intervention called OBtAIN-PD. This intervention has the potential to improve the lives of people with Parkinson's and anxiety. However, balancing this with research monitoring, management, and governance was challenging. After some reflection, it was realised that this area requires further development and strict processes will be needed to monitor and support recruitment. The practical implications of this will be explored during the feasibility cluster RCT.

Occupational therapists who contributed to developing the intervention and the RCT protocol in the GCM, logic modelling, and PPIE consultation reported that a 'lead in' time to become familiar with the study processes would be beneficial (Lovegrove, Bannigan, Marsden, et al., in press; Lovegrove et al., 2024). Therefore a recommendation for a future pragmatic RCT would be to include this as part of the study's overall timescales.

This particular PhD project was financially supported by an HEE-NIHR Clinical Doctoral Research Fellowship award (NIHR301565). The fellowship is a research training award dedicated to health and social care professionals, except for doctors and dentists, who aspire to build careers combining research and research leadership with ongoing practice and professional development. This fellowship is part of the Integrated Clinical and Academic (ICA) pathway. Despite being thoroughly enjoyable, this pathway still faces critical issues such as a lack of clear entry points, no model of career progression, and insufficient post-doctoral posts (Baltruks & Callaghan, 2018; Trusson et al., 2019). These closely mirror previous analyses of factors hypothesised as critical to establishing a robust and sustainable clinical academic career pathway for nurses, midwifery and the allied health professions (Latter et al., 2009; Westwood et al., 2018). Current pay systems, infrastructure and differing cultures in healthcare organisations and universities create barriers. These factors present substantial obstacles to early career researchers, and support from employers to remain research-active can be felt lacking (Avery et al., 2022). Employment conditions, especially salaries and pensions, could usefully be examined to ensure they don't disadvantage non-medical health professionals pursuing a clinical academic career (Baltruks & Callaghan, 2018). Alongside the ongoing debate around the concerns and issues raised by this career path, which is outside the scope of this thesis to discuss in detail, there remains

an appetite in the UK and many parts of the world to explore formalised training pathways underpinned by infrastructure and funding.

7.11 Implications for Future Research

After completing the OBtAIN-PD project, it is essential to consider how its contributions will inform future research.

Outcome measurement

The scoping review highlighted that current occupational therapy interventions for anxiety primarily utilise impairment-based outcome measures, such as the GAD-7, rather than occupational therapy-specific measures. The occupational therapists in the OBtAIN-PD project emphasised the importance of using occupation therapy-specific measures and provided suggestions to incorporate them into the OBtAIN-PD. These suggestions might have been missed if only people with Parkinson's and care partners were involved, as these groups may not consider service level factors such as outcome measurement. This approach is in keeping with the MRC's framework for developing and evaluating complex interventions. Future research studies developing complex interventions should follow this framework and be vigilant to include both those delivering and receiving interventions in the co-production.

Delivery mode

Most occupational therapy interventions for anxiety are delivered in group settings. People with Parkinson's desire personalised intervention for individual goals. This will help tailor interventions more effectively based on a person's needs. Future research aimed at designing occupational therapy treatments for anxiety must avoid group sessions as the

default delivery method. Instead, an individual approach involving tailoring and goal setting should be considered to improve the effectiveness and impact of occupational therapy interventions for adults with anxiety. Ultimately, clinical trials of occupational therapy interventions for adults with anxiety, including economic evaluations, are needed.

Stakeholder involvement

The OBtAIN-PD components were identified collaboratively with people with Parkinson's, care partners, and occupational therapists using GCM. This study used a structured methodology for data collection that prioritised participant input. Most participants had Parkinson's and anxiety. Future complex intervention development studies should consider using the GCM methodology to involve stakeholders meaningfully in providing consensus on a given topic during the development phase (Lovegrove et al., 2024; Skivington et al., 2021).

Recruitment

In the GCM study, the number of care partners and occupational therapists who participated was less than those with Parkinson's. Additionally, most of our sample were females. Although digital delivery was a convenient way to conduct this research, it might have impacted the engagement of potential participants. In future research, the recruitment strategy and sampling frame may need to be changed to address such imbalances, such as sampling through community organisations or oversampling methods (Bonevski et al., 2014; Valerio et al., 2016). Merely providing a convenient data collection method may not be enough to ensure participant engagement.

The logic modelling study encountered challenges like those of previous studies. However, the logic modelling approach presented a unique opportunity to co-create the OBtAIN-PD

with a diverse group of stakeholders. Logic model frameworks proved valuable for identifying the necessary inputs for successful intervention, the crucial implementation processes, and the outcomes to determine the intervention's impact. Future research should consider using logic modelling frameworks in similar co-creation partnerships. However, the study also experienced imbalances in its sample, and future studies should develop recruitment strategies that actively target seldom-heard groups. Relying on peer referral methods and convenient data collection methods such as snowball sampling may yield limited results.

7.12 Conclusion- final thoughts

An occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD) has been co-produced with people with Parkinson's, care partners, and occupational therapists. The OBtAIN-PD aims to help people with Parkinson's manage their anxiety and engage in meaningful activities. A feasibility cluster randomised controlled trial (RCT) protocol has been presented (Chapter 5). This feasibility RCT aims to determine the feasibility of conducting a fully powered RCT within an NHS setting while also testing the feasibility and fidelity of the OBtAIN-PD. A fully powered RCT is planned to assess the clinical and cost-effectiveness of the OBtAIN-PD based on the results of the feasibility trial.

The research presented in this PhD thesis has contributed to the body of knowledge by synthesising and drawing together the existing evidence of occupational therapy interventions for anxiety (Chapter 2). The project has involved various stakeholders in identifying the necessary components for the new intervention (Chapter 3). The OBtAIN-PD was co-produced based on findings from relevant studies and previous research (Chapter 4). As far as we know, this is the first research program to utilise group concept mapping and

logic modelling methodologies for co-producing a new intervention designed explicitly for people with Parkinson's. It is based on what they believe they need to lead a fulfilling life while managing anxiety. This PhD programme has allowed me to acquire research skills that will support my clinical academic career. The next step in the OBtAIN-PD journey is to conduct the feasibility cluster RCT based on the presented protocol (Chapter 5).

Fortunately, the COVID-19 pandemic has not substantially impacted my PhD journey. Data collection for the group concept mapping and logic modelling studies was completed online. These approaches protected participants and the research team from COVID-19 and were also entirely appropriate for these methodologies. The COVID-19 pandemic prompted the adoption of new skills, such as conducting remote informed consent, which will shape future RCT development.

The journey described in this thesis has been enjoyable, and it is the realisation of nine years of my early clinical academic career. I am enthusiastic about developing my skills and conducting research to improve the lives of individuals with Parkinson's and other groups. Developing and testing innovative strategies is essential in meeting the rapidly increasing numbers and complications arising from Parkinson's in health and social care. The OBtAIN-PD may be a potentially important strategy for helping people with Parkinson's to live well with anxiety.

References

Aarsland, D., Batzu, L., Halliday, G. M., Geurtsen, G. J., Ballard, C., Ray Chaudhuri, K., & Weintraub, D. (2021). Parkinson disease-associated cognitive impairment. *Nature Reviews Disease Primers*, 7(1), 47. <u>https://doi.org/10.1038/s41572-021-00280-3</u>

Aarsland, D., & Kurz, M. W. (2010). The epidemiology of dementia associated with Parkinson disease. *Journal of the Neurological Sciences*, *289*(1-2), 18-22. <u>https://doi.org/10.1016/j.jns.2009.08.034</u>

Abraham, C., Denford, S., Smith, J. R., Dean, S., Greaves, C., Lloyd, J., Tarrant, M., White, M. P., & Wyatt, K. (2015). Designing interventions to change health-related behaviour. *Complex interventions in health: an overview of research methods. Oxford: Routledge*, 103-110.

Agren, T., Engman, J., Frick, A., Björkstrand, J., Larsson, E. M., Furmark, T., & Fredrikson, M. (2012). Disruption of reconsolidation erases a fear memory trace in the human amygdala. *Science*, *337*(6101), 1550-1552. <u>https://doi.org/10.1126/science.1223006</u>

Ahern, L., Timmons, S., Lamb, S. E., & McCullagh, R. (2024). A systematic review of Behaviour Change Interventions to improve exercise self-efficacy and adherence in people with Parkinson's disease using the Theoretical Domains Framework. *J Frailty Sarcopenia Falls*, *9*(1), 66-68. <u>https://doi.org/10.22540/jfsf-09-066</u>

Ahn, S., Springer, K., & Gibson, J. S. (2022). Social withdrawal in Parkinson's disease: A scoping review. *Geriatr Nurs*, *48*, 258-268. <u>https://doi.org/10.1016/j.gerinurse.2022.10.010</u>

Al-Worafi, Y. M. (2023a). Healthcare Access and Equitable Access Services in Developing Countries. In Y. M. Al-Worafi (Ed.), *Handbook of Medical and Health Sciences in Developing Countries : Education, Practice, and Research* (pp. 1-24). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-74786-2_217-1</u>

Al-Worafi, Y. M. (2023b). Healthcare Facilities in Developing Countries: Infrastructure. In Y.M. Al-Worafi (Ed.), Handbook of Medical and Health Sciences in Developing Countries :
Education, Practice, and Research (pp. 1-21). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-74786-2_209-1</u>

Alafuzoff, I., & Hartikainen, P. (2017). Alpha-synucleinopathies. *Handb Clin Neurol*, *145*, 339-353. <u>https://doi.org/10.1016/b978-0-12-802395-2.00024-9</u>

Alderwick, H., Hutchings, A., Briggs, A., & Mays, N. (2021). The impacts of collaboration between local health care and non-health care organizations and factors shaping how they work: a systematic review of reviews. *BMC Public Health*, *21*(1), 753. https://doi.org/10.1186/s12889-021-10630-1

Alvarado-Bolaños, A., Cervantes-Arriaga, A., Rodríguez-Violante, M., Llorens-Arenas, R., Calderón-Fajardo, H., Millán-Cepeda, R., Leal-Ortega, R., Estrada-Bellmann, I., & Zuñiga-Ramírez, C. (2015). Convergent validation of EQ-5D-5L in patients with Parkinson's disease. *Journal of the Neurological Sciences*, *358*(1), 53-57. <u>https://doi.org/https://doi.org/10.1016/j.jns.2015.08.010</u>

Alvarez, R. P., Biggs, A., Chen, G., Pine, D. S., & Grillon, C. (2008). Contextual fear conditioning in humans: cortical-hippocampal and amygdala contributions. *J Neurosci*, *28*(24), 6211-6219. <u>https://doi.org/10.1523/jneurosci.1246-08.2008</u>

Ambrosio, L., Navarta-Sanchez, M. V., & Portillo, M. C. (2014). Living with Parkinson's disease in the community: improving assessments and interventions. *Primary Health Care*, *24*(10), 26-29.

https://search.ebscohost.com/login.aspx?direct=true&AuthType=athens&db=cin20&AN=10 3773791&site=ehost-live

Anesi, G. L., & Kerlin, M. P. (2021). The impact of resource limitations on care delivery and outcomes: routine variation, the coronavirus disease 2019 pandemic, and persistent shortage. *Curr Opin Crit Care*, *27*(5), 513-519. https://doi.org/10.1097/mcc.00000000000859 Angulo, J., Fleury, V., Peron, J. A., Penzenstadler, L., Zullino, D., & Krack, P. (2019). Shame in Parkinson'S Disease: A Review [Review]. *Journal of Parkinsons Disease Print*, *9*(3), 489-499. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medl&AN=3108</u> <u>1792</u>

Anzaldi, K., & Shifren, K. (2019). Optimism, Pessimism, Coping, and Depression: A Study on Individuals With Parkinson's Disease. *Int J Aging Hum Dev, 88*(3), 231-249. <u>https://doi.org/10.1177/0091415018763401</u>

Apostol, C., Cranwell, K., & Hitch, D. (2021). Evaluating a multidimensional strategy to improve the professional self-care of occupational therapists working with people with life limiting illness. *BMC Palliative Care*, *20*(1), 2.

https://doi.org/https://dx.doi.org/10.1186/s12904-020-00695-x

Arain, M., Campbell, M. J., Cooper, C. L., & Lancaster, G. a. (2010). What is a pilot or feasibility study? A review of current practice and editorial policy. 1-7. <u>https://doi.org/10.1186/1471-2288-10-67</u>

Arakawa, N., & Bader, L. R. (2022). Consensus development methods: Considerations for national and global frameworks and policy development. *Research in Social and Administrative Pharmacy*, *18*(1), 2222-2229.

https://doi.org/https://doi.org/10.1016/j.sapharm.2021.06.024

Arbesman, M., & Logsdon, D. W. (2011). Occupational Therapy Interventions for Employment and Education for Adults With Serious Mental Illness: A Systematic Review. *The American Journal of Occupational Therapy*, *65*(3), 238-246. <u>https://doi.org/10.5014/ajot.2011.001289</u>

Arbuckle, M. R., Foster, F. P., Talley, R. M., Covell, N. H., & Essock, S. M. (2020). Applying Motivational Interviewing Strategies to Enhance Organizational Readiness and Facilitate Implementation Efforts. *Qual Manag Health Care*, *29*(1), 1-6. <u>https://doi.org/10.1097/qmh.0000000000234</u> Arlinghaus, K. R., & Johnston, C. A. (2018). Advocating for Behavior Change With Education. *Am J Lifestyle Med*, *12*(2), 113-116. <u>https://doi.org/10.1177/1559827617745479</u>

Arlinghaus, K. R., & Johnston, C. A. (2019). The Importance of Creating Habits and Routine. *Am J Lifestyle Med*, *13*(2), 142-144. <u>https://doi.org/10.1177/1559827618818044</u>

Ashley, L., Marti, J., Jones, H., Velikova, G., & Wright, P. (2015). Illness perceptions within 6 months of cancer diagnosis are an independent prospective predictor of health-related quality of life 15 months post-diagnosis. *Psycho-Oncology*, *24*(11), 1463-1470.

Avery, M., Westwood, G., & Richardson, A. (2022). Enablers and barriers to progressing a clinical academic career in nursing, midwifery and allied health professions: A cross-sectional survey. *Journal of Clinical Nursing*, *31*(3-4), 406-416. https://doi.org/https://doi.org/10.1111/jocn.15673

Bailey, R. R. (2019). Goal Setting and Action Planning for Health Behavior Change. *Am J Lifestyle Med*, *13*(6), 615-618. <u>https://doi.org/10.1177/1559827617729634</u>

Baker, R., Camosso-Stefinovic, J., Gillies, C., Shaw, E. J., Cheater, F., Flottorp, S., Robertson, N., Wensing, M., Fiander, M., Eccles, M. P., & et al. (2015). Tailored interventions to address determinants of practice. *Cochrane Database of Systematic Reviews*(4). https://doi.org/10.1002/14651858.CD005470.pub3

Baldereschi, M., Di Carlo, A., Rocca, W. A., Vanni, P., Maggi, S., Perissinotto, E., Grigoletto, F., Amaducci, L., & Inzitari, D. (2000). Parkinson's disease and parkinsonism in a longitudinal study: two-fold higher incidence in men. ILSA Working Group. Italian Longitudinal Study on Aging. *Neurology*, *55*(9), 1358-1363. <u>https://doi.org/10.1212/wnl.55.9.1358</u>

Ballanger, B., Klinger, H., Eche, J., Lerond, J., Vallet, A. E., Le Bars, D., Tremblay, L., Sgambato-Faure, V., Broussolle, E., & Thobois, S. (2012). Role of serotonergic 1A receptor dysfunction in depression associated with Parkinson's disease. *Mov Disord*, *27*(1), 84-89. <u>https://doi.org/10.1002/mds.23895</u> Baltruks, D., & Callaghan, P. (2018). Nursing, midwifery and allied health clinical academic research careers in the UK. *Council of Deans for Health (CoDH)*.

Bannigan, K., Boniface, G., Doherty, P., Nicol, M., Porter-Armstrong, A., & Scudds, R. (2008). Priorities for Occupational Therapy Research in the United Kingdom: Executive Summary of the POTTER Project. *British Journal of Occupational Therapy*, *71*(1), 13-16. https://doi.org/10.1177/030802260807100103

Bar, M. A., & Jarus, T. (2015). The Effect of Engagement in Everyday Occupations, Role Overload and Social Support on Health and Life Satisfaction among Mothers. *Int J Environ Res Public Health*, *12*(6), 6045-6065. <u>https://doi.org/10.3390/ijerph120606045</u>

Barbalat, G., van den Bergh, D., & Kossakowski, J. J. (2019). Outcome measurement in mental health services: insights from symptom networks. *BMC Psychiatry*, *19*(1), 202. <u>https://doi.org/10.1186/s12888-019-2175-7</u>

Barone, P., Antonini, A., Colosimo, C., Marconi, R., Morgante, L., Avarello, T. P., Bottacchi, E., Cannas, A., Ceravolo, G., Ceravolo, R., Cicarelli, G., Gaglio, R. M., Giglia, R. M., Iemolo, F., Manfredi, M., Meco, G., Nicoletti, A., Pederzoli, M., Petrone, A., & Pisani, A. (2009). The PRIAMO study: A multicenter assessment of nonmotor symptoms and their impact on quality of life in Parkinson's disease. *Movement Disorders*, *24*(11), 1641-1649. https://doi.org/10.1002/mds.22643

Batzu, L., Podlewska, A., Gibson, L., Chaudhuri, K. R., & Aarsland, D. (2024). A general clinical overview of the non-motor symptoms in Parkinson's disease: Neuropsychiatric symptoms. *Int Rev Neurobiol*, *174*, 59-97. <u>https://doi.org/10.1016/bs.irn.2023.11.001</u>

Baum, C. M., & Law, M. (1997). Occupational therapy practice: Focusing on occupational performance. *The American Journal of Occupational Therapy*, *51*(4), 277-288.

Baxter, A. J., Scott, K. M., Vos, T., & Whiteford, H. A. (2013). Global prevalence of anxiety disorders: a systematic review and meta-regression. *Psychological medicine*, *43*(5), 897-910. https://doi.org/10.1017/S003329171200147X Beck, J. S. (2020). Cognitive behavior therapy: Basics and beyond. Guilford Publications.

Bellg, A. J., Borrelli, B., Resnick, B., Hecht, J., Minicucci, D. S., Ory, M., Ogedegbe, G., Orwig,
D., Ernst, D., & Czajkowski, S. (2004). Enhancing treatment fidelity in health behavior change studies: best practices and recommendations from the NIH Behavior Change Consortium. *Health Psychol*, 23(5), 443-451. <u>https://doi.org/10.1037/0278-6133.23.5.443</u>

Ben-Shlomo, Y., Darweesh, S., Llibre-Guerra, J., Marras, C., San Luciano, M., & Tanner, C. (2024). The epidemiology of Parkinson's disease. *Lancet*, *403*(10423), 283-292. <u>https://doi.org/10.1016/s0140-6736(23)01419-8</u>

Benharoch, J., & Wiseman, T. (2004). Participation in occupations: some experiences of people with Parkinson's disease. *British Journal of Occupational Therapy*, *67*(9), 380-387. <u>http://search.ebscohost.com/login.aspx?direct=true&db=cin20&AN=2005087064&site=eho</u> <u>st-live</u>

Berry, S. A., Doll, M. C., McKinley, K. E., Casale, A. S., & Bothe, A., Jr. (2009). ProvenCare: quality improvement model for designing highly reliable care in cardiac surgery. *Qual Saf Health Care*, *18*(5), 360-368. <u>https://doi.org/10.1136/qshc.2007.025056</u>

Betrouni, N., Alazard, E., Bayot, M., Carey, G., Derambure, P., Defebvre, L., Leentjens, A. F., Delval, A., & Dujardin, K. (2022). Anxiety in Parkinson's disease: A resting-state high density EEG study. *Neurophysiol Clin*, *52*(3), 202-211. <u>https://doi.org/10.1016/j.neucli.2022.01.001</u>

Beyeler, A., & Dabrowska, J. (2020). Chapter 3 - Neuronal diversity of the amygdala and the bed nucleus of the stria terminalis. In J. H. Urban & J. A. Rosenkranz (Eds.), *Handbook of Behavioral Neuroscience* (Vol. 26, pp. 63-100). Elsevier. https://doi.org/https://doi.org/10.1016/B978-0-12-815134-1.00003-9

Bhalsing, K. S., Abbas, M. M., & Tan, L. C. S. (2018). Role of physical activity in Parkinson's disease. *Annals of Indian Academy of Neurology*, *21*(4), 242-249.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=emed19&AN=62 4816272

Bianchini, E., Della Gatta, F., Virgilio, M., Alborghetti, M., Salvetti, M., Giubilei, F., & Pontieri,
F. E. (2022). Validation of the Canadian Occupational Performance Measure in Italian
Parkinson's Disease Clients. *Physical & Occupational Therapy in Geriatrics*, 40(1), 26-37.
https://doi.org/10.1080/02703181.2021.1942392

Biundo, R., Fiorenzato, E., & Antonini, A. (2017). Nonmotor Symptoms and Natural History of Parkinson's Disease: Evidence From Cognitive Dysfunction and Role of Noninvasive Interventions [Review]. *International Review of Neurobiology*, *133*, 389-415. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med14&AN=288</u> 02926

Biundo, R., Weis, L., & Antonini, A. (2016). Cognitive decline in Parkinson's disease: the complex picture. *npj Parkinson's Disease*, *2*(1), 16018-16018. <u>https://doi.org/10.1038/npjparkd.2016.18</u>

Bleijenberg, N., de Man-van Ginkel, J. M., Trappenburg, J. C. A., Ettema, R. G. A., Sino, C. G., Heim, N., Hafsteindóttir, T. B., Richards, D. A., & Schuurmans, M. J. (2018). Increasing value and reducing waste by optimizing the development of complex interventions: Enriching the development phase of the Medical Research Council (MRC) Framework. *Int J Nurs Stud, 79*, 86-93. <u>https://doi.org/10.1016/j.ijnurstu.2017.12.001</u>

Bloem, B. R., Henderson, E. J., Dorsey, E. R., Okun, M. S., Okubadejo, N., Chan, P., Andrejack, J., Darweesh, S. K. L., & Munneke, M. (2020). Integrated and patient-centred management of Parkinson's disease: a network model for reshaping chronic neurological care. *Lancet Neurology*, *19*(7), 623-634. <u>https://doi.org/10.1016/S1474-4422(20)30064-8</u>

Bloem, B. R., Okun, M. S., & Klein, C. (2021). Parkinson's disease. *The Lancet*, *397*(10291), 2284-2303. <u>https://doi.org/10.1016/S0140-6736(21)00218-X</u>

Blundell, E. K., Grover, L. E., Stott, J., & Schrag, A. (2023). The experience of Anxiety for people with Parkinson's disease. *NPJ Parkinsons Dis*, *9*(1), 75. https://doi.org/10.1038/s41531-023-00512-1

Boaz, A., Hanney, S., Borst, R., O'Shea, A., & Kok, M. (2018). How to engage stakeholders in research: design principles to support improvement. *Health Res Policy Syst*, *16*(1), 60. <u>https://doi.org/10.1186/s12961-018-0337-6</u>

Boaz, A., Hanney, S., Jones, T., & Soper, B. (2015). Does the engagement of clinicians and organisations in research improve healthcare performance: a three-stage review. *BMJ Open*, *5*(12), e009415-e009415. <u>https://doi.org/10.1136/bmjopen-2015-009415</u>

Bocchio, M., McHugh, S. B., Bannerman, D. M., Sharp, T., & Capogna, M. (2016). Serotonin, Amygdala and Fear: Assembling the Puzzle. *Front Neural Circuits*, *10*, 24. <u>https://doi.org/10.3389/fncir.2016.00024</u>

Bogosian, A., Hurt, C. S., Hindle, J. V., McCracken, L. M., Vasconcelos, E. S. D. A., Axell, S., Tapper, K., Stevens, J., Hirani, P. S., Salhab, M., Ye, W., & Cubi-Molla, P. (2022). Acceptability and Feasibility of a Mindfulness Intervention Delivered via Videoconferencing for People With Parkinson's. *J Geriatr Psychiatry Neurol*, *35*(1), 155-167.

https://doi.org/10.1177/0891988720988901

Bogosian, A., Rixon, L., & Hurt, C. S. (2020). Prioritising target non-pharmacological interventions for research in Parkinson's disease: achieving consensus from key stakeholders. *Research Involvement and Engagement*, *6*, 1-9. https://doi.org/http://dx.doi.org/10.1186/s40900-020-00212-7

Bond, F. W., & Dryden, W. (2002). *Handbook of brief cognitive behaviour therapy*. Wiley Online Library.

Bonevski, B., Randell, M., Paul, C., Chapman, K., Twyman, L., Bryant, J., Brozek, I., & Hughes, C. (2014). Reaching the hard-to-reach: a systematic review of strategies for improving health

and medical research with socially disadvantaged groups. *BMC Medical Research Methodology*, 14(1), 42. <u>https://doi.org/10.1186/1471-2288-14-42</u>

Boote, J., Baird, W., & Sutton, A. (2011). Public involvement in the systematic review process in health and social care: A narrative review of case examples. *Health Policy*, 102(2-3), 105-116. <u>https://doi.org/10.1016/j.healthpol.2011.05.002</u>

Bourgouin, P. A., Rahayel, S., Gaubert, M., Postuma, R. B., Montplaisir, J., Carrier, J., Monchi, O., Pelletier, A., & Gagnon, J. F. (2019). Gray matter substrates of depressive and anxiety symptoms in idiopathic REM sleep behavior disorder. *Parkinsonism Relat Disord*, *62*, 163-170. <u>https://doi.org/10.1016/j.parkreldis.2018.12.020</u>

Bowen, D. J., Kreuter, M., Spring, B., Cofta-Woerpel, L., Linnan, L., Weiner, D., Bakken, S., Kaplan, C. P., Squiers, L., Fabrizio, C., & Fernandez, M. (2009). How we design feasibility studies. *Am J Prev Med*, *36*(5), 452-457. <u>https://doi.org/10.1016/j.amepre.2009.02.002</u>

Bradley, T., Campbell, E., Dray, J., Bartlem, K., Wye, P., Hanly, G., Gibson, L., Fehily, C., Bailey, J., Wynne, O., Colyvas, K., & Bowman, J. (2022). Systematic review of lifestyle interventions to improve weight, physical activity and diet among people with a mental health condition. *Systematic Reviews*, *11*(1), 198. <u>https://doi.org/10.1186/s13643-022-</u> 02067-3

Brandão, M. L., & Coimbra, N. C. (2019). Understanding the role of dopamine in conditioned and unconditioned fear. *Rev Neurosci*, *30*(3), 325-337. <u>https://doi.org/10.1515/revneuro-</u> 2018-0023

Bratan, T., Aichinger, H., Brkic, N., Rueter, J., Apfelbacher, C., Boyer, L., & Loss, J. (2021). Impact of the COVID-19 pandemic on ongoing health research: an ad hoc survey among investigators in Germany. *BMJ Open*, *11*(12), e049086. <u>https://doi.org/10.1136/bmjopen-</u> <u>2021-049086</u> Braun, V., & Clarke, V. (2014). What can "thematic analysis" offer health and wellbeing researchers? *International Journal of Qualitative Studies on Health and Well-being*, *9*, 26152-26152. https://doi.org/10.3402/qhw.v9.26152

Braun, V., & Clarke, V. (2023). Thematic Analysis. In N. K. Denzin, Y. S. Lincoln, M. D. Giardina, & G. S. Cannella (Eds.), *The SAGE Handbook of Qualitative Research* (6th ed., pp. 385-402). SAGE Publications.

Broen, M. P. G., Narayen, N. E., Kuijf, M. L., Dissanayaka, N. N. W., & Leentjens, A. F. G.
(2016). Prevalence of anxiety in Parkinson's disease: A systematic review and meta-analysis. *Movement Disorders*, *31*(8), 1125-1133. <u>https://doi.org/10.1002/mds.26643</u>

Broomfield, N. M., Laidlaw, K., Hickabottom, E., Murray, M. F., Pendrey, R., Whittick, J. E., & Gillespie, D. C. (2011). Post-stroke depression: the case for augmented, individually tailored cognitive behavioural therapy. *Clin Psychol Psychother*, *18*(3), 202-217. <u>https://doi.org/10.1002/cpp.711</u>

Bryant. (2022). *Creek's occupational therapy and mental health* (Sixth edition / [edited by] Wendy Bryant, DipCOT, MSC, PGCert (TLHE), PdD, Jon Fieldhouse, BA(Hons), DipCOT, MScOT, PGCert (THLE), DPhil, Nicola Ann Plastow, BSc (OccTher), PGCert (TLHE), MSc (PHS), PHd (Clinical sciences). ed.).

Bystritsky, A., Khalsa, S. S., Cameron, M. E., & Schiffman, J. (2013). Current diagnosis and treatment of anxiety disorders. *P t, 38*(1), 30-57.

Bystritsky, A., & Kronemyer, D. (2014). Stress and anxiety: Counterpart elements of the stress/anxiety complex. *Psychiatric Clinics of North America*, *37*(4), 489-518.

Çalışkan, G., & Stork, O. (2019). Hippocampal network oscillations at the interplay between innate anxiety and learned fear. *Psychopharmacology*, *236*(1), 321-338. https://doi.org/10.1007/s00213-018-5109-z Campbell, M. K., & Quintiliani, L. M. (2006). Tailored Interventions in Public Health:Where Does Tailoring Fit in Interventions to Reduce Health Disparities? *American Behavioral Scientist*, *49*(6), 775-793. <u>https://doi.org/10.1177/0002764205283807</u>

Cantrill, J. A., Sibbald, B., & Buetow, S. (1996). The Delphi and nominal group techniques in health services research. *The International Journal of Pharmacy Practice*, *4*, 7.

Carey, G., Gormezoglu, M., de Jong, J. J. A., Hofman, P. A. M., Backes, W. H., Dujardin, K., & Leentjens, A. F. G. (2021). Neuroimaging of Anxiety in Parkinson's Disease: A Systematic Review. *Mov Disord*, *36*(2), 327-339. <u>https://doi.org/10.1002/mds.28404</u>

Carey, G., Lopes, R., Viard, R., Betrouni, N., Kuchcinski, G., Devignes, Q., Defebvre, L., Leentjens, A. F. G., & Dujardin, K. (2020). Anxiety in Parkinson's disease is associated with changes in the brain fear circuit [Research Support, Non-U.S. Gov't]. *Parkinsonism & Related Disorders, 80*, 89-97.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=prem&AN=3297 9785

Carl, E. C., Smitaa, P., Natalie, I., Caroline, E. R., Francis, D., Rebecca, W., Keith, W., Marion, F. W., & Catherine, M. S. (2016). Physiotherapy and Occupational Therapy vs No Therapy in Mild to Moderate Parkinson Disease: A Randomized Clinical Trial. *JAMA Neurology*, *73*(3), 291-299. <u>https://doi.org/10.1001/jamaneurol.2015.4452</u>

Carroll, C., Clarke, C. E., Grosset, D., Rather, A., Mohamed, B., Parry, M., Reddy, P., Fackrell, R., & Chaudhuri, K. R. (2024). Addressing Comorbidities in People with Parkinson's Disease: Considerations From An Expert Panel. *Journal of Parkinson's Disease*, *14*, 53-63. <u>https://doi.org/10.3233/JPD-230168</u>

Carter, S. A., Tong, A., Craig, J. C., Teixeira-Pinto, A., & Manera, K. E. (2023). Consensus Methods for Health Research in a Global Setting. In P. Liamputtong (Ed.), *Handbook of Social Sciences and Global Public Health* (pp. 959-984). Springer International Publishing. <u>https://doi.org/10.1007/978-3-031-25110-8_65</u> Castellanos, R., Yildiz Spinel, M., Phan, V., Orengo-Aguayo, R., Humphreys, K. L., & Flory, K. (2020). A Systematic Review and Meta-Analysis of Cultural Adaptations of Mindfulness-Based Interventions for Hispanic Populations. *Mindfulness*, *11*(2), 317-332. https://doi.org/10.1007/s12671-019-01210-x

Chagas, M. H. N., Oliveira, T. H. G. F., Linares, I. M. P., Balarini, F. B., Chagas, N. M. S., Tumas, V., & Crippa, J. A. S. (2017). Can anxiety increase tremors in patients with Parkinson's disease? An experimental model. *Archives of Clinical Psychiatry (São Paulo)*, 44.

Chan, A. W., Tetzlaff, J. M., Altman, D. G., Laupacis, A., Gøtzsche, P. C., Krleža-Jerić, K.,
Hróbjartsson, A., Mann, H., Dickersin, K., Berlin, J. A., Doré, C. J., Parulekar, W. R.,
Summerskill, W. S., Groves, T., Schulz, K. F., Sox, H. C., Rockhold, F. W., Rennie, D., & Moher,
D. (2013). SPIRIT 2013 statement: defining standard protocol items for clinical trials. *Ann Intern Med*, *158*(3), 200-207. <u>https://doi.org/10.7326/0003-4819-158-3-201302050-00583</u>

Chaudhuri, K. R., & Schapira, A. H. V. (2009). Non-motor symptoms of Parkinson's disease: dopaminergic pathophysiology and treatment. *The Lancet Neurology*, *8*(5), 464-474. <u>https://doi.org/10.1016/S1474-4422(09)70068-7</u>

Chaudhuri, K. R., Yates, L., & Martinez-Martin, P. (2005). The non-motor symptom complex of Parkinson's disease: a comprehensive assessment is essential. *Current Neurology and Neuroscience Reports*, *5*(4), 275-283.

Chen, J., & Marsh, L. (2014). Anxiety in Parkinson's disease: identification and management. *Therapeutic Advances in Neurological Disorders*, 7(1), 52-59. <u>https://doi.org/10.1177/</u>

Chen, Z., Fan, V. S., Belza, B., Pike, K., & Nguyen, H. Q. (2017). Association between Social Support and Self-Care Behaviors in Adults with Chronic Obstructive Pulmonary Disease. *Ann Am Thorac Soc*, *14*(9), 1419-1427. <u>https://doi.org/10.1513/AnnalsATS.201701-026OC</u>

Cherian, A., & Divya, K. P. (2020). Genetics of Parkinson's disease. *Acta neurologica Belgica*, *120*(6), 1297-1305. <u>https://doi.org/10.1007/s13760-020-01473-5</u>

Chesworth, C., Duffy, R., Hodnett, J., & Knight, A. (2002). Measuring Clinical Effectiveness in Mental Health: is the Canadian Occupational Performance an appropriate Measure? *British Journal of Occupational Therapy*, *65*(1), 30-34.

Christine, A. M. (1990). A Philosophy Unique to Occupational Therapy. *British Journal of Occupational Therapy*, *53*(9), 379-380. <u>https://doi.org/10.1177/030802269005300910</u>

Chugh-Gupta, N., Baldassarre, F. G., & Vrkljan, B. H. (2013). A systematic review of yoga for state anxiety: considerations for occupational therapy [Review Systematic Review]. *Canadian Journal of Occupational Therapy - Revue Canadienne d Ergotherapie, 80*(3), 150-170.

https://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med10&AN =24224228

Clark, D. M. (1986). A cognitive approach to panic. *Behav Res Ther*, *24*(4), 461-470. https://doi.org/10.1016/0005-7967(86)90011-2

Clark, D. M. (2018). Realizing the Mass Public Benefit of Evidence-Based Psychological Therapies: The IAPT Program. *Annual Review of Clinical Psychology*, *14*(1), 159-183. <u>https://doi.org/10.1146/annurev-clinpsy-050817-084833</u>

Clark, F., Jackson, J., Carlson, M., Chou, C.-P., Cherry, B. J., Jordan-Marsh, M., Knight, B. G., Mandel, D., Blanchard, J., Granger, D. A., Wilcox, R. R., Lai, M. Y., White, B., Hay, J., Lam, C., Marterella, A., & Azen, S. P. (2012). Effectiveness of a lifestyle intervention in promoting the well-being of independently living older people: results of the Well Elderly 2 Randomised Controlled Trial. *Journal of Epidemiology and Community Health*, *66*(9), 782-790. <u>https://doi.org/10.1136/jech.2009.099754</u>

The Concept System® groupwisdom™. In. (2020). (Version Build 2019.24.01) [Web-based Platform]. Concept Systems, Incorporated,. <u>https://www.groupwisdom.tech</u>

Conner, M., & Norman, P. (2017). Health behaviour: Current issues and challenges. *Psychology & Health*, *32*(8), 895-906. <u>https://doi.org/10.1080/08870446.2017.1336240</u>

Cook, K., & Bergeron, K. (2019). Using Group Concept Mapping to Engage a Hard-to-Reach Population in Research: Young Adults With Life-Limiting Conditions. *International Journal of Qualitative Methods*, *18*. <u>https://doi.org/10.1177/1609406919891315</u>

Cook, T. D. (1979). Qualitative and Quantitative Methods in Evaluation. Sage.

Correll, T., Gentile, J., & Correll, A. (2023). Healthy Lifestyle Interventions Augmenting Psychotherapy in Anxiety and PTSD. *Innov Clin Neurosci*, *20*(7-9), 18-26.

Costa, F. P., Diaferia, G., & Behlau, M. (2016). Communicative aspects and coping strategies in patients with Parkinson's disease. *Codas*, *28*(1), 46-52. <u>https://doi.org/10.1590/2317-1782/20162015054</u>

Coto-Lesmes, R., Fernández-Rodríguez, C., & González-Fernández, S. (2020). Acceptance and Commitment Therapy in group format for anxiety and depression. A systematic review. *Journal of Affective Disorders*, *263*, 107-120.

Coventry, P. A., Dickens, C., & Todd, C. (2014). How does mental-physical multimorbidity express itself in lived time and space? A phenomenological analysis of encounters with depression and chronic physical illness. *Social Science and Medicine*, *118*(C), 108-118. <u>https://doi.org/10.1016/j.socscimed.2014.07.068</u>

Craig, P., Di Ruggiero, E., Frohlich, K. L., Mykhalovskiy, E., White, M., Campbell, R., Cummins, S., Edwards, N., Hunt, K., & Kee, F. (2018). Taking account of context in population health intervention research: guidance for producers, users and funders of research.

Craig, P., Dieppe, P., Macintyre, S., Michie, S., Nazareth, I., Petticrew, M., & Medical Research Council, G. (2008). Developing and evaluating complex interventions: the new Medical Research Council guidance. *BMJ (Clinical research ed.)*, *337*, a1655-a1655. https://doi.org/10.1136/bmj.a1655 Creighton, C. (1992). The origin and evolution of activity analysis. *Am J Occup Ther*, *46*(1), 45-48.

Criaud, M., Kim, J. H., Zurowski, M., Lobaugh, N., Chavez, S., Houle, S., & Strafella, A. P. (2021). Anxiety in Parkinson's disease: Abnormal resting activity and connectivity. *Brain Res*, *1753*, 147235. <u>https://doi.org/10.1016/j.brainres.2020.147235</u>

Crotty, M. (1998). *The foundations of social research: Meaning and perspective in the research process.* Sage Publications.

Cruz Rivera, S., Kyte, D. G., Aiyegbusi, O. L., Keeley, T. J., & Calvert, M. J. (2017). Assessing the impact of healthcare research: A systematic review of methodological frameworks. *PLoS Med*, *14*(8), e1002370. <u>https://doi.org/10.1371/journal.pmed.1002370</u>

CSDH. (2008). Closing the gap in a generation: health equity through action on the social determinants of health. Final Report of the Commission on Social Determinants of Health [Report].

Cuhls, K. (2023). The Delphi Method: An Introduction. In M. Niederberger & O. Renn (Eds.), Delphi Methods In The Social And Health Sciences: Concepts, applications and case studies (pp. 3-27). Springer Fachmedien Wiesbaden. <u>https://doi.org/10.1007/978-3-658-38862-1_1</u>

Cup, E. H. C., Scholte Op Reimer, W. J. M., Thijssen, M. C. E., & van Kuyk-Minis, M. A. H. (2003). Reliability and validity of the Canadian Occupational Performance Measure in stroke patients. *Clinical Rehabilitation*, *17*(4), 402. <u>https://doi.org/10.1191/0269215503cr6350a</u>

CYPH, INC. v. ZOOM VIDEO COMMUNICATIONS, INC, (Dist. Court, ND California 2022).

Dan, R., Růžička, F., Bezdicek, O., Růžička, E., Roth, J., Vymazal, J., Goelman, G., & Jech, R. (2017). Separate neural representations of depression, anxiety and apathy in Parkinson's disease. *Sci Rep*, *7*(1), 12164. <u>https://doi.org/10.1038/s41598-017-12457-6</u>

Datta, J., & Petticrew, M. (2013). Challenges to evaluating complex interventions: a content analysis of published papers. *BMC Public Health*, *13*(1), 568. <u>https://doi.org/10.1186/1471-</u>2458-13-568

de Lau, L. M., & Breteler, M. M. (2006). Epidemiology of Parkinson's disease. *5*, 525-535. http://dx.doi.org/10.1016/S1474-4422(06)70471-9

De Micco, R., Satolli, S., Siciliano, M., Di Nardo, F., Caiazzo, G., Russo, A., Giordano, A., Esposito, F., Tedeschi, G., & Tessitore, A. (2021). Connectivity Correlates of Anxiety Symptoms in Drug-Naive Parkinson's Disease Patients. *Mov Disord*, *36*(1), 96-105. <u>https://doi.org/10.1002/mds.28372</u>

de Oliveira, A. R., Reimer, A. E., de Macedo, C. E., de Carvalho, M. C., Silva, M. A., & Brandão, M. L. (2011). Conditioned fear is modulated by D2 receptor pathway connecting the ventral tegmental area and basolateral amygdala. *Neurobiol Learn Mem*, *95*(1), 37-45. <u>https://doi.org/10.1016/j.nlm.2010.10.005</u>

Deane, K. H. O., Flaherty, H., Daley, D. J., Pascoe, R., Penhale, B., Clarke, C. E., Sackley, C., & Storey, S. (2014). Priority setting partnership to identify the top 10 research priorities for the management of Parkinson's disease. *BMJ Open*, *4*(12), e006434-e006434.

https://doi.org/10.1136/bmjopen-2014-006434

 Dedding, C., Cardol, M., Eyssen, I. C. J. M., Dekker, J., & Beelen, A. (2004). Validity of the Canadian Occupational Performance Measure: a client-centred outcome measurement. *Clinical Rehabilitation*, *18*(6), 660.
 <u>https://doi.org/10.1191/0269215504cr746oa</u>

Delgadillo, J., Dawson, A., Gilbody, S., & Böhnke, J. R. (2017). Impact of long-term medical conditions on the outcomes of psychological therapy for depression and anxiety. *The British Journal of Psychiatry*, *210*(1), 47-53. <u>https://doi.org/10.1192/bjp.bp.116.189027</u>

Dettmann, L. M., Adams, S., & Taylor, G. (2022). Investigating the prevalence of anxiety and depression during the first COVID-19 lockdown in the United Kingdom: Systematic review

and meta-analyses. *British Journal of Clinical Psychology*, *61*(3), 757-780. <u>https://doi.org/https://doi.org/10.1111/bjc.12360</u>

Devaney, L., & Henchion, M. (2018). Who is a Delphi 'expert'? Reflections on a bioeconomy expert selection procedure from Ireland. *Futures*, *99*, 45-55. <u>https://doi.org/https://doi.org/10.1016/j.futures.2018.03.017</u>

Di Bona, L., Wenborn, J., Field, B., Hynes, S. M., Ledgerd, R., Mountain, G., & Swinson, T. (2017). Enablers and challenges to occupational therapists' research engagement: A qualitative study. *British Journal of Occupational Therapy, 80*(11), 642-650. https://doi.org/10.1177/0308022617719218

Diagnostic and statistical manual of mental disorders : DSM-5™. (2013). (5th edition. ed.). American Psychiatric Publishing, a division of American Psychiatric Association.

Dijkman, B. L., Luttik, M. L., Van der Wal-Huisman, H., Paans, W., & van Leeuwen, B. L. (2022). Factors influencing family involvement in treatment decision-making for older patients with cancer: A scoping review. *Journal of Geriatric Oncology*, *13*(4), 391-397. https://doi.org/https://doi.org/10.1016/j.jgo.2021.11.003

Dilgul, M., McNamee, P., Orfanos, S., Carr, C. E., & Priebe, S. (2018). Why do psychiatric patients attend or not attend treatment groups in the community: A qualitative study. *PLoS ONE*, *13*(12), e0208448. <u>https://doi.org/10.1371/journal.pone.0208448</u>

Dissanayaka, N. N., Sellbach, A., Matheson, S., O'Sullivan, J. D., Silburn, P. A., Byrne, G. J., Marsh, R., & Mellick, G. D. (2010). Anxiety disorders in Parkinson's disease: prevalence and risk factors [Research Support, Non-U.S. Gov't]. *Movement Disorders*, *25*(7), 838-845. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med8&AN=2046</u> <u>1800</u>

Dissanayaka, N. N., White, E., O'Sullivan, J. D., Marsh, R., Pachana, N. A., & Byrne, G. J. (2014). The clinical spectrum of anxiety in Parkinson's disease [Research Support, Non-U.S. Gov't Review]. *Movement Disorders*, *29*(8), 967-975.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med11&AN=250
43800

Dissanayaka, N. N. W., Lawson, R. A., Yarnall, A. J., Duncan, G. W., Breen, D. P., Khoo, T. K., Barker, R. A., & Burn, D. J. (2017). Anxiety is associated with cognitive impairment in newlydiagnosed Parkinson's disease. *Parkinsonism Relat Disord*, *36*, 63-68. https://doi.org/10.1016/j.parkreldis.2017.01.001

Donohoe, H. M., & Needham, R. D. (2009). Moving best practice forward: Delphi characteristics, advantages, potential problems, and solutions. *International Journal of Tourism Research*, *11*(5), 415-437. <u>https://doi.org/https://doi.org/10.1002/jtr.709</u>

Dorsey, E. R., Constantinescu, R., Thompson, J. P., Biglan, K. M., Holloway, R. G., Kieburtz, K., Marshall, F. J., Ravina, B. M., Schifitto, G., Siderowf, A., & Tanner, C. M. (2007). Projected number of people with Parkinson disease in the most populous nations, 2005 through 2030. *Neurology*, *68*(5), 384-386. <u>https://doi.org/10.1212/01.wnl.0000247740.47667.03</u>

Dorsey, E. R., Elbaz, A., Nichols, E., Abd-Allah, F., Abdelalim, A., Adsuar, J. C., Ansha, M. G., Brayne, C., Choi, J.-Y. J., Collado-Mateo, D., Dahodwala, N., Do, H. P., Edessa, D., Endres, M., Fereshtehnejad, S.-M., Foreman, K. J., Gankpe, F. G., Gupta, R., Hankey, G. J.,...Murray, C. J. L. (2018). Global, regional, and national burden of Parkinson's disease, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. *The Lancet Neurology*, *17*(11), 939-953. <u>https://doi.org/10.1016/s1474-4422(18)30295-3</u>

Dorsey, E. R., Sherer, T., Okun, M. S., & Bloem, B. R. (2018). The Emerging Evidence of the Parkinson Pandemic. *J Parkinsons Dis*, 8(s1), S3-s8. <u>https://doi.org/10.3233/jpd-181474</u>

Dujardin, K., & Sgambato, V. (2020). Neuropsychiatric Disorders in Parkinson's Disease: What Do We Know About the Role of Dopaminergic and Non-dopaminergic Systems? [Review]. *Frontiers in Neuroscience*, *14*. <u>https://doi.org/10.3389/fnins.2020.00025</u>

Duncan, G. W., Khoo, T. K., Yarnall, A. J., O'Brien, J. T., Coleman, S. Y., Brooks, D. J., Barker, R. A., & Burn, D. J. (2014). Health-related quality of life in early Parkinson's disease: The impact

of nonmotor symptoms. *Movement Disorders*, 29(2), 195-202. https://doi.org/10.1002/mds.25664

Duncan, R. P., & Earhart, G. M. (2011). Measuring participation in individuals with Parkinson disease: relationships with disease severity, quality of life, and mobility. *Disability and Rehabilitation*, *33*(15-16), 1440-1446. <u>https://doi.org/10.3109/09638288.2010.533245</u>

Edelman, A., Marten, R., Montenegro, H., Sheikh, K., Barkley, S., Ghaffar, A., Dalil, S., & Topp, S. M. (2021). Modified scoping review of the enablers and barriers to implementing primary health care in the COVID-19 context. *Health Policy and Planning*, *36*(7), 1163-1186. <u>https://doi.org/10.1093/heapol/czab075</u>

Egan, S. J., Laidlaw, K., & Starkstein, S. (2015). Cognitive Behaviour Therapy for Depression and Anxiety in Parkinson's Disease. *Journal of Parkinsons Disease*, *5*(3), 443-451. <u>https://doi.org/10.3233/JPD-150542</u>

Ehgoetz Martens, K. a., Ellard, C. G., & Almeida, Q. J. (2014). Does Anxiety Cause Freezing of Gait in Parkinson's Disease? *PLoS ONE*, *9*(9), e106561-e106561. https://doi.org/10.1371/journal.pone.0106561

Ehring, T., & Watkins, E. R. (2008). Repetitive negative thinking as a transdiagnostic process. *International journal of cognitive therapy*, *1*(3), 192-205.

Eklund, M. (2013). Anxiety, Depression, and Stress Among Women in Work Rehabilitation for Stress-Related Disorders [Article]. *International Journal of Mental Health*, *42*(4), 37-47. https://doi.org/10.2753/IMH0020-7411420402

Eklund, M., & Erlandsson, L. K. (2013). Quality of life and client satisfaction as outcomes of the Redesigning Daily Occupations (ReDO) programme for women with stress-related disorders: a comparative study. *Work*, *46*(1), 51-58. <u>https://doi.org/10.3233/wor-121524</u>

Eklund, M., Wästberg, B. A., & Erlandsson, L. K. (2013). Work outcomes and their predictors in the Redesigning Daily Occupations (Re DO) rehabilitation programme for women with

stress-related disorders [Article]. *Australian Occupational Therapy Journal, 60*(2), 85-92. https://doi.org/10.1111/1440-1630.12019

Ellis, T., Boudreau, J. K., DeAngelis, T. R., Brown, L. E., Cavanaugh, J. T., Earhart, G. M., Ford,
M. P., Foreman, K. B., & Dibble, L. E. (2013). Barriers to Exercise in People With Parkinson
Disease. *Physical Therapy*, *93*(5), 628-636. <u>https://doi.org/10.2522/ptj.20120279</u>

Emamzadeh, F. N., & Surguchov, A. (2018). Parkinson's Disease: Biomarkers, Treatment, and Risk Factors. *Front Neurosci*, *12*, 612. <u>https://doi.org/10.3389/fnins.2018.00612</u>

Emerson, R. W. (2015). Convenience Sampling, Random Sampling, and Snowball Sampling: How Does Sampling Affect the Validity of Research? *Journal of Visual Impairment & Blindness*, *109*(2), 164-168.

http://library.gcu.edu:2048/login?url=http://search.ebscohost.com/login.aspx?direct=true& db=ccm&AN=2012938256&site=eds-live&scope=site

Engin, E., Smith, K. S., Gao, Y., Nagy, D., Foster, R. A., Tsvetkov, E., Keist, R., Crestani, F., Fritschy, J. M., Bolshakov, V. Y., Hajos, M., Heldt, S. A., & Rudolph, U. (2016). Modulation of anxiety and fear via distinct intrahippocampal circuits. *Elife*, *5*, e14120. https://doi.org/10.7554/eLife.14120

Epley, E., Wolske, J., Lee, J., Mirza, M., & Fisher, G. (2021). Habits and Health Promotion in Occupational Therapy: A Scoping Review. *Annals of International Occupational Therapy*, 4(4), e269-e277. <u>https://doi.org/doi:10.3928/24761222-20210921-04</u>

Erlandsson, L.-K. (2013). The Redesigning Daily Occupations (ReDO)-Program: Supporting Women With Stress-Related Disorders to Return to Work—Knowledge Base, Structure, and Content [Journal Article]. *Occupational Therapy in Mental Health*, *29*(1), 85-101. <u>https://doi.org/10.1080/0164212X.2013.761451</u>

Erro, R., Pappatà, S., Amboni, M., Vicidomini, C., Longo, K., Santangelo, G., Picillo, M., Vitale, C., Moccia, M., Giordano, F., Brunetti, A., Pellecchia, M. T., Salvatore, M., & Barone, P. (2012). Anxiety is associated with striatal dopamine transporter availability in newly diagnosed untreated Parkinson's disease patients. *Parkinsonism Relat Disord*, *18*(9), 1034-1038. <u>https://doi.org/10.1016/j.parkreldis.2012.05.022</u>

Evans, R. G., & Stoddart, G. L. (2017). Producing health, consuming health care. In *Why are some people healthy and others not*? (pp. 27-64). Routledge.

Eyssen, I. C., Steultjens, M. P., Oud, T. A., Bolt, E. M., Maasdam, A., & Dekker, J. (2011). Responsiveness of the Canadian occupational performance measure. *J Rehabil Res Dev*, *48*(5), 517-528. <u>https://doi.org/10.1682/jrrd.2010.06.0110</u>

Eyssen, I. C. J. M., Beelen, A., Dedding, C., Cardol, M., & Dekker, J. (2005). The reproducibility of the Canadian Occupational Performance Measure. *Clinical Rehabilitation*, *19*(8), 888-894. <u>https://doi.org/10.1191/0269215505cr8830a</u>

Fang, C., Lv, L., Mao, S., Dong, H., & Liu, B. (2020). Cognition Deficits in Parkinson's Disease: Mechanisms and Treatment. *Parkinsons Dis, 2020*, 2076942. <u>https://doi.org/10.1155/2020/2076942</u>

Farchione, T. J., Fairholme, C. P., Ellard, K. K., Boisseau, C. L., Thompson-Hollands, J., Carl, J. R., Gallagher, M. W., & Barlow, D. H. (2012). Unified protocol for transdiagnostic treatment of emotional disorders: a randomized controlled trial. *Behavior Therapy*, *43*(3), 666-678.

Farombi, T. H., Owolabi, M. O., & Ogunniyi, A. (2016). Falls and Their Associated Risks in Parkinson's Disease Patients in Nigeria. *Journal of movement disorders*, *9*(3), 160-165. <u>https://doi.org/10.14802/jmd.16011</u>

Farquhar, M., Preston, N., Evans, C. J., Grande, G., Short, V., Benalia, H., Higginson, I. J., & Todd, o. b. o. M., Chris. (2013). Mixed Methods Research in the Development and Evaluation of Complex Interventions in Palliative and End-of-Life Care: Report on the MORECare Consensus Exercise. *Journal of Palliative Medicine*, *16*(12), 1550-1560. <u>https://doi.org/10.1089/ipm.2012.0572</u> Fernández-Baizán, C., García, M. P. F., Díaz-Cáceres, E., Menéndez-González, M., Arias, J. L., & Méndez, M. (2022). Activities of Daily Living in Parkinson Disease: Are They Related to Short Term and Working Memory and Visuospatial Abilities? *Topics in Geriatric Rehabilitation*, *38*(4).

https://journals.lww.com/topicsingeriatricrehabilitation/fulltext/2022/10000/activities of daily living in parkinson disease .7.aspx

Fernie, B. A., Spada, M. M., & Brown, R. G. (2019). Motor fluctuations and psychological distress in Parkinson's disease. *Health Psychology*, *38*(6), 518-526. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med16&AN=309</u> 73744

Ferreira, R. M., Alves, W. M. G. d. C., Lima, T. A., Alves, T. G. G., Filho, P. A. M. A., Pimentel, C. P., Sousa, E. C., & Cortinhas-Alves, E. A. (2018). The effect of resistance training on the anxiety symptoms and quality of life in elderly people with Parkinson's disease: A randomized controlled trial. *Arquivos de Neuro-Psiquiatria*, *76*(8), 499-506. https://doi.org/http://dx.doi.org/10.1590/0004-282x20180071

Fineberg, N. A., Haddad, P. M., Carpenter, L., Gannon, B., Sharpe, R., Young, A. H., Joyce, E., Rowe, J., Wellsted, D., Nutt, D. J., & Sahakian, B. J. (2013). The size, burden and cost of disorders of the brain in the UK. *Journal of Psychopharmacology*, *27*(9), 761-770. <u>https://doi.org/10.1177/0269881113495118</u>

Fisher, A. G. (2013). Occupation-centred, occupation-based, occupation-focused: same, same or different? *Scand J Occup Ther*, *20*(3), 162-173. https://doi.org/10.3109/11038128.2012.754492

Fisher, A. G., & Marterella, A. (2019). Powerful practice: A model for authentic occupational therapy. (*No Title*).

Flavell, J. H. (1979). Metacognition and cognitive monitoring: A new area of cognitive– developmental inquiry. *American psychologist*, *34*(10), 906. Flores-Cuadrado, A., Ubeda-Bañon, I., Saiz-Sanchez, D., de la Rosa-Prieto, C., & Martinez-Marcos, A. (2015). α -Synuclein staging in the amygdala of a Parkinson's disease model: cell types involved. *Eur J Neurosci*, 41(1), 137-146. <u>https://doi.org/10.1111/ejn.12763</u>

Flores-Cuadrado, A., Ubeda-Bañon, I., Saiz-Sanchez, D., & Martinez-Marcos, A. (2017). α-Synucleinopathy in the Human Amygdala in Parkinson Disease: Differential Vulnerability of Somatostatin- and Parvalbumin-Expressing Neurons. *J Neuropathol Exp Neurol*, *76*(9), 754-758. <u>https://doi.org/10.1093/jnen/nlx054</u>

Floresco, S. B., & Tse, M. T. (2007). Dopaminergic regulation of inhibitory and excitatory transmission in the basolateral amygdala-prefrontal cortical pathway. *J Neurosci, 27*(8), 2045-2057. <u>https://doi.org/10.1523/jneurosci.5474-06.2007</u>

Forsyth, A., Deane, F. P., & Williams, P. (2015). A lifestyle intervention for primary care patients with depression and anxiety: A randomised controlled trial. *Psychiatry Research*, *230*(2), 537-544. <u>https://doi.org/https://doi.org/10.1016/j.psychres.2015.10.001</u>

Fosha, D. (2000). Meta-therapeutic processes and the affects of transformation: Affirmation and the healing affects. *Journal of Psychotherapy Integration*, *10*, 71-97.

Fox, J., Erlandsson, L.-K., & Shiel, A. (2019). A systematic review and narrative synthesis of occupational therapy-led interventions for individuals with anxiety and stress-related disorders. *Occupational Therapy in Mental Health*, *35*(2), 179-204. <u>https://doi.org/10.1080/0164212x.2018.1516172</u>

Fox, J., Erlandsson, L. K., & Shiel, A. (2021). A feasibility study of the Redesigning Daily Occupations (ReDOTM-10) programme in an Irish context. *Scandinavian Journal of Occupational Therapy*, 1-15.

https://doi.org/https://dx.doi.org/10.1080/11038128.2021.1882561

Friedman, A. J., Cosby, R., Boyko, S., Hatton-Bauer, J., & Turnbull, G. (2011). Effective Teaching Strategies and Methods of Delivery for Patient Education: A Systematic Review and Practice Guideline Recommendations. *Journal of Cancer Education*, *26*(1), 12-21. https://doi.org/10.1007/s13187-010-0183-x

Frost, H., Campbell, P., Maxwell, M., O'Carroll, R. E., Dombrowski, S. U., Williams, B., Cheyne, H., Coles, E., & Pollock, A. (2018). Effectiveness of Motivational Interviewing on adult behaviour change in health and social care settings: A systematic review of reviews. *PLoS ONE*, *13*(10), e0204890. <u>https://doi.org/10.1371/journal.pone.0204890</u>

Frost, R., Rait, G., Wheatley, A., Wilcock, J., Robinson, L., Harrison Dening, K., Allan, L., Banerjee, S., Manthorpe, J., Walters, K., & team, t. P. S. p. (2020). What works in managing complex conditions in older people in primary and community care? A state-of-the-art review. *Health & Social Care in the Community*, *28*(6), 1915-1927. https://doi.org/https://doi.org/10.1111/hsc.13085

G.B.D. Neurology Collaborators. (2019). Global, regional, and national burden of neurological disorders, 1990-2016: a systematic analysis for the Global Burden of Disease Study 2016 [Research Support, N.I.H., Extramural Research Support, U.S. Gov't, Non-P.H.S.]. *Lancet Neurology*, *18*(5), 459-480.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medc&AN=3087 9893

Galaviz, K. I., Narayan, K. M. V., Lobelo, F., & Weber, M. B. (2018). Lifestyle and the Prevention of Type 2 Diabetes: A Status Report. *Am J Lifestyle Med*, *12*(1), 4-20. <u>https://doi.org/10.1177/1559827615619159</u>

Gallo, K. P., & Barlow, D. H. (2012). Factors involved in clinician adoption and nonadoption of evidence-based interventions in mental health. *Clinical Psychology: Science and Practice*, *19*(1), 93-106. <u>https://doi.org/10.1111/j.1468-2850.2012.01276.x</u>

Garcia-Ptacek, S., & Kramberger, M. G. (2016). Parkinson Disease and Dementia. *J Geriatr Psychiatry Neurol*, *29*(5), 261-270. <u>https://doi.org/10.1177/0891988716654985</u>

Gaudet, P. (2002). Measuring the impact of Parkinson's disease: an occupational therapy perspective. *Canadian Journal of Occupational Therapy*, *69*(2), 104-113. <u>https://search.ebscohost.com/login.aspx?direct=true&AuthType=athens&db=cin20&AN=10</u> 6943446&site=ehost-live

Geleto, A., Chojenta, C., Musa, A., & Loxton, D. (2018). Barriers to access and utilization of emergency obstetric care at health facilities in sub-Saharan Africa: a systematic review of literature. *Systematic Reviews*, 7(1), 183. <u>https://doi.org/10.1186/s13643-018-0842-2</u>

George-Paschal, L., Krusen, N. E., & Fan, C.-W. (2022). Psychometric evaluation of the Relative Mastery Scale: An Occupational Adaptation instrument. *OTJR: Occupational Therapy Journal of Research*, *42*(2), 154-161. <u>https://doi.org/10.1177/15394492211060877</u>

Gerritzen, E. V., Lee, A. R., McDermott, O., Coulson, N., & Orrell, M. (2022). Online Peer Support for People With Parkinson Disease: Narrative Synthesis Systematic Review. *JMIR Aging*, *5*(3), e35425. <u>https://doi.org/10.2196/35425</u>

Gesell, S. B., Prvu Bettger, J., Lawrence, R. H., Li, J., Hoffman, J., Lutz, B. J., Grudzen, C., Johnson, A. M., Krishnan, J. A., Hsu, L. L., Zwart, D., Williams, M. V., & Schnipper, J. L. (2021). Implementation of Complex Interventions: Lessons Learned From the Patient-Centered Outcomes Research Institute Transitional Care Portfolio. *Medical Care*, *59*, S344-S354. <u>https://doi.org/10.1097/mlr.00000000001591</u>

Ghielen, I., Koene, P., Twisk, J. W., Kwakkel, G., Heuvel, O. A. v. d., & Wegen, E. E. v. (2020). The association between freezing of gait, fear of falling and anxiety in Parkinson's disease: a longitudinal analysis. *Neurodegenerative Disease Management*, *10*(3), 159-168. <u>https://doi.org/10.2217/nmt-2019-0028</u>

Ghielen, I., van Wegen, E. E. H., Rutten, S., de Goede, C. J. T., Houniet-de Gier, M., Collette,
E. H., Burgers-Bots, I. A. L., Twisk, J. W. R., Kwakkel, G., Vermunt, K., van Vliet, B., Berendse,
H. W., & van den Heuvel, O. A. (2017). Body awareness training in the treatment of wearing-off related anxiety in patients with Parkinson's disease: Results from a pilot randomized

controlled trial. J Psychosom Res, 103, 1-8.

https://doi.org/10.1016/j.jpsychores.2017.09.008

Giddings, L. S., & Grant, B. M. (2007). A Trojan horse for positivism?: a critique of mixed methods research. *ANS Adv Nurs Sci*, *30*(1), 52-60. <u>https://doi.org/10.1097/00012272-</u> 200701000-00006

Gilboa, Y., & Helmer, A. (2020). Self-Management Intervention for Attention and Executive Functions Using Equine-Assisted Occupational Therapy Among Children Aged 6–14 Diagnosed with Attention Deficit/Hyperactivity Disorder. *The Journal of Alternative and Complementary Medicine*, *26*(3), 239-246. <u>https://doi.org/10.1089/acm.2019.0374</u>

Gilmartin, M. R., Balderston, N. L., & Helmstetter, F. J. (2014). Prefrontal cortical regulation of fear learning. *Trends Neurosci*, *37*(8), 455-464. <u>https://doi.org/10.1016/j.tins.2014.05.004</u>

Gogolla, N. (2017). The insular cortex. *Curr Biol*, *27*(12), R580-r586. https://doi.org/10.1016/j.cub.2017.05.010

Goltz, F., van der Heide, A., & Helmich, R. C. (2024). Alleviating Stress in Parkinson's Disease: Symptomatic Treatment, Disease Modification, or Both? *J Parkinsons Dis*. <u>https://doi.org/10.3233/jpd-230211</u>

Gómez-Soria, I., Ferreira, C., Oliván-Blázquez, B., Aguilar-Latorre, A., & Calatayud, E. (2023). Effects of cognitive stimulation program on cognition and mood in older adults, stratified by cognitive levels: A randomized controlled trial. *Archives of Gerontology and Geriatrics*, *110*, 104984. <u>https://doi.org/https://doi.org/10.1016/j.archger.2023.104984</u>

Gracey, F., Longworth, C., & Psaila, K. (2016). A provisional transdiagnostic cognitive behavioural model of post brain injury emotional adjustment. *Neuro-Disability and Psychotherapy*, *3*(2), 154-185.

Graham, C. D., Gouick, J., Krahé, C., & Gillanders, D. (2016). A systematic review of the use of Acceptance and Commitment Therapy (ACT) in chronic disease and long-term conditions.

Clinical Psychology Review, 46, 46-58.

https://doi.org/https://doi.org/10.1016/j.cpr.2016.04.009

Grajo, L., Boisselle, A., & DaLomba, E. (2018). Occupational Adaptation as a Construct: A Scoping Review of Literature. *The Open Journal of Occupational Therapy*, *6*(1). <u>https://doi.org/10.15453/2168-6408.1400</u>

Green, S., & Lambert, R. (2017). A systematic review of health economic evaluations in occupational therapy. *British Journal of Occupational Therapy*, *80*(1), 5-19. <u>https://doi.org/10.1177/0308022616650898</u>

Greene, J. C. (2008). Is Mixed Methods Social Inquiry a Distinctive Methodology? *Journal of Mixed Methods Research*, 2(1), 7-22. <u>https://doi.org/10.1177/1558689807309969</u>

Greene, J. C., Caracelli, V. J., & Graham, W. F. (1989). Toward a Conceptual Framework for Mixed-Method Evaluation Designs. *Educational Evaluation and Policy Analysis*, *11*(3), 255-274. <u>https://doi.org/10.3102/01623737011003255</u>

Greenhalgh, T. (2013). Why do we always end up here? Evidence-based medicine's conceptual cul-de-sacs and some off-road alternative routes. *Int J Prosthodont*, *26*(1), 11-15.

Greenhalgh, T., & Papoutsi, C. (2018). Studying complexity in health services research: desperately seeking an overdue paradigm shift. *BMC medicine*, *16*(1), 95. https://doi.org/10.1186/s12916-018-1089-4

Gregorio, M. V., Lopez, L. L., De Dios, N. A., Casimiro, R., & Ramos, V. (2023). Rural Healthcare Service Delivery in Pantabangan, Nueva Ecija, Philippines: Basis for Rural Health Program and Service Planning. *The Quest: Journal of Multidisciplinary Research and Development*, 2(3).

Grogans, S. E., Bliss-Moreau, E., Buss, K. A., Clark, L. A., Fox, A. S., Keltner, D., Cowen, A. S., Kim, J. J., Kragel, P. A., MacLeod, C., Mobbs, D., Naragon-Gainey, K., Fullana, M. A., & Shackman, A. J. (2023). The nature and neurobiology of fear and anxiety: State of the science and opportunities for accelerating discovery. *Neuroscience & Biobehavioral Reviews*, 151, 105237. <u>https://doi.org/https://doi.org/10.1016/j.neubiorev.2023.105237</u>

Gunnarsson, A. B., Jansson, J. A., & Eklund, M. (2006). The Tree Theme Method in psychosocial occupational therapy: a case study [Case Reports Research Support, Non-U.S. Gov't]. *Scandinavian Journal of Occupational Therapy*, *13*(4), 229-240. <u>https://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med6&AN=</u> <u>17203673</u>

Gunnarsson, A. B., Wagman, P., Hakansson, C., & Hedin, K. (2015). The Tree Theme Method R (TTM), an occupational therapy intervention for treating depression and anxiety: study protocol of a randomized controlled trial [Randomized Controlled Trial Research Support, Non-U.S. Gov't]. *BMC psychology*, *3*, 40. <u>https://doi.org/https://dx.doi.org/10.1186/s40359-015-0097-9</u>

Gunnarsson, B., Wagman, P., Hedin, K., & Hakansson, C. (2018). Treatment of depression and/or anxiety - outcomes of a randomised controlled trial of the tree theme method R versus regular occupational therapy [Comparative Study Randomized Controlled Trial]. *BMC psychology*, *6*(1), 25. <u>https://doi.org/https://dx.doi.org/10.1186/s40359-018-0237-0</u>

Haahr, A., Groos, H., & Sørensen, D. (2021). 'Striving for normality' when coping with Parkinson's disease in everyday life: A metasynthesis. *International Journal of Nursing Studies*, *118*, 103923. <u>https://doi.org/https://doi.org/10.1016/j.ijnurstu.2021.103923</u>

Haddaway, N. R., Page, M. J., Pritchard, C. C., & McGuinness, L. A. (2022). PRISMA2020: An R package and Shiny app for producing PRISMA 2020-compliant flow diagrams, with interactivity for optimised digital transparency and Open Synthesis [https://doi.org/10.1002/cl2.1230]. *Campbell Systematic Reviews*, *18*(2), e1230. https://doi.org/https://doi.org/10.1002/cl2.1230

Hallion, L. S., & Ruscio, A. M. (2013). Should uncontrollable worry be removed from the definition of GAD? A test of incremental validity. *Journal of abnormal psychology*, *122*(2), 369-375. <u>https://doi.org/10.1037/a0031731</u>

Han, J. S., McMahan, R. W., Holland, P., & Gallagher, M. (1997). The role of an amygdalonigrostriatal pathway in associative learning. *J Neurosci*, *17*(10), 3913-3919. <u>https://doi.org/10.1523/jneurosci.17-10-03913.1997</u>

Hanagasi, H. A., Tufekcioglu, Z., & Emre, M. (2017). Dementia in Parkinson's disease. J Neurol Sci, 374, 26-31. <u>https://doi.org/10.1016/j.jns.2017.01.012</u>

Hand, C., Law, M., & McColl, M. A. (2011). Occupational Therapy Interventions for Chronic Diseases: A Scoping Review. *The American Journal of Occupational Therapy*, *65*(4), 428-436. <u>https://doi.org/10.5014/ajot.2011.002071</u>

Hanna, K. K., & Cronin-Golomb, A. (2012). Impact of anxiety on quality of life in parkinson's disease. *Parkinson's Disease*, 2012. <u>https://doi.org/10.1155/2012/640707</u>

Hanssen, D. J. C., Ras, A., & Rosmalen, J. G. M. (2021). Barriers and facilitators to the implementation of interventions for medically unexplained symptoms in primary care: A modified Delphi study. *Journal of Psychosomatic Research*, *143*, 110386. <u>https://doi.org/https://doi.org/10.1016/j.jpsychores.2021.110386</u>

Harding, A. J., Stimson, E., Henderson, J. M., & Halliday, G. M. (2002). Clinical correlates of selective pathology in the amygdala of patients with Parkinson's disease [Research Support, Non-U.S. Gov't]. *Brain*, *125*(Pt 11), 2431-2445.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med4&AN=1239 0970

Hardison, M. E., & Roll, S. C. (2016). Mindfulness Interventions in Physical Rehabilitation: A Scoping Review. *Am J Occup Ther*, *70*(3), 7003290030p7003290031-7003290039. https://doi.org/10.5014/ajot.2016.018069

Hardison, M. E., Unger, J., & Roll, S. C. (2021). Hand Therapy Patients' Psychosocial Symptomology and Interests in Mindfulness: A Cross-Sectional Study. *Canadian Journal of*

Occupational Therapy - Revue Canadienne d Ergotherapie, 84174211060120. <u>https://doi.org/https://dx.doi.org/10.1177/00084174211060120</u>

Harris, P. A., Taylor, R., Minor, B. L., Elliott, V., Fernandez, M., O'Neal, L., McLeod, L., Delacqua, G., Delacqua, F., Kirby, J., Duda, S. N., & Consortium, R. E. (2019). The REDCap consortium: Building an international community of software platform partners. *J Biomed Inform*, *95*, 103208. <u>https://doi.org/10.1016/j.jbi.2019.103208</u>

Hawe, P. (2015). Lessons from Complex Interventions to Improve Health. *Annual Review of Public Health*, *36*(1), 307-323. <u>https://doi.org/10.1146/annurev-publhealth-031912-114421</u>

Hayano, F., Nakamura, M., Asami, T., Uehara, K., Yoshida, T., Roppongi, T., Otsuka, T., Inoue, T., & Hirayasu, Y. (2009). Smaller amygdala is associated with anxiety in patients with panic disorder. *Psychiatry Clin Neurosci, 63*(3), 266-276. <u>https://doi.org/10.1111/j.1440-1819.2009.01960.x</u>

Hayes-Skelton, S. A., Roemer, L., Orsillo, S. M., & Borkovec, T. D. (2013). A contemporary view of applied relaxation for generalized anxiety disorder. In (Vol. 42, pp. 292-302): Taylor & Francis.

He, Z., Tang, X., Yang, X., Guo, Y., George, T. J., Charness, N., Quan Hem, K. B., Hogan, W., & Bian, J. (2020). Clinical Trial Generalizability Assessment in the Big Data Era: A Review. *Clin Transl Sci*, *13*(4), 675-684. <u>https://doi.org/10.1111/cts.12764</u>

Heim, E., Harper Shehadeh, M., Van't Hof, E., Carswell, K., Maercker, A., & Kirmayer, L. (2019). Cultural adaptation of scalable interventions. *Cultural clinical psychology and PTSD*, 201-218.

Heuzenroeder, L., Donnelly, M., Haby, M. M., Mihalopoulos, C., Rossell, R., Carter, R., Andrews, G., & Vos, T. (2004). Cost-Effectiveness of Psychological and Pharmacological Interventions for Generalized Anxiety Disorder and Panic Disorder. *Australian & New Zealand Journal of Psychiatry*, *38*(8), 602-612. <u>https://doi.org/10.1080/j.1440-</u>

<u>1614.2004.01423.x</u>

Ho, H., Jose, I., Cheesman, M., Garrison, C., Bishop, K., Taber, S., Witt, J., & Sin, M. K. (2021). Depression and Anxiety Management in Parkinson Disease. *J Neurosci Nurs*, *53*(4), 170-176. <u>https://doi.org/10.1097/jnn.00000000000596</u>

Hoffmann, T. C., Glasziou, P. P., Boutron, I., Milne, R., Perera, R., Moher, D., Altman, D. G., Barbour, V., Macdonald, H., Johnston, M., Lamb, S. E., Dixon-Woods, M., McCulloch, P., Wyatt, J. C., Chan, A. W., & Michie, S. (2014). Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. *BMJ*, *348*, g1687. <u>https://doi.org/10.1136/bmj.g1687</u>

Hofmann, S., Sawyer, A., Witt, A., & Oh, D. (2010). The effect of mindfulness- based therapy on anxiety and depression: A meta-analytic review. *Journal of Consulting and Clinical Psychology*, *78*(2), 169-183. <u>https://doi.org/10.1037/a0018555</u>

Hofmann, S. G., Ellard, K. K., & Siegle, G. J. (2012). Neurobiological correlates of cognitions in fear and anxiety: a cognitive-neurobiological information-processing model. *Cognition & Emotion*, *26*(2), 282-299. <u>https://doi.org/10.1080/02699931.2011.579414</u>

Hogan, L.-M., Björklund Carlstedt, A., & Wagman, P. (2023). Occupational therapy and stress-related exhaustion – a scoping review. *Scandinavian Journal of Occupational Therapy*, *30*(7), 1047-1063. <u>https://doi.org/10.1080/11038128.2023.2207802</u>

Horkoff, M. J. (2012). Adolescents, anxiety, and occupational performance. Psychology.

Horton, A., Hebson, G., & Holman, D. (2021). A longitudinal study of the turning points and trajectories of therapeutic relationship development in occupational and physical therapy. BMC Health Serv Res, 21(1), 97. <u>https://doi.org/10.1186/s12913-021-06095-y</u>

Horváth, K., Aschermann, Z., Kovács, M., Makkos, A., Harmat, M., Janszky, J., Komoly, S., Karádi, K., & Kovács, N. (2017). Changes in Quality of Life in Parkinson's Disease: How Large Must They Be to Be Relevant? *Neuroepidemiology*, *48*(1-2), 1-8.

https://doi.org/10.1159/000455863

HSJ Parkinson's UK Excellence Network. (no date). *Spend on hospital admissions for people with Parkinson's*. Retrieved 06 December from <u>https://parkinsonsuk.hsj.co.uk/spend-hospital-admissions-people-parkinsons</u>

Hsu, C.-C., & Sandford, B. (2007). The Delphi Technique: Making Sense Of Consensus. *Practical Assessment, Research and Evaluation*, *12*.

Hughes, S., & Burch, S. (2020). 'I'm not just a number on a sheet, I'ma person': Domiciliary care, self and getting older. *Health & Social Care in the Community*, *28*(3), 903-912.

Hultqvist, J., Sahlström, T., Timpka, J., Henriksen, T., Nyholm, D., Odin, P., & Eklund, M. (2020). Everyday Occupations and Other Factors in Relation to Mental Well-Being among Persons with Advanced Parkinson's Disease. *Occupational Therapy in Health Care*, *34*(1), 1-18. <u>https://doi.org/10.1080/07380577.2019.1692269</u>

Hunter, H., Lovegrove, C., Haas, B., Freeman, J., & Gunn, H. (2019). Experiences of people with Parkinson's disease and their views on physical activity interventions: a qualitative systematic review. *JBI Database System Rev Implement Rep*, *17*(4), 548-613. https://doi.org/10.11124/jbisrir-2017-003901

Hurley, D. A., Keogh, A., Mc Ardle, D., Hall, A. M., Richmond, H., Guerin, S., Magdalinski, T.,
& Matthews, J. (2019). Evaluation of an E-Learning Training Program to Support
Implementation of a Group-Based, Theory-Driven, Self-Management Intervention For
Osteoarthritis and Low-Back Pain: Pre-Post Study [Original Paper]. J Med Internet Res, 21(3),
e11123. https://doi.org/10.2196/11123

Hurt, C. S., Landau, S., Burn, D. J., Hindle, J. V., Samuel, M., Wilson, K., & Brown, R. G. (2012).
Cognition, coping, and outcome in Parkinson's disease. *International Psychogeriatrics*, 24(10), 1656-1663. <u>https://doi.org/10.1017/S1041610212000749</u>

Hwang, H. C., & Matsumoto, D. (2020). Cross-Cultural Emotional Expression. *The Wiley Encyclopedia of Personality and Individual Differences: Clinical, Applied, and Cross-Cultural Research*, 257-261.

Iliffe, S., Bourne, R., Patterson, L., & Manthorpe, J. (2021). Resource allocation in the NHS: Shifting the balance towards the community. *Renewal: a Journal of Social Democracy*, *29*(2), 52-62.

International handbook of occupational therapy interventions. (2015). (2nd ed. ed.). Springer International Publishing, Cham. <u>https://doi.org/http://dx.doi.org/10.1007/978-3-319-08141-0</u>

Iqbal, S., & Pipon-Young, L. (2009). The Delphi method. *Psychologist*, 22, 598-601.

Islam, M. S., Azim, F., Saju, H., Zargaran, A., Shirzad, M., Kamal, M., Fatema, K., Rehman, S., Azad, M. A. M., & Ebrahimi-Barough, S. (2021). Pesticides and Parkinson's disease: Current and future perspective. *Journal of Chemical Neuroanatomy*, *115*, 101966. <u>https://doi.org/https://doi.org/10.1016/j.jchemneu.2021.101966</u>

Ismail, G., & Taliep, N. (2023). The Delphi Method. In P. Liamputtong (Ed.), *Handbook of Social Sciences and Global Public Health* (pp. 1-19). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-96778-9_66-1</u>

Jackson, J. C., Watts, J., Henry, T. R., List, J.-M., Forkel, R., Mucha, P. J., Greenhill, S. J., Gray, R. D., & Lindquist, K. A. (2019). Emotion semantics show both cultural variation and universal structure. *Science*, *366*(6472), 1517-1522.

Jacob, J., Stankovic, M., Spuerck, I., & Shokraneh, F. (2022). Goal setting with young people for anxiety and depression: What works for whom in therapeutic relationships? A literature review and insight analysis. *BMC psychology*, *10*(1), 171. <u>https://doi.org/10.1186/s40359-022-00879-5</u>

James, T., Mukadam, N., Sommerlad, A., Guerra Ceballos, S., & Livingston, G. (2021). Culturally tailored therapeutic interventions for people affected by dementia: a systematic review and new conceptual model. *Lancet Healthy Longev*, *2*(3), e171-e179. https://doi.org/10.1016/s2666-7568(21)00001-5

Janak, P. H., & Tye, K. M. (2015). From circuits to behaviour in the amygdala. *Nature*, *517*(7534), 284-292. <u>https://doi.org/10.1038/nature14188</u>

Jankovic, J., & Tan, E. K. (2020). Parkinson's disease: etiopathogenesis and treatment. Journal of Neurology, Neurosurgery & Psychiatry, 91(8), 795-808. <u>https://doi.org/10.1136/jnnp-2019-322338</u>

Jankowska, M. (2019). Cultural modifications of cognitive behavioural treatment of social anxiety among culturally diverse clients: a systematic literature review. *The Cognitive Behaviour Therapist*, *12*, e7.

Johansson, A., & Björklund, A. (2016). The impact of occupational therapy and lifestyle interventions on older persons' health, well-being, and occupational adaptation. *Scandinavian Journal of Occupational Therapy*, *23*(3), 207-219. https://doi.org/10.3109/11038128.2015.1093544

Johansson, G., Eklund, M., & Erlandsson, L. K. (2012). Everyday hassles and uplifts among women on long-term sick-leave due to stress-related disorders. *Scand J Occup Ther*, *19*(3), 239-248. <u>https://doi.org/10.3109/11038128.2011.569942</u>

Johnsen, T. J., & Friborg, O. (2015). The effects of cognitive behavioral therapy as an antidepressive treatment is falling: A meta-analysis. *Psychological Bulletin*, *141*(4), 747-768. <u>https://doi.org/10.1037/bul0000015</u>

Johnson, R. B., Onwuegbuzie, A. J., & Turner, L. A. (2007). Toward a Definition of Mixed Methods Research. *Journal of Mixed Methods Research*, 1(2), 112-133. https://doi.org/10.1177/1558689806298224 Johnson, S. U., Ulvenes, P. G., Øktedalen, T., & Hoffart, A. (2019). Psychometric Properties of the General Anxiety Disorder 7-Item (GAD-7) Scale in a Heterogeneous Psychiatric Sample. *Front Psychol*, *10*, 1713. <u>https://doi.org/10.3389/fpsyg.2019.01713</u>

Johnston, M., Carey, R. N., Connell Bohlen, L. E., Johnston, D. W., Rothman, A. J., de Bruin, M., Kelly, M. P., Groarke, H., & Michie, S. (2020). Development of an online tool for linking behavior change techniques and mechanisms of action based on triangulation of findings from literature synthesis and expert consensus. *Translational Behavioral Medicine*, *11*(5), 1049-1065. <u>https://doi.org/10.1093/tbm/ibaa050</u>

Joling, M., van den Heuvel, O. A., Berendse, H. W., Booij, J., & Vriend, C. (2018). Serotonin transporter binding and anxiety symptoms in Parkinson's disease. *J Neurol Neurosurg Psychiatry*, *89*(1), 89-94. <u>https://doi.org/10.1136/jnnp-2017-316193</u>

Jones, M. D., Casson, S. M., Barry, B. K., Li, S. H., Valenzuela, T., Cassar, J., Lamanna, C., Lloyd, A. R., & Sandler, C. X. (2023). eLearning improves allied health professionals' knowledge and confidence to manage medically unexplained chronic fatigue states: A randomized controlled trial. *Journal of Psychosomatic Research*, *173*, 111462. https://doi.org/https://doi.org/10.1016/j.jpsychores.2023.111462

Jonsson, H. (2008). A new direction in the conceptualization and categorization of occupation. *Journal of Occupational Science*, *15*(1), 3-8. <u>https://doi.org/10.1080/14427591.2008.9686601</u>

Kadam, U., Croft, P., McLeod, J., & Hutchinson, M. (2001). A qualitative study of patients' views on anxiety and depression. *British Journal of General Practice*, *51*(466), 375-380.

Kalia, L. V., & Lang, A. E. (2015). Parkinson's disease. *The Lancet, 386*(9996), 896-912. https://doi.org/10.1016/S0140-6736(14)61393-3

Kandola, A., & Stubbs, B. (2020). Exercise and Anxiety. In J. Xiao (Ed.), *Physical Exercise for Human Health* (pp. 345-352). Springer Singapore. <u>https://doi.org/10.1007/978-981-15-1792-1_23</u>

Kane, M., & Trochim, W. (2007). *Concept Mapping for Planning and Evaluation*. https://doi.org/10.4135/9781412983730

Kang, E., Friz, D., Lipsey, K., & Foster, E. R. (2022). Empowerment in people with Parkinson's disease: A scoping review and qualitative interview study. *Patient Education and Counseling*, *105*(10), 3123-3133. <u>https://doi.org/https://doi.org/10.1016/j.pec.2022.06.003</u>

Kano, O., Ikeda, K., Cridebring, D., Takazawa, T., Yoshii, Y., & Iwasaki, Y. (2011). Neurobiology of depression and anxiety in Parkinson's disease. *Parkinson's Disease, 2011*, 143547-143547. <u>https://doi.org/10.4061/2011/143547</u>

Karl, J. A., Johnson, F. N., Bucci, L., & Fischer, R. (2022). In search of mindfulness: a review and reconsideration of cultural dynamics from a cognitive perspective. *Journal of the Royal Society of New Zealand*, *52*(2), 168-191.

Karran, E. L., Cashin, A. G., Barker, T., Boyd, M. A., Chiarotto, A., Dewidar, O., Mohabir, V., Petkovic, J., Sharma, S., Tejani, S., Tugwell, P., & Moseley, G. L. (2023). Using PROGRESS-plus to identify current approaches to the collection and reporting of equity-relevant data: a scoping review. *Journal of Clinical Epidemiology*, *163*, 70-78.

https://doi.org/10.1016/j.jclinepi.2023.09.017

Katz, N., Karpin, H., Lak, A., Furman, T., & Hartman-Maeir, A. (2003). Participation in occupational performance: Reliability and validity of the Activity Card Sort. *OTJR: Occupation, Participation and Health, 23*(1), 10-17.

Kessler, R. C., Angermeyer, M., Anthony, J. C., de Graaf, R., Demyttenaere, K., Gasquet, I., de Girolamo, G., Gluzman, S., Gureje, O., Haro, J. M., Kawakami, N., Karam, A., Levinson, D., Medina Mora, M. E., Oakley Browne, M. A., Posada-Villa, J., Stein, D. J., Adley Tsang, C. H., Aguilar-Gaxiola, S.,...Üstün, T. B. (2007). Lifetime prevalence and age-of-onset distributions of mental disorders in the World Health Organization's World Mental Health Survey Initiative. *World Psychiatry*, *6*(3), 9. Keyworth, C., Epton, T., Goldthorpe, J., Calam, R., & Armitage, C. J. (2020). Perceptions of receiving behaviour change interventions from GPs during routine consultations: a qualitative study. *PLoS ONE*, *15*(5), e0233399.

Khodyakov, D., Grant, S., Kroger, J., & Bauman, M. (2023). *RAND Methodological Guidance for Conducting and Critically Appraising Delphi Panels*. RAND Corporation. <u>https://doi.org/10.7249/TLA3082-1</u>

Khodyakov, D., Grant, S., Kroger, J., Gadwah-Meaden, C., Motala, A., & Larkin, J. (2023). Disciplinary trends in the use of the Delphi method: A bibliometric analysis. *PLoS ONE*, *18*(8), e0289009. <u>https://doi.org/10.1371/journal.pone.0289009</u>

Kielhofner, G. (2007). Respecting both the "occupation" and the "therapy" in our field. *Am J Occup Ther*, *61*(4), 479-482. <u>https://doi.org/10.5014/ajot.61.4.479</u>

Kienast, T., Hariri, A. R., Schlagenhauf, F., Wrase, J., Sterzer, P., Buchholz, H. G., Smolka, M. N., Gründer, G., Cumming, P., Kumakura, Y., Bartenstein, P., Dolan, R. J., & Heinz, A. (2008).
Dopamine in amygdala gates limbic processing of aversive stimuli in humans. *Nature neuroscience*, *11*(12), 1381-1382. <u>https://doi.org/10.1038/nn.2222</u>

Kim, E. J., Horovitz, O., Pellman, B. A., Tan, L. M., Li, Q., Richter-Levin, G., & Kim, J. J. (2013). Dorsal periaqueductal gray-amygdala pathway conveys both innate and learned fear responses in rats. *Proc Natl Acad Sci U S A*, *110*(36), 14795-14800. <u>https://doi.org/10.1073/pnas.1310845110</u>

Kirmayer, L. J. (2015). Mindfulness in cultural context. In (Vol. 52, pp. 447-469): Sage Publications Sage UK: London, England.

Kirsh, B., Martin, L., Hultqvist, J., & Eklund, M. (2019). Occupational Therapy Interventions in Mental Health: A Literature Review in Search of Evidence. *Occupational Therapy in Mental Health*, *35*(2), 109-156. <u>https://doi.org/10.1080/0164212X.2019.1588832</u>
Kishita, N., & Laidlaw, K. (2017). Cognitive behaviour therapy for generalized anxiety disorder: Is CBT equally efficacious in adults of working age and older adults? *Clin Psychol Rev*, *52*, 124-136. <u>https://doi.org/10.1016/j.cpr.2017.01.003</u>

Kitchiner, N. J., Edwards, D., Wood, S., Sainsbury, S., Hewin, P., Burnard, P., & Bisson, J. I. (2009). A randomized controlled trial comparing an adult education class using cognitive behavioural therapy ("stress control"), anxiety management group treatment and a waiting list for anxiety disorders [Article]. *Journal of Mental Health*, *18*(4), 307-315. https://doi.org/10.1080/09638230802052153

Kitwood, T. (2011). Dementia reconsidered: The person comes first. *Adult lives: A life course perspective, 89*.

Klein, J. G., Brown, G. T., & Lysyk, M. (2000). Replication Research: A Purposeful Occupation Worth Repeating. *Canadian Journal of Occupational Therapy*, *67*(3), 155-161. <u>https://doi.org/10.1177/000841740006700310</u>

Kobayashi, E., Himuro, N., Mitani, Y., Tsunashima, T., Nomura, K., & Chiba, S. (2023). Feasibility and informativeness of the Canadian occupational performance measure for identifying priorities in patients with Parkinson's disease. *Physiotherapy theory and practice*, *39*(3), 607-614.

Koerts, J., König, M., Tucha, L., & Tucha, O. (2016). Working capacity of patients with Parkinson's disease – A systematic review. *Parkinsonism & Related Disorders*, *27*, 9-24. <u>https://doi.org/https://doi.org/10.1016/j.parkreldis.2016.03.017</u>

Kopchia, K. L., Altman, H. J., & Commissaris, R. L. (1992). Effects of lesions of the central nucleus of the amygdala on anxiety-like behaviors in the rat. *Pharmacol Biochem Behav*, *43*(2), 453-461. <u>https://doi.org/10.1016/0091-3057(92)90176-g</u>

Kumaresan, M., & Khan, S. (2021). Spectrum of non-motor symptoms in Parkinson's disease. *Cureus*, *13*(2).

Kwok, J. Y. Y., Kwan, J. C. Y., Auyeung, M., Mok, V. C. T., Lau, C. K. Y., Choi, K. C., & Chan, H. Y. L. (2019). Effects of Mindfulness Yoga vs Stretching and Resistance Training Exercises on Anxiety and Depression for People With Parkinson Disease. *JAMA Neurology*, *76*(7), 755. https://doi.org/http://dx.doi.org/10.1001/jamaneurol.2019.0534

Laar, A., Silva de Lima, A. L., Maas, B. R., Bloem, B. R., & de Vries, N. M. (2023). Successful implementation of technology in the management of Parkinson's disease: Barriers and facilitators. *Clinical Parkinsonism & Related Disorders*, *8*, 100188. <u>https://doi.org/https://doi.org/10.1016/j.prdoa.2023.100188</u>

LaBar, K. S., Gatenby, J. C., Gore, J. C., LeDoux, J. E., & Phelps, E. A. (1998). Human amygdala activation during conditioned fear acquisition and extinction: a mixed-trial fMRI study. *Neuron*, *20*(5), 937-945. <u>https://doi.org/10.1016/s0896-6273(00)80475-4</u>

Lambert, R., Caan, W., & McVicar, A. (2008). Influences of lifestyle and general practice (GP) care on the symptom profile of people with panic disorder. *Journal of Public Mental Health*, *7*(2), 18-24. <u>https://doi.org/10.1108/17465729200800011</u>

Lambert, R. A., Harvey, I., & Poland, F. (2007). A pragmatic, unblinded randomised controlled trial comparing an occupational therapy-led lifestyle approach and routine GP care for panic disorder treatment in primary care [Comparative Study Randomized Controlled Trial Research Support, Non-U.S. Gov't]. *Journal of Affective Disorders, 99*(1-3), 63-71.

https://ovidsp.ovid.com/ovidweb.cgi?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med6&AN= 17014912

Lambert, R. A., Lorgelly, P., Harvey, I., & Poland, F. (2010). Cost-effectiveness analysis of an occupational therapy-led lifestyle approach and routine general practitioner's care for panic disorder [Comparative Study Multicenter Study Randomized Controlled Trial Research Support, Non-U.S. Gov't]. *Social Psychiatry & Psychiatric Epidemiology*, *45*(7), 741-750. https://doi.org/https://dx.doi.org/10.1007/s00127-009-0114-5 Lang, P. J., Davis, M., & Öhman, A. (2000). Fear and anxiety: animal models and human cognitive psychophysiology. *Journal of Affective Disorders*, *61*(3), 137-159. <u>http://ac.els-cdn.com/S0165032700003438/1-s2.0-S0165032700003438-main.pdf?_tid=d3ed7ebc-118e-11e4-bfa7-00000aacb35d&acdnat=1406026766_b0814d7fd1e10b714e64ece74da3d4ba</u>

Larsson-Lund, M., & Nyman, A. (2017). Participation and occupation in occupational therapy models of practice: A discussion of possibilities and challenges. *Scand J Occup Ther*, *24*(6), 393-397. <u>https://doi.org/10.1080/11038128.2016.1267257</u>

Lather, P. (2006). Paradigm proliferation as a good thing to think with: teaching research in education as a wild profusion. *International Journal of Qualitative Studies in Education*, *19*(1), 35-57. <u>https://doi.org/10.1080/09518390500450144</u>

Latter, S., Clark, J. M., Geddes, C., & Kitsell, F. (2009). Implementing a clinical academic career pathway in nursing; criteria for success and challenges ahead. *Journal of Research in Nursing*, *14*(2), 137-148.

Laver Fawcett, A. (2014). Routine standardised outcome measurement to evaluate the effectiveness of occupational therapy interventions: essential or optional? *Ergoterapeuten*, *4*, 28-37.

Law, M., Polatajko, H., Pollock, N., Mccoll, M. A., Carswell, A., & Baptiste, S. (1994). Pilot Testing of the Canadian Occupational Performance Measure: Clinical and Measurement Issues. *Canadian Journal of Occupational Therapy*, *61*(4).

Lawson, A., & Beckett, A. E. (2021). The social and human rights models of disability: towards a complementarity thesis. *The International Journal of Human Rights*, *25*(2), 348-379. <u>https://doi.org/10.1080/13642987.2020.1783533</u>

Laxminarayan, R., Mills, A. J., Breman, J. G., Measham, A. R., Alleyne, G., Claeson, M., Jha, P., Musgrove, P., Chow, J., Shahid-Salles, S., & Jamison, D. T. (2006). Advancement of global health: key messages from the Disease Control Priorities Project. *Lancet*, *367*(9517), 1193-1208. <u>https://doi.org/10.1016/s0140-6736(06)68440-7</u> Lebuda, I., & Benedek, M. (2023). A systematic framework of creative metacognition. *Physics of Life Reviews*, *46*, 161-181. https://doi.org/https://doi.org/10.1016/j.plrev.2023.07.002

Ledgerd, R., & World Federation of Occupational Therapists, n. (2020). WFOT report: WFOT human resources project 2018 and 2020. *World Federation of Occupational Therapists Bulletin*, *76*(2), 69-74. <u>https://doi.org/10.1080/14473828.2020.1821475</u>

LeDoux, J. (1998). Fear and the brain: where have we been, and where are we going? *Biol Psychiatry*, 44(12), 1229-1238. <u>https://doi.org/10.1016/s0006-3223(98)00282-0</u>

LeDoux, J., & Pine, D. (2016). Using Neuroscience to Help Understand Fear and Anxiety: A Two-System Framework. *American Journal of Psychiatry*, *173*(11), 1083-1093. <u>https://doi.org/10.1176/appi.ajp.2016.16030353</u>

Lee, A., & Gilbert, R. M. (2016). Epidemiology of Parkinson Disease. *Neurologic Clinics*, 34(4), 955-965. <u>https://doi.org/10.1016/j.ncl.2016.06.012</u>

Leland, N. E., Crum, K., Phipps, S., Roberts, P., & Gage, B. (2015). Advancing the value and quality of occupational therapy in health service delivery. *Am J Occup Ther*, *69*(1), 6901090010p6901090011-6901090017. <u>https://doi.org/10.5014/ajot.2015.691001</u>

Lemire, S., Kwako, A., Nielsen, S. B., Christie, C. A., Donaldson, S. I., & Leeuw, F. L. (2020). What Is This Thing Called a Mechanism? Findings From a Review of Realist Evaluations. *New Directions for Evaluation*, *2020*(167), 73-86.

https://doi.org/https://doi.org/10.1002/ev.20428

Levati, S., Campbell, P., Frost, R., Dougall, N., Wells, M., Donaldson, C., & Hagen, S. (2016). Optimisation of complex health interventions prior to a randomised controlled trial: a scoping review of strategies used. *Pilot and Feasibility Studies*, *2*(1), 17. <u>https://doi.org/10.1186/s40814-016-0058-y</u> LeWitt, P. A., & Chaudhuri, K. R. (2020). Unmet needs in Parkinson disease: Motor and nonmotor [Research Support, Non-U.S. Gov't]. *Parkinsonism & Related Disorders, 80 Suppl 1,* S7-S12.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=prem&AN=3334 9582

Lim, S.-Y., Fox, S. H., & Lang, A. E. (2009). Overview of the extranigral aspects of Parkinson disease. *Archives of Neurology*, *66*(2), 167-172.

Lin, Y.-P., Lin, I.-I., Chiou, W.-D., Chang, H.-C., Chen, R.-S., Lu, C.-S., & Chang, Y.-J. (2023). The Executive-Function-Related Cognitive–Motor Dual Task Walking Performance and Task Prioritizing Effect on People with Parkinson's Disease. *Healthcare*, *11*(4), 567. https://www.mdpi.com/2227-9032/11/4/567

Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. Sage.

Lindquist, K. A., MacCormack, J. K., & Shablack, H. (2015). The role of language in emotion: predictions from psychological constructionism. *Front Psychol*, *6*, 444. <u>https://doi.org/10.3389/fpsyg.2015.00444</u>

Lintel, H., Corpuz, T., Paracha, S.-u.-R., & Grossberg, G. T. (2021). Mood Disorders and Anxiety in Parkinson's Disease: Current Concepts. *Journal of Geriatric Psychiatry and Neurology*, *34*(4), 280-288. <u>https://doi.org/10.1177/08919887211018267</u>

Lister, T. (2020). Nutrition and Lifestyle Interventions for Managing Parkinson's Disease: A Narrative Review. *J Mov Disord*, *13*(2), 97-104. <u>https://doi.org/10.14802/jmd.20006</u>

Liu, C.-j., Chang, W.-P., & Chang, M. C. (2018). Occupational Therapy Interventions to Improve Activities of Daily Living for Community-Dwelling Older Adults: A Systematic Review. *The American Journal of Occupational Therapy*, *72*(4), 7204190060p7204190061-7204190060p7204190011. <u>https://doi.org/10.5014/ajot.2018.031252</u> Locke, A., Kirst, N., & Shultz, C. (2015). Diagnosis and Management of Generalized Anxiety Disorder and Panic Disorder in Adults. *American Family Physician*, *91*(9), 617-624.

Logsdon, R. G., McCurry, S. M., & Teri, L. (2007). Evidence-Based Interventions to Improve Quality of Life for Individuals with Dementia. *Alzheimers care today*, *8*(4), 309-318.

Lopez, A. (2011). Posttraumatic stress disorder and occupational performance: building resilience and fostering occupational adaptation. *Work*, *38*(1), 33-38. https://doi.org/10.3233/WOR-2011-1102

Lork, K., Holmgren, K., & Danielsson, L. (2021). A short work-directed rehabilitation to promote work capacity while depressed and anxious: a qualitative study of workers' experiences. *Disabil Rehabil*, *43*(17), 2487-2496.

https://doi.org/10.1080/09638288.2019.1704893

Lovegrove, C. (2021). Characteristics of Occupational Therapy Interventions for Community-Dwelling Adults with Anxiety: A Scoping Review protocol. In (October 22nd 2021 ed.). OSF.

Lovegrove, C., & Bannigan, K. (2021). What is the lived experience of anxiety for people with Parkinson's? A phenomenological study. *PLoS ONE*, *16*(4), e0249390. <u>https://doi.org/10.1371/journal.pone.0249390</u>

Lovegrove, C., Bannigan, K., Cheeseman, D., & Latour, J. (2017). The involvement of people with Parkinson's in designing a study of the lived experience of anxiety. *British Journal of Occupational Therapy*, 1-8. <u>https://doi.org/10.1177/0308022617700654</u>

Lovegrove, C., Bannigan, K., Hayward, C., Ingram, W., Bailey, M., Aspinall, P., Hosking, J., Sturkenboom, I., & Marsden, J. (in press). Evaluating the occupation-based complex intervention for living well with anxiety and Parkinson's disease (OBtAIN-PD) in community rehabilitation teams in the UK: A feasibility cluster randomised controlled trial protocol. *BMJ Open*. Lovegrove, C., Bannigan, K., Marsden, J., & Sturkenboom, I. (in press). Developing an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD) using logic modelling. *BMJ Open*.

Lovegrove, C., & Marsden, J. (2020). Service evaluation of an acute Parkinson's therapy pathway between hospital and home. *International Journal of Therapy and Rehabilitation*, *27*(7), 1-6. <u>https://doi.org/10.12968/ijtr.2019.0047</u>

Lovegrove, C., Marsden, J., Smith, M., Sturkenboom, I., & Bannigan, K. (2023). Characteristics of Occupational Therapy Interventions for Community-Dwelling Adults With Anxiety: Protocol for a Scoping Review [Protocol]. *JMIR Res Protoc, 12*, e41230. <u>https://doi.org/10.2196/41230</u>

Lovegrove, C., Marsden, J., Smith, M., Sturkenboom, I., & Bannigan, K. (in press). Characteristics of occupational therapy interventions for community-dwelling adults with anxiety: a scoping review. *JMIR Aging*.

Lovegrove, C., Sturkenboom, I., Marsden, J., & Bannigan, K. (2024). Concept Mapping to Define Components for an Occupation-Based Intervention for Parkinson's Disease and Anxiety. *J Parkinsons Dis*, *14*(1), 181-195. <u>https://doi.org/10.3233/jpd-230150</u>

Lovegrove, C., Sturkenboom, I., Marsden, J., & Bannigan, K. (in press). Concept mapping to define components for an occupation-based intervention for Parkinson's & anxiety. *Journal of Parkinson's Disease*.

Löwe, B., Decker, O., Müller, S., Brähler, E., Schellberg, D., Herzog, W., & Herzberg, P. Y. (2008). Validation and Standardization of the Generalized Anxiety Disorder Screener (GAD-7) in the General Population. *Medical Care*, *46*(3), 266-274. <u>https://doi.org/10.1097/MLR.0b013e318160d093</u>

Mackenzie, L., Coppola, S., Alvarez, L., Cibule, L., Maltsev, S., Loh, S. Y., Mlambo, T., Ikiugu, M. N., Pihlar, Z., Sriphetcharawut, S., Baptiste, S., & Ledgerd, R. (2017). International

Occupational Therapy Research Priorities. *OTJR (Thorofare N J), 37*(2), 72-81. https://doi.org/10.1177/1539449216687528

MacMahon, D. G., & Thomas, S. (1998). Practical approach to quality of life in Parkinson's disease: the nurse's role. *Journal of Neurology*, *245*(1), 19-22. http://www.ncbi.nlm.nih.gov/pubmed/9617718

Maguire, T., Garvey, L., Ryan, J., Olasoji, M., & Willets, G. (2022). Using the Nominal Group Technique to determine a nursing framework for a forensic mental health service: A discussion paper. *International Journal of Mental Health Nursing*, *31*(4), 1030-1038. <u>https://doi.org/10.1111/inm.13023</u>

Mahon, D. (2023). Transcultural Considerations. In *Alcohol Use: Assessment, Withdrawal Management, Treatment and Therapy: Ethical Practice* (pp. 33-55). Springer.

Maillet, A., Krack, P., Lhommee, E., Metereau, E., Klinger, H., Favre, E., Le Bars, D., Schmitt, E., Bichon, A., Pelissier, P., Fraix, V., Castrioto, A., Sgambato-Faure, V., Broussolle, E., Tremblay, L., & Thobois, S. (2016). The prominent role of serotonergic degeneration in apathy , anxiety and depression in de novo Parkinson ' s disease. *Brain*, *139*, 2486-2502. <u>https://doi.org/10.1093/aww190</u>

Malhotra, A., & Baker, J. (2023). Group Therapy. In *StatPearls*. StatPearls Publishing Copyright © 2023, StatPearls Publishing LLC.

Malm, C., Andersson, S., Kylén, M., Iwarsson, S., Hanson, E., & Schmidt, S. M. (2021). What motivates informal carers to be actively involved in research, and what obstacles to involvement do they perceive? *Research Involvement and Engagement*, *7*(1), 80. <u>https://doi.org/10.1186/s40900-021-00321-x</u>

Manera, K., Hanson, C. S., Gutman, T., & Tong, A. (2019). Consensus Methods: Nominal Group Technique. In P. Liamputtong (Ed.), *Handbook of Research Methods in Health Social Sciences* (pp. 737-750). Springer Singapore. <u>https://doi.org/10.1007/978-981-10-5251-</u>

<u>4 100</u>

Manger, S. (2019). Lifestyle interventions for mental health. *Aust J Gen Pract, 48*(10), 670-673. <u>https://doi.org/10.31128/ajgp-06-19-4964</u>

Manzoni, G. M., Pagnini, F., Castelnuovo, G., & Molinari, E. (2008). Relaxation training for anxiety: A ten-years systematic review with meta-analysis. *BMC Psychiatry*, *8*. <u>https://doi.org/10.1186/1471-244X-8-41</u>

Mathieson, A., Grande, G., & Luker, K. (2019). Strategies, facilitators and barriers to implementation of evidence-based practice in community nursing: a systematic mixed-studies review and qualitative synthesis. *Prim Health Care Res Dev, 20*, e6. <u>https://doi.org/10.1017/s1463423618000488</u>

Matuska, K. M., & Christiansen, C. H. (2008). A proposed model of lifestyle balance. *Journal of Occupational Science*, 15(1), 9-19. <u>https://doi.org/10.1080/14427591.2008.9686602</u>

Maydeu-Olivares, A., & Garcia-Forero, C. (2010). Goodness-of-Fit Testing. In *International Encyclopedia of Education*.

McAllister, S., Simpson, A., Tsianakas, V., Canham, N., De Meo, V., Stone, C., & Robert, G. (2021). Developing a theory-informed complex intervention to improve nurse–patient therapeutic engagement employing Experience-based Co-design and the Behaviour Change Wheel: an acute mental health ward case study. *BMJ Open*, *11*(5), e047114.

McLaughlin, D., Hasson, F., George Kernohan, W., Waldron, M., McLaughlin, M., Cochrane, B., & Chambers, H. (2011). Living and coping with Parkinson's disease: Perceptions of informal carers. *Palliative medicine*, *25*(2), 177-182.

https://doi.org/http://dx.doi.org/10.1177/0269216310385604

McLean, G., Lawrence, M., Simpson, R., & Mercer, S. W. (2017). Mindfulness-based stress reduction in Parkinson's disease: a systematic review. *BMC Neurology*, *17*(1), 92-92. <u>https://doi.org/10.1186/s12883-017-0876-4</u> McMillan, S. S., Kelly, F., Sav, A., Kendall, E., King, M. A., Whitty, J. A., & Wheeler, A. J. (2014). Using the Nominal Group Technique: how to analyse across multiple groups. *Health Services and Outcomes Research Methodology*, *14*(3), 92-108. https://doi.org/10.1007/s10742-014-0121-1

McMillan, S. S., King, M., & Tully, M. P. (2016). How to use the nominal group and Delphi techniques. *International Journal of Clinical Pharmacy*, *38*, 7.

Medical Research Council. (2008). Developing and evaluating complex interventions: new guidance. *Medical Research Council*.

Mendorf, S., Teschner, U., Lehmann, T., Prell, T., & Mühlhammer, H. M. (2023). Tailored interventions to improve adherence to medication in elderly patients with Parkinson's disease: a study protocol for a randomized controlled trial (AdhCare). *Trials*, *24*(1), 668. <u>https://doi.org/10.1186/s13063-023-07663-9</u>

Mental Health Foundation. (2014). *Living with Anxiety: Understanding the role and impact of anxiety in our lives*. M. H. Foundation.

Menza, M. A., Palermo, B., DiPaola, R., Sage, J. I., & Ricketts, M. H. (1999). Depression and anxiety in Parkinson's disease: possible effect of genetic variation in the serotonin transporter. *J Geriatr Psychiatry Neurol*, *12*(2), 49-52. <u>https://doi.org/10.1177/089198879901200202</u>

Mercieca-Bebber, R., King, M. T., Calvert, M. J., Stockler, M. R., & Friedlander, M. (2018). The importance of patient-reported outcomes in clinical trials and strategies for future optimization. *Patient Relat Outcome Meas*, *9*, 353-367. <u>https://doi.org/10.2147/prom.S156279</u>

Meyer, A. (1922). The Philosophy of Occupational Therapy. *Arch Occupat Ther*, *68*(283), 421-423. <u>https://doi.org/10.1192/bjp.68.283.421</u>

Milham, M. P., Nugent, A. C., Drevets, W. C., Dickstein, D. P., Leibenluft, E., Ernst, M., Charney, D., & Pine, D. S. (2005). Selective reduction in amygdala volume in pediatric anxiety disorders: a voxel-based morphometry investigation. *Biol Psychiatry*, *57*(9), 961-966. <u>https://doi.org/10.1016/j.biopsych.2005.01.038</u>

Miller, W. R., & Rollnick, S. (2012). *Motivational interviewing: Helping people change*. Guilford press.

Mills, T., Lawton, R., & Sheard, L. (2019). Advancing complexity science in healthcare research: the logic of logic models. *BMC Med Res Methodol*, *19*(1), 55. <u>https://doi.org/10.1186/s12874-019-0701-4</u>

Milne-Ives, M., Carroll, C., & Meinert, E. (2022). Self-management Interventions for People With Parkinson Disease: Scoping Review. *J Med Internet Res*, *24*(8), e40181. <u>https://doi.org/10.2196/40181</u>

Moberg, J., & Kramer, M. (2015). A brief history of the cluster randomised trial design. *J R Soc Med*, *108*(5), 192-198. <u>https://doi.org/10.1177/0141076815582303</u>

Mohammadi, A., Abasi, I., Soleimani, M., Moradian, S. T., Yahyavi, T., & Zarean, M. (2019). Cultural Aspects of Social Anxiety Disorder: A Qualitative Analysis of Anxiety Experiences and Interpretation. *Iranian Journal of Psychiatry*, *14*(1), 33-29.

Moher, D., Shamseer, L., Clarke, M., Ghersi, D., Liberati, A., Petticrew, M., Shekelle, P., Stewart, L. A., & Group, P.-p. (2015). Preferred Reporting Items for Systematic Review and Meta-Analysis Protocols (PRISMA-P) 2015 statement. *Systematic Reviews*, *4*(1), 1-1. <u>https://doi.org/10.1186/2046-4053-4-1</u>

Möhler, R., Bartoszek, G., Köpke, S., & Meyer, G. (2012). Proposed criteria for reporting the development and evaluation of complex interventions in healthcare (CReDECI): guideline development. *Int J Nurs Stud*, *49*(1), 40-46. <u>https://doi.org/10.1016/j.ijnurstu.2011.08.003</u>

Möhler, R., Köpke, S., & Meyer, G. (2015) Criteria for Reporting the Development and Evaluation of Complex Interventions in Healthcare: revised guideline (CReDECI2). *Trials, 16(204)*. doi: 10.1186/s13063-015-0709-y.

Moll, S. E., Gewurtz, R. E., Krupa, T. M., Law, M. C., Lariviere, N., & Levasseur, M. (2015). "Do-Live-Well": a Canadian framework for promoting occupation, health, and well-being. *Can J Occup Ther*, *82*(1), 9-23. <u>https://doi.org/10.1177/0008417414545981</u>

Moore, A., Grime, J., Campbell, P., & Richardson, J. (2013). Troubling stoicism: Sociocultural influences and applications to health and illness behaviour. *Health:*, *17*(2), 159-173.

Moore, G. F., Audrey, S., Barker, M., Bond, L., Bonell, C., Hardeman, W., Moore, L., O'Cathain, A., Tinati, T., Wight, D., & Baird, J. (2015). Process evaluation of complex interventions: Medical Research Council guidance. *BMJ*, *350*, h1258. https://doi.org/10.1136/bmj.h1258

Moore, G. F., Evans, R. E., Hawkins, J., Littlecott, H., Melendez-Torres, G. J., Bonell, C., & Murphy, S. (2019a). From complex social interventions to interventions in complex social systems: Future directions and unresolved questions for intervention development and evaluation. *Evaluation (Lond)*, *25*(1), 23-45. <u>https://doi.org/10.1177/1356389018803219</u>

Moore, G. F., Evans, R. E., Hawkins, J., Littlecott, H., Melendez-Torres, G. J., Bonell, C., & Murphy, S. (2019b). From complex social interventions to interventions in complex social systems: future directions and unresolved questions for intervention development and evaluation. *Evaluation*, *25*(1), 23-45.

Morley, D., Selai, C., & Thompson, A. (2012). The self-report Barthel Index: preliminary validation in people with Parkinson's disease. *Eur J Neurol*, *19*(6), 927-929. <u>https://doi.org/10.1111/j.1468-1331.2011.03592.x</u>

Morris, D. M., & Jenkins, G. R. (2018). Preparing Physical and Occupational Therapists to Be Health Promotion Practitioners: A Call for Action. *International Journal of Environmental Research and Public Health*, *15*(2), 392. <u>https://www.mdpi.com/1660-4601/15/2/392</u> Morris, Z. S., Wooding, S., & Grant, J. (2011). The answer is 17 years, what is the question: understanding time lags in translational research. *J R Soc Med*, *104*(12), 510-520. <u>https://doi.org/10.1258/jrsm.2011.110180</u>

Morton, K., Beauchamp, M., Prothero, A., Joyce, L., Saunders, L., Spencer-Bowdage, S., Dancy, B., & Pedlar, C. (2015). The effectiveness of motivational interviewing for health behaviour change in primary care settings: a systematic review. *Health psychology review*, *9*(2), 205-223.

Munn, Z., Aromataris, E., Tufanaru, C., Stern, C., Porritt, K., Farrow, J., Lockwood, C., Stephenson, M., Moola, S., Lizarondo, L., McArthur, A., Peters, M., Pearson, A., & Jordan, Z. (2019). The development of software to support multiple systematic review types: the Joanna Briggs Institute System for the Unified Management, Assessment and Review of Information (JBI SUMARI). *JBI Evidence Implementation*, *17*(1), 36-43. <u>https://doi.org/10.1097/xeb.00000000000152</u>

Munn, Z., Peters, M. D. J., Stern, C., Tufanaru, C., McArthur, A., & Aromataris, E. (2018). Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. *BMC Medical Research Methodology*, *18*(1), 143. <u>https://doi.org/10.1186/s12874-018-0611-x</u>

Murray-Mendes, S., Martinez, A. R., & Hackett, K. L. (2021). Identifying Occupational Therapy Research Priorities in Trinidad and Tobago: A Group Concept Mapping Study. *Occup Ther Int, 2021*, 9970566. <u>https://doi.org/10.1155/2021/9970566</u>

Muskin, P. (2023). *What are anxiety disorders?* American Psychiatric Association. Retrieved 19 July from <u>https://www.psychiatry.org/patients-families/anxiety-disorders/what-are-anxiety-disorders#section 13</u>

Myers-Schulz, B., & Koenigs, M. (2012). Functional anatomy of ventromedial prefrontal cortex: implications for mood and anxiety disorders. *Mol Psychiatry*, *17*(2), 132-141. <u>https://doi.org/10.1038/mp.2011.88</u> Nag, N., & Jelinek, G. A. (2019). A Narrative Review of Lifestyle Factors Associated with Parkinson's Disease Risk and Progression [Research Support, Non-U.S. Gov't Review]. *Neurodegenerative Diseases*, *19*(2), 51-59.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medl&AN=3148
7721

Naisola-Ruiter, V. (2022). The Delphi technique: a tutorial. *Research in Hospitality Management*, *12*(1), 91-97. <u>https://doi.org/10.1080/22243534.2022.2080942</u>

Nambisan, S., Agarwal, R., & Tanniru, M. (1999). Organizational Mechanisms for Enhancing User Innovation in Information Technology. *MIS Quarterly*, *23*(3), 365-395. <u>https://doi.org/10.2307/249468</u>

Napier-Raman, S., Rosas, S., Hossain, S. Z., Mpofu, E., Lee, M.-J., Liamputtong, P., Dune, T., & Mapedzahama, V. (2023). Concept Mapping Method. In P. Liamputtong (Ed.), *Handbook of Social Sciences and Global Public Health* (pp. 1-25). Springer International Publishing. https://doi.org/10.1007/978-3-030-96778-9_63-1

National Collaborating Centre for Chronic, C. (2006). *National Institute for Health and Clinical Excellence (NICE) Guidelines – Parkinson's Disease: Diagnosis and Management in Primary and Secondary Care*. Royal College of Physicians.

National Collaborating Centre for Mental Health. (2018). The Improving Access to Psychological Therapies Manual [Manual].

National Institute for Health Research. (2012). *Briefing notes for researchers: public involvement in NHS, public health and social care research*. <u>http://www.invo.org.uk/wp-content/uploads/2014/11/9938 INVOLVE Briefing Notes WEB.pdf</u>

Naylor, C., Parsonage, M., McDaid, D., Knapp, M., Fossey, M., & Galea, A. (2012). *Long-term conditions and mental health: the cost of co-morbidities*. E. King's Fund (London.

Nelson, D. L., & Mathiowetz, V. (2004). Randomized Controlled Trials To Investigate Occupational Therapy Research Questions. *The American Journal of Occupational Therapy*, *58*(1), 24-34. <u>https://doi.org/10.5014/ajot.58.1.24</u>

Ng, S. Y.-E., Chia, N. S.-Y., Abbas, M. M., Saffari, E. S., Choi, X., Heng, D. L., Xu, Z., Tay, K.-Y., Au, W.-L., Tan, E.-K., & Tan, L. C.-S. (2021). Physical Activity Improves Anxiety and Apathy in Early Parkinson's Disease: A Longitudinal Follow-Up Study [Brief Research Report]. *Frontiers in Neurology*, *11*. <u>https://doi.org/10.3389/fneur.2020.625897</u>

NICE. (2011). Generalised anxiety disorder and panic disorder in adults: management (NICE clinical guideline 113). <u>https://www.nice.org.uk/guidance/cg113/resources/generalised-anxiety-disorder-and-panic-disorder-in-adults-management-pdf-35109387756997</u>

NICE. (2018). Parkinson's disease Quality standard. NICE guidelines (QS164). https://www.nice.org.uk/guidance/qs164/resources/parkinsons-disease-pdf-75545600441029

NICE. (2022). *Parkinson's disease in adults (NICE guideline 71)*. London, GB Retrieved from https://www.nice.org.uk/guidance/ng71

Niederberger, M., Köberich, S., & Network, m. o. t. D. (2021). Coming to consensus: the Delphi technique. *European Journal of Cardiovascular Nursing*, *20*(7), 692-695. <u>https://doi.org/10.1093/eurjcn/zvab059</u>

Niederman, F., Brancheau, J. C., & Wetherbe, J. C. (1991). Information Systems Management Issues for the 1990s. *MIS Quarterly*, *15*(4), 475-500. <u>https://doi.org/10.2307/249452</u>

Nielsen, K. T., Klokker, L., Guidetti, S., & Waehrens, E. E. (2018). Identifying, organizing and prioritizing ideas on how to enhance ADL ability. *Scand J Occup Ther*, 1-12. <u>https://doi.org/10.1080/11038128.2018.1424235</u> Nielsen, S. S., & Christensen, J. R. (2018). Occupational therapy for adults with overweight and obesity: mapping interventions involving occupational therapists. *Occupational Therapy International, 2018*.

Nieto-Escamez, F., Obrero-Gaitán, E., & Cortés-Pérez, I. (2023). Visual Dysfunction in Parkinson's Disease. *Brain Sci*, *13*(8). <u>https://doi.org/10.3390/brainsci13081173</u>

Nilsen, P., & Bernhardsson, S. (2019). Context matters in implementation science: a scoping review of determinant frameworks that describe contextual determinants for implementation outcomes. *BMC Health Services Research*, *19*(1), 189. <u>https://doi.org/10.1186/s12913-019-4015-3</u>

Nodehi, Z., Mehdizadeh, H., Azad, A., Mehdizadeh, M., Reyhanian, E., Saberi, Z. S., Meimandi, M., Soltanzadeh, A., Roohi-Azizi, M., Vasaghi-Gharamaleki, B., Parnianpour, M., Khalaf, K., & Taghizadeh, G. (2021). Anxiety and cognitive load affect upper limb motor control in Parkinson's disease during medication phases. *Annals of the New York Academy of Sciences*, *1494*(1), 44-58. <u>https://doi.org/https://doi.org/10.1111/nyas.14564</u>

Nolan, T., & Berwick, D. (2006). All-or-None Measurement Raises the Bar on Performance. *Jama*, *295*(10), 1168-1170.

O'Cathain, A., Croot, L., Duncan, E., Rousseau, N., Sworn, K., Turner, K. M., Yardley, L., & Hoddinott, P. (2019). Guidance on how to develop complex interventions to improve health and healthcare. *BMJ Open*, *9*(8). <u>https://doi.org/10.1136/bmjopen-2019-029954</u>

O'Neill, J., Tabish, H., Welch, V., Petticrew, M., Pottie, K., Clarke, M., Evans, T., Pardo Pardo, J., Waters, E., White, H., & Tugwell, P. (2014). Applying an equity lens to interventions: using PROGRESS ensures consideration of socially stratifying factors to illuminate inequities in health. *Journal of Clinical Epidemiology*, *67*(1), 56-64. https://doi.org/10.1016/j.jclinepi.2013.08.005

OECD. (no date). *Health care use - length of hospital stay - OECD data*. Retrieved 06 December from <u>https://data.oecd.org/healthcare/length-of-hospital-stay.htm</u>

Office for National Statistics. (2019). *Exploring the UK's digital divide*.

https://www.ons.gov.uk/peoplepopulationandcommunity/householdcharacteristics/homei nternetandsocialmediausage/articles/exploringtheuksdigitaldivide/2019-03-04

Office for National Statistics. (2020). 2011 Census: Aggregate Data. In: UK Data Service.

Oosterwijk, C. S., Vriend, C., Berendse, H. W., van der Werf, Y. D., & van den Heuvel, O. A. (2018). Anxiety in Parkinson's disease is associated with reduced structural covariance of the striatum. *J Affect Disord*, *240*, 113-120. <u>https://doi.org/10.1016/j.jad.2018.07.053</u>

Ost, L. G., & Breitholtz, E. (2000). Applied relaxation vs. cognitive therapy in the treatment of generalized anxiety disorder. *Behaviour Research and Therapy*, *38*(8), 777-790. <u>http://www.ncbi.nlm.nih.gov/pubmed/10937426</u>

Ouzzani, M., Hammady, H., Fedorowicz, Z., & Elmagarmid, A. (2016). Rayyan-a web and mobile app for systematic reviews. *Syst Rev*, *5*(1), 210. <u>https://doi.org/10.1186/s13643-016-0384-4</u>

Padilla-Coreano, N., Bolkan, S. S., Pierce, G. M., Blackman, D. R., Hardin, W. D., Garcia-Garcia, A. L., Spellman, T. J., & Gordon, J. A. (2016). Direct Ventral Hippocampal-Prefrontal Input Is Required for Anxiety-Related Neural Activity and Behavior. *Neuron*, *89*(4), 857-866. <u>https://doi.org/10.1016/j.neuron.2016.01.011</u>

Padilla, R. (2011). Effectiveness of Interventions Designed to Modify the Activity Demands of the Occupations of Self-Care and Leisure for People With Alzheimer's Disease and Related Dementias. *The American Journal of Occupational Therapy*, *65*(5), 523-531. https://doi.org/10.5014/ajot.2011.002618

Park, J. J. H., Mogg, R., Smith, G. E., Nakimuli-Mpungu, E., Jehan, F., Rayner, C. R., Condo, J., Decloedt, E. H., Nachega, J. B., Reis, G., & Mills, E. J. (2021). How COVID-19 has fundamentally changed clinical research in global health. *Lancet Glob Health*, *9*(5), e711e720. <u>https://doi.org/10.1016/S2214-109X(20)30542-8</u> Parker, E. L., Banfield, M., Fassnacht, D. B., Hatfield, T., & Kyrios, M. (2021). Contemporary treatment of anxiety in primary care: a systematic review and meta-analysis of outcomes in countries with universal healthcare. *BMC Family Practice*, *22*(1), 92. https://doi.org/10.1186/s12875-021-01445-5

Parkinson's UK. (2017). *The Cost of Parkinson's*. <u>https://www.parkinsons.org.uk/sites/default/files/2017-</u> 07/CS2547%20Cost%20of%20Parkinson%27s%20report%202017%20-%20UK 1.pdf

Parkinson's UK. (2018). The incidence and prevalence of Parkinson's in the UK. Results from the Clinical Practice Research Datalink.

https://www.parkinsons.org.uk/professionals/resources/methods-used-estimate-incidenceand-prevalence-parkinsons-uk-report

Parkinson's UK. (2014). *Parkinson's and You*. Retrieved February from <u>https://s3-eu-west-</u> <u>1.amazonaws.com/puk-live-1-d8-ie/2017-03/Parkinson%27s%20and%20you.pdf</u>

Patel, V., Saxena, S., Lund, C., Thornicroft, G., Baingana, F., Bolton, P., Chisholm, D., Collins,
P. Y., Cooper, J. L., Eaton, J., Herrman, H., Herzallah, M. M., Huang, Y., Jordans, M. J. D.,
Kleinman, A., Medina-Mora, M. E., Morgan, E., Niaz, U., Omigbodun, O.,...UnÜtzer, J. (2018).
The Lancet Commission on global mental health and sustainable development. *The Lancet*, *392*(10157), 1553-1598. <u>https://doi.org/10.1016/S0140-6736(18)31612-X</u>

Pelicioni, P. H. S., Menant, J. C., Latt, M. D., & Lord, S. R. (2019). Falls in Parkinson's Disease Subtypes: Risk Factors, Locations and Circumstances. *International Journal of Environmental Research and Public Health*, *16*(12), 2216. <u>https://www.mdpi.com/1660-4601/16/12/2216</u>

Penninx, B. W., Pine, D. S., Holmes, E. A., & Reif, A. (2021). Anxiety disorders. *Lancet*, *397*(10277), 914-927. <u>https://doi.org/10.1016/S0140-6736(21)00359-7</u>

Penzo, M. A., Robert, V., Tucciarone, J., De Bundel, D., Wang, M., Van Aelst, L., Darvas, M., Parada, L. F., Palmiter, R. D., He, M., Huang, Z. J., & Li, B. (2015). The paraventricular thalamus controls a central amygdala fear circuit. *Nature*, *519*(7544), 455-459. https://doi.org/10.1038/nature13978

Persch, A. C., & Page, S. J. (2013). Protocol development, treatment fidelity, adherence to treatment, and quality control. *The American Journal of Occupational Therapy*, *67*(2), 146-153.

Peters, M., Godrey, C., McInerney, P., Munn, Z., Tricco, A., & Khalil, H. (2020). Chapter 11: Scoping Reviewes (2020 version). In E. Aromataris & Z. Munn (Eds.), *JBI Manial for Evidence Synthesis*. JBI. <u>https://doi.org/https://doi.org/10.46658/JBIMES-20-12</u>

Peters, M., Marnie, C., Tricco, A. C., Pollock, D., Munn, Z., Alexander, L., McInerney, P., Godfrey, C. M., & Khalil, H. (2020). Updated methodological guidance for the conduct of scoping reviews. *JBI Evidence Synthesis*, *18*(10), 2119-2126. <u>https://doi.org/10.11124/jbies-20-00167</u>

Pfeiffer, H. C. V., Løkkegaard, A., Zoetmulder, M., Friberg, L., & Werdelin, L. (2014). Cognitive impairment in early-stage non-demented Parkinson's disease patients. *Acta Neurologica Scandinavica*, *129*(5), 307-318.

https://doi.org/https://doi.org/10.1111/ane.12189

Pighills, A. C., Plummer, D., Harvey, D., & Pain, T. (2013). Positioning occupational therapy as a discipline on the research continuum: Results of a cross-sectional survey of research experience. *Australian Occupational Therapy Journal*, *60*(4), 241-251. https://doi.org/10.1111/1440-1630.12057

Piotrowski, M. C., Lunsford, J., & Gaynes, B. N. (2021). Lifestyle psychiatry for depression and anxiety: Beyond diet and exercise. *Lifestyle Medicine*, *2*(1). <u>https://doi.org/10.1002/lim2.21</u>

Pisegna, J., Anderson, S., & Krok-Schoen, J. L. (2022). Occupational Therapy Interventions to Address Depressive and Anxiety Symptoms in the Physical Disability Inpatient Rehabilitation

Setting: A Systematic Review. *Am J Occup Ther*, 76(1). https://doi.org/10.5014/ajot.2022.049068

Plano Clark, V. L., & Ivankova, N. V. (2016). *Mixed Methods Research* (1 ed.). Sage Publications Inc.

Poerbodipoero, S. J., Sturkenboom, I. H., van Hartingsveldt, M. J., Nijhuis-van der Sanden, M. W. G., & Graff, M. J. (2016). The construct validity of the Dutch version of the activity card sort. *Disability & Rehabilitation*, *38*(19), 1943-1951. https://doi.org/10.3109/09638288.2015.1107779

Pontone, G. M., Dissanayka, N., Apostolova, L., Brown, R. G., Dobkin, R., Dujardin, K., Friedman, J. H., Leentjens, A. F. G., Lenze, E. J., Marsh, L., Mari, L., Monchi, O., Richard, I. H., Schrag, A., Strafella, A. P., Vernaleo, B., Weintraub, D., & Mari, Z. (2019). Report from a multidisciplinary meeting on anxiety as a non-motor manifestation of Parkinson's disease. *NPJ Parkinsons Dis*, *5*, 30. <u>https://doi.org/10.1038/s41531-019-0102-8</u>

Pontone, G. M., McDaniels, B., Keener, A. M., & Subramanian, I. (2023). A Wellness Prescription for Parkinson's: Mid to Late-Stage Disease. *The American Journal of Geriatric Psychiatry*, *31*(9), 737-747. <u>https://doi.org/https://doi.org/10.1016/j.jagp.2023.02.049</u>

Pontone, G. M., & Mills, K. A. (2021). Optimal Treatment of Depression and Anxiety in Parkinson's Disease. *The American Journal of Geriatric Psychiatry*, *29*(6), 530-540. <u>https://doi.org/https://doi.org/10.1016/j.jagp.2021.02.037</u>

Pontone, G. M., Williams, J. R., Anderson, K., Chase, G., Goldstein, S., Grill, S., Hirsch, E. S., Little, J. T., Margolis, R. L., Rabins, P. V., & Marsh, L. (2009). Prevalence of Anxiety Disorders and Anxiety Subtypes in Patients With Parkinson's Disease. *Movement Disorders, 24*(9), 1333-1338. <u>https://doi.org/10.1002/mds.22611.Prevalence</u>

Popper, K. R. (1959). The logic of scientific discovery. Basic Books.

Prado, L., Hadley, R., & Rose, D. (2020). Taking Time: A Mixed Methods Study of Parkinson's Disease Caregiver Participation in Activities in Relation to Their Wellbeing. *Parkinson's Disease (20420080)*, 1-18. https://doi.org/10.1155/2020/7370810

Prediger, R. D., Matheus, F. C., Schwarzbold, M. L., Lima, M. M., & Vital, M. A. (2012). Anxiety in Parkinson's disease: a critical review of experimental and clinical studies [Research Support, Non-U.S. Gov't Review]. *Neuropharmacology*, *62*(1), 115-124. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med9&AN=2190</u> <u>3105</u>

Prior, S. (1998a). Anxiety Management: Results of a Follow-up Study. *British Journal of Occupational Therapy*, *61*(6), 284-285. <u>https://doi.org/10.1177/030802269806100615</u>

Prior, S. (1998b). Determining the Effectiveness of a Short-Term Anxiety Management Course. *British Journal of Occupational Therapy*, *61*(5), 207-213. <u>https://doi.org/10.1177/030802269806100507</u>

Prior, S. J., & Campbell, S. (2018). Patient and Family Involvement: A Discussion of Co-Led Redesign of Healthcare Services. *J Particip Med*, *10*(1), e5. <u>https://doi.org/10.2196/jopm.8957</u>

Pugh, Z. H., Choo, S., Leshin, J. C., Lindquist, K. A., & Nam, C. S. (2022). Emotion depends on context, culture and their interaction: evidence from effective connectivity. *Soc Cogn Affect Neurosci*, *17*(2), 206-217. <u>https://doi.org/10.1093/scan/nsab092</u>

Pusswald, G., Fleck, M., Lehrner, J., Haubenberger, D., Weber, G., & Auff, E. (2012). The "Sense of Coherence" and the coping capacity of patients with Parkinson disease. *Int Psychogeriatr*, *24*(12), 1972-1979. <u>https://doi.org/10.1017/S1041610212001330</u>

Pyatak, E., King, M., Vigen, C. L. P., Salazar, E., Diaz, J., Schepens Niemiec, S. L., Blanchard, J., Jordan, K., Banerjee, J., & Shukla, J. (2019). Addressing Diabetes in Primary Care: Hybrid Effectiveness–Implementation Study of Lifestyle Redesign[®] Occupational Therapy. *The* American Journal of Occupational Therapy, 73(5), 7305185020p7305185021-7305185020p7305185012. <u>https://doi.org/10.5014/ajot.2019.037317</u>

Pyatak, E. A., Carandang, K., Rice Collins, C., & Carlson, M. (2022). Optimizing Occupations, Habits, and Routines for Health and Well-Being With Lifestyle Redesign(R): A Synthesis and Scoping Review. *Am J Occup Ther*, *76*(5). <u>https://doi.org/10.5014/ajot.2022.049269</u>

QSR International. (2012). *Nvivo*. In QSR International. <u>https://ict-</u> webtools.plymouth.ac.uk/workathome/QSR/NVivo11.aspx

Rachman, S. (2012). Health anxiety disorders: A cognitive construal. *Behaviour Research and Therapy*, *50*(7), 502-512. <u>https://doi.org/https://doi.org/10.1016/j.brat.2012.05.001</u>

Radder, D. L. M., Sturkenboom, I. H., van Nimwegen, M., Keus, S. H., Bloem, B. R., & de Vries, N. M. (2016). Physical Therapy and Occupational Therapy in Parkinson's Disease. *International Journal of Neuroscience*, 7454(December), 1-42. https://doi.org/10.1080/00207454.2016.1275617

Rajan, R., Brennan, L., Bloem, B. R., Dahodwala, N., Gardner, J., Goldman, J. G., Grimes, D.
A., Iansek, R., Kovács, N., McGinley, J., Parashos, S. A., Piemonte, M. E. P., & Eggers, C.
(2020). Integrated Care in Parkinson's Disease: A Systematic Review and Meta-Analysis. *Movement Disorders*, 35(9), 1509-1531. <u>https://doi.org/10.1002/mds.28097</u>

Ramaswamy, R., Shidhaye, R., & Nanda, S. (2018). Making complex interventions work in low resource settings: developing and applying a design focused implementation approach to deliver mental health through primary care in India. *Int J Ment Health Syst*, *12*, 5. <u>https://doi.org/10.1186/s13033-018-0181-7</u>

Rawat, C. S., & Pandey, S. (2022). Parkinson's Disease–An Introduction. In S. P. Arjunan & D. K. Kumar (Eds.), *Techniques for Assessment of Parkinsonism for Diagnosis and Rehabilitation* (pp. 1-24). Springer Singapore. <u>https://doi.org/10.1007/978-981-16-3056-9_1</u>

Ray, M. H., Russ, A. N., Walker, R. A., & McDannald, M. A. (2020). The Nucleus Accumbens Core is Necessary to Scale Fear to Degree of Threat. *J Neurosci*, *40*(24), 4750-4760. <u>https://doi.org/10.1523/jneurosci.0299-20.2020</u>

Read, J., Cable, S., Löfqvist, C., Iwarsson, S., Bartl, G., & Schrag, A. (2019). Experiences of health services and unmet care needs of people with late-stage Parkinson's in England: A qualitative study. *PLoS ONE*, *14*(12).

https://doi.org/http://dx.doi.org/10.1371/journal.pone.0226916

Read, J., Frost, R., Walters, K., Tuijt, R., Manthorpe, J., Maydon, B., Pigott, J., Schrag, A., & Davies, N. (2022). Transitions and challenges for people with Parkinson's and their family members: A qualitative study. *PLoS ONE*, *17*(7), e0268588. https://doi.org/10.1371/journal.pone.0268588

Reisine, T. D., Fields, J. Z., & Yamamura, H. I. (1977). Neurotransmitter receptor alterations in Parkinson's disease. *Life Sci*, *21*(3), 335-343. <u>https://doi.org/10.1016/0024-</u> 3205(77)90514-8

Reitz, S. M., Scaffa, M. E., & Dorsey, J. (2020). Occupational Therapy in the Promotion of Health and Well-Being. *Am J Occup Ther*, *74*(3), 7403420010p7403420011-7403420010p7403420014. <u>https://doi.org/10.5014/ajot.2020.743003</u>

Ren, C., & Tao, Q. (2020). Neural Circuits Underlying Innate Fear. *Adv Exp Med Biol*, *1284*, 1-7. <u>https://doi.org/10.1007/978-981-15-7086-5_1</u>

Renedo, A., Marston, C. A., Spyridonidis, D., & Barlow, J. (2015). Patient and Public Involvement in Healthcare Quality Improvement: How organizations can help patients and professionals to collaborate. *Public Management Review*, *17*(1), 17-34. <u>https://doi.org/10.1080/14719037.2014.881535</u>

Reynolds, G. O., Saint-Hilaire, M., Thomas, C. A., Barlow, D. H., & Cronin-Golomb, A. (2020). Cognitive-Behavioral Therapy for Anxiety in Parkinson's Disease. *Behavior Modification*, 44(4), 552-579. <u>https://doi.org/10.1177/0145445519838828</u> Richard, I. H., Frank, S., McDermott, M. P., Wang, H., Justus, A. W., LaDonna, K. A., & Kurlan, R. (2004). The ups and downs of Parkinson disease: a prospective study of mood and anxiety fluctuations. *Cogn Behav Neurol*, *17*(4), 201-207.

Richardson, A. L. (2023). Hope and anxiety. *Current Opinion in Psychology*, *53*, 101664. <u>https://doi.org/https://doi.org/10.1016/j.copsyc.2023.101664</u>

Rintala, T. M., Jaatinen, P., Paavilainen, E., & Astedt-Kurki, P. (2013). Interrelation between adult persons with diabetes and their family: a systematic review of the literature. *J Fam Nurs*, *19*(1), 3-28. <u>https://doi.org/10.1177/1074840712471899</u>

Rippe, J. M. (2019). Lifestyle Strategies for Risk Factor Reduction, Prevention, and Treatment of Cardiovascular Disease. *Am J Lifestyle Med*, *13*(2), 204-212. <u>https://doi.org/10.1177/1559827618812395</u>

Robinson, O. C. (2014). Sampling in Interview-Based Qualitative Research: A Theoretical and Practical Guide. *Qualitative Research in Psychology*, *11*(1), 25-41. <u>https://doi.org/10.1080/14780887.2013.801543</u>

Rodriguez, B. F., Bruce, S. E., Pagano, M. E., & Keller, M. B. (2005). Relationships among psychosocial functioning, diagnostic comorbidity, and the recurrence of generalized anxiety disorder, panic disorder, and major depression. *J Anxiety Disord*, *19*(7), 752-766. https://doi.org/10.1016/j.janxdis.2004.10.002

Roper, A., Pacas Fronza, G., Dobkin, R. D., Beaudreau, S. A., Mitchell, L. K., Pachana, N. A., Thangavelu, K., & Dissanayaka, N. N. (2024). A Systematic Review of Psychotherapy Approaches for Anxiety in Parkinson's Disease. *Clin Gerontol*, *47*(2), 188-214. <u>https://doi.org/10.1080/07317115.2022.2074814</u>

Rosas, S. (2020). Conversations About Group Concept Mapping: Applications, Examples, and Enhancements.

Rosas, S. R. (2017). Group concept mapping methodology: toward an epistemology of group conceptualization, complexity, and emergence. *Quality & Quantity*, *51*(3), 1403-1416. <u>https://doi.org/10.1007/s11135-016-0340-3</u>

Rosen, J. B., & Schulkin, J. (1998). From normal fear to pathological anxiety. *Psychological Review*, *105*(2), 325-350. <u>https://doi.org/10.1037/0033-295X.105.2.325</u>

Rosengren, L., Forsberg, A., Brogårdh, C., & Lexell, J. (2021). Social Belonging as the Main Concern for Achieving Life Satisfaction When Adapting to Parkinson's Disease. *International Journal of Environmental Research and Public Health*, *18*(16), 8653. <u>https://www.mdpi.com/1660-4601/18/16/8653</u>

Rosier, C., Williams, H., & Ryrie, I. (1998). Anxiety management groups in a community mental health team. *British Journal of Occupational Therapy*, *61*(5), 203-206. <u>http://bjo.sagepub.com/</u>

Rosland, A.-M., Kieffer, E., Israel, B., Cofield, M., Palmisano, G., Sinco, B., Spencer, M., & Heisler, M. (2008). When Is Social Support Important? The Association of Family Support and Professional Support with Specific Diabetes Self-management Behaviors. *Journal of General Internal Medicine*, *23*(12), 1992-1999. https://doi.org/10.1007/s11606-008-0814-7

Royal College of Occupational Therapists. (2018). *Occupational therapy for people with Parkinson's*. <u>https://www.parkinsons.org.uk/sites/default/files/2018-</u> <u>11/Occupational%20therapy%20for%20people%20with%20Parkinson%27s%202nd%20editi</u> <u>on%202018.pdf</u>

Royal College of Occupational Therapists. (2019). *Royal College of Occupational Therapists' Research and Development Strategy 2019-2024*.

Rozeske, R. R., Valerio, S., Chaudun, F., & Herry, C. (2015). Prefrontal neuronal circuits of contextual fear conditioning. *Genes Brain Behav*, *14*(1), 22-36. https://doi.org/10.1111/gbb.12181 Ruiz, M. A., Zamorano, E., Garcia-Campayo, J., Pardo, A., Freire, O., & Rejas, J. (2011). Validity of the GAD-7 scale as an outcome measure of disability in patients with generalized anxiety disorders in primary care. *J Affect Disord*, *128*(3), 277-286. <u>https://doi.org/10.1016/j.jad.2010.07.010</u>

Rutter, L. A., & Brown, T. A. (2017). Psychometric Properties of the Generalized Anxiety Disorder Scale-7 (GAD-7) in Outpatients with Anxiety and Mood Disorders. *J Psychopathol Behav Assess*, *39*(1), 140-146. <u>https://doi.org/10.1007/s10862-016-9571-9</u>

Ryan, R. M., Patrick, H., Deci, E. L., & Williams, G. C. (2008). Facilitating health behaviour change and its maintenance: Interventions based on self-determination theory. *European Health Psychologist*, *10*(1), 2-5.

Sainsbury, A., Seebass, G., Bansal, A., & Young, J. B. (2005). Reliability of the Barthel Index when used with older people. *Age and Ageing*, *34*(3), 228-232. <u>https://doi.org/10.1093/ageing/afi063</u>

Salkovskis, P. M., Clark, D. M., Hackmann, A., Wells, A., & Gelder, M. G. (1999). An experimental investigation of the role of safety-seeking behaviours in the maintenance of panic disorder with agoraphobia. *Behav Res Ther*, *37*(6), 559-574.

https://doi.org/10.1016/s0005-7967(98)00153-3

Sanders, M. J., Wiltgen, B. J., & Fanselow, M. S. (2003). The place of the hippocampus in fear conditioning. *Eur J Pharmacol*, *463*(1-3), 217-223. <u>https://doi.org/10.1016/s0014-</u> 2999(03)01283-4

Schell, S., Luke, D., Schooley, M., Elliott, M., Herbers, S., Mueller, N., & Bunger, A. (2013). Public health program capacity for sustainability: a new framework. *Implementation Science*, 8(15), 1-9.

Schulz, K. F., Altman, D. G., & Moher, D. (2010). CONSORT 2010 statement: updated guidelines for reporting parallel group randomised trials. *BMJ*, *340*, c332. https://doi.org/10.1136/bmj.c332 Scobbie, L., Dixon, D., & Wyke, S. (2011). Goal setting and action planning in the rehabilitation setting: development of a theoretically informed practice framework. *Clinical Rehabilitation*, *25*(5), 468-482. <u>https://doi.org/10.1177/0269215510389198</u>

Scott, A. H., Butin, D. N., Tewfik, D., Burkhardt, A., Mandel, D., & Nelson, L. (2001). Occupational Therapy as a Means to Wellness with the Elderly. *Physical & Occupational Therapy in Geriatrics*, *18*(4), 3-22. <u>https://doi.org/10.1080/J148v18n04_02</u>

Seritan, A. L., Rienas, C., Duong, T., Delucchi, K., & Ostrem, J. L. (2019). Ages at Onset of Anxiety and Depressive Disorders in Parkinson's Disease. *J Neuropsychiatry Clin Neurosci*, *31*(4), 346-352. <u>https://doi.org/10.1176/appi.neuropsych.18090201</u>

Sermeus, W. (2015). Modelling Process and Outcomes in Complex Interventions. In D. A. Richards & I. R. Hallberg (Eds.), *Complex Interventions in Health* (pp. 111-120). Routledge.

Shah, R., Read, J., Davies, N., Nimmons, D., Pigott, J., Schrag, A., Walters, K., & Armstrong,
M. (2022). People with Parkinson's perspectives and experiences of self-management:
Qualitative findings from a UK study. *PLoS ONE*, *17*(9), e0273428.
https://doi.org/10.1371/journal.pone.0273428

Shechner, T., Hong, M., Britton, J. C., Pine, D. S., & Fox, N. A. (2014). Fear conditioning and extinction across development: Evidence from human studies and animal models. *Biological psychology*, *100*, 1-12. <u>https://doi.org/https://doi.org/10.1016/j.biopsycho.2014.04.001</u>

Sherrington, C., Fairhall, N., Kwok, W., Wallbank, G., Tiedemann, A., Michaleff, Z. A., Ng, C. A. C. M., & Bauman, A. (2020). Evidence on physical activity and falls prevention for people aged 65+ years: systematic review to inform the WHO guidelines on physical activity and sedentary behaviour. *International Journal of Behavioral Nutrition and Physical Activity*, *17*(1), 144. <u>https://doi.org/10.1186/s12966-020-01041-3</u>

Shin, L. M., & Liberzon, I. (2010). The neurocircuitry of fear, stress, and anxiety disorders. *Neuropsychopharmacology*, *35*(1), 169-191. <u>https://doi.org/10.1038/npp.2009.83</u>

Shulman, L. M., Taback, R. L., Rabinstein, A. A., & Weiner, W. J. (2002). Non-recognition of depression and other non-motor symptoms in Parkinson's Disease. *Parkinsonism and Related Disorders*, *8*, 193-197.

Simuni, T., & Fernandez, H. H. (2013). Anxiety in Parkinson's Disease. In R. F. Pfeiffer & I. Bodis-Wollner (Eds.), *Parkinson's Disease and Nonmotor Dysfunction* (pp. 17-29). Humana Press. <u>https://doi.org/10.1007/978-1-60761-429-6_2</u>

Sinaise, M. K., Tran, A., Johnson, H. M., Vedder, L. S., Hoppe, K. K., & Lauver, D. (2023). Concepts from behavioral theories can guide clinicians in coaching for behavior change. *Patient Education and Counseling*, *106*, 188-193.

Sjodahl Hammarlund, C., Nilsson, M. H., Idvall, M., Rosas, S. R., & Hagell, P. (2014). Conceptualizing and prioritizing clinical trial outcomes from the perspectives of people with Parkinson's disease versus health care professionals: a concept mapping study. *Qual Life Res*, *23*(6), 1687-1700. <u>https://doi.org/10.1007/s11136-013-0614-3</u>

Skivington, K., Matthews, L., Simpson, S. A., Craig, P., Baird, J., Blazeby, J. M., Boyd, K. A.,
Craig, N., French, D. P., McIntosh, E., Petticrew, M., Rycroft-Malone, J., White, M., & Moore,
L. (2021). A new framework for developing and evaluating complex interventions: update of
Medical Research Council guidance. *BMJ*, *374*, n2061. <u>https://doi.org/10.1136/bmj.n2061</u>

Slade, S. C., Philip, K., & Morris, M. E. (2018). Frameworks for embedding a research culture in allied health practice: a rapid review. *Health Research Policy and Systems*, *16*(1), 29. https://doi.org/10.1186/s12961-018-0304-2

Spitzer, R., Kroenke, K., Williams, J., & Lowe, B. (2006). A Brief Measure for Assessing Generalized Anxiety Disorder: The GAD-7. *JAMA Internal Medicine*, *166*(10), 1092-1097. <u>https://doi.org/doi:10.1001/archinte.166.10.1092</u>

Steimer, T. (2002). The biology of fear- and anxiety-related behaviors. *Dialogues in clinical neuroscience*, *4*(3), 231-249. <u>https://doi.org/10.31887/DCNS.2002.4.3/tsteimer</u>

Street, J., & Richard, L. (2013). How clinician–patient communication contributes to health improvement: modeling pathways from talk to outcome. *Patient Education and Counseling*, *92*(3), 286-291.

Strom, J. L., & Egede, L. E. (2012). The impact of social support on outcomes in adult patients with type 2 diabetes: a systematic review. *Curr Diab Rep*, *12*(6), 769-781. <u>https://doi.org/10.1007/s11892-012-0317-0</u>

Sturkenboom, I., Graff, M., Borm, G., Adang, E., Nijhuis-van der Sanden, M., Bloem, B., Munneke, M., & George-Hyslop, P. (2013). Effectiveness of occupational therapy in Parkinson's disease: study protocol for a randomized controlled trial. *Trials*, *14*(1), 34-34. <u>https://doi.org/10.1186/1745-6215-14-34</u>

Sturkenboom, I., Graff, M., Borm, G., Veenhuizen, Y., Bloem, B., Munneke, M., & Nijhuis-van der Sanden, M. (2013). The impact of occupational therapy in Parkinson's disease: a randomized controlled feasibility study. *Clin Rehabil*, *27*(2), 99-112. https://doi.org/10.1177/0269215512448382

Sturkenboom, I., Graff, M., Hendriks, J., Veenhuizen, Y., Munneke, M., Bloem, B., & Nijhuisvan der Sanden, M. (2014). Efficacy of occupational therapy for patients with Parkinson's disease: a randomised controlled trial [Multicenter Study Randomized Controlled Trial Research Support, Non-U.S. Gov't]. *Lancet Neurology*, *13*(6), 557-566. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med11&AN=247</u> 26066

Sturkenboom, I., Thijssen, M., Gons-van Elsacker, J., Jansen, I., Maasdam, A., Schulten, M., Vijver-Visser, D., Steultjens, E., Bloem, B., & Munneke, M. (2011). *Guidelines for Occupational Therapy in Parkinson's Disease Rehabilitation*. ParkinsonNet/NPF.

Subbiah, V. (2023). The next generation of evidence-based medicine. *Nature Medicine*, 29(1), 49-58. <u>https://doi.org/10.1038/s41591-022-02160-z</u>

Swann, C., Rosenbaum, S., Lawrence, A., Vella, S. A., McEwan, D., & Ekkekakis, P. (2021). Updating goal-setting theory in physical activity promotion: a critical conceptual review. *Health psychology review*, *15*(1), 34-50. https://doi.org/10.1080/17437199.2019.1706616

Talebi, A. H., Ypinga, J. H. L., De Vries, N. M., Nonnekes, J., Munneke, M., Bloem, B. R., Heskes, T., Ben-Shlomo, Y., & Darweesh, S. K. L. (2023). Specialized Versus Generic Allied Health Therapy and the Risk of Parkinson's Disease Complications. *Mov Disord*, *38*(2), 223-231. <u>https://doi.org/10.1002/mds.29274</u>

Tamagawa, R., Garland, S., Vaska, M., & Carlson, L. E. (2012). Who benefits from psychosocial interventions in oncology? A systematic review of psychological moderators of treatment outcome. *Journal of Behavioral Medicine*, *35*, 658-673.

Tan, Q. Y., Roberts, H. C., Fraser, S. D. S., Amar, K., & Ibrahim, K. (2023). What are the modifiable factors of treatment burden and capacity among people with Parkinson's disease and their caregivers: A qualitative study. *PLoS ONE*, *18*(3), e0283713. https://doi.org/10.1371/journal.pone.0283713

Tang, W., Kochubey, O., Kintscher, M., & Schneggenburger, R. (2020). A VTA to Basal Amygdala Dopamine Projection Contributes to Signal Salient Somatosensory Events during Fear Learning. *J Neurosci, 40*(20), 3969-3980. <u>https://doi.org/10.1523/jneurosci.1796-</u> <u>19.2020</u>

Taylor, E. (2020). We Agree, Don't We? The Delphi Method for Health Environments Research. *HERD: Health Environments Research & Design Journal*, *13*(1), 11-23. https://doi.org/10.1177/1937586719887709

The EndNote Team. (2013). EndNote. In (Version EndNote 20) [64 bit]. Clarivate.

The King's Fund. (2023). *Key facts and figures about the NHS*. Retrieved 06 December from <u>https://www.kingsfund.org.uk/audio-video/key-facts-figures-nhs</u>

The King's Fund. (2015). Mental Health Under Pressure.

Thompson, B. (2009). Mindfulness-Based Stress Reduction for People with Chronic Conditions. *British Journal of Occupational Therapy*, *72*(9), 405-410. https://doi.org/10.1177/030802260907200907

Tinaz, S., Kamel, S., Aravala, S. S., Sezgin, M., Elfil, M., & Sinha, R. (2021). Distinct neural circuits are associated with subclinical neuropsychiatric symptoms in Parkinson's disease. *J Neurol Sci*, *423*, 117365. <u>https://doi.org/10.1016/j.jns.2021.117365</u>

Todorova, A., Jenner, P., & Chaudhuri, K. R. (2014). Non-motor Parkinson's: integral to motor Parkinson's, yet often neglected. *Practical neurology*, *14*(5), 310-322. <u>https://doi.org/10.1136/practneurol-2013-000741</u>

Tolin, D. F., Davies, C. D., Moskow, D. M., & Hofmann, S. G. (2020). Biofeedback and Neurofeedback for Anxiety Disorders: A Quantitative and Qualitative Systematic Review. In Y.-K. Kim (Ed.), *Anxiety Disorders: Rethinking and Understanding Recent Discoveries* (pp. 265-289). Springer Singapore. <u>https://doi.org/10.1007/978-981-32-9705-0_16</u>

Tolosa, E., Garrido, A., Scholz, S. W., & Poewe, W. (2021). Challenges in the diagnosis of Parkinson's disease. *The Lancet Neurology*, *20*(5), 385-397. <u>https://doi.org/10.1016/S1474-4422(21)00030-2</u>

Torres, E. R. S., Stanojlovic, M., Zelikowsky, M., Bonsberger, J., Hean, S., Mulligan, C., Baldauf, L., Fleming, S., Masliah, E., Chesselet, M. F., Fanselow, M. S., & Richter, F. (2021). Alpha-synuclein pathology, microgliosis, and parvalbumin neuron loss in the amygdala associated with enhanced fear in the Thy1-aSyn model of Parkinson's disease. *Neurobiol Dis*, *158*, 105478. <u>https://doi.org/10.1016/j.nbd.2021.105478</u>

Toussaint, A., Hüsing, P., Gumz, A., Wingenfeld, K., Härter, M., Schramm, E., & Löwe, B. (2020). Sensitivity to change and minimal clinically important difference of the 7-item Generalized Anxiety Disorder Questionnaire (GAD-7). *J Affect Disord*, *265*, 395-401. https://doi.org/10.1016/j.jad.2020.01.032 Tovote, P., Fadok, J. P., & Lüthi, A. (2015). Neuronal circuits for fear and anxiety. *Nat Rev Neurosci*, *16*(6), 317-331. <u>https://doi.org/10.1038/nrn3945</u>

Tricco, A. C., Lillie, E., Zarin, W., O'Brien, K. K., Colquhoun, H., Levac, D., Moher, D., Peters, M. D. J., Horsley, T., Weeks, L., Hempel, S., Akl, E. A., Chang, C., McGowan, J., Stewart, L., Hartling, L., Aldcroft, A., Wilson, M. G., Garritty, C.,...Straus, S. E. (2018). PRISMA Extension for Scoping Reviews (PRISMA-ScR): Checklist and Explanation. *Ann Intern Med*, *169*(7), 467-473. <u>https://doi.org/10.7326/M18-0850</u>

Trochim, W. M., & McLinden, D. (2017). Introduction to a special issue on concept mapping. *Eval Program Plann*, *60*, 166-175. <u>https://doi.org/10.1016/j.evalprogplan.2016.10.006</u>

Trusson, D., Rowley, E., & Bramley, L. (2019). A mixed-methods study of challenges and benefits of clinical academic careers for nurses, midwives and allied health professionals. *BMJ Open*, *9*(10), e030595. <u>https://doi.org/10.1136/bmjopen-2019-030595</u>

Tuntland, H., Aaslund, M. K., Langeland, E., Espehaug, B., & Kjeken, I. (2016). Psychometric properties of the Canadian Occupational Performance Measure in home-dwelling older adults. *J Multidiscip Healthc*, *9*, 411-423. <u>https://doi.org/10.2147/JMDH.S113727</u>

Tweeddale, P. M., Rowbottom, I., & McHardy, G. J. (1994). Breathing retraining: effect on anxiety and depression scores in behavioural breathlessness. *Journal of Psychosomatic Research*, *38*(1), 11-21. <u>http://www.ncbi.nlm.nih.gov/pubmed/8126685</u>

Twomey, C., Cieza, A., & Baldwin, D. S. (2017). Utility of functioning in predicting costs of care for patients with mood and anxiety disorders: a prospective cohort study. *Int Clin Psychopharmacol*, *32*(4), 205-212. <u>https://doi.org/10.1097/YIC.00000000000178</u>

Uebelacker, L. A., Epstein-Lubow, G., Lewis, T., Broughton, M. K., & Friedman, J. H. (2014). A survey of Parkinson's disease patients: most bothersome symptoms and coping preferences. *J Parkinsons Dis*, *4*(4), 717-723. <u>https://doi.org/10.3233/jpd-140446</u>

Uyeshiro Simon, A., & Collins, C. E. R. (2017). Lifestyle Redesign[®] for Chronic Pain Management: A Retrospective Clinical Efficacy Study. *The American Journal of Occupational Therapy*, *71*(4), 7104190040p7104190041-7104190040p7104190047. <u>https://doi.org/10.5014/ajot.2017.025502</u>

Vahedian-Shahroodi, M., Mansourzadeh, A., Shariat Moghani, S., & Saeidi, M. (2023). Using the Nominal Group Technique in Group Decision-Making: A Review. *Medical Education Bulletin*, *4*(4), 837-845. <u>https://doi.org/10.22034/meb.2024.434656.1090</u>

Valerio, M. A., Rodriguez, N., Winkler, P., Lopez, J., Dennison, M., Liang, Y., & Turner, B. J. (2016). Comparing two sampling methods to engage hard-to-reach communities in research priority setting. *BMC Med Res Methodol*, *16*(1), 146. <u>https://doi.org/10.1186/s12874-016-0242-z</u>

Van de Ven, A. H., & Delbecq, A. L. (1972). The nominal group as a research instrument for exploratory health studies. *American journal of public health*, *62*(3), 337-342. <u>https://doi.org/10.2105/ajph.62.3.337</u>

van der Heide, A., Meinders, M. J., Speckens, A. E. M., Peerbolte, T. F., Bloem, B. R., & Helmich, R. C. (2021). Stress and Mindfulness in Parkinson's Disease: Clinical Effects and Potential Underlying Mechanisms. *Movement Disorders*, *36*(1), 64-70. <u>https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=medp&AN=3309</u> <u>4858</u>

van der Heide, A., Speckens, A. E. M., Meinders, M. J., Rosenthal, L. S., Bloem, B. R., & Helmich, R. C. (2021). Stress and mindfulness in Parkinson's disease - a survey in 5000 patients. *Npj Parkinsons Disease*, *7*(1), 7.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=prem&AN=3346 2213

van der Steen, J. T., Lennaerts, H., Hommel, D., Augustijn, B., Groot, M., Hasselaar, J., Bloem, B. R., & Koopmans, R. (2019). Dementia and Parkinson's Disease: Similar and Divergent Challenges in Providing Palliative Care [Review]. *Frontiers in neurology [electronic resource]*.

10, 54.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=prem6&AN=309
15012

van Mierlo, T. J., Chung, C., Foncke, E. M., Berendse, H. W., & van den Heuvel, O. A. (2015). Depressive symptoms in Parkinson's disease are related to decreased hippocampus and amygdala volume. *Mov Disord*, *30*(2), 245-252. <u>https://doi.org/10.1002/mds.26112</u>

Vansteenkiste, M., & Sheldon, K. M. (2006). There's nothing more practical than a good theory: Integrating motivational interviewing and self-determination theory. *British Journal of Clinical Psychology*, *45*(1), 63-82.

Varela, C., Young, S., Mkandawire, N., Groen, R. S., Banza, L., & Viste, A. (2019). TRANSPORTATION BARRIERS TO ACCESS HEALTH CARE FOR SURGICAL CONDITIONS IN MALAWI a cross sectional nationwide household survey. *BMC Public Health*, *19*(1), 264. <u>https://doi.org/10.1186/s12889-019-6577-8</u>

Veenhuizen, Y., Cup, E. H. C., Jonker, M. A., Voet, N. B. M., van Keulen, B. J., Maas, D. M., Heeren, A., Groothuis, J. T., van Engelen, B. G. M., & Geurts, A. C. H. (2019). Selfmanagement program improves participation in patients with neuromuscular disease A randomized controlled trial [Article]. *Neurology*, *93*(18), E1720-E1731. https://doi.org/10.1212/wnl.00000000008393

Vos, T., Barber, R. M., Bell, B., Bertozzi-Villa, A., Biryukov, S., Bolliger, I., Charlson, F., Davis, A., Degenhardt, L., Dicker, D., Duan, L., Erskine, H., Feigin, V. L., Ferrari, A. J., Fitzmaurice, C., Fleming, T., Graetz, N., Guinovart, C., Haagsma, J.,...Murray, C. J. L. (2015). Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. *The Lancet, 386*(9995), 743-800.

https://doi.org/10.1016/s0140-6736(15)60692-4

Vriend, C., Boedhoe, P. S., Rutten, S., Berendse, H. W., van der Werf, Y. D., & van den Heuvel, O. A. (2016). A smaller amygdala is associated with anxiety in Parkinson's disease: a combined FreeSurfer-VBM study. *J Neurol Neurosurg Psychiatry*, *87*(5), 493-500. https://doi.org/10.1136/jnnp-2015-310383

Wagner, C. C., & Ingersoll, K. S. (2012). *Motivational interviewing in groups*. Guilford Press.

Walder, K., Molineux, M., Bissett, M., & Whiteford, G. (2021). Occupational adaptation – analyzing the maturity and understanding of the concept through concept analysis. *Scandinavian Journal of Occupational Therapy*, *28*(1), 26-40. https://doi.org/10.1080/11038128.2019.1695931

Walker, J. E., & Howland, J. (1991). Falls and Fear of Falling Among Elderly Persons Living in the Community: Occupational Therapy Interventions. *The American Journal of Occupational Therapy*, *45*(2), 119-122. <u>https://doi.org/10.5014/ajot.45.2.119</u>

Walker, L. (2023). What Next? Learning from the findings of the Royal College of Occupational Therapists workforce survey 2022–2023. *British Journal of Occupational Therapy*, *0*(0), 03080226231206679. <u>https://doi.org/10.1177/03080226231206679</u>

Wallace, D. D., Carlson, R. G., & Ohrt, J. H. (2021). Culturally adapted cognitive-behavioral therapy in the treatment of panic episodes and depression in an African American woman: a clinical case illustration. *Journal of Mental Health Counseling*, *43*(1), 40-58.

Walsh, K., & Bennett, G. (2001). Parkinson's disease and anxiety. *Post Graduate Medical Journal*(77), 89-93.

Wang, Y., Zhang, S., Yang, H., Zhang, X., He, S., Wang, J., & Li, J. (2021). Altered cerebellum functional network on newly diagnosed drug-naïve Parkinson's disease patients with anxiety. *Transl Neurosci*, *12*(1), 415-424. <u>https://doi.org/10.1515/tnsci-2020-0192</u>

Ward, J. H. (1963). Hierarchical Grouping to Optimize an Objective Function. *Journal of the American Statistical Association*, *58*(301), 236-244. https://doi.org/10.1080/01621459.1963.10500845 Wearne, S., Greenhill, J., Berryman, C., Sweet, L., & Tietz, L. (2011). An online course in clinical education: Experiences of Australian clinicians [Journal Article]. *Australian family physician*, *40*(12), 1000-1003.

https://search.informit.org/doi/10.3316/informit.687011578684978

Weernink, M. G. M., van Til, J. A., Groothuis-Oudshoorn, C. G. M., & MJ, I. J. (2017). Patient and Public Preferences for Treatment Attributes in Parkinson's Disease. *The Patient: Patient-Centered Outcomes Research*, *10*(6), 763-772.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=med14&AN=285 08354

Wegner, D. M., Schneider, D. J., Carter, S. R., & White, T. L. (1987). Paradoxical effects of thought suppression. *Journal of Personality and Social Psychology*, *53*(1), 5.

Weintraub, D., Newberg, A. B., Cary, M. S., Siderowf, A. D., Moberg, P. J., Kleiner-Fisman, G., Duda, J. E., Stern, M. B., Mozley, D., & Katz, I. R. (2005). Striatal dopamine transporter imaging correlates with anxiety and depression symptoms in Parkinson's disease. *J Nucl Med*, *46*(2), 227-232.

Wells, A. (1995). Meta-Cognition and Worry: A Cognitive Model of Generalized Anxiety Disorder. *Behavioural and Cognitive Psychotherapy*, *23*(3), 301-320. <u>https://doi.org/10.1017/S1352465800015897</u>

Wells, A. (1999). A Cognitive Model of Generalized Anxiety Disorder. *Behavior Modification*, *23*(4), 526-555.

Wells, A. (2010). Metacognitive Theory and Therapy for Worry and Generalized Anxiety Disorder: Review and Status. *Journal of Experimental Psychopathology*, 1(1). <u>https://doi.org/10.5127/jep.007910</u>

Wells, A. (2011). Metacognitive therapy for anxiety and depression. Guilford press.
Welsby, E., Berrigan, S., & Laver, K. (2019). Effectiveness of occupational therapy intervention for people with Parkinson's disease: Systematic review. *Aust Occup Ther J*. https://doi.org/10.1111/1440-1630.12615

Westwood, G., Richardson, A., Latter, S., Macleod Clark, J., & Fader, M. (2018). Building clinical academic leadership capacity: sustainability through partnership. *J Res Nurs*, *23*(4), 346-357. <u>https://doi.org/10.1177/1744987117748348</u>

Whitfield, G. (2010). Group cognitive–behavioural therapy for anxiety and depression. *Advances in Psychiatric Treatment*, *16*(3), 219-227. <u>https://doi.org/10.1192/apt.bp.108.005744</u>

Whittle, N., Fadok, J., MacPherson, K. P., Nguyen, R., Botta, P., Wolff, S. B. E., Müller, C., Herry, C., Tovote, P., Holmes, A., Singewald, N., Lüthi, A., & Ciocchi, S. (2021). Central amygdala micro-circuits mediate fear extinction. *Nature Communications*, *12*(1), 4156. <u>https://doi.org/10.1038/s41467-021-24068-x</u>

Whitty, E., Mansour, H., Aguirre, E., Palomo, M., Charlesworth, G., Ramjee, S., Poppe, M.,
Brodaty, H., Kales, H. C., Morgan-Trimmer, S., Nyman, S. R., Lang, I., Walters, K., Petersen, I.,
Wenborn, J., Minihane, A. M., Ritchie, K., Huntley, J., Walker, Z., & Cooper, C. (2020).
Efficacy of lifestyle and psychosocial interventions in reducing cognitive decline in older
people: Systematic review. *Ageing research reviews*, *62*, 101113.
https://doi.org/https://doi.org/10.1016/j.arr.2020.101113

Willekens, B., Perrotta, G., Cras, P., & Cools, N. (2018). Into the moment: Does mindfulness affect biological pathways in multiple sclerosis? *Frontiers in Behavioral Neuroscience*, *12 (no pagination)*, Article 103.

https://ovidsp.ovid.com/athens?T=JS&CSC=Y&NEWS=N&PAGE=fulltext&D=emed19&AN=62 2815278

Willems, R. A., Lechner, L., Verboon, P., Mesters, I., Kanera, I. M., & Bolman, C. A. W. (2017). Working mechanisms of a web-based self-management intervention for cancer survivors: a randomised controlled trial. *Psychology & Health*, *32*(5), 605-625. https://doi.org/10.1080/08870446.2017.1293054

Willis, G. L., Moore, C., & Armstrong, S. M. (2012). Breaking away from dopamine deficiency: an essential new direction for Parkinson's disease. *Reviews in the Neurosciences*, *23*(4), 403-428.

Wimpenny, K., Savin-Baden, M., & Cook, C. (2014). A qualitative research synthesis examining the effectiveness of interventions used by occupational therapists in mental health. *The British Journal of Occupational Therapy*, *77*(6), 276-288. <u>https://doi.org/10.4276/030802214X14018723137959</u>

Wood, J., Henderson, W., & Foster, E. R. (2022). Occupational Therapy Practice Guidelines for People With Parkinson's Disease. *The American Journal of Occupational Therapy*, *76*(3). <u>https://doi.org/10.5014/ajot.2022.763001</u>

World Federation of Occupational Therapists. (2019). *About Occupational Therapy*. <u>https://www.wfot.org/about-occupational-therapy</u>

World Health Organization. (2001). *International classification of functioning, disability, and health : ICF*. In Version 1.0. Geneva : World Health Organization, [2001] ©2001. <u>https://search.library.wisc.edu/catalog/999977181002121</u>

Wressle, E., Engstrand, C., & Granérus, A.-K. (2007). Living with Parkinson's disease: Elderly patients and relatives perspective on daily living. *Australian Occupational Therapy Journal*, *54*(2), 131-139. <u>https://doi.org/10.1111/j.1440-1630.2006.00610.x</u>

Wu, Z., Pandigama, D. H., Wrigglesworth, J., Owen, A., Woods, R. L., Chong, T. T.-J., Orchard,
S. G., Shah, R. C., Sheets, K. M., McNeil, J. J., Murray, A. M., & Ryan, J. (2023). Lifestyle
Enrichment in Later Life and Its Association With Dementia Risk. *JAMA Network Open*, 6(7),
e2323690-e2323690. <u>https://doi.org/10.1001/jamanetworkopen.2023.23690</u>

Yang, G., Schmiel, L., Zhou, M., Cintina, I., Spencer, D., & Hogan, P. (2019). *Economic Burden* and Future Impact of Parkinson's Disease Final Report. T. M. J. F. F. f. P. s. Research.

Yerxa, E. J. (1992). Some implications of occupational therapy's history for its epistemology, values, and relation to medicine. *Am J Occup Ther*, *46*(1), 79-83. https://doi.org/10.5014/ajot.46.1.79

Yerxa, E. J. (1998). Health and the human spirit for occupation. *Am J Occup Ther*, *52*(6), 412-418. <u>https://doi.org/10.5014/ajot.52.6.412</u>

Yıldız, E. (2020). The effects of acceptance and commitment therapy on lifestyle and behavioral changes: A systematic review of randomized controlled trials. *Perspectives in Psychiatric Care*, *56*(3), 657-690.

York, D., Borkovec, T. D., Vasey, M., & Stern, R. (1987). Effects of worry and somatic anxiety induction on thoughts, emotion and physiological activity. *Behaviour Research and Therapy*, *25*(6), 523-526. <u>https://doi.org/https://doi.org/10.1016/0005-7967(87)90060-X</u>

Younossi, Z. M., Zelber-Sagi, S., Henry, L., & Gerber, L. H. (2023). Lifestyle interventions in nonalcoholic fatty liver disease. *Nature Reviews Gastroenterology & Hepatology*, *20*(11), 708-722. <u>https://doi.org/10.1038/s41575-023-00800-4</u>

Zarrindast, M.-R., & Khakpai, F. (2015). The Modulatory Role of Dopamine in Anxiety-like Behavior. *Archives of Iranian medicine*, *18*(9), 591-603. <u>https://doi.org/0151809/AIM.009</u>

Zelikowsky, M., Hersman, S., Chawla, M. K., Barnes, C. A., & Fanselow, M. S. (2014). Neuronal ensembles in amygdala, hippocampus, and prefrontal cortex track differential components of contextual fear. *J Neurosci*, *34*(25), 8462-8466. <u>https://doi.org/10.1523/jneurosci.3624-13.2014</u>

Zhang, P., Gao, Y., Hu, Y., Luo, Y., Wang, L., Wang, K., Tian, H., & Jin, M. (2023). Altered Fractional Amplitude of Low-Frequency Fluctuation in Anxious Parkinson's Disease. *Brain Sci*, *13*(1). <u>https://doi.org/10.3390/brainsci13010087</u> Zhou, N., Smith, K. V., Stelzer, E., Maercker, A., Xi, J., & Killikelly, C. (2023). How the bereaved behave: a cross-cultural study of emotional display behaviours and rules. *Cognition and Emotion*, *37*(5), 1023-1039. <u>https://doi.org/10.1080/02699931.2023.2219046</u>

Research Outputs

Publications in peer-reviewed journals

Lovegrove CJ., Sturkenboom I., Foster M., Marsden J., and Bannigan K. (2023) What community-based interventions are used to improve the participation, quality of life, and wellbeing in adults with Parkinson's disease? A scoping review protocol. *JMIR Research Protocols*

Lovegrove CJ., Sturkenboom I., Foster M., Marsden J., and Bannigan K. (*in press*) What community-based interventions are used to improve the participation, quality of life, and wellbeing in adults with Parkinson's disease? A scoping review. *JMIR Aging*

Lovegrove CJ., Sturkenboom I., Marsden J., and Bannigan K. (2024) Concept mapping to define components for an occupation-based intervention for Parkinson's & anxiety. *Journal of Parkinson's Disease*

Lovegrove CJ., Marsden J., Bannigan K., and Sturkenboom I. (*in press*) Developing an occupation-based complex intervention for living well with anxiety and Parkinson's Disease (OBtAIN-PD): a logic modelling approach. *BMJ Open*

Lovegrove CJ., Bannigan K., Hayward C., Ingram W., Bailey M., Aspinall P., Hosking J., Sturkenboom I., and Marsden J. (*in press*) Evaluating the occupation-based complex intervention for living well with anxiety and Parkinson's disease (OBtAIN-PD): A feasibility cluster randomised controlled trial protocol. *BMJ Open*

Conference abstracts and presentations

"Developing an occupation-based complex intervention for living well with anxiety in Parkinson's (OBtAIN-PD): A feasibility cluster randomised controlled trial", Faculty of Health Postgraduate Research Showcase, UK, 2023

"Evaluating the Occupation-Based Complex Intervention for living well with anxiety and Parkinson's Disease (OBtAIN-PD): a feasibility cluster randomised controlled trial protocol", (poster) World Parkinson's Congress, Barcelona, Spain, 2023

"Developing an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD): a logic modelling approach", (poster) World Parkinson's Congress, Barcelona, Spain, 2023

"Defining the components of an occupation-based intervention for people with Parkinson's with anxiety using Group Concept Mapping", (poster) World Parkinson's Congress, Barcelona, Spain, 2023

"I am the expert in my own body: meaningful stakeholder involvement in intervention development", Neuro Convention, Birmingham, UK, 2023

"I feel like I'm going to die: why anxiety in Parkinson's is more than just worrying", Neuro Convention, Birmingham, UK, 2023

"Developing an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD)", Glasgow Caledonian University Research Showcase, Scotland, UK, 2023

"What is the lived experience of anxiety for people with Parkinson's? A phenomenological study", (poster) World Federation of Occupational Therapists International Congress, Paris, France, 2022

"Developing an occupation-based complex intervention for living well with anxiety and Parkinson's (OBtAIN-PD)", World Federation of Occupational Therapists International Congress, Paris, France, 2022

Notable Awards

Extraordinary Endeavour (nominated) and Exceptional Care (nominated)- Extraordinary People Awards, Royal Devon University Healthcare NHS Foundation Trust, 2023

Royal College of Occupational Therapists Research Champion, 2023-present

CPD Event Award (nominated), Glasgow Caledonian University Occupational Therapy Society Awards, 2022

Appendices

S20	Limit S19 to all adult/English language only	84
S19	S9 AND S12 AND S18	159
S18	S13 OR S14 OR S15 OR S16 OR S17	2,507,632
S17	TI (interven* or group* or home or homes or goal* or treat* or manag*) OR AB (interven*	2,348,880
	or group* or home or homes or goal* or treat* or manag*)	
S16	(MH "Psychosocial Intervention")	66
S15	(MH "Goals and Objectives") OR (MH "Goal-Setting") OR (MH "Goal Attainment")	20,292
S14	(MH "Group Processes+")	247,999
S13	(MH "Support Groups+") OR (MH "Group Exercise")	13,368
S12	\$10 OR \$11	205,440
S11	TI "quality of life" OR AB "quality of life"	129,603
S10	(MH "Quality of Life+") OR (MH "Psychological Well-Being")	150,321
S9	S4 AND S8	1,167
S8	S5 OR S6 OR S7	210,945
S7	(MH "Stress, Psychological+")	89,483
S6	TI anxi* or (stress* n2 relat*) OR AB anxi* (stress* n2 relat*)	89,982
S5	(MH "Anxiety+") OR (MH "Anxiety Disorders+")	89,584
S4	S1 OR S2 OR S3	45,119
S3	TI "occupational therap*" OR AB "occupational therap*"	24,287
S2	(MH "Occupational Therapists") OR (MH "British Association and College of Occupational	10,258
	Therapists")	
S1	(MH "Occupational Therapy+") OR (MH "Occupational Therapy Practice")	29,362

Appendix 1: Protocol search (Chapter 2)

Note: This search was run on CINAHL (via Ebsco) on 9 July, 2021.

Appendix 2: Data extraction tool (Chapter 2)

ltem	Item	Where located?	
Number			
		Primary paper	Other
		(page or appendix	details
		number)	
1	Brief name		
	(Name of phrase that describes the interventions)		
2	Why		
	(Rationale, theory, or goal of the elements		
	essential to the intervention)		
3	What		
	(Describe any physical of informational materials		
	used in the intervention and how they were		
	provided)		
4	Procedures		
	(Describe each procedure, activity, and/or process		
	used in the intervention)		
5	Who provided		
	(For each provider, describe their expertise,		
	background, and any specific training given)		
6	How		
	(Describe modes of delivery)		
7	Where		
	(Describe type(s) of location(s) where the		
	intervention occurred)		
8	When and how much		

	(Describe the number of times the intervention	
	was delivered and over what period of time)	
9	Tailoring	
	(If the intervention was planned to be	
	personalised, titrated, or adapted, then describe	
	what, why, when, and how)	
10	Modifications	
	(Describe any changes during the course of the	
	study/ intervention)	
11	Planned	
	(If intervention adherence or fidelity was	
	assessed, describe how and by whom; if any	
	strategies were used to maintain or improve	
	fidelity, describe them.)	
12	Actual	
	(If intervention adherence or fidelity was	
	assessed, describe the extent to which the	
	intervention was delivered as planned.)	

Appendix 3: PRISMA-SCr checklist (Chapter 2)

Preferred Reporting Items for Systematic reviews and Meta-Analyses extension for Scoping Reviews (PRISMA-ScR) Checklist

SECTION	ITEM	PRISMA-ScR CHECKLIST ITEM	REPORTED
TITLE			
Title	1	Identify the report as a scoping review.	1
ABSTRACT			
Structured summary	2	Provide a structured summary that includes (as applicable): background, objectives, eligibility criteria, sources of evidence, charting methods, results, and conclusions that relate to the review questions and objectives.	1
INTRODUCTION			
Rationale	3	Describe the rationale for the review in the context of what is already known. Explain why the review questions/objectives lend themselves to a scoping review approach.	2
Objectives	4	Provide an explicit statement of the questions and objectives being addressed with reference to their key elements (e.g., population or participants, concepts, and context) or other relevant key elements used to conceptualize the review questions and/or objectives.	4
METHODS		· · · · · ·	11
Protocol and registration	5	Indicate whether a review protocol exists; state if and where it can be accessed (e.g., a Web address); and if available, provide registration information, including the registration number.	7
Eligibility criteria	6	Specify characteristics of the sources of evidence used as eligibility criteria (e.g., years considered, language, and publication status), and provide a rationale.	5
Information sources*	7	Describe all information sources in the search (e.g., databases with dates of coverage and contact with authors to identify additional sources), as well as the date the most recent search was executed.	7
Search	8	Present the full electronic search strategy for at least 1 database, including any limits used, such that it could be repeated.	11
Selection of sources of evidence†	9	State the process for selecting sources of evidence (i.e., screening and eligibility) included in the scoping review.	7
Data charting process‡	10	Describe the methods of charting data from the included sources of evidence (e.g., calibrated forms or forms that have been tested by the team before their use, and whether data charting was done independently or in duplicate) and any processes for obtaining and confirming data from investigators.	8
Data items	11	List and define all variables for which data were sought and any assumptions and simplifications made.	8
Critical appraisal of individual sources of evidence§	12	If done, provide a rationale for conducting a critical appraisal of included sources of evidence; describe the methods used and how this information was used in any data synthesis (if appropriate).	8
Synthesis of results	13	Describe the methods of handling and summarizing the data that were charted.	8



St. Michael's Inspired Care. Inspiring Science.

SECTION	ITEM	PRISMA-ScR CHECKLIST ITEM	REPORTED ON PAGE #
RESULTS			
Selection of sources of evidence	14	Give numbers of sources of evidence screened, assessed for eligibility, and included in the review, with reasons for exclusions at each stage, ideally using a flow diagram.	N/a for protocol
Characteristics of sources of evidence	15	For each source of evidence, present characteristics for which data were charted and provide the citations.	N/a for protocol
Critical appraisal within sources of evidence	16	If done, present data on critical appraisal of included sources of evidence (see item 12).	N/a for protocol
Results of individual sources of evidence	17	For each included source of evidence, present the relevant data that were charted that relate to the review guestions and objectives.	N/a for protocol
Synthesis of results	18	Summarize and/or present the charting results as they relate to the review questions and objectives.	N/a for protocol
DISCUSSION			
Summary of evidence	19	Summarize the main results (including an overview of concepts, themes, and types of evidence available), link to the review questions and objectives, and consider the relevance to key groups.	N/a for protocol
Limitations	20	Discuss the limitations of the scoping review process.	N/a for protocol
Conclusions	21	Provide a general interpretation of the results with respect to the review questions and objectives, as well as potential implications and/or next steps.	N/a for protocol
FUNDING			
Funding	22	Describe sources of funding for the included sources of evidence, as well as sources of funding for the scoping review. Describe the role of the funders of the scoping review.	1

JBI = Joanna Briggs Institute; PRISMA-ScR = Preferred Reporting Items for Systematic reviews and Meta-Analyses extension for Scoping Reviews. * Where *sources of evidence* (see second footnote) are compiled from, such as bibliographic databases, social media

* Where sources of evidence (see second footnote) are compiled from, such as bibliographic databases, social media platforms, and Web sites.

† A more inclusive/heterogeneous term used to account for the different types of evidence or data sources (e.g., quantitative and/or qualitative research, expert opinion, and policy documents) that may be eligible in a scoping review as opposed to only studies. This is not to be confused with *information sources* (see first footnote).
‡ The frameworks by Arksey and O'Malley (6) and Levac and colleagues (7) and the JBI guidance (4, 5) refer to the process of data extraction in account review as data charting.

The franktions by rules of maney (b) and cerve and conceptes (r) and the bir guarder (4, s) rule to the process of data extraction in a scoping review as data charting. § The process of systematically examining research evidence to assess its validity, results, and relevance before using it to inform a decision. This term is used for items 12 and 19 instead of "risk of bias" (which is more applicable to systematic reviews of interventions) to include and acknowledge the various sources of evidence that may be used in a scoping review (e.g., quantitative and/or qualitative research, expert opinion, and policy document).

From: Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, et al. PRISMA Extension for Scoping Reviews (PRISMAScR): Checklist and Explanation. Ann Intern Med. 2018;169:467–473. doi: 10.7326/M18-0850.



St. Michael's Inspired Care. Inspiring Science.

Appendix 4: Scoping review searches (Chapter 2)

MeSH descriptor:	803
[Occupational Therapy]	
explode all trees	
MeSH descriptor:	9
[Occupational Therapists]	
explode all trees	
"occupational therap*":ti,ab	0
#1 or #2	807
MeSH descriptor: [Anxiety]	8588
explode all trees	
MeSH descriptor: [Anxiety	7380
Disorders] explode all trees	
MeSH descriptor: [Stress,	6544
Psychological] explode all	
trees	
(anxi* or "stress	54014
related"):ti,ab	
#6 or #7 or #8	60914
#4 and #9	39
	MeSH descriptor: [Occupational Therapy] explode all trees MeSH descriptor: [Occupational Therapists] explode all trees "occupational therap*":ti,ab #1 or #2 MeSH descriptor: [Anxiety] explode all trees MeSH descriptor: [Anxiety] Disorders] explode all trees MeSH descriptor: [Stress, Psychological] explode all trees (anxi* or "stress related"):ti,ab #6 or #7 or #8 #4 and #9

Cochrane Database 22nd November, 2021

Psychology & Behavioural Sciences Collection (Ebsco) 22nd November, 2021

S6	S1 AND S5	167
S5	S2 OR S3	512,658
S4	S1 AND S2 AND S3	40
S3	interven* or group* or home or homes or goal* or treat* or manag*	492,112
S2	"quality of life" or wellbeing or well-being or "well being"	53,757
S1	"occupational therap*" AND (anxi* or (stress* n2 relat*))	202

Embase search (via Ovid) 22nd November, 2021

1	exp occupational therapy/	23427
2	exp occupational therapist/	7413
3	"occupational therap*".ti,ab.	21712
4	1 or 2 or 3	33376
5	exp anxiety/ or exp anxiety disorder/	469346
6	exp mental stress/	170892
7	(anxi* or (stress* adj2 relat*)).ti,ab.	380003
8	5 or 6 or 7	716621
9	4 and 8	1962
10	exp "quality of life"/	555402
11	exp wellbeing/	98573
12	("quality of life" or wellbeing or well-being or "well being").ti,ab.	611658
13	10 or 11 or 12	795297
14	exp support group/	13440
15	exp group process/	9119
16	exp motivation/	148347

17	exp psychosocial intervention/	619
18	(interven* or group* or home or homes or goal* or treat* or	13844975
	manag*).ti,ab.	
19	14 or 15 or 16 or 17 or 18	13927713
20	9 and 13 and 19	509
21	limit 20 to (english language and exclude medline journals)	81
22	13 or 19	14163426
23	9 and 22	1647
24	limit 23 to (english language and exclude medline journals)	239

Medline search (via Ovid) 22nd November, 2021

1	exp Occupational Therapy/	14007
2	exp Occupational Therapists/	483
3	"occupational therap*".ti,ab.	14862
4	1 or 2 or 3	21027
5	exp Anxiety/ or exp Anxiety Disorders/	171357
6	exp Stress, Psychological/	143341
7	(anxi* or (stress* adj2 relat*)).ti,ab.	274197
8	5 or 6 or 7	452223
9	4 and 8	678
10	exp "Quality of Life"/	227009
11	(wellbeing or "quality of life").ti,ab.	335407
12	10 or 11	399343
13	exp Self-Help Groups/	10514
14	((group* adj3 exercis*) or (support* adj2 group*)).ti,ab.	31803
15	exp Group Processes/	184737
16	exp Goals/	18452
17	((goal adj2 set*) or (attain* adj3 goal*)).ti,ab.	9796
18	exp Psychosocial Intervention/	433
19	(interven* or group* or home or homes or treat* or manag*).ti,ab.	10229161
20	13 or 14 or 15 or 16 or 17 or 18 or 19	10359816

21	9 and 12 and 20	100
22	limit 21 to english language	95
23	12 or 20	10468152
24	9 and 23	512
25	limit 24 to english language	466

PubMed Search 22nd November, 2021

#17	#7 and #16 Filters: English Sort by: Most Recent	486
#16	#10 or #13 Filters: English Sort by: Most Recent	9,600,000
#15	#7 and #10 and #13 Filters: English Sort by: Most Recent	92
#14	#7 and #10 and #13 Sort by: Most Recent	97
#13	#11 or #12 Sort by: Most Recent	10,565,089
#12	interven*[Title/Abstract] OR group*[Title/Abstract] OR	10,446,377
	home[Title/Abstract] OR homes[Title/Abstract] OR	
	goal*[Title/Abstract] OR treat*[Title/Abstract] OR	
	manag*[Title/Abstract] Sort by: Most Recent	
#11	((("Self-Help Groups"[Mesh]) OR "Group Processes"[Mesh])	211,636
	OR "Goals"[Mesh]) OR "Psychosocial Intervention"[Mesh] Sort	
	by: Most Recent	
#10	#8 or #9 Sort by: Most Recent	376,593
#9	"quality of life"[Title/Abstract] Sort by: Most Recent	312,891
#8	("Quality of Life"[Mesh]) Sort by: Most Recent	227,016
#7	#3 and #6 Sort by: Most Recent	702
#6	#4 or #5 Sort by: Most Recent	442,025
#5	anxi*[Title/Abstract] OR "stress related"[Title/Abstract] Sort	260,329
	by: Most Recent	
#4	("Anxiety"[Mesh] OR "Anxiety Disorders"[Mesh]) OR "Stress,	298417
	Psychological"[Mesh] Sort by: Most Recent	
#3	#1 or #2 Sort by: Most Recent	21,546
#2	"occupational therap*"[Title/Abstract] Sort by: Most Recent	16,256
#1	("Occupational Therapy"[Mesh]) OR "Occupational	14,247
	Therapists"[Mesh] Sort by: Most Recent	

Trip Advanced search 22nd November, 2021

ALL of these words: anxiety quality in the Document ANY of these words: intervention group groups home homes goal goals manage management managing treat treating treatment in theDocument THIS exact phrase: occupational therapy in the Document Excluding Words: child children adolescents teenagers in the Document

174 results

Proquest Dissertations & Theses

25th November, 2021

345 results

"occupational therapies" OR "occupational therapist" OR "occupational therapists" OR "occupational therapy") AND (anxi* or (stress* NEAR/2 relat*)) AND ("quality of life" OR wellbeing) AND (support OR group* OR goal* OR intervention* OR home or homes OR treat* OR manag*) AND adult* NOT (child* OR adolescen* OR teen*) 345 results

PsycArticles search 25th November, 2021 (via Proquest)

S3	(MAINSUBJECT.EXACT.EXPLODE("Occupational Therapy") OR	20
	MAINSUBJECT.EXACT.EXPLODE("Occupational Therapists") OR	
	noft("occupational therap*")) AND	
	((MAINSUBJECT.EXACT.EXPLODE("Anxiety Management") OR	
	MAINSUBJECT.EXACT.EXPLODE("Anxiety Disorders") OR	
	MAINSUBJECT.EXACT.EXPLODE("Anxiety")) OR	
	MAINSUBJECT.EXACT("Psychological Stress") OR noft(anxi* OR	
	(stress* NEAR/2 relat*)))	
S2	(MAINSUBJECT.EXACT.EXPLODE("Anxiety Management") OR	15,139
	MAINSUBJECT.EXACT.EXPLODE("Anxiety Disorders") OR	
	MAINSUBJECT.EXACT.EXPLODE("Anxiety")) OR	
	MAINSUBJECT.EXACT("Psychological Stress") OR noft(anxi* or	
	(stress* NEAR/2 relat*))	
S1	MAINSUBJECT.EXACT.EXPLODE("Occupational Therapy") OR	258
	MAINSUBJECT.EXACT.EXPLODE("Occupational Therapists") OR	
	noft("occupational therap*")	

Appendix 5: Abbreviations used in the scoping review (Chapter 2)

- ADIS-IV: Anxiety Disorders Interview Schedule
- BAI: Beck Anxiety Inventory
- BDI-II: Beck's Depression Inventory
- CBT: Cognitive Behavioural Therapy
- CINAHL: Cumulated Index to Nursing and Allied Health Literature
- COPM: Canadian Occupational Performance Measure
- CSQ: Client Satisfaction Questionnaire
- DASS: Depression, Anxiety, and Stress Scales
- DSM: Diagnostic and Statistical Manual of Mental Disorders
- EQ-5D: EuroQol- 5 Dimension
- EQ-VAS: EuroQol Visual Analogue Scale
- GAD: Generalised Anxiety Disorder
- GHQ-28: General Health Questionnaire-28
- HADS: Hospital Anxiety and Depression Scale
- Haq-II: Helping Alliance Questionnaire
- ICER: Incremental Cost-Effectiveness Ratio
- JBI: Joanna Briggs Institute
- JBI EBP database: Joanna Briggs Institute Evidence-Based Practice database
- JBI SUMARI: Joanna Briggs Institute System for the Unified Management of the Assessment and Review of Information
- MADRS-S: Montgomery–Åsberg Depression Rating Scale
- MANSA: Manchester Short Assessment of Quality of Life
- MEDLINE: Medical Literature Analysis and Retrieval System Online
- NIHR: National Institute for Health Research
- OBQ: Occupational Balance Questionnaire
- OVal-pd: Occupational Value Assessment with predefined items
- PANAS: Positive and Negative Affect Schedule
- PLOS: Public Library of Science
- PMS: Pearson Mastery Scale
- PRISMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses

- PRISMA-P: Preferred Reporting Items for Systematic Reviews and Meta-Analyses Protocols
- PRISMA-ScR: Preferred Reporting Items for Systematic Review and Meta-Analyses Extension for Scoping Reviews
- PSS: Perceived Stress Scale
- QALY: Quality-adjusted life year
- RCT: Randomised Controlled Trial
- ReDO[®]-10: Redesigning Daily Occupations[®] programme
- SCL-90-r: Symptom Checklist-90-Revised
- SDO: Satisfaction with Daily Occupations
- SF-36: Short Form Health Survey Questionnaire
- SOC: Send of Coherence tool
- STAI: State-Trait Anxiety Inventory
- QCRI: Qatar Computing Research Institute
- TIDieR: Template for intervention description and replication
- TTM[®]: Tree Theme Method[®]
- UK: United Kingdom
- USD: United States Dollar
- WEIS-SR: Work Environment Impact Scale- Self Rating
- WHODAS 2.0: World Health Organization Disability Assessment Schedule 2.0
- WRS: Work Role Self-Assessment
- WSAS: Work And Social Adjustment Scale

		Importance	Feasibility Scale
		Scale	[2.5152]-
One th	ing that would help a person with Parkinson's to live	[2.2222]-	[4.5758]
	well with anxiety might be	[4.8000]	Median =
		Median = 2.4	2.2879
		n = 46	n = 33
	Full Map Go-Zone		
	R=0.44181457652678		
#	Statements	Average Rating	Average Rating
3	To have access to meditation classes.	3.4444	3.3333
5	Singing.	3.1957	3.5152
6	Going to a Parkinson's dance class.	2.8913	2.9394
8	Tai-chi.	3.0217	3.3030
11	Shiatsu.	2.5652	2.8182
12	Qi Gong.	2.6739	2.7576
14	Meditation.	3.3636	3.6250
17	To get help with planning your finances.	3.2667	3.1515
18	One to one support to improve self-confidence.	3.7333	2.9394
19	Small focus groups to improve self-confidence.	3.4565	3.2727
29	Having a one-to-one session before joining a group.	3.1087	3.1818
38	Free access to apps like headspace.	3.2391	3.5152
46	Using cognitive behaviour therapy.	3.4667	3.4375
49	To increase assertiveness.	3.5000	3.3226
53	Mentor/lifestyle coaches who can advise on	3.4565	2.8485
	matters unrelated to medications.		
57	A matched and paired Buddy to discuss	3.6222	2.9091
62	A hiefeedback gadget	2 0111	2 6070
05	A bioleeuback gauget.	2.9111	2.0970
64	hours.	3.4545	2.5152
72	Reviewing your diet with a specialist.	3.7111	3.5152
74	A blog/magazine to share experiences and good PD news for the local area.	3.4348	3.3939
82	An app that can divert your attention when you feel anxious.	3.3696	3.3333
84	A little set of laminated cue cards that can be used as a calming strategy.	2.7609	3.5313
87	Having immediate access to a calming phone app.	3.0222	3.5758
88	Having a designated person to contact for reassurance in a moment of panic.	3.7333	2.9375
93	A visual infographic with ideas and quick fixes for dealing with anxiety.	3.3043	3.3333
	The opportunity to talk openly in a group of people		
96	with Parkinsons led by an experienced group worker.	3.5435	3.3636
100	Wearing an invisible disability lanyard when out.	2.7333	3.3030
104	To write poetry.	2.2222	2.6970

Appendix 6: Group Concept Mapping statements (Chapter 3)

106	Intense exercise. 3.4565 3.27		3.2727
109	Visualisation.	3.1111	3.6250
110	Swimming.	3.0217	3.3636
112	To meditate in a PD group.	2.7333	3.0000
113	Doing yoga.	3.3182	3.5152
115	Literature you could give to an employer.	3.2955	3.5758
117	Having a "buddy" they can text or call when they're feeling particularly anxious.	3.7556	3.0909
9	Timely access to services when needed.	4.6304	3.4242
25	To continue in any type of work for as long as possible.	3.8913	3.4545
26	Having access to specialist outpatient services.	4.3778	3.4545
30	Finding the right support group for you.	4.1957	3.3030
34	Having peer support.	3.9773	3.5455
37	A designated phone line which you can call any time.	3.9130	3.0606
39	Free person centred counselling as soon as diagnosed.	4.2609	2.9688
40	Having a structured support network with a list of who is responsible for what.	3.8261	3.5455
41	To have a dedicated helper who can get to know the patient individually.	3.8261	2.8485
42	Increasing general awareness of PD in society.	4.2000	3.6061
44	To speak to other people with Parkinson's of a similar age and stage in disease.	4.0000	3.5758
50	To manage potential causes.	4.1111	3.3939
51	To identify potential causes.	3.9333	3.4242
52	To turn anxiety into a driver for action and motivation.	4.1333	3.2727
55	Focused support with sleep problems.	4.2727	3.6061
59	Access to specialist fitness trainers who understand the physical effects of PD on the body.	4.1957	3.1875
60	A tool to help isolate the source of the anxiety.	3.8261	3.2424
66	To make available counselling for more than the current allocation of 6 weeks.	4.2500	2.9394
67	To ensure that GPs receive training in movement and non movement symptoms.	4.4565	3.3333
69	Having rapid access to a neurologist.	4.1522	2.6970
75	Access to qualified PD nurses for day to day support.	4.5000	3.3438
76	To attend a specific Parkinson's exercise classes.	3.8261	3.6061
83	Having a designated contact to talk through your feelings of anxiety.	3.9348	3.0000
99	Letting other people know how you feel.	4.0000	3.5455
103	Help easing back into it all in reconnecting after lockdown.	4.0435	3.3030
116	Letting others know what makes me anxious.	3.9130	3.5758
2	To plan your day.	3.7442	4.1515

4	Learning how to meditate at home for a few	3.6000	3.9394
7	Listoning to calming music	2 6667	1 2020
/	An ann to take the person through a breathing	5.0007	4.5555
20	exercise.	3.3478	4.0303
23	Access to easy to understand books.	3.2609	3.7576
32	Having a favourite "comfort" book close by.	2.8222	3.8788
58	Strategies to help with eating in public.	3.6304	3.7273
68	Creating of a list of every day necessary activities.	3.5111	4.1212
86	Having a practised mantra to repeat when anxiety sets in.	3.2609	3.8485
89	Play a game.	2.9111	3.8182
90	A reading list of recommended books on living with anxiety.	3.1778	4.0909
94	Recorded relaxation messages that could be easily listened to.	3.2222	3.7879
95	A card in your purse/ wallet with a preprepared mantra that can be read in moments of anxiety.	2.9111	4.0303
98	To have a daily five to ten minute nap in the afternoon.	3.1957	3.8788
118	Using mindfulness.	3.6818	4.0606
1	To simplify your life.	4.0000	3.8485
10	Support to help people develop their own coping strategies.	4.2609	3.6667
13	Exploring different ways to relax.	4.2609	4.2424
15	Having knowledge about the condition.	4.5000	4.5455
16	Knowing where to go to get specific information and support.	4.6000	4.2121
21	To have a regular exercise routine you enjoy.	4.5000	4.2121
22	Breathing exercises to help you to relax.	4.2000	4.5758
24	Access to easy to understand current research articles.	3.8478	3.6667
27	Taking time out of the day to relax.	4.3043	4.0606
28	Get a medication review to make sure you are on the right ones.	4.5217	4.1212
31	To think of ways that you can continue to interact with others e.g.online, face to face, by, phone, etc.	4.0870	4.0909
33	Doing things that make you laugh.	4.4348	4.0000
35	Not putting off taking some rescue medication.	3.9783	3.8182
36	Engaging with a creative activity.	4.0222	3.8485
43	Understanding the personal drivers that create anxiety.	4.2609	3.7879
	Use the 5		
45	A's; Acknowledge, Accept, Allow, Adapt, & Adjust.	3.8478	3.9394
47	To engage in meaningful activities.	4.4222	4.0606
48	To understand how cognition is affected by anxiety.	4.1087	3.8788
54	Planning for the future.	4.1957	3.9394
56	To develop coping strategies for social situations.	4.3111	3.9394

61	Honesty from PD clinicians about the likely progress of the condition in order to prepare mentally.	4.2609	3.7813
62	Training in simple breathing techniques which can be used at any time.	4.1304	4.1515
65	More support for newly diagnosed people.	4.5652	3.7879
70	A booklet that guides users through changes they can make to live well with Parkinsons.	4.0444	4.2424
71	To live in the moment.	4.4667	3.9697
73	To practice clean sleeping habits.	4.1136	3.9063
77	Keeping active.	4.8000	4.3030
78	Maintaining social interests e.g. sports.	4.5111	3.8788
79	Access to up to date research.	4.0000	4.0909
80	Having a balanced diet.	4.2444	4.3030
81	Better information and support regarding the later stages of Parkinson's Disease.	4.0667	4.1212
85	To devise a 5-minute exercise program that can be done anywhere.	3.8444	4.1515
91	To take a walk in the open air if possible.	4.4130	4.5152
92	Encouragement to share anxiety with partners and close friends.	4.1304	3.8182
97	To get involved in projects to advance the understanding, treatment, and elimination of Parkinsons.	4.1304	3.9697
101	Planning ahead to give yourself plenty of time.	4.3696	4.3030
102	Understanding how PD anxiety is different from other anxiety.	4.3333	3.8438
105	To try new things.	3.9783	3.6970
107	Finding out what triggers your anxiety.	4.3913	3.8750
108	Learning to live in the moment.	4.3261	4.1212
111	Accepting my condition rather than being frustrated by what I can't do.	4.4565	3.9688
114	Trying not to do too much especially with time restraints	4.2222	3.9394
119	Enagingin hobbies.	4.3913	3.9091

Appendix 7: Logic modelling information pack (Chapter 4)

Ethics approval: 3115 Version 1.2 20/04/22 IRAS Project ID: 302129



Information pack

Title of Project: Developing an occupation-based complex intervention for living well with anxiety and Parkinson's Disease (OBtAIN-PD): a logic modelling approach

Name of Researcher: Mr Chris Lovegrove

Dear participant,

Thank you for agreeing to participate in our research study to identify the things that people with Parkinson's, their care partners, and occupational therapists feel should be included in a new intervention to help people with Parkinson's live well with anxiety.

You will soon be receiving the details of the session that you will be taking part in. This will be run by the lead researcher, Chris Lovegrove. Prior to your session, please read this information pack. It contains summaries of the research team's recent research findings. Reading this information will help to stimulate your ideas for the main session.

When you are reading this information, please keep the following prompt in mind to help direct your thinking:

I think that an intervention to help people with Parkinson's with anxiety should include...

If you have any questions about the information in this pack or would like any further details, please do not hesitate to contact the lead researcher at <u>Christopher.lovegrove@postgrad.plymouth.ac.uk</u>

Yours sincerely,

Chris Lovegrove NIHR Clinical Doctoral Research Fellow Occupational Therapist



Study 1: What is the lived experience of anxiety for people with Parkinson's? A phenomenological study

The full open-access article can be found here.

Purpose

Anxiety is a common non-motor symptom of Parkinson's and there is no specific pharmacological intervention for people with Parkinson's who experience anxiety. Yet there is little specific research documenting how individuals with this condition experience anxiety. It is important to explore the experiences of people with Parkinson's to identify potential issues in developing future non-pharmacological interventions. This study explored the lived experience of anxiety for people with Parkinson's.

Materials and Methods

This study explored the lived experience of anxiety for people with Parkinson's by describing the findings from an interview with individuals; a technique called descriptive phenomenology. Six participants were recruited through the charity Parkinson's UK, using a maximum variation sampling strategy. Face to face interviews were conducted. Data analysis employed thematic analysis.

Results

Three key themes encapsulated the data: Finding ways to cope to "Try not to let it rule your life", highlights that finding ways to cope with anxiety is important but needs to be tailored to the individual. Amplifies symptoms "It's emotionally draining it it's also physically draining", highlights how anxiety makes people's existing Parkinson's symptoms worse, including their cognitive symptoms. "Anxiety is a funny thing" experienced in myriad ways, discusses how anxiety is usually experienced as a negative, ubiquitous presence, though some people found a positive aspect to it. A model of the experience of PWP experience of anxiety was proposed and can be seen in the article.

Conclusions

Anxiety is a complex experience constructed of interlinked parts affecting people with Parkinson's in lots of different ways. Researchers and healthcare professionals should take these findings into account when designing future studies and interventions. Future interventions should be based on the experiences of those living with anxiety and not healthcare professional or researcher assumptions.



Study 2: Defining the components of an occupation-based intervention for people with Parkinson's with anxiety using Group Concept Mapping

This research study is now closed. An article is currently being written by the research team and will be submitted to a scientific journal.

Purpose

Anxiety is a common symptom of Parkinson's disease. There is currently no gold standard treatment for anxiety in Parkinson's. People with Parkinson's (PWPs) have expressed that new interventions should be focused on 'doing' as opposed to talking therapies. This study engaged with key stakeholder groups to identify the key components required for the future co-production of an anxiety-focused intervention in Parkinson's. The aim of this study was to define the components of an occupation-based intervention for PWPs with anxiety by identifying, organising, and prioritising ideas on how to live well with anxiety in Parkinson's, according to PWPs, their carers, and occupational therapists.

Methods and Materials

Participants were invited to take part in an online group concept mapping process with five stages. These are called brainstorming, idea synthesis, sorting activity, rating activity, and analysis. Based on participants' responses a cluster map, pattern match, and 'go-zone' charts were created through a process called multivariate statistical analysis.

Results

Eighty-three people participated, with 64 taking part in more than one activity. Participants consisted of 72 people with Parkinson's, six carers, and five occupational therapists. Following the 'idea synthesis' activity, 119 statements were included in the final map with eight clusters. There was significant agreement between the importance and feasibility rating activities. 'Go-zone' charts (charts that show what things should be focused on in the future) highlight the statements considered priorities for intervention development. The final clusters, with example statements, were:

- · Exercise (Keeping active, intense exercises, to have a regular exercise routine you enjoy)
- Lifestyle Changes (To engage in meaningful activities, planning for the future, engaging in hobbies)
- · Self help (To plan your day, exploring different ways to relax, using mindfulness)



- Coping (To develop coping situations for social situations, finding out what triggers your anxiety, planning ahead to give yourself plenty of time)
- Access to information (Access to up to date research, literature you could give to an employer, having knowledge about the condition)
- Professional help (Knowing where to get specific information and support, to understand how cognition is affected by anxiety, understanding of PD anxiety is different from other anxiety)
- Peers and groups (Support to help people develop their own coping strategies, having a one-to-one session before joining a group, having a structured support network with a list of who is responsible for what)
- Support from others (Strategies to help with eating in public, one to one support to improve self confidence, help easing back into it all in reconnecting after lockdown)

Conclusion

This study used a systematic approach to identify important concepts for possible inclusion in a new intervention to help people with Parkinson's to live well with anxiety.



Study 3: Characteristics of occupational therapy interventions for community-dwelling adults with anxiety: a scoping review

This review has now finished. An article is currently being written by the research team and will be submitted to a scientific journal.

Purpose

Anxiety is linked to decreased life quality and wellbeing. Developing an overview of studies of occupational therapy interventions for people with anxiety is a necessary first step to understanding the existing knowledge and increasing the impact of future interventions: This scoping review identified: i) what occupational therapy interventions exist for adults with anxiety; ii) the intervention characteristics, including outcomes used and impact observed.

Methods and Materials

The review considered all studies and publications of occupational therapy that include communitydwelling adults with a diagnosis of anxiety who are aged 18 years and older and diagnosed with an anxiety disorder. Multiple databases were searched. Potentially relevant studies were retrieved in full and assessed against the inclusion criteria. Study selection was completed by two independent reviewers. Data were extracted using a data extraction tool.

Results

The literature searches found 2,058 articles. After the assessment process was completed, 19 articles were completed in the final review. Group sessions were a commonly used type of intervention. However, individual sessions were also delivered and valued by the people who used them. Lifestyle management and cognitive behavioural approaches were the most used intervention types. Some approaches used exercises like yoga to help people with anxiety. One study used a creative arts approach to support people with anxiety. Few of the studies used a scientifically rigorous approach. Of the studies that delivered or tested a treatment, these were conducted in a GP surgery or a hospital setting. None were conducted in the person's own home or in the community.

Conclusions

This review highlights the need for further well-conducted research studies into occupational therapy interventions for community-dwelling adults with anxiety. It also provides a snapshot of the characteristics of current occupational therapy interventions for this group of people.

Appendix 8: OBtAIN-PD intervention manual (Chapter 4)



Introduction

This manual will provide guidance for the provision of the **Occupation-Based Complex Intervention for living well with anxiety and Parkinson's Disease (OBtAIN-PD)** for the OBtAIN-PD feasibility cluster randomized controlled trial (RCT).

If your team has not been allocated to provide the OBtAIN-PD and you are providing usual care occupational therapy only, please stop reading this manual and inform the research team by emailing Jonathan.marsden@plymouth.ac.uk

This manual provides guidance and structure on how to deliver the OBtAIN-PD by an Occupational Therapist for a person with Parkinson's with anxiety but is not prescriptive as to the content of individual sessions.

OB004 OBtAIN-PD manual v1.1 24/03/2023



Development of the OBtAIN-PD

The OBtAIN-PD project has its roots in clinical practice. The chief investigator (CI, Chris Lovegrove) was working in an inpatient ward for neurology with people with Parkinson's whose treatment was medically optimized, yet their participation in meaningful occupation remained substantially impeded. On assessment, these people were experiencing uncontrolled anxiety that was exacerbating their Parkinson's symptoms. A literature review revealed that there is currently no 'gold-standard' treatment for anxiety in Parkinson's and that current treatments seem to be less effective for people with Parkinson's. The CI became interested in exploring the development of a new intervention to help people with Parkinson's to live well with anxiety.

Formally spanning 2015 to 2023, the OBtAIN-PD project has consisted of four studies (excluding this feasibility trial) that have involved over 120 people with Parkinson's, care partners, and occupational therapists. These studies are:

- A foundational qualitative research study exploring the lived experience of people with Parkinson's (Lovegrove & Bannigan, 2021)
- A group concept mapping study to identify the key components of what would become the OBtAIN-PD (article in peer review)
- A scoping review to identify what occupational therapy interventions for adults with anxiety
 already exist, and what the characteristics of those interventions are (article in peer review)
- A study using an approach called 'logic modelling' to use the information from the above studies to co-produce the initial version of the OBtAIN-PD (article to be submitted for peer review in February/ March 2023)

Alongside these studies, we have conducted continuous Patient & Public Involvement (PPI) in the form of several consultations to involve people with Parkinson's, care partners, and occupational therapists in the development of our research studies. Using PPI can improve the quality and real-world relevance of research studies. We have conducted a range of PPI consultations to support this project, including a published article (Lovegrove *et al., 2017*) and an interactive session at the 2022 Royal College of Occupational Therapists' Annual Conference. This session was attended by over 100 occupational therapists.

The aim of this research project is to conduct a cluster randomized feasibility trial of the OBtAIN-PD in a real-world practice setting. We aim to use the data from this study to inform the design and planning of a future definitive trial, and to further develop and refine the OBtAIN-PD.

OB004 OBtAIN-PD manual v1.1 24/03/2023



Overview of OBtAIN-PD sessions

The number of treatment sessions provided will depend on the person with Parkinson's goals.

The OBtAIN-PD trial protocol suggests eight one-hour sessions over a 10-week period based on published evidence. This is number of sessions **is not** prescriptive and is suggested for trial planning purposes. As the occupational therapist delivering the OBtAIN-PD you can decide how many sessions a patient receives (and over what time frame) to meet their goal based on your clinical decision-making,. For example, you may deliver all eight sessions (or more if you decide it is required) to meet the persons goal, or you may deliver only three sessions. There is no strict rule of how many sessions should be provided.

The below table provides an overview of activities performed the trial for the participants receiving the OBtAIN- PD based on the **suggested** eight one-hour sessions over 10 weeks as outlined in the trial protocol, and who will deliver the different activities. The different activities will be described in more detail in later sections.

Activities in the table are colour-coded according to who performs them. Green-shaded activities are performed by the community rehabilitation teams. Amber-shaded activities are performed by the CI (or delegated members of the research team).

OB004 OBtAIN-PD manual v1.1 24/03/2023



Overview of activities (OBtAIN-PD)

Session/ Activity	Time	Activity (confirmed & suggested)	Delivered by?
Referral sent to the community rehabilitation team (CRT)			Normal referral routes- consultants, Parkinson's specialist nurses, clinics, other AHPs
Referral triaged as per local protocol	As per local protocol	 Referral triaged and prioritized First appointment booked Study details discussed 	CRT clinician
Screening log completed, information pack sent to participant if they are interested	5 minutes maximum per potential participant	 Complete screening log If participant interested, send email of postal information pack (provided) 	CRT clinician
Details sent to chief investigator	1-2 minutes	Participant details emailed to chief investigator Or CRT clinician sends details to CI at Christopher.lovegrove@nhs.net	Participant or CRT clinician
Screening and informed consent process completed	30 minutes	 Questions answered Informed consent documentation completed Participant details entered to trial database 	Chief Investigator (Chris Lovegrove)
Baseline outcome measures taken	One week prior to first appointment Clinical outcomes measures (60 minutes) Patient-reported outcome measures (15 minutes)	 COPM eACS3 GAD-7 PDQ-39 EQ-5D-5L Barthel Index Baseline clinical outcomes sent to the treating clinician 	Chief Investigator (Chris Lovegrove) Participant- Patient- reported outcome measures
Session 1	60 minutes (or as required by the patient), complete the	 Initial assessment and goal discussion 	CRT occupational therapist

OB004 OBtAIN-PD manual v1.1 24/03/2023

Btain-PD

	therapist contact form (max. 5 minutes)	 complete the OBtAIN-PD goal planning sheet Provide 'Parkinson's & Anxiety' information sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	
Session 2	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Provide necessary information sheet (exercise, sleep, or diet & nutrition) if indicated Amend goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
Session 3	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Provide necessary information sheet (exercise, sleep, or diet & nutrition) if indicated Amend the goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
Session 4	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Provide necessary information sheet (exercise, sleep, or diet & nutrition) if indicated Amend the goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
Session 5	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Amend the goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist

OB004 OBtAIN-PD manual v1.1 24/03/2023

Btain-PD

Session 6	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Amend the goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
Session 7	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Treatment Amend the goal sheet Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
Session 8	60 minutes (or as required by the patient), complete the therapist contact form (max. 5 minutes)	 Review goal planning Conclusion Complete Therapist contact sheet (see below) Report adverse events (if needed, see below) 	CRT occupational therapist
12-week (from baseline assessments) follow-up	Clinical outcomes measures (60 minutes) Patient-reported outcome measures (15 minutes)	As per baseline	Chief Investigator (Chris Lovegrove) Participant- Patient- reported outcome measures
24-week (from baseline assessments) follow-up	Clinical outcomes measures (60 minutes) Patient-reported outcome measures (15 minutes)	As per baseline	Chief Investigator (Chris Lovegrove) Participant- Patient- reported outcome measures
Qualitative interview	60- 90 minutes	Semi-structured interview at a date and location convenient to the participant/ clinician	Chief Investigator (Chris Lovegrove)

A summary of both the clinical and patient-reported outcome measures is provided in appendix 1.

OB004 OBtAIN-PD manual v1.1 24/03/2023



1. Initial assessment- delivered by CRT occupational therapist

Objectives

- Identify the person's occupational performance history, and how their current performance may differ from the past.
- Identify what values, roles, and interests give meaning to their activities.
- Identifying/clarifying the patient's perceived problems in activities and participation and priorities.
- Identifying/clarifying occupational therapy needs of care partner (if necessary).
- Examining and analysing aspects of the person, the activity, and the environment that hinder and promote meaningful participation. Identifying facilitating factors is important as they are used during treatment.
- Identify what factors influence the person's anxiety.
- Identify how anxiety affects the person's performance.
- Observe current occupational performance.
- Identify what the person wants to change and what they want to remain the same.
- Analysing the possibilities of changing the person, the activity, and the environment to positively influence participation.

Timing

The delivery of the initial assessment is flexible, depending on what suits the patient or carer best, and the occupational therapist's own style i.e., face-to-face, online, or telephone. Please follow the structure presented below when delivering the initial assessment. Some patients and care partners will want to rapidly do something and talk more about their issues along the way. Others will prefer to have an in-depth discussion first. As per usual practice, the initial interview will provide a complete picture of the referred issue to aid the occupational therapist in developing a treatment plan.

The chief investigator, Chris Lovegrove, will provide the outcomes of the baseline clinical outcome measures (COPM, eACS3) to the occupational therapist prior to the initial assessment appointment. During the initial assessment, the clinician will perform any additional clinical assessments that they deem necessary and check with the person that these are still the most important issues to address in sessions. This is important as goal setting is a core principle of the OBtAIN-PD approach.

As part of the OBtAIN-PD intervention, please complete the "OBtAIN-PD goal planning sheet". Ideally, this should be completed by the patient (if they are able) and retained by them. The purpose of this is to promote 'ownership' of the goals by the patient and to use this as a tool for future therapy sessions. Goals should be related to managing the impact of anxiety and engaging in meaningful activities. Paper copies of the goal planning sheet will be provided by the research team, as well as a link to an online version should more copies be required quickly. The goal planning sheet should be reviewed at the beginning and end of each session and adjusted accordingly, providing a new sheet if needed.

OB004 OBtAIN-PD manual v1.1 24/03/2023



An example of a completed goal planning sheet with dummy data is included below.

.1 30.09.2022 RAS ID: 318175	
OBtAIN-P	D goal planning
Лу long-term goal is:	
To be able to be able to Walk my o panic attack	dog along the sea front without having a
will accomplish this goal by the fol	lowing date:
31st March 2023	
his goal will be accomplished by ta	king the following steps:
(What short-term goals will you use to help attend an exercise class at least two times o	you achieve your long-term goal? Example: "I will a week")
 I will cut down the amount cups each day. 	of coffee that I drink from four to two
 I will stop drinking caffeir 	re after midday.
• I will walk my dog to the e	nd of my road with my wife.
he times that I feel I am at risk of r	not completing my goal and steps are:
(Situations that you know causes you have a "I have difficulty going to bed at a regular t	increased challenges with staying consistent. Example: ime because I feel pressured to go out")
 I tend to get a coffee whene because that is my 'go to' a 	ver I go out with my family or friends Irink.
 I usually drink more coffer 	e when I don't sleep well.
• When there are lots of other	people out walking their dogs.
 When I don't know the rout 	te that we are going to walk.

OB004 OBtAIN-PD manual v1.1 24/03/2023



Figure 1. Example of a completed goal planning sheet. This example is based on a real-life example given by a person with Parkinson's.

During the initial assessment and discussion on goals, your communications should focus on 'guiding' people in a compassionate manner to empower them to identify the steps to achieve their goals themselves. The purpose of this is to evoke motivation within the person to make positive change. To achieve this, when working with a person with Parkinson's the following communications skills should be used:

- Open-ended questions: using open-ended questions allows to find out more about the
 person's perspectives and ideas about how to change. These questions are also crucial in
 building and strengthening a collaborative therapeutic rapport. Example: "How would you like
 things to be different?"
- Affirming: this can be achieved through recognising and commenting on the person's strengths and abilities. Research in approaches such as motivational interviewing has shown that it helps rapport building by acknowledging the person's emotions and feelings. Example: "This sounds like this has been really challenging. No wonder you have felt overwhelmed."
- **Reflective listening:** Sometimes known as 'active listening', the purpose of reflective listening is to clarify what the person is saying for both the purposes of understanding correctly and to reflect to the person that they are being listened to. This can be useful in helping the person to consider change. *Example: "So I can check that I have understood what you have told me, you said..."*
- **Summarising:** this is closely linked to reflective listening. During summarising, the clinician reflects what the person has told them using their own words. This has the effect of the person 'hearing themselves' talk about change, which can be an effective motivator.

During the initial assessment, it is important to listen for 'change talk' to help guide the goal and treatment planning. Change talk are phrases that signal that the person is more willing, able, or ready to make a change. The role of the clinician is to support the client to generate their change talk in a collaborative way, and to avoid imposing it. Change talk can occur in several forms and can be exemplified by statements that indicate a desire for, ability to, reasons for, and a need to change. Some examples are included below:

- Desire statements indicate a desire to make a change e.g., "Getting more exercise would make me feel so much better about myself."
- Ability statements reflect the person's belief in their ability to make change e.g., "With some help, I think I might be able to cut back on how much I eat."
- Reason statements reflect the person's reasons for wanting to change e.g., "I have to control my anxiety so I can spend time with my grandchildren."

OB004 OBtAIN-PD manual v1.1 24/03/2023


• Need statements indicate a need for change that might be more emotional than 'reason' statements (which tend to be more objective). E.g., "My anxiety has to change, or my marriage will fall apart."

During the session, reflecting this language back to the person when summarising can help to strengthen the person's commitment and motivation to change.

Some further examples of these communication skills are provided below.

	Examples
Open-ended questions	 "What makes you think it might be time for a change?" "What brought you here today?" "What happens when you [insert risky/problem/unhealthy behaviour]?" "What was that like for you?" "What's different about (quitting smoking, improving your exercise, diet, etc.) this time?"
Affirming	 "That's a good suggestion." "I appreciate that you are willing to meet with me today." "It sounds like it has been very hard to cut down on coffee." "It's clear that you're really aware of how important it is to get enough sleep."
Reflective listening	 "It sounds like" "It seems as if" "What I hear you saying" "I get the sense that" "I get the sense that this has been difficult" "What I hear you saying is that your [insert risky/problem/unhealthy behaviour] is affecting you by"
Summarising	 "It sounds like you are concerned that you are concerned about your anxiety because it is stopping you from spending quality time with your grandchildren." "You have been talking about improving your diet and losing weight as you feel it will help your anxiety. It seems you have started to recognise the less good things about being overweight. It's easy to understand why you want to work on this."

OB004 OBtAIN-PD manual v1.1 24/03/2023



When completing the goal planning sheet, these communication skills should be used to support the person to complete the *"this goal will be accomplished by taking the following steps"* box. Examples of language to use to help the person decide on these steps include:

- "What changes were you thinking about making?"
- "How would you like things to turn out?"
- "After reviewing all of this, what's the next step for you?"
- "It sounds like things can't stay the same. What do you think you might do to change this?"
- "What are you going to do next?"
- "What do you think will help at this point?"

This is not an exhaustive list of phrases. You may have phrases that you routinely use in clinical practice. It is OK to use these as long as the conversation remains focused on guiding and eliciting from the person what they plan to do, rather than giving instruction or advising.

"What if the person wants a quick fix?"

It is common for people using health and care services to seek answers or fixes. OBtAIN-PD is intended to empower people with Parkinson's to find their own solutions to living well with anxiety. In keeping with this, a simple phrase to help remind the person of their autonomy can be useful. An example would be; "You are the expert on yourself, so I'm not sure I am the best person to judge what will work for you. But I can give you an idea of what research evidence shows us and what other people have done in your situation."

During the initial session, you should complete the standard paperwork that is required for your service. Following the session, you must complete a therapist contact sheet as described in section 3.

As part of the session, provide the person with the information sheet titled "Parkinson's & Anxiety V1". This sheet has been designed to provide introductory information about anxiety and Parkinson's, something that people with Parkinson's feel is important to receive at the beginning of the intervention to help them understand what is going on. Please talk through this sheet with the person, using some of the following questions to guide the conversation:

- Would you mind me telling you a little more about anxiety in Parkinson's?
- (After reading the sheet) Are there any things raised in the information sheet that remind you of your anxiety?
- What things contribute to your anxiety?
- How does your anxiety change throughout the day?
- How does the environment affect your anxiety?
- How does your anxiety affect the things that you do?

You should document the session in the patient's medical notes as per usual practice.

OB004 OBtAIN-PD manual v1.1 24/03/2023



2. Treatment sessions

Purpose

To assist and support the patient and care partner in optimising meaningful participation by addressing barriers to engaging in meaningful occupation (as identified by the person with Parkinson's).

Treatment provision and timing

Treatment may take the form of education, advising and coaching, instructing, practising skills, and giving feedback. The potential content of treatment sessions is expanded on below.

The number of treatment sessions provided by the occupational therapist depends on the reason for referral, the patient's goals and the types of interventions delivered. This is a clinical decision made by the treating occupational therapist and is not prescribed.

The OBtAIN-PD approach is designed to be flexible in its application. Occupational therapy intervention should be tailored to each person; this is what creates dynamic and meaningful outcomes. Each person's treatment goal will be different, but the OBtAIN-PD provides a framework towards delivering this treatment, developed in co-production with people with Parkinson's, carer partners, and occupational therapists.

Information sheets

During the intervention, you may need to provide the person with extra information to help them adapt their roles, routines, and lifestyles to help them live well with anxiety. Three information sheets are available for you to give out; *"Exercise"*, *"Sleep"*, and *"Diet & nutrition"*. These information sheets are based on currently published NHS guidelines. A summary of each information sheet is provided below.

OB004 OBtAIN-PD manual v1.1 24/03/2023

Btain-PD

Information sheet	Summary
Exercise	 Explains what exercise and physical activity is. Provides evidence-based recommendations for the amount per week (2.5 hours, 150 minutes). Gives examples of non-exercise activity
Sleep	 Explains what sleep is and why it is important. Provides NHS recommendations of how much sleep is needed per night (seven to nine hours). Provides suggestions on how to improve sleep for the person to refer to.
Diet & nutrition	 Provides general rules for following a healthy diet based on NHS recommendations. Provides information on medication, adapted cutlery, and swallowing problems to help inform the person's thinking.

The subjects and contents of these information sheets have been developed based on the thoughts and opinions of people with Parkinson's, care partners, and occupational therapists on what they felt should be included in the OBtAIN-PD. The information sheets **should not** be provided unless the topic has been identified by the person with Parkinson's, and **should not** be provided as a treatment. These information sheets are intended to aid the delivery of OBtAIN-PD by providing information and supporting discussion between the person and the occupational therapist.

OB004 OBtAIN-PD manual v1.1 24/03/2023



Treatment

As part of the initial discussion, you will have supported the person to identify how they will accomplish their goal. Each treatment session should start by reviewing these steps and how the person has managed to work towards them.

Treatment should focus on directly addressing the person's own individual goal. For example, the goal used on the example goal planning sheet:

"To be able to walk my dog along the sea front without having a panic attack."

As opposed to 'talking' therapies such as cognitive behavioural therapy (CBT), OBtAIN-PD specifically focuses on engaging the person in their chosen meaningful occupation to remove barriers to participation. People with Parkinson's have expressed that they want their participation barriers addressed in pragmatic and direct ways.

Thus, this goal could be addressed in a variety of ways:

- Identifying the triggers that result in the person feeling anxiety and looking for ways to reduce them.
- Identifying parts of the activity that the person is already doing and encouraging them to complete that daily to increase their sense of mastery and confidence.
- Supporting the person to plan ahead to reduce potential stressors e.g., timing the walk when there are fewer people around.
- Graded exposure- supporting the person to engage in the activity at the location (in this case, a walk at the sea front), and gradually increasing this each session in collaboration with the individual.
- Situational use of relaxation strategies, such as breathing techniques, during the activity to
 prevent anxiety from escalating.
- Cueing strategies, such as auditory or visual cueing, to help with the management of freezing episodes or the initiation of movement.
- Cognitive support strategies, such as setting up reminder or prompts on mobile device, or planning novel (new and interesting) activities to promote cognitive initiation.

This is not an exhaustive list of the treatments provided by occupational therapists delivering the **OBtAIN-PD**. The list provides suggestions of the sorts of treatment activities that an occupational therapist may provide when working with a person with Parkinson's with anxiety.

You may use different treatments from the list above based on the individual's needs. Please provide usual occupational therapy intervention as needed e.g., providing compensatory equipment to reduce falls risk in ADLs. Usual occupational therapy intervention should form a maximum of 50% of

OB004 OBtAIN-PD manual v1.1 24/03/2023

🗘 Btain-Pd

the session. For a description of usual care occupational therapy, please refer to the usual care manual (v1.1).

Session close and planning

The final part of each treatment session should focus on reviewing what has occurred during the treatment session, reviewing this in relation to the participant's goal, and planning the next session (if required). This process should include:

- Reviewing the person's experience of the session.
- Using the previously identified communication strategies to help the person identify what went well (and what didn't go so well).
- Plan the next session in collaboration with the person. For example, if the person walked their dog for 5 minutes, they may aim for 10 minutes in the next session.
- Updating the goal planning sheet if required.
- Confirm the date and time for the next treatment session.

Final session

If the person has completed their goal, and they do not require any further intervention, please review the goal planning sheet.

As part of the closing discussion, discuss with the person the steps that they have taken to accomplish their goal, and how they have addressed the potential barriers that they faced towards addressing their goal. Goal formation is a critical component in fostering behaviour change; closing the intervention by focusing on goal accomplishment can be a supportive factor in maintaining behaviour change.

To help maintain behaviour change, please discuss the following with the person:

- What are some habits that prevent you from doing the things that you want to?
- How do you react when you are not able to do the things that you want to?
- What helps you to develop new habits and routines?
- What are some barriers to making new habits?
- What are some unexpected things that may interfere with forming new habits and stop you from doing what you want to do? What steps can you take to prepare for such things?

Remember to tick the 'discharge' box on the therapist contact sheet (see below). This will indicate to the research team that this is the final treatment session.

Please discharge the person as per your team's process.

3. Therapist contact sheet and Adverse Event Monitoring

Monitoring how interventions are delivered is an important part of a clinical trial. This information is used to establish the average length of a treatment session, how many sessions are required on average, the most common treatments used, and any extra resources that might have been needed.

OB004 OBtAIN-PD manual v1.1 24/03/2023



To help collect this information, the treating clinician will be required to complete a 'Treating Therapist Contact Sheet' (*appendix 2*).

This form has been designed to be completed remotely on a mobile work device in a matter of minutes. Once completed, the treating occupational therapist will be able to submit it directly to the trial database from the device. It is important that the contact sheet is completed promptly at the end of the session so that the information is as accurate as possible.

Paper copies will be provided so that the treating occupational therapist can log the session in the event of a device failure, or a mobile or internet provider fault. The information on the paper forms will need to be transferred to the online form as soon as is feasibly possible and submitted. Once submitted, the paper copy can be disposed of in a secure confidential waste bin.

Reportable adverse events (such as injurious falls, new panic attacks, or increases in freezing episodes) in a therapy session can be reported directly to the Peninsula Clinical Trials Unit (PenCTU) at the number provided on the form. This information will then be passed on to a co-investigator (Professor Jon Marsden) who will investigate in line with trial processes.

To further support the monitoring of adverse events, the participants will be required to complete a falls, adverse event, and social cost log every four weeks. This will be delivered electronically and completed online. A copy of this log can be seen in *appendix 3*.

Appendix 1: Outcome measures

Outcome measure	Use	Delivered by?
The Canadian	The standard Canadian Occupational Performance	Chief investigator
Occupational	Measure (COPM) is a valid measure of a person's	(Chris Lovegrove) at
Performance	self-perception of performance in everyday living.	baseline, 12-week,
Measure (COPM)	The COPM is a client-centred outcome measure for	

OB004 OBtAIN-PD manual v1.1 24/03/2023



	individuals to identify and prioritise everyday issues	and 24-week follow-
	that restrict their participation in everyday living. The	up.
	COPM is the proposed primary outcome for the	
	future definitive trial	
Activity Card Sort	An assessment of a person's perceived level of	Chief investigator
(eACS3)	participation with demonstrated	(Chris Lovegrove) at
198 ² - 1982	applications in clinical practice and research.	baseline, 12-week,
	100 (00)	and 24-week follow-
		up.
Generalised	A valid and reliable 7-item instrument used to assess	Participant at
Anxiety Disorder	the severity of generalised anxiety disorder. The	baseline, 12-week,
assessment scale	GAD-7 is commonly used in the 'Improving Access to	and 24-week follow-
(GAD-7)	Psychological Therapies' pathway and will provide a	up.
	viable way of comparing the OBtAIN-PD against	
	other interventions in the future. A score of ≥10	
	identifies a level of anxiety that has a direct impact	
	on the quality of life and is a recommended cut-off	
	that identifies a need for further clinical evaluation.	
European Quality	Evaluation of health-related quality of life. This	Participant at
of Life-5	measure has been used in trials with people with	baseline, 12-week,
dimensions (EQ-	Parkinson's and has been psychometrically validated	and 24-week follow-
5D-5L)	for this population. The EQ-5D-5L can be used to	up.
	cost-utility analyses.	
The Parkinson's	This condition-specific questionnaire, a patient	Participant at
Disease	reported measure of health status and quality of life,	baseline, 12-week,
Questionnaire	has been psychometrically evaluated in people with	and 24-week follow-
(PDQ-39)	Parkinson's.	up.
Barthel Index	A self-reported scale used to measure a person's	Participant at
	performance in activities of daily living and a widely	baseline, 12-week,
	used measure in research.	and 24-week follow-
		up.
Fall, adverse event,	A commonly used tool in research for measuring falls	Participants at
and social cost log	frequency and freezing of gait in people with	baseline and then
	Parkinson's, and will be used to measure adverse	every four weeks
	event (AE) rates in this study including AEs that are	after this.
	not falls such as panic attacks. Data regarding	
	societal costs will be collected, such as unplanned	
	loss of workdays for the participants and carers, and	
	personal expenditure on support services.	

OB004 OBtAIN-PD manual v1.1 24/03/2023



Appendix 2 Treating Therapist Contact Sheet

Treating Therapists Contact Sheet V1 2.09.2022 IRAS ID: 318175 Sponsor: University of Plymouth Chief Investigator: Chris Lovegrove	(insert trial logo here)
Participant study number:	
Participant initials:	
Date:	
Time: H H : M M AM D PM D	
Session number:	
Type of contact: Home visit Online consultation Telepho	one consultation \Box
First visit 🗆 🛛 Follow-up 🗆 🛛 Dischar	ge 🗆
Location of contact: (Please enter location here)	
Location of contact: (Please enter location here) Did you provide any equipment: Yes No	
Location of contact: (Please enter location here) Did you provide any equipment: Yes □ No □ What did you prescribe: (Please enter here)	
Location of contact: (Please enter location here) Did you provide any equipment: Yes No What did you prescribe: (Please enter here) What was covered in the session (brief summary or bullet points):	
Location of contact: (Please enter location here) Did you provide any equipment: Yes What did you prescribe: (Please enter here) What was covered in the session (brief summary or bullet points): (Please enter here)	
Location of contact: (Please enter location here) Did you provide any equipment: Yes What did you prescribe: (Please enter here) What was covered in the session (brief summary or bullet points): (Please enter here) Total participant contact time: H H H H	

(Please enter here)

Safety Monitoring

If you become aware of any serious adverse event that has occurred because of the intervention delivery, please call the PenCTU Trial Managers on 01752 315256

OB004 OBtAIN-PD manual v1.1 24/03/2023

Any other comments:



.

Version 1.0 28.11.2022 IRAS Project ID: 318175 (Insert trial logo)

OBtAIN-PD DIARY

Monthly diary for week commencing: ____/ ___/ ____/ _____

Region: LiveWell South West Royal Devon

Participant ID: __ __ __

OB004 OBtAIN-PD manual v1.1 24/03/2023



(Insert trial logo)

Version 1.0 28.11.2022 IRAS Project ID: 318175 Example Completed Diary

Number of Falls	None	Less than usual	About the same ⊠	More than usual 🗆	A lot more than usual
Number of falls that caused you an injury	None ⊠	Less than usual	About the same □	More than usual 🗆	A lot more than usual □
Freezing of Gait episodes	None	Less than usual ⊠	About the same	More than usual 🗆	A lot more than usual
Panic Attacks	None	Less than usual ⊠	About the same □	More than usual 🗆	A lot more than usual □
Unplanned GP appointment	None ⊠	Less than usual	About the same	More than usual 🗆	A lot more than usual
Unplanned hospital attendance	None	Less than usual	About the same	More than usual ⊠	A lot more than usual
Unplanned social services visit	None ⊠	Less than usual	About the same	More than usual 🗆	A lot more than usual
If you are in paid employment, have you had to reduce the number of hours you worked this week due	No. I wo the sam hours []	orked e]	Yes. I have had to reduce my working hours		Yes. I have had to stop work completely

OB004 OBtAIN-PD manual v1.1 24/03/2023



(Insert trial logo)

Version 1.0 28.11.2022 IRAS Project ID: 318175						
to Parkinson's and anxiety?						
If you are not employed: due to your Parkinson's and anxiety, have you had to reduce the number of hours per week you spend carrying out your normal daily activities (e.g., gardening, housework, social activity)?	No 🗆	Yes. I have had to reduce my normal daily activities a little IX Yes, a little IX to ca			Yes. I have stopped my normal daily activities completely	
Over the last week, has a relative or friend taken time off work to look after you?	Not at all 🗆				Yes. They had o stop work completely	
Have you incurred any other costs because of your Parkinson's and anxiety this week?	Yes 🗆	Yes 🗆		Ν	1o 🗆	
If yes, please list them here.	£20- raised toile £4- parking at he Cancelled socia	t seat ospital I event wit	h friends			

OB004 OBtAIN-PD manual v1.1 24/03/2023



(Insert trial logo)

Version 1.0 28.11.2022 IRAS Project ID: 318175

4 weekly log

Number of Falls	None	Less than usual	About the same	More than usual	A lot more than usual
Number of falls that caused you an injury	None	Less than usual	About the same	More than usual 🗆	A lot more than usual
Freezing of Gait	None	Less than usual	About the same	More than usual 🗆	A lot more than usual
Panic Attacks	None	Less than usual □	About the same	More than usual 🗆	A lot more than usual □
Unplanned GP appointment	None	Less than usual	About the same	More than usual 🗆	A lot more than usual □
Unplanned hospital attendance	None	Less than usual	About the same	More than usual 🗆	A lot more than usual □
Unplanned social services visit	None	Less than usual	About the same	More than usual 🗆	A lot more than usual □
If you are in paid employment, have you had to reduce the number of hours you			Yes. I have had to reduce my working hours		Yes. I have had to stop

OB004 OBtAIN-PD manual v1.1 24/03/2023



(Insert trial logo) Version 1.0 28.11.2022 IRAS Project ID: 318175 worked this week due No. I worked work to Parkinson's and the same completely anxiety? hours If you are not No 🗆 Yes. I have had to Yes. I have employed: due to your stopped my reduce my normal Parkinson's and normal daily daily activities a anxiety, have you had activities little 🗆 to reduce the number completely \Box of hours per week you spend carrying out your normal daily activities (e.g., gardening, housework, social activity)? Over the last week, Not at all \Box Yes, a little 🗆 Yes. They had has a relative or friend to stop work taken time off work to completely \Box look after you? Have you incurred any other costs because of Yes 🗆 No 🗆 your Parkinson's and anxiety this week? If yes, please list them here.

OB004 OBtAIN-PD manual v1.1 24/03/2023

OBtAIN-PD information sheet (Parkinson's & Anxiety) V.1 30.09.2022 IRAS ID: 318175

Parkinson's and Anxiety

People with Parkinson's may be more prone to feeling anxious than other people due to the brain chemicals involved in Parkinson's. Living with a long-term condition can be stressful, and combined with the chemical brain changes in Parkinson's, this may mean people with Parkinson's are more prone to anxiety than other people.

What are the main symptoms of anxiety?

Anxiety can cause both psychological and physical symptoms. Some of the psychological symptoms can be:

- Having a sense of impending doom, panic, danger, or dread.
- Trouble concentrating.
- Trouble thinking about anything other than the present worry.
- Still feeling worried once the situation has passed.

There are also physical symptoms of anxiety. These can include:

- Feeling restless, tense, or nervous.
- Sweating.
- Dizziness.
- Feeling breathless, or rapid breathing (hyperventilation).
- A racing heart (palpitations).
- Trembling.
- Stomach cramps, nausea, and indigestion.

When people become anxious for long periods of time, this can affect their life and cause feelings of hopelessness. It may affect things like their work and social life.

How does anxiety affect people with Parkinson's?

OBtAIN-PD information sheet (Parkinson's & Anxiety) V.1 30.09.2022 IRAS ID: 318175

In Parkinson's, anxiety is likely caused by a combination of things like changes in brain chemicals, stressful life events, and genetics (i.e., a family history of anxiety). The concerns and worries of living with a long-term condition like Parkinson's can also contribute to anxiety.

Some people with Parkinson's experience anxiety when their medications wear 'off'. Taking the medication can improve their anxiety symptoms but they can come back before the next dose is due. This is called 'wearing off'.

Anxiety can make people with Parkinson's experience the following:

- Their tremor might get worse.
- They might experience more 'freezing' when they walk. This means that they are at a greater risk of falling over.
- They might feel stiffer and find it harder to move.

How can I manage my anxiety?

Regular physical activity can help you to feel less anxious. If you feel that you would like to learn more about this, ask your occupational therapist for the information sheet called 'Physical Activity and Exercise'.

Eating a healthy, balanced diet can also help manage the symptoms of anxiety. If you would like to learn more about this, please ask your occupational therapist for the information sheet called 'Diet and Nutrition'.

Sleep is important for your physical and mental health. If you would like to learn more about this, please ask your occupational therapist for the information sheet called 'Sleep Hygiene'.

The OBtAIN-PD intervention is designed to help you manage your anxiety and take part in the things that you want to do. Your occupational therapist will work with you to develop a goal and help you to come up with a plan to work on this. They will help you to come up with this plan yourself using an approach called 'coaching' to help empower you.

OBtAIN-PD OBtAIN-PD information sheet (Diet and Nutrition) V.1 30.09.2022 IRAS ID: 318175

Diet and Nutrition

If you have Parkinson's, there is no particular diet that you should follow unless you have been prescribed one by a registered healthcare professional. It is recommended to you maintain as healthy a diet as possible.

The general rules for following a healthy diet are:

- Eat a varied diet, that is do not eat a few of the same foods all the time.
- Eat at least five portions of fruit and vegetables a day.
- Eat regular meals.
- Make sure to drink plenty of fluids.
- Eat healthy portions of food (avoid eating too much in one sitting).
- Eat the right balance of carbohydrates (carbs), proteins, and fats.

If you are worried about your diet, your occupational therapist can help you to look at the choices you are making. If they feel it is needed, they might refer you to a dietitian.

Food and medication

When you take your medication should always be discussed with your consultant, GP, or Parkinson's nurse.

Some people with Parkinson's can feel sick after taking their medication. Having a simple snack like a plain cracker can help with this. You may find that taking your medication with water can help to reduce the sick feeling.

For some people with Parkinson's, protein (found mainly in meat, fish, eggs, pulses, and dairy) may interfere with the effects of their levodopa medication. It is generally advised that you should take your Parkinson's medication 30-45 minutes before meals.

If you want to review the timing of your medication to reduce the impact on your life and mealtimes, you should see your consultant, GP, Parkinson's nurse, or ask to be referred to a dietitian.

You should not stop eating protein completely as it is important to help your body recover and fight infection.

OBTAIN-PD OBTAIN-PD information sheet (Diet and Nutrition) V.1 30.09.2022 IRAS ID: 318175

Should I buy special cutlery?

The physical symptoms of Parkinson's can make it more difficult for you to get food or drink to your mouth. There are many types of adapted cutlery and utensils available that might be able to help you.

Special cutlery can help you if you have reduced grip, muscle weakness, or a tremor. This cutlery might have special handles or be shaped differently.

Special cups and mugs can help you to prevent spills when you are drinking. These cups and mugs may have extra handles, be shaped differently, or have a lid.

There are special plates available that can be helpful, particularly if you find it difficult to get food on your cutlery. These plates can help keep food warmer if it takes you a long time to eat, you can also get plates with high lips to stop spills, or non-slip mats to stop plates sliding around.

There are other aids available that can help with things like opening tins, bottles, and jars.

Your occupational therapist can help to advise you on the right equipment for your needs. They might refer you to a place called the 'Independent Living Centre' if you need more specialist advice.

Swallowing problems

Some people with Parkinson's may find that they have problems swallowing when they are eating or drinking.

If you notice any problems with your swallowing (such as coughing when you eat or drink, finding it hard to start swallowing so food stays in your mouth, or find it difficult to keep food and drink in your mouth) then it is very important that you talk to your GP as soon as possible.

They can refer you to a speech and language therapist who will be able to give you specialist advice.



Physical Activity and Exercise

There is now lots of research evidence that shows that being active can help manage the symptoms of Parkinson's. Being physically active has a range of positive benefits both physically and mentally, including on anxiety.

How can physical activity and exercise help me?

It is easier to live well with Parkinson's if you are more physically active. Research shows that being more active can be as important as your medication in helping to manage your Parkinson's symptoms. The current recommendation is that being active for 2.5 hours a week can help to manage symptoms.

Regular physically activity will help to improve your strength, balance, and posture. It can also help with sleep, tiredness, your mood, and your mental health (including anxiety).

The activities that you can do might be affected by your current Parkinson's symptoms. Your occupational therapist can help you to work out what you might be able to do, or can refer you on to someone who can.

What about exercise?

Exercise is an activity that is performed to specifically improve physical fitness and overall wellness. Examples of exercise can include things like boxing, weightlifting, and exercise classes. There might be Parkinson's exercise classes in your area, if you would prefer to exercise with other people with Parkinson's.

Get active doing something that you enjoy

The right physical activity is different for everyone. Being active is nearly always most effective when you push yourself and get out of breath.

OBTAIN-PD information sheet (PA and Exercise) V.1 30.09.2022 IRAS ID: 318175

Whatever activity you decide to do, it is important to do it regularly. If you enjoy the physical activity that you are doing then you are more likely to keep doing it.

Some people do not like the idea of doing exercise. There are lots of other activities that you could do to help you be more active. Examples of these activities could be:

- Gardening
- Walking/ hiking
- Dancing
- Housework
- Playing with children/ grandchildren

Your occupational therapist can help you to identify activities that you enjoy that could be used as regular physical activity. They may refer you to another professional if you need more specific advise, such as your GP who might be able to supply you with a gym prescription.

OBTAIN-PD OBTAIN-PD information sheet (Sleep) V.1 30.09.2022 IRAS ID: 318175

Sleep Hygiene

Sleep is a naturally occurring and essential process for human beings. It is important for both physical recovery, mental restoration, and processing of memories.

It is important to get enough sleep and rest. Restful periods help the brain to refresh and cleanse itself. This helps us to absorb and process new information better.

The NHS recommends that adults need seven to nine hours of sleep per night. The amount of sleep that you need is individual to you and can be dependent on a range of other factors, like how much physical activity you do. Having a consistent cycle of sleeping and being awake is very important

When sleeping, fluid flows through the brain that removes waste products that are associated with conditions like Parkinson's. Falling and staying asleep at night can be more difficult if you have Parkinson's. 'Sleep hygiene' is the term used to describe good sleep habits.

Sleep hygiene involves doing things to promote good sleep and avoiding things that interfere with your sleep. Below are some examples of sleep hygiene that you can use to help you. You may want to discuss these with your occupational therapist:

- Try to keep to a regular routine- try to stick to a regular pattern of bed times and waking up.
- Avoid caffeine before you go to bed- it is important to not drink too much caffeine during the day, and you may consider using decaffeinated drinks.

OBTAIN-PD OBTAIN-PD information sheet (Sleep) V.1 30.09.2022 IRAS ID: 318175

- Avoid smoking- Nicotine is a stimulant like caffeine and the effects are similar. Avoid smoking around bedtime or during the day, even if it feels like it relaxes you.
- Get regular exercise- but try to avoid vigorous exercising within two hours of going to bed as the effects of this may make it harder to fall asleep. Regular exercise can help to improve certain sleep problems.
- Use your bedroom mainly for sleeping- Keeping your bedroom for sleep can help your mind to associate it with activities that lead to sleep. Try to avoid things like watching television or using electronic devices with screens in bed.
- Keep your bedroom calm and comfortable- avoid excessively hot or cold temperatures as this may disturb your sleep and make you restless. Keep your curtains or blinds closed to reduce light coming into the room. You might want to try an eye mask if this is difficult.
- Try to relax before going to bed- If you are in a relaxed mindset before you go to bed, then you may find it easier to drift off to sleep. Other things that might help you relax could be taking a warm bath or shower, listening to soothing music, meditating, reading, or trying relaxation exercises.
- Don't stay in bed if you cannot sleep- if you can't fall asleep within 20 minutes of going to bed or you wake early and can't get back to sleep, get out of bed and try a relaxing activity until you become drowsy.

You might have some other ideas that you might want to discuss with your occupational therapist. They can help you to identify specific issues and help you to come up with your own solutions to address them.

Appendix 11: OBtAIN-PD therapist contact form (Chapter 5)

Treating Therapists Contact Sheet V1 2.09.2022 IRAS ID: 318175 Sponsor: University of Plymouth Chief Investigator: Chris Lovegrove	(insert trial logo here)
Participant study number:	
Participant initials:	
Date: D D M M Y Y	
Time: H H : M M AM PM D	
Session number:	
Type of contact: Home visit Online consultation Telephor	ne consultation \Box
First visit 🗆 🛛 Follow-up 🗆 🔹 Discharg	e 🗆
Location of contact: (Please enter location here)	
Did you provide any equipment: Yes 🗌 No 🗌	
What did you prescribe: (Please enter here)	

What was covered in the session (brief summary or bullet points):

(Please enter here)							
Total participant contact time:	Н	Н	:	Μ	M		
Total administration time:	Н	Н	:	M	Μ		
Any other comments: (Please ente	r here)						

Safety Monitoring

If you become aware of any serious adverse event that has occurred because of the intervention delivery, please call the PenCTU Trial Managers on 01752 315256



OBtAIN-PD [Occupational Therapists] Interview Schedule

The guide acts as a prompt, covering the pertinent questions and areas to probe.

Introductions and confirmation of consent process and voluntary nature of the research. Remind participant that they are free to end the interview at any time. Check that the participant is happy for the interview to be recorded (video and text transcription on Microsoft Teams) and **start recording.**

[Read to participant] The purpose of today's interview is to find out how you experienced delivering either the OBtAIN-PD intervention or usual occupational therapy care. We would also like to find out your experience of being in the OBtAIN-PD study, such as completing the training, screening participants, and using the systems. You will not be identifiable in the research report, publications, or feedback to NHS staff or patients, so please speak freely. We are interested in both positive and less positive experiences to help us understand how to optimise the OBtAIN-PD intervention and research going forward. Do you have any questions before we start? [Time for response] Are you happy to continue? [Time for response]

1. Background

Aims: - To help the interviewee settle into the interview and to provide some context for subsequent questions/responses.

[Ask participant] Could you tell me a little about your history of what you do (your work or role) with the community rehabilitation team.

Ask participant to confirm the team that they are in, and if they were delivering the OBtAIN-PD or usual occupational therapy care. Probe on experience, role within the service, length of time in team, any prior research experience.

2. Experiences of OBtAIN-PD training

Aims: To understand how acceptable the training programme is for occupational therapists - their engagement with the content and their confidence to deliver the intervention/study.

[Ask participant, both groups] Could you tell me how you found the OBtAIN-PD study training?

[Ask participant, OBtAIN-PD group only] How did you find the training video?

[Ask participant, both groups] How did you find the trial manual(s)?

Probe on how they felt before, during and after the training and their experience of the content/activities/delivery. Usual care OTs received usual care manual and training at the SIV.

OBtAIN-PD interview schedule (Clinicians) v1.0 23.11.2023, IRAS ID: 318175 Page 1 of 4





OBtAIN-PD clinicians received usual care AND OBtAIN-PD manual, SIV, plus a training video.

[If participant delivered OBtAIN-PD, ask this] Could you tell me how you found the OBtAIN-PD intervention training?

[If participant delivered OBtAIN-PD, ask this] How confident were you in delivering the OBtAIN-PD?

Probe on how they felt before, during and after the training and their experience of the content/activities/delivery.

3. Experiences of delivering OBtAIN-PD/ usual occupational therapy care

Aims: To explore aspects of delivery – what worked, what didn't work so well, delivery challenges, how challenges were overcome etc.

[Ask participant] Could you tell me how delivery of [the OBtAIN-PD/ usual occupational therapy care]

went for you?

Probes: whether experiences changed over time (beginning, middle and end), what worked, what was challenging, how they overcame these.

[Ask participant] Could you tell me if there were any staffing issues with delivering [the OBtAIN-PD/ usual occupational therapy care]?

[Ask participant] How do you think the intervention was for the client?

Probes: whether they also delivered usual care, whether the commitment to delivering OBtAIN-PD took them away from other duties, and, if so, the impact.

[Ask participant] What factors do you think influenced the outcome of the intervention? Probes: client factors (such as health, confusion, other stresses, willingness), intervention factors (design, access to materials), own factors (belief in intervention, experience levels)

[Ask participant] Could you tell me how you think the participants responded to [the OBtAIN-PD/ usual occupational therapy care]? Probes: levels of engagement, attrition

OBtAIN-PD interview schedule (Clinicians) v1.0 23.11.2023, IRAS ID: 318175 Page 2 of 4





[Ask all participants] How did you find recruiting to the study?

[Ask participant if their team did not recruit any participants] Could you tell me why you think your

team didn't recruit participants?

Probes: what could be done to help improve recruitment

4. Acceptability of Trial Processes

Aims: To understand facilitator experience of being involved in the trial.

[Ask participant] Could you tell me how you felt about being involved in the trial?

Probes: quality of interactions with colleagues/researchers, were there any fears/concerns/issues and were they were sufficiently allayed.

[Ask participant] How did you feel about the screening process for participants?

Probes: whether they did this, when they completed the forms and how much of a burden these were. What was it like talking about the trial with potential participants?

[Ask participant] How did you find using the electronic system for recording the interventions?

Probes: whether they did this, when they completed the online forms and how much of a burden these were. Were there any problems, if so what was it like getting help. Were the issues rectified.

5. Contamination

Aims: To ascertain if occupational therapists who also deliver usual care import aspects of OBtAIN-PD $\ensuremath{\mathsf{PD}}$

[Ask participant if they delivered usual care] Do you have any sense of what was involved in

OBtAIN-PD? Did you use any of this in delivering usual care?

Probe: how did they find out about it, did they get access to materials, did they use it at all.

Reinforce that there is no repercussion and not about blame.

[Ask participant if they delivered OBtAIN-PD] Did you draw upon aspects of the OBtAIN-PD when

you delivered usual care? If so, what were they?

Probe: which aspects did they use and why they chose these. Did they discuss the intervention with colleagues from other teams. Were they advised by colleagues to do anything different, and did they do this. Key components of OBtAIN-PD include; bespoke goal setting sheets, bespoke information sheets (Parkinson's and anxiety, sleep, exercise, or diet & nutrition), the OBtAIN-PD manual to

OBtAIN-PD interview schedule (Clinicians) v1.0 23.11.2023, IRAS ID: 318175 Page 3 of 4





structure conversations (e.g. used specific prompts from the manual).

6. Suggestions for improving the OBtAIN-PD intervention/ study

Aim: To support the optimisation of the OBtAIN-PD intervention and research programme.

[Ask participant] What would you change about the OBtAIN-PD study? Probes: why these refinements would help

[Ask participant if they delivered OBtAIN-PD] What would you change about the OBtAIN-PD intervention?

Probes: why these refinements would help

[Ask participant] What would you change about training or resources to help you deliver [intervention]?

Probe: why these refinements would help

[Ask participant] Have you any comments/thoughts about how training could be improved?

[Ask participant] Would you like to share anything else about your experiences?

Thank the participant for their time and for sharing their experiences in the interview. State that you will now stop the recording and transcription.

STOP THE RECORDING NOW BEFORE PROCEEDING.

After recording has stopped, please ask the participant if they have any further questions or comment. If the participant offers something that might be valuable, or something else has come to mind, ask them for permission to restart the recording and transcription. Ask them to repeat what they just told you and thank them again for their time and stop the recording/transcription.

Tell the participant that if anything comes to mind following the interview, they can email this to the interviewer. Ask the participant if they would be willing to comment on a summary of the analysis. Thank the participant for their time.

OBtAIN-PD interview schedule (Clinicians) v1.0 23.11.2023, IRAS ID: 318175 Page 4 of 4